UC Riverside

UC Riverside Previously Published Works

Title

Para-hyper-Kähler Geometry of the Deformation Space of Maximal Globally Hyperbolic Anti-de Sitter Three-Manifolds

Permalink

https://escholarship.org/uc/item/5cw1h4x9

Journal

Memoirs of the American Mathematical Society, 306(1546)

ISSN

0065-9266

Authors

Mazzoli, Filippo Seppi, Andrea Tamburelli, Andrea

Publication Date

2025-02-01

DOI

10.1090/memo/1546

Copyright Information

This work is made available under the terms of a Creative Commons Attribution License, available at https://creativecommons.org/licenses/by/4.0/

Peer reviewed

PARA-HYPERKÄHLER GEOMETRY OF THE DEFORMATION SPACE OF MAXIMAL GLOBALLY HYPERBOLIC ANTI-DE SITTER THREE-MANIFOLDS

FILIPPO MAZZOLI, ANDREA SEPPI, AND ANDREA TAMBURELLI

ABSTRACT. In this paper we study the para-hyperKähler geometry of the deformation space of MGHC anti-de Sitter structures on $\Sigma \times \mathbb{R}$, for Σ a closed oriented surface. We show that a neutral pseudo-Riemannian metric and three symplectic structures coexist with an integrable complex structure and two para-complex structures, satisfying the relations of para-quaternionic numbers. We show that these structures are directly related to the geometry of MGHC manifolds, via the Mess homeomorphism, the parameterization of Krasnov-Schlenker by the induced metric on K-surfaces, the identification with the cotangent bundle $T^*\mathcal{T}(\Sigma)$, and the circle action that arises from this identification. Finally, we study the relation to the natural para-complex geometry that the space inherits from being a component of the $\mathbb{P}\mathrm{SL}(2,\mathbb{B})$ -character variety, where \mathbb{B} is the algebra of para-complex numbers, and the symplectic geometry deriving from Goldman symplectic form

Contents

1.	Introduction	2
1.1.	Motivation and state-of-the-art	2
1.2.	Deformation space of MGHC AdS manifolds	3
1.3.	Para-hyperKähler structures	4
1.4.	Parameterizations of $\mathcal{MGH}(\Sigma)$	5
1.5.	The circle action	7
1.6.	The character variety	8
1.7.	Outline of techniques and proofs	10
Acknowledgments		14
2.	Preliminaries on Anti-de Sitter geometry	15
2.1.	Maximal surfaces	15
2.2.	Cotangent bundle of Teichmüller space	16
2.3.	Mess' parameterization	18
2.4.	Constant curvature surfaces and circle action	20
2.5.	An equivalent model for $\mathcal{MS}(\Sigma)$	21
3.	The toy model: genus 1	23

Date: October 14, 2022.

3.1. Space of linear almost-complex structures	23
3.2. The tangent space of $T^*\mathcal{J}(\mathbb{R}^2)$	24
3.3. A para-hyperKähler structure on $T^*\mathcal{J}(\mathbb{R}^2)$	26
3.4. Liouville form on $T^*\mathcal{J}(\mathbb{R}^2)$	27
3.5. Relation with $\mathcal{MGH}(T^2)$	30
3.6. A formal Mess homeomorphism	33
3.7. The circle action	36
3.8. A one-parameter family of maps	39
4. The general case: genus ≥ 2	40
4.1. The group of (Hamiltonian) symplectomorphisms and its Lie	e algebra 40
4.2. The Teichmüller space as a symplectic quotient	41
4.3. The construction of $\mathcal{MS}_0(\Sigma, \rho)$	46
4.4. The para-hyperKähler structure of $\mathcal{MS}_0(\Sigma, \rho)$	48
4.5. The proof of Theorem L	50
4.6. The proof of Proposition K	62
5. Geometric interpretations	
5.1. The cotangent bundle parametrization	69
5.2. The Mess homeomorphism	70
5.3. The circle action on $\mathcal{MS}(\Sigma)$	71
5.4. Para-complex geometry of the $\mathbb{P}SL(2,\mathbb{B})$ -character variety	73
6. Symplectic reduction	83
6.1. The moment maps on $T^*\mathcal{J}(\mathbb{R}^2)$	83
6.2. Donaldson's construction	85
6.3. The moment maps on $T^*\mathcal{J}(\Sigma)$	87
6.4. The symplectic quotient	90
Appendix A. Para-complex geometry	
References	100

1. Introduction

The main purpose of this paper is to study the geometry of the deformation space of maximal globally hyperbolic Cauchy-compact three-dimensional Anti-de Sitter three-manifolds. In short, our results show that these deformation spaces are endowed with a mapping-class group invariant para-hyperKähler metric, and then provide geometric interpretations to each element that constitutes the para-hyperKähler structure.

1.1. Motivation and state-of-the-art

Since the pioneering work of Mess of 1990 [Mes07], maximal globally hyperbolic Anti-de Sitter manifolds in dimension three have been largely studied, motivated on the one hand by

the striking analogies with quasi-Fuchsian manifolds, and on the other by the deep relations with Teichmüller theory. See also, among others, [ABB+07, BB09, Bar15, Bar18, BS20].

In particular, the deformation space of maximal globally hyperbolic Anti-de Sitter manifolds, which in this paper we will denote by $\mathcal{MGH}(\Sigma)$, where Σ is a closed surface of genus ≥ 2 , is intimately related with the Teichmüller space $\mathcal{T}(\Sigma)$. This has been first observed by Mess, who provided a parametrization of $\mathcal{MGH}(\Sigma)$ by the product $\mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$; other parametrizations again by $\mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$ or by $T^*\mathcal{T}(\Sigma)$ were introduced in [KS07]. The latter relies on existence and uniqueness results for maximal surfaces, as shown in [BBZ07, BS10]. A further understanding of the geometry of Anti-de Sitter manifolds and their deformation space has been obtained by the study of geometric invariants such as the convex core ([BS12, SS18]), its width ([BS10, Sep19]), its volume ([BST17]), by means of surfaces with curvature conditions ([BBZ07, ABBZ12, BS18, Tam19, CT19]), and by the symplectic geometric approach ([BMS13, BMS15, BS19]).

Stepping back to the parallel with quasi-Fuchsian hyperbolic manifolds, recently Donaldson highlighted the existence of a natural $hyperK\ddot{a}hler$ structure on a neighborhood of the Fuchsian locus in the deformation space of almost-Fuchsian manifolds, seen as a neighborhood of the zero section in the cotangent bundle $T^*\mathcal{T}(\Sigma)$. See [Don03, Hod05, Tra18, Tra19]. The purpose of this paper is to develop a similar approach for maximal globally hyperbolic Anti-de Sitter manifolds, and to demonstrate that the natural structure that appears in this setting is a $para-hyperK\ddot{a}hler$ structure. For more details on para-Kähler and para-hyperKähler geometry, see [CFG96, GMV01, AMT09a, Vac12]. We will see that this structure recovers many of the geometric constructions that have been introduced before, so as to elucidate the global picture and the relations between different approaches. We now give the fundamental definitions and state our main results.

1.2. Deformation space of MGHC AdS manifolds

We give here the standard definition of maximal globally hyperbolic Cauchy compact Antide Sitter manifolds (in short, MGHC AdS). A Cauchy surface in a Lorentzian manifold is an embedded hypersurface that intersects every inextensible causal curve exactly in one point; a Lorentzian manifold admitting a Cauchy surface is called globally hyperbolic. It is moreover maximal if every isometric embedding in another globally hyperbolic manifold sending a Cauchy surface to a Cauchy surface is surjective. Finally, a MGHC AdS manifold is a maximal globally hyperbolic Lorentzian manifold of constant sectional curvature -1 admitting a closed Cauchy surface. A simple example of MGHC AdS manifolds are Fuchsian manifolds, whose metric G can be written globally as a warped product

$$G = -dt^2 + \cos^2(t)h , \qquad (1.1)$$

for $t \in (-\pi/2, \pi/2)$ and h a hyperbolic metric on a closed manifold. In this case the Cauchy surface t = 0 is totally geodesic.

A classical fact in Lorentzian geometry (see [Ger70, BE81, BS03]) is that globally hyperbolic Lorentzian manifolds are diffeomorphic to $\Sigma \times \mathbb{R}$, where Σ is a Cauchy surface, and any two Cauchy surfaces are diffeomorphic. In this work we will only consider three-dimensional AdS manifolds whose Cauchy surfaces are closed. Hence from now on we will fix a closed oriented surface Σ . We then define the *deformation space* of MGHC AdS manifolds as follows:

$$\mathcal{MGH}(\Sigma) := \{G \mid G \text{ is a MGHC AdS metric on } \Sigma \times \mathbb{R}\}/\mathrm{Diff}_0(\Sigma \times \mathbb{R}) ,$$

where the group $\mathrm{Diff}_0(\Sigma \times \mathbb{R})$ of diffeomorphisms isotopic to the identity acts by pull-back of G. It turns out that $\mathcal{MGH}(S^2)$ is empty, $\mathcal{MGH}(T^2)$ is a four-dimensional manifold, while if Σ is a surface of genus ≥ 2 , then $\mathcal{MGH}(\Sigma)$ has dimension $6|\chi(\Sigma)|$. Observe moreover that there is a natural action of the mapping class-group $MCG(\Sigma) = \mathrm{Diff}_+(\Sigma)/\mathrm{Diff}_0(\Sigma)$ on $\mathcal{MGH}(\Sigma)$, again by pull-back. When Σ has genus ≥ 2 , the deformation space $\mathcal{MGH}(\Sigma)$ contains the Fuchsian locus $\mathcal{F}(\Sigma)$, namely those manifolds whose metric is of the form (1.1), which is $MCG(\Sigma)$ -invariant and naturally identified to the Teichmüller space $\mathcal{T}(\Sigma)$.

1.3. Para-hyperKähler structures

We now introduce the notion of para-hyperKähler structure and state our first result. Recall that a pseudo-Kähler structure on a manifold M consists of a pair (\mathbf{g}, \mathbf{I}) where \mathbf{g} is a pseudo-Riemannian metric and \mathbf{I} is an integrable almost complex structure (i.e. \mathbf{I} is a section of the bundle of endomorphisms of the tangent bundle, that satisfies $\mathbf{I}^2 = -1$ and is induced by a complex structure on M) such that $\mathbf{g}(\mathbf{I}v, w) = -\mathbf{g}(v, \mathbf{I}w)$ and the 2-form $\omega_{\mathbf{I}}(\cdot, \cdot) := \mathbf{g}(\cdot, \mathbf{I}\cdot)$ is closed (hence a symplectic form). Similarly, a para-Kähler structure is a pair (\mathbf{g}, \mathbf{P}) , where \mathbf{P} is an integrable almost para-complex structure, which means that

- $P^2 = 1$:
- the **P**-eigenspaces of 1 and -1 have the same dimension;
- the distributions on M given by the 1 and -1 eigenspaces of \mathbf{P} are integrable;

and **P** is such that $\mathbf{g}(\mathbf{P}v, w) = -\mathbf{g}(v, \mathbf{P}w)$ and the 2-form $\omega_{\mathbf{P}}(\cdot, \cdot) := \mathbf{g}(\cdot, \mathbf{P}\cdot)$ is closed.

Observe that a direct consequence of the existence of a para-Kähler structure is that $\mathbf{g}(\mathbf{P}\cdot,\mathbf{P}\cdot)=-\mathbf{g}(\cdot,\cdot)$, hence \mathbf{g} is necessarily of neutral signature. Moreover the condition that $d\omega_{\mathbf{I}}=0$ (resp. $d\omega_{\mathbf{P}}=0$) is known to be equivalent to $\nabla\mathbf{I}=0$ (resp. $\nabla\mathbf{P}=0$), for ∇ the Levi-Civita connection of \mathbf{g} . We finally give the definition of para-hyperKähler structure:

Definition. A para-hyperKähler structure on a manifold M is the data $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$, where (\mathbf{g}, \mathbf{I}) is a pseudo-Kähler structure, (\mathbf{g}, \mathbf{J}) and (\mathbf{g}, \mathbf{K}) are para-Kähler structures, and $(\mathbf{I}, \mathbf{J}, \mathbf{K})$ satisfy the para-quaternionic relations.

By para-quaternionic relations we mean the identities $\mathbf{I}^2 = -1$, $\mathbf{J}^2 = \mathbf{K}^2 = 1$ — which are implicitly assumed by the condition that \mathbf{I} (resp. \mathbf{J} , \mathbf{K}) is a complex (resp. paracomplex) structure — and moreover $\mathbf{IJ} = -\mathbf{JI} = \mathbf{K}$.

We remark that, given a para-hyperKähler structure $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$, a complex symplectic form is defined by:

$$\omega_{\mathbf{I}}^{\mathbb{C}} := \omega_{\mathbf{J}} + i\omega_{\mathbf{K}} .$$

It is complex in the sense that it is a \mathbb{C} -valued symplectic form and satisfies $\omega_{\mathbf{I}}^{\mathbb{C}}(\mathbf{I}v, w) = \omega_{\mathbf{I}}^{\mathbb{C}}(v, \mathbf{I}w) = i\omega_{\mathbf{I}}^{\mathbb{C}}(v, w)$. Similarly, one has two *para-complex symplectic forms* defined by

$$\omega_{\mathbf{J}}^{\mathbb{B}} := \omega_{\mathbf{I}} + \tau \omega_{\mathbf{K}}$$
 and $\omega_{\mathbf{K}}^{\mathbb{B}} := \omega_{\mathbf{I}} - \tau \omega_{\mathbf{J}}$,

where we denote by $\mathbb{B} = \mathbb{R} \oplus \tau \mathbb{R}$ the algebra of para-complex numbers, i.e. $\tau^2 = 1$. Again, these are para-complex in the sense that $\omega_{\mathbf{J}}^{\mathbb{B}}(\mathbf{J}v, w) = \omega_{\mathbf{J}}^{\mathbb{B}}(v, \mathbf{J}w) = \tau \omega_{\mathbf{J}}^{\mathbb{B}}(v, w)$ and $\omega_{\mathbf{K}}^{\mathbb{B}}(\mathbf{K}v, w) = \omega_{\mathbf{K}}^{\mathbb{B}}(v, \mathbf{K}w) = \tau \omega_{\mathbf{K}}^{\mathbb{B}}(v, w)$.

Only manifolds of dimension 4n can support a para-hyperKähler structure. Our first result is that $\mathcal{MGH}(\Sigma)$, whose dimension is four if Σ has genus one and $6|\chi(\Sigma)|$ otherwise, does support a very natural one.

Theorem A. Let Σ be a closed oriented surface of genus $\geqslant 1$. Then $\mathcal{MGH}(\Sigma)$ admits a $MCG(\Sigma)$ -invariant para-hyperKähler structure $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$. When Σ has genus $\geqslant 2$, the Fuchsian locus $\mathcal{F}(\Sigma)$ is totally geodesic and (\mathbf{g}, \mathbf{I}) restricts to (a multiple of) the Weil-Petersson Kähler structure of Teichmüller space.

The para-hyperKähler structure of $\mathcal{MGH}(\Sigma)$ is extremely natural from the point of view of AdS geometry, in the sense that all the elements that constitute the para-hyperKähler structure have (at least one) interpretation in terms of the geometry of MGHC AdS manifolds. We now state and explain all these interpretations.

1.4. Parameterizations of $\mathcal{MGH}(\Sigma)$

The first interpretation is in terms of the cotangent bundle of Teichmüller space. There is a natural map

$$\mathcal{F}: \mathcal{MGH}(\Sigma) \to T^*\mathcal{T}(\Sigma)$$
,

which associates to a MGHC AdS manifold $(\Sigma \times \mathbb{R}, G)$ the pair (J,q), where J is the (almost-)complex structure of the first fundamental form of the unique maximal Cauchy surface in (M,G), and q is the holomorphic quadratic differential whose real part is the second fundamental form. The map \mathcal{F} is a $(MCG(\Sigma)$ -equivariant) diffeomorphism if Σ has genus ≥ 2 ; for genus one it is a diffeomorphism onto the complement of the zero section. The cotangent bundle $T^*\mathcal{T}(\Sigma)$ is naturally a complex symplectic manifold; our first geometric interpretation is the fact that the map \mathcal{F} is anti-holomorphic and preserves the complex symplectic forms up to conjugation.

Theorem B. Let Σ be a closed oriented surface of genus ≥ 1 . Then

$$\mathcal{F}^*(\mathcal{I}_{T^*\mathcal{T}(\Sigma)}, \Omega_{T^*\mathcal{T}(\Sigma)}^{\mathbb{C}}) = \left(-\mathbf{I}, -\frac{i}{2}\overline{\omega}_{\mathbf{I}}^{\mathbb{C}}\right) ,$$

where $\mathcal{I}_{T^*\mathcal{T}(\Sigma)}$ denotes the complex structure of $T^*\mathcal{T}(\Sigma)$ and $\Omega^{\mathbb{C}}_{T^*\mathcal{T}(\Sigma)}$ its complex symplectic form.

Let us assume (until the end of this section) that Σ has genus ≥ 2 . In [Mes07], Mess proved that $\mathcal{MGH}(\Sigma)$ is parameterized by the product of two copies of the Teichmüller space of Σ , by a map

$$\mathcal{M}: \mathcal{MGH}(\Sigma) \to \mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$$
,

that essentially gives (under the isomorphism between the isometry group of AdS space and $\mathbb{P}SL(2,\mathbb{R})\times\mathbb{P}SL(2,\mathbb{R})$), the left and right components of the holonomy map of a MGHC AdS manifold (M,G). The manifold $\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)$ is easily a para-complex manifold, where the para-complex structure $\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$ is the endomorphism of the tangent bundle for which the integral submanifolds of the distribution of 1-eigenspaces are the slices $\mathcal{T}(\Sigma)\times\{*\}$, and those for the (-1)-eigenspaces are the slices $\{*\}\times\mathcal{T}(\Sigma)$. It has moreover a para-complex symplectic form compatible with $\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$:

$$\Omega_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)} := \frac{1}{2} (\pi_l^* \Omega_{WP} + \pi_r^* \Omega_{WP}) + \frac{\tau}{2} (\pi_l^* \Omega_{WP} - \pi_r^* \Omega_{WP})$$

where Ω_{WP} is the Weil-Petersson symplectic form and π_l, π_r denote the projections on the left and right factor.

Theorem C. Let Σ be a closed oriented surface of genus ≥ 2 . Then

$$\mathcal{M}^*(\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}, 4\Omega_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}^{\mathbb{B}}) = (\mathbf{J}, \omega_{\mathbf{J}}^{\mathbb{B}}) ,$$

where $\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$ denotes the para-complex structure of $\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)$ and $\Omega^{\mathbb{B}}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$ its para-complex symplectic form.

Combining Theorems B and C in a particular case, we see that $(1/2)\omega_{\mathbf{K}}$ equals on the one hand the pull-back by \mathcal{M} of the symplectic form $\pi_l^*\Omega_{WP} - \pi_r^*\Omega_{WP}$, and on the other hand the pull-back by \mathcal{F} of minus the real part of $\Omega_{T^*\mathcal{T}(\Sigma)}^{\mathbb{C}}$ (i.e. the natural real symplectic form of the cotangent bundle). This identity has been proved in [SS18, Theorem 1.14], by completely different methods.

There is another parameterization of $\mathcal{MGH}(\Sigma)$ by the product of two copies of the Teichmüller space of Σ , which has been introduced in [KS07]. It is given by the map

$$C: \mathcal{MGH}(\Sigma) \to \mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$$
,

which associates to (M, G) the first fundamental forms of the two Cauchy surfaces (one future-convex, one past-convex) of constant intrinsic curvature -2. These two Cauchy surfaces of constant curvature are unique ([BBZ11, BS18]), and we rescale their first fundamental forms by a factor so as to consider them as hyperbolic metrics. We show:

Theorem D. Let Σ be a closed oriented surface of genus ≥ 2 . Then

$$\mathcal{C}^*(\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}, 4\Omega^{\mathbb{B}}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}) = (\mathbf{K}, \omega^{\mathbb{B}}_{\mathbf{K}}) ,$$

where $\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$ denotes the para-complex structure of $\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)$ and $\Omega^{\mathbb{B}}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$ its para-complex symplectic form.

We remark that there are formal analogues of Theorem C and Theorem D in genus one (see Section 3.6 and 3.8), but the corresponding maps $\mathcal{M}, \mathcal{C} : \mathcal{MGH}(T^2) \to \mathcal{T}(T^2) \times \mathcal{T}(T^2)$ do not have the same geometric interpretation (namely, the holonomy map or the constant curvature surfaces) as in the higher genus case, which is why we restricted to genus ≥ 2 when stating these results here.

1.5. The circle action

We now move on to studying a circle action on $\mathcal{MGH}(\Sigma)$. Using the diffeomorphism $\mathcal{F}: \mathcal{MGH}(\Sigma) \to T^*\mathcal{T}(\Sigma)$, the circle action on $T^*\mathcal{T}(\Sigma)$ given by $e^{i\theta} \cdot (J,q) = (J,e^{i\theta}q)$ (where J is an almost-complex structure on Σ and q a holomorphic quadratic differential) induces an action of S^1 on $\mathcal{MGH}(\Sigma)$. Let us denote by $R_{\theta}: \mathcal{MGH}(\Sigma) \to \mathcal{MGH}(\Sigma)$ the corresponding self-diffeomorphism. For genus ≥ 2 , this action of S^1 induces an action on $\mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$ by means of the map \mathcal{M} . The so obtained S^1 -action on $\mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$ has been studied in [BMS13, BMS15] under the name of landslide flow.

It will be relevant to introduce the function

$$\mathcal{A}: \mathcal{MGH}(\Sigma) \to \mathbb{R}$$

which associates to a MGHC AdS manifold the area of its unique maximal Cauchy surface. It is easy to see that A is constant on the orbits of the circle action. We show:

Theorem E. Let Σ be a closed oriented surface of genus $\geqslant 1$. The circle action on $\mathcal{MGH}(\Sigma)$ is Hamiltonian with respect to $\omega_{\mathbf{I}}$, and satisfies

$$R_{\theta}^* \mathbf{g} = \mathbf{g}$$
 $R_{\theta}^* \omega_{\mathbf{I}} = \omega_{\mathbf{I}}$ $R_{\theta}^* \omega_{\mathbf{I}}^{\mathbb{C}} = e^{-i\theta} \omega_{\mathbf{I}}^{\mathbb{C}}$.

When Σ has genus ≥ 2 , the function \mathcal{A} is a Hamiltonian function for the vector field that is the infinitesimal generator of the circle action.

We remark that, in terms of the (para-)complex structures $\mathbf{I}, \mathbf{J}, \mathbf{K}$, the pull-back relations of Theorem E read:

$$R_{\theta}^* \mathbf{I} = \mathbf{I}$$
 $R_{\theta}^* \mathbf{J} = \cos(\theta) \mathbf{J} + \sin(\theta) \mathbf{K}$ $R_{\theta}^* \mathbf{K} = -\sin(\theta) \mathbf{J} + \cos(\theta) \mathbf{K}$. (1.2)

In [BMS15], Bonsante, Mondello and Schlenker showed that the landslide flow is Hamiltonian with respect to the symplectic form $\pi_l^*\Omega_{WP} + \pi_r^*\Omega_{WP}$. As a consequence of Theorem C and the first part of Theorem E, we thus recover (by independent methods) their results and include it in a more general context.

The map $\mathcal{A}: \mathcal{MGH}(\Sigma) \to \mathbb{R}$ that encodes the area of the maximal Cauchy surface is also applied in the following context. Given a para-Kähler structure (\mathbf{g}, \mathbf{P}) on a manifold M, a para-Kähler potential is a smooth function $\rho: M \to \mathbb{R}$ such that $\omega_{\mathbf{P}} = (\tau/2)\overline{\partial}_{\mathbf{P}}\partial_{\mathbf{P}}\rho$. We then prove:

Theorem F. Let Σ be a closed oriented surface of genus $\geqslant 1$. Then the para-Kähler structures (\mathbf{g}, \mathbf{J}) and (\mathbf{g}, \mathbf{K}) admit a para-Kähler potential, which coincides up to a constant with a Hamiltonian function for the circle action.

One could alternatively have used the map \mathcal{C} to induce a circle action on $\mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$. However, the obtained action is the same as when using \mathcal{M} (i.e. the landslide flow), as a consequence of the observation that $\mathcal{M} = \mathcal{C} \circ R_{-\pi/2}$. By this relation, Theorem D immediately follows from Theorem C and Theorem E.

In fact we can define a one-parameter family of maps

$$C_{\theta}: \mathcal{MGH}(\Sigma) \to \mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$$
,

simply defined by $C_{\theta} = C \circ R_{\theta}$. An immediate consequence of our previous Theorems D and E is the following identity:

$$C_{\theta}^{*}(\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}, 4\Omega_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}^{\mathbb{B}}) = (\cos\theta\mathbf{K} - \sin\theta\mathbf{J}, \omega_{\mathbf{I}} - \tau(\cos(\theta)\omega_{\mathbf{J}} + \sin(\theta)\omega_{\mathbf{K}})) . \quad (1.3)$$

The maps \mathcal{C}_{θ} have the following interpretation purely in terms of harmonic maps and Teichmüller theory. From the theory of harmonic maps between hyperbolic surfaces ([Sam78, Wol89, Wol91b, Wol91a, Min92]), Teichmüller space admits a parameterization by the vector space of holomorphic quadratic differentials $H^0(\Sigma, \mathcal{K}_J^2)$ with respect to a fixed complex structure J on Σ . The construction goes as follows. To a holomorphic quadratic differential q, we associate the hyperbolic metric $h_{(J,q)}$ on Σ (unique up to isotopy) such that the (unique) harmonic map $(\Sigma, J) \to (\Sigma, h)$ isotopic to the identity has Hopf differential q. We now let J vary over $\mathcal{T}(\Sigma)$. Then the map

$$\mathcal{H}_{\theta} := \mathcal{C}_{\theta} \circ \mathcal{F}^{-1} : T^*\mathcal{T}(\Sigma) \to \mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$$

can be interpreted as follows:

$$\mathcal{H}_{\theta}(J,q) = (h_{(J-e^{i\theta}q)}, h_{(Je^{i\theta}q)})$$
.

There is a completely analogous construction in genus one, by replacing hyperbolic surfaces by flat tori. As a consequence of Equation (1.3), we obtain:

Theorem G. Let Σ be a closed oriented surface of genus ≥ 1 . Then

$$\operatorname{Im} \mathcal{H}_{\theta}^{*}(2\Omega_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}^{\mathbb{B}}) = -\operatorname{Re}(ie^{i\theta}\Omega_{T^{*}\mathcal{T}(\Sigma)}^{\mathbb{C}}).$$

We remark that the statement above is expressed purely in terms of Teichmüller theory, and is independent of Anti-de Sitter geometry.

1.6. The character variety

Let us now consider the character variety of the fundamental group $\pi_1(\Sigma)$ in the isometry group of AdS space. We have already observed that the isometry group is isomorphic to $\mathbb{P}SL(2,\mathbb{R}) \times \mathbb{P}SL(2,\mathbb{R})$; using the model of Hermitian matrices ([Dan13, Dan14]), it can be

described as the Lie group $\mathbb{P}SL(2,\mathbb{B})$, where as usual \mathbb{B} denotes the algebra of para-complex numbers. Using the (para-complex) Killing form, the character variety $\chi(\pi_1(S), \mathbb{P}SL(2,\mathbb{B}))$ is endowed with a para-complex symplectic form $\Omega_{Gol}^{\mathbb{B}}$, which is defined by adapting the work of Goldman ([Gol84]) to this context. It is para-complex with respect to the para-complex structure \mathcal{T} induced by multiplication by τ . It can be checked that, under the isomorphism $\mathbb{P}SL(2,\mathbb{R}) \times \mathbb{P}SL(2,\mathbb{R}) \cong \mathbb{P}SL(2,\mathbb{B})$, the para-complex structure \mathcal{T} corresponds to the para-complex structure \mathcal{P} for which the integral distributions of the 1 and -1 eigenspaces are the horizontal and vertical slices.

Hence if we denote by

$$\mathcal{H}ol: \mathcal{MGH}(\Sigma) \to \chi(\pi_1(S), \mathbb{P}SL(2, \mathbb{B}))$$

the map that associates to a MGHC AdS manifold its holonomy representation, we obtain the following corollary of Theorem C:

Corollary H. Let Σ be a closed oriented surface of genus ≥ 2 . Then

$$\mathcal{H}ol^*(\mathcal{T}, 4\Omega_{Gol}^{\mathbb{B}}) = (\mathbf{J}, \omega_{\mathbf{J}}^{\mathbb{B}}) .$$

We conclude the overview of our results by a concrete description of the para-complex symplectic structure $\omega_{\mathbf{J}}^{\mathbb{B}}$. In [Tam20], the third author introduced \mathbb{B} -valued Fenchel-Nielsen coordinates. Roughly speaking, these are defined as follows. Let $\rho = (\rho_+, \rho_-) : \pi_1(\Sigma) \to \mathbb{P}\mathrm{SL}(2,\mathbb{R}) \times \mathbb{P}\mathrm{SL}(2,\mathbb{R})$ be the holonomy representation of a MGHC AdS manifold. Since both ρ_- and ρ_+ are Fuchsian representations, $\rho_{\pm}(\alpha)$ are loxodromic elements for any nontrivial $\alpha \in \pi_1(\Sigma)$. As a consequence, we can associate to α a principal axis $\tilde{\alpha}$ in AdS space, which is the spacelike geodesic with endpoints in $\mathbb{RP}^1 \times \mathbb{RP}^1$ given by the pair of attracting and the pair of repelling fixed points of $\rho_{\pm}(\alpha)$. Then the Fenchel-Nielsen coordinates of ρ are $(\ell_{\rho}^{\mathbb{B},j}, \operatorname{tw}_{\rho}^{\mathbb{B},j})$ (for γ_j a pants decomposition of Σ), where $\ell_{\rho}^{\mathbb{B},j}$ are para-complex numbers whose real part corresponds to the translation length and imaginary part to the bending angle of $\rho(\gamma_j)$ on the principal axis $\tilde{\gamma}_j$; a similar interpretation can be given for the (paracomplex) twist coordinates $\operatorname{tw}_{\rho}^{\mathbb{B},j}$.

These coordinates are an analogue of the complex Fenchel-Nielsen coordinates on the space of quasi-Fuchsian manifolds, which are Darboux coordinates ([Wol83],[Pla01],[PP08]). Here we show that an analogous result holds for $\omega_{\mathbf{J}}^{\mathbb{B}}$, which we recall corresponds (up to a multiplicative constant) both to the para-complex sympletic form on $\mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$ and to the Goldman form $\Omega_{Gol}^{\mathbb{B}}$.

Theorem I. The \mathbb{B} -valued Fenchel-Nielsen coordinates are para-holomorphic for \mathbf{J} , and are Darboux coordinates with respect to the para-complex symplectic form $\omega_{\mathbf{J}}^{\mathbb{B}}$.

In other words, we can express the symplectic form $\omega_{\mathbf{J}}^{\mathbb{B}}$, which coincides up to a multiplicative constant with the para-complex Goldman form $\Omega_{Gol}^{\mathbb{B}}$, as

$$\omega_{\mathbf{J}}^{\mathbb{B}} = \frac{1}{4} \sum_{j=1}^{n} d\ell_{\rho}^{\mathbb{B}, j} \wedge d \operatorname{tw}_{\rho}^{\mathbb{B}, j}$$

where $\ell_{\rho}^{\mathbb{B},j}$ and $\operatorname{tw}_{\rho}^{\mathbb{B},j}$ are the \mathbb{B} -valued length and twist parameters on the curve γ_j in a pants decomposition of Σ (where $n = (3/2)|\chi(\Sigma)|$ is the number of such curves).

Finally, we give a formula for the value of the symplectic form $\omega_{\mathbf{J}}^{\mathbb{B}}$ along two twist deformations, generalizing Wolpert's cosine formula. For this purpose, given $\alpha, \beta \in \pi_1(\Sigma)$ two intersecting closed curves, the principal axes of $\rho(\alpha)$ and $\rho(\beta)$ admit a common orthogonal geodesic of timelike type. Then we define the \mathbb{B} -valued angle as the para-complex number whose imaginary part is the signed timelike distance between the two axes, and the real part is the angle between one principal axis and the parallel transport of the other, along the common orthogonal geodesic.

Theorem J. Let $\rho = (\rho_+, \rho_-) : \pi_1(\Sigma) \to \mathbb{P}SL(2, \mathbb{R}) \times \mathbb{P}SL(2, \mathbb{R})$ be the holonomy of a MGHC AdS manifold, and let α, β be non-trivial simple closed curves. Then

$$\omega_{\mathbf{J}}^{\mathbb{B}} \left(\frac{\partial}{\partial t w_{\rho}^{\mathbb{B}, \alpha}}, \frac{\partial}{\partial t w_{\rho}^{\mathbb{B}, \beta}} \right) = \frac{1}{4} \sum_{\rho \in \alpha \cap \beta} \cos \left(d^{\mathbb{B}} (\tilde{\alpha}_{\rho}, \tilde{\beta}_{\rho}) \right) ,$$

where $\tilde{\alpha}_{\rho}$ and $\tilde{\beta}_{\rho}$ are the principal axes of $\rho(\alpha)$ and $\rho(\beta)$ on AdS space, and $d^{\mathbb{B}}(\tilde{\alpha}_{\rho}, \tilde{\beta}_{\rho})$ is their \mathbb{B} -valued angle.

Here the cosine of a para-complex number is formally defined by the power series expansion.

1.7. Outline of techniques and proofs

We now provide an overview of the techniques that we apply in the proofs.

The "toy model": genus one

After reviewing the necessary preliminaries on the deformation space of Anti-de Sitter three-manifolds in Section 2, we complete in Section 3 the proof of our results from Theorem A to Theorem G (or their formal analogues, for Theorems B and C) in genus one. The reason is that the methods used to prove these results in genus one also provide the fundamental step on which the proofs in higher genus rely. Let us give a quick outline.

Denote by $\mathcal{J}(\mathbb{R}^2)$ the space of linear almost-complex structures on \mathbb{R}^2 compatible with the standard orientation, namely those endomorphisms J of \mathbb{R}^2 such that $J^2 = -1$ and that $\{v, Jv\}$ is a positive basis. This space is naturally identified to \mathbb{H}^2 , in such a way that the natural transitive $\mathrm{SL}(2,\mathbb{R})$ action by conjugation on $\mathcal{J}(\mathbb{R}^2)$ corresponds to the $\mathrm{SL}(2,\mathbb{R})$ action on the upper half-space model. We construct an explicit $\mathrm{SL}(2,\mathbb{R})$ -invariant para-hyperKähler structure on the cotangent bundle $T^*\mathcal{J}(\mathbb{R}^2)$, which is in turn identified to the space of pairs (J,σ) where σ is a symmetric bilinear form satisfying $\sigma(J\cdot,J\cdot)=-\sigma$, and give geometric interpretations to each object of the para-hyperKähler structure. For instance, the complex symplectic form $\omega_{\mathbf{I}}^{\mathbb{C}}$ coincides up to conjugation and a multiplicative factor with the complex symplectic form of $T^*\mathcal{J}(\mathbb{R}^2)$, and the almost-complex structure \mathbf{I} is compatible with its complex structure. Also, the restriction to the zero section is

identified, up to a factor, with the area form of \mathbb{H}^2 . All the theorems until Theorem G have an analogue that we prove in this context.

The construction of the para-hyperKähler structure on $\mathcal{MGH}(T^2)$ (Theorem A), as well as all the other statements in genus one, follow immediately by identifying (using the first and second fundamental form of the unique maximal surface) the deformation space $\mathcal{MGH}(T^2)$ with the complement of the zero section in $T^*\mathcal{J}(\mathbb{R}^2)$, in such a way that the mapping class group action corresponds to the action of $\mathrm{SL}(2,\mathbb{Z}) < \mathrm{SL}(2,\mathbb{R})$ (Lemma 3.11.)

Setup in higher genus

Let us now move on from genus one to genus ≥ 2 . The proof of Theorem A, and of the geometric interpretations from Theorem B to Theorem G, rely on the following construction, which is developed in Section 4. Fix an area form ρ on Σ . Then we consider the space of all pairs (J, σ) where (similarly to $T^*\mathcal{J}(\mathbb{R}^2)$ above) J is an almost-complex structure on Σ and σ a smooth symmetric bilinear form satisfying $\sigma(J\cdot,J\cdot)=-\sigma$. This infinite-dimensional space, that is denoted by $T^*\mathcal{J}(\Sigma)$, can be endowed formally with three symplectic structures $\omega_{\mathbf{I}}$, $\omega_{\mathbf{J}}$, $\omega_{\mathbf{K}}$. Indeed the tangent space $T_p\Sigma$ at every point has a parahyperKähler structure induced by choosing an area-preserving linear isomorphism between $T_p\Sigma$ and \mathbb{R}^2 , which induces an identification between $T^*\mathcal{J}(\mathbb{R}^2)$ and $T^*\mathcal{J}(T_p\Sigma)$. Since the para-hyperKähler structure on $T^*\mathcal{J}(\mathbb{R}^2)$ is $\mathrm{SL}(2,\mathbb{R})$ -invariant, the induced structures on $T^*\mathcal{J}(T_p\Sigma)$ do not depend on the chosen area-preserving linear isomorphism. Then one can formally integrate each symplectic form over Σ , evaluated on first-order deformations $(\dot{J}, \dot{\sigma})$.

To make this construction more formal, $T^*\mathcal{J}(\Sigma)$ is identified with the space of smooth sections of the bundle

$$P(T^*\mathcal{J}(\mathbb{R}^2)) = P \times T^*\mathcal{J}(\mathbb{R}^2) / \mathrm{SL}(2,\mathbb{R}),$$

where P is the $\mathrm{SL}(2,\mathbb{R})$ -principal bundle over Σ whose fiber over $p \in \Sigma$ is the space of linear isomorphisms between \mathbb{R}^2 and $T_p\Sigma$ that pull-back the area form ρ on $T_p\Sigma$ to the standard area form of \mathbb{R}^2 . Hence one can introduce the formal symplectic forms

$$(\omega_{\mathbf{X}})_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) := \int_{\Sigma} \hat{\omega}_{\mathbf{X}}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) \,\rho, \tag{1.4}$$

for $\mathbf{X} = \mathbf{I}, \mathbf{J}, \mathbf{K}$, where $\hat{\omega}_{\mathbf{X}}$ is the symplectic form induced on the vertical space at every point, and analogously for the formal pseudo-Riemannian metric

$$\mathbf{g}_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) := \int_{\Sigma} \hat{\mathbf{g}}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) \rho , \qquad (1.5)$$

on the infinite-dimensional space of sections. Similarly, one can define the endomorphisms \mathbf{I} , \mathbf{J} and \mathbf{K} by applying pointwise those that have been defined on $T^*\mathcal{J}(\mathbb{R}^2)$ under the identification as above. (See the discussion on symplectic reduction below for more details.)

Now, recalling that every MGHC AdS manifold contains a unique maximal Cauchy surface, the essential point is to determine the space of solutions to the Gauss-Codazzi equations for maximal surfaces in AdS space, as a subspace of sections (J, σ) . Given an almost-complex structure J, one can build a Riemannian metric $g_J = \rho(\cdot, J \cdot)$. However, in general this metric cannot be realised as the first fundamental form of a maximal surface in a MGHC AdS manifold, because the area form of g_J coincides with ρ , hence all metrics g_J have the same area, which is not the case for maximal surfaces. Instead, inspired by a similar "change of variables" used in the construction of the hyperKähler structure on a neighborhood of the Fuchsian locus in the space of quasi-Fuchsian manifolds ([Don03, Hod05, Tra19]), we define the metric $h = (1 + f(\|\sigma\|_{g_J}))g_J$, where $\|\sigma\|_{g_J}$ is the norm of the 2-tensor σ with respect to the metric g_J , and $f(t) = \sqrt{1+t^2}$. Then imposing the Gauss-Codazzi equations on I = h and $I = h^{-1}\sigma$, we determine a Symp₀(Σ, ρ)-invariant subspace of the space of smooth section (J, σ) , which we denote by $\mathcal{MS}_0(\Sigma, \rho)$, whose quotient $\mathcal{MS}_0(\Sigma, \rho)$ by the action of $\operatorname{Symp}_0(\Sigma, \rho)$ is identified to $\mathcal{MGH}(\Sigma)$ thanks to the existence and uniqueness result for maximal surfaces, together with a standard application of Moser's trick. We remark that, unlike the construction in the quasi-Fuchsian setting where the correct function to apply the change of variables is $f(t) = \sqrt{1-t^2}$, hence only defined for $t = \|\sigma\|_{g_J} < 1$, in the AdS setting this change of variables permits to recover all the maximal surfaces in MGHC AdS manifolds, and not only a neighborhood of the Fuchsian locus.

A distinguished complement to the orbit

In order to induce a para-hyperKähler structure on $\mathcal{MS}_0(\Sigma, \rho)$, by restriction of the symplectic forms $\omega_{\mathbf{I}}$, $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$ and of the metric \mathbf{g} , we need to construct a distribution of subspaces of the tangent space of $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$ which are isomorphic at every point (J, σ) to the tangent space to the quotient $\mathcal{MGH}(\Sigma)$, and invariant for the action of $\operatorname{Symp}_0(\Sigma, \rho)$. The construction of such invariant distribution, which is denoted by $V_{(J,\sigma)}$ and defined as the space of solutions of a system of partial differential equations, constitutes the main technical difficulty of Section 4. The defining equations can be formulated in several equivalent ways, as in the following technical statement, whose proof is done in Section 4.6 by overcoming a number of technical difficulties.

Proposition K. Given $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, and $(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$, the following conditions are equivalent:

i) the pair $(\dot{J}, \dot{\sigma})$ satisfies

$$\begin{cases} \operatorname{div}_{g}(f^{-1}g^{-1}\dot{\sigma}_{0}) = -f^{-1}\langle\nabla^{g}_{J\bullet}\sigma \mid \dot{J}\rangle, \\ \operatorname{div}_{g}\dot{J} = -f^{-2}\langle\nabla^{g}_{J\bullet}\sigma, \dot{\sigma}_{0}\rangle. \end{cases}$$
(V1)

ii) the endomorphisms $Q^{\pm} = Q^{\pm}(\dot{J}, \dot{\sigma}) := f^{-1}g^{-1}\dot{\sigma}_0 \pm \dot{J}$ satisfy

$$\begin{cases} \operatorname{div}_{g}(Q^{+}JJ_{l}) = -\langle \nabla^{g}_{J\bullet}\sigma \mid Q^{+}\rangle, \\ \operatorname{div}_{g}(Q^{-}JJ_{r}) = \langle \nabla^{g}_{J\bullet}\sigma \mid Q^{-}\rangle, \end{cases}$$
(V2)

where J_l and J_r denote the almost complex structures of the components of the Mess map \mathcal{M} ;

iii) the endomorphisms Q^{\pm} satisfy

$$\begin{cases} \operatorname{div}_{g} Q^{+} = -f^{-1} \langle \nabla_{J \bullet}^{g} \sigma \mid Q^{+} \rangle, \\ \operatorname{div}_{g} Q^{-} = +f^{-1} \langle \nabla_{J \bullet}^{g} \sigma \mid Q^{-} \rangle. \end{cases}$$
(V3)

Defining $V_{(J,\sigma)}$ as the space of solutions to some (hence all) the conditions V1-V3, the fundamental result is the following, whose proof again contains a number of technical difficulties and is done in Section 4.5.

Theorem L. For every pair (J, σ) lying in $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, the vector space $V_{(J,\sigma)}$ is contained inside $T_{(J,\sigma)}\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, it is invariant by the action of \mathbf{I} , \mathbf{J} and \mathbf{K} , and it defines a Symp (Σ, ρ) -invariant distribution $\mathcal{V} = \{V_{(J,\sigma)}\}_{(J,\sigma)}$ on $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$. Moreover, the natural projection $\pi \colon \widetilde{\mathcal{MS}}_0(\Sigma, \rho) \to \mathcal{MS}_0(\Sigma, \rho)$ induces a linear isomorphism $d\pi_{(J,\sigma)} \colon V_{(J,\sigma)} \to T_{[J,\sigma]}\mathcal{MS}_0(\Sigma, \rho)$.

Once these steps are achieved, the proofs of all the results from Theorem A to Theorem G mostly follow from their analogues that are showed in Section 3, by applying the same computations pointwise, and recognizing the geometric interpretations in terms of the geometry of MGHC AdS manifolds either in a similar fashion to what has been done in the case of the torus (for instance for Theorems B, E and Theorem F), or by recalling the known constructions of Mess homeomorphism and constant curvature surfaces explained in Section 2 (Theorems C and D). These proofs are provided in the first part of Section 5. The second part then contains the proofs of Theorem I and Theorem J, which mainly rely on the isomorphism $\mathbb{P}SL(2,\mathbb{B}) \cong \mathbb{P}SL(2,\mathbb{R}) \times \mathbb{P}SL(2,\mathbb{R})$ for the isometry group of Anti-de Sitter space, and some geometric constructions in the $\mathbb{P}SL(2,\mathbb{R})$ model.

Relation with symplectic reduction

Although the proofs of the results stated above are essentially self-contained and completed entirely in Sections 3, 4 and 5, in Section 6 we include a discussion with the aims, on the one hand, of highlighting the relations with symplectic reduction, which is also the motivation that led us to the definition of the distribution $V_{(J,\sigma)}$ as in Proposition K and Theorem L; on the other hand, of explaining the additional difficulties that do not permit to apply directly the strategy of symplectic reduction to obtain our results.

The starting point is a general theorem of Donaldson ([Don03, Theorem 9], see Theorem 6.2 for the statement with minimal hypothesis that we apply in our setting). In short,

it turns out that the $SL(2,\mathbb{R})$ -action on $\mathcal{T}^*\mathcal{J}(\mathbb{R}^2)$ is Hamiltonian with respect to any of the three symplectic forms constituting the para-hyperKähler structure on $\mathcal{T}^*\mathcal{J}(\mathbb{R}^2)$, with moment maps that we denote by $\eta_{\rm I}$, $\eta_{\rm J}$ and $\eta_{\rm K}$. Donaldson's theorem gives a formula to compute a corresponding map (that we call $\mu_{\mathbf{I}}$, $\mu_{\mathbf{J}}$ and $\mu_{\mathbf{K}}$) for the action on $T^*\mathcal{J}(\Sigma)$ of the group $\operatorname{Ham}(\Sigma, \rho)$ of Hamiltonian symplectomorphisms of (Σ, ρ) . It turns out that $\mu_{\rm J}$ and $\mu_{\mathbf{K}}$ are moment maps for the action of $\operatorname{Ham}(\Sigma, \rho)$, whereas $\mu_{\mathbf{I}}$ needs to be modified by an adding a scalar multiple of ρ . We denote this new map by $\tilde{\mu}_{\mathbf{I}}$. Moreover, although $\operatorname{Ham}(\Sigma, \rho)$ is a proper normal subgroup of $\operatorname{Symp}_0(\Sigma, \rho)$, $\mu_{\mathbf{J}}$ and $\mu_{\mathbf{K}}$ can actually be promoted to moment maps $\tilde{\mu}_{\mathbf{J}}$ and $\tilde{\mu}_{\mathbf{K}}$ for the action of $\operatorname{Symp}_{0}(\Sigma, \rho)$; $\tilde{\mu}_{\mathbf{I}}$ cannot, but it still satisfies some additional properties that make it "almost" a moment map for $\operatorname{Symp}_0(\Sigma, \rho)$. We compute explicit formulas for these three maps, and show that the kernel of $\tilde{\mu}_{\rm J} + i\tilde{\mu}_{\rm K}$ consists precisely of the pairs (J, σ) such that σ is the real part of a holomorphic quadratic differential on (Σ, J) . The intersection with the kernel of the third map $\tilde{\mu}_{\mathbf{I}}$ is then precisely the space $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$. Hence one is tempted to apply the symplectic reduction in order to induce a pseudo-Riemannian metric and three symplectic stuctures, in the quotient $\tilde{\mu}_{\mathbf{I}}^{-1}(0) \cap \tilde{\mu}_{\mathbf{J}}^{-1}(0) \cap \tilde{\mu}_{\mathbf{K}}^{-1}(0) / \operatorname{Symp}_{0}(\Sigma, \rho).$

However, at this point the usual construction by which the quotient inherits a hyper-Kähler structure fails because of the fact that our metric on each fiber is not positive definite. Hence we do not have a natural Hilbert space structure on the tangent space to the space of sections $T^*\mathcal{J}(\Sigma)$. Concretely, this means that we cannot take the orthogonal complement to the tangent space to the $\operatorname{Symp}_0(\Sigma, \rho)$ -orbit as a distribution which is invariant by the actions of \mathbf{I} , \mathbf{J} and \mathbf{K} and isomorphic to the tangent space to the quotient. Nevertheless, inspired by the properties satisfied in the hyperKähler setting, we prove the following characterization of $V_{(J,\sigma)}$, which is the main result of Section 6:

Theorem M. For every $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, $V_{(J,\sigma)}$ is the largest subspace of $T_{(J,\sigma)}\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$ that is:

- invariant under I, J and K;
- **g**-orthogonal to $T_{(J,\sigma)}(\operatorname{Symp}_0(\Sigma,\rho)\cdot(J,\sigma))$

As said before, Theorem M is not applied in the proof of any of the previous results; nevertheless, it serves as a motivation for Proposition K and Theorem L, namely the two technical tools which play an essential role in passing from the para-hyperKähler structure in genus one (or more precisely, on $T^*\mathcal{J}(\mathbb{R}^2)$) to that for higher genus surfaces.

Acknowledgments

We would like to thank the participants of the reading group on hyperKähler structures on quasi-Fuchsian space held at the University of Luxembourg in the Fall 2018, for many valuable discussions. The second author is grateful to Andy Sanders for interesting discussions.

The third author acknowledges support from the U.S. National Science Foundation under grant NSF-DMS:2005501.

2. Preliminaries on Anti-de Sitter Geometry

In this preliminary section, we introduce the necessary notions concerning maximal globally hyperbolic Cauchy compact Anti-de Sitter three-manifolds (in short, MGHC AdS manifold) and their deformation space.

2.1. Maximal surfaces

The starting point of our construction comes from the role of maximal surfaces in MGHC AdS manifolds. Recall that a *maximal surface* in a Lorentzian three-manifold is a spacelike surface (i.e. its first fundamental form is a Riemannian metric) whose mean curvature vanishes identically. Then we have the following existence and uniqueness result:

Theorem 2.1 ([BBZ07, BS10]). Any MGHC AdS three-manifold admits a unique maximal Cauchy surface.

This result permits to obtain a parameterization of $\mathcal{MGH}(\Sigma)$ by means of embedding data of maximal surfaces. Recall that the embedding data of a spacelike surface in a Lorentzian manifold consists of the pair (h, B), where h is the first fundamental form and B the shape operator, and these satisfy the Gauss-Codazzi equations

$$\begin{cases} K_h = -1 - \det B, \\ \mathrm{d}^{\nabla^h} B = 0, \end{cases}$$
 (GC)

where the exterior derivative $\mathrm{d}^{\nabla^h}B$ is the $T\Sigma$ -valued 2-form

$$(\operatorname{d}^{\nabla^h}B)(X,Y) = (\nabla^h_XB)Y - (\nabla^h_YB)X \ ,$$

for X, Y tangent vector fields on Σ , ∇^h being the Levi-Civita connection of h and K_h its curvature. By definition, the surface is maximal if and only if B is traceless. Conversely, every pair (h, B) satisfying (GC), with h a Riemannian metric and B a traceless h-self-adjoint tensor, represents the embedding data of a maximal surface in a MGHC AdS manifold diffeomorphic to $\Sigma \times \mathbb{R}$, whose metric has the following explicit expression in a tubular neighbourhood of the surface (namely for $t \in (-\epsilon, \epsilon)$):

$$-dt^2 + h((\cos(t)\mathbb{1} + \sin(t)B)\cdot, (\cos(t)\mathbb{1} + \sin(t)B)\cdot) \ .$$

See [BS20, Lemma 6.2.2, Proposition 6.2.3]. Moreover the above correspondence is natural by the actions of Diff(Σ). Motivated by these observations, let us introduce the space of embedding data of maximal surfaces:

$$\mathcal{MS}(\Sigma) := \left\{ (h,B) \middle| \begin{array}{l} h \text{ is a Riemannian metric} \\ B \text{ is traceless and } h\text{-self-adjoint} \\ \text{equations (GC) are satisfied} \end{array} \right\} / \text{Diff}_0(\Sigma) \ ,$$

where $\mathrm{Diff}_0(\Sigma)$ denotes the group of diffeomorphisms isotopic to the identity. In summary we have:

Proposition 2.2. Let Σ be a closed surface of genus $g \ge 1$. There is a $MCG(\Sigma)$ -invariant homeomorphism between $\mathcal{MGH}(\Sigma)$ and $\mathcal{MS}(\Sigma)$, given by the embedding data of the unique maximal Cauchy surface.

In the remainder of the paper, we will often implicitly identify $\mathcal{MGH}(\Sigma)$ with $\mathcal{MS}(\Sigma)$.

2.2. Cotangent bundle of Teichmüller space

Maximal surfaces also permit to obtain a parameterization of $\mathcal{MGH}(\Sigma)$ by means of the cotangent bundle of Teichmüller space. Recall that the Teichmüller space of the surface Σ is defined as:

$$\mathcal{T}^{\mathfrak{c}}(\Sigma) := \{ J \in \Gamma(\operatorname{End}(T\Sigma)) \mid J^2 = -1, (v, J(v)) \text{ is an oriented frame} \}$$

We use the notation $\mathcal{T}^{\mathfrak{c}}$ to highlight that this is the Teichmüller space defined in terms of (almost-)complex structures J, to distinguish with the other incarnations of Teichmüller space (see Section 2.3 below).

To explain this parameterization, let us first provide several equivalent descriptions of the Codazzi equation $d^{\nabla^h}B = 0$, which we summarize in the following statement:

Lemma 2.3. Let h be a Riemannian metric on Σ , and denote by J the almost-complex structure induced by h. Assume a (1,1) tensor B and a (2,0) tensor σ are related by $\sigma = hB$ (i.e. $\sigma(\cdot, \cdot) = h(\cdot, B \cdot)$). Then B is h-self-adjoint and traceless if and only if σ is the real part of a quadratic differential, which can be expressed as $q = \sigma - i\sigma(\cdot, J \cdot)$. If this holds, then the following conditions are equivalent:

- i) B is h-Codazzi, i.e. $d^{\nabla^h}B = 0$;
- ii) σ is the real part of a holomorphic quadratic differential on (Σ, J) ;
- iii) σ is h-divergence-free;
- iv) for every tangent vector field X on Σ we have $\nabla^h_{JX}\sigma = (\nabla^h_X\sigma)(\cdot,J\cdot)$.

A standard reference for these equivalences is [Tro92]; see also [Hop51, Tau04, KS07].

Remark 2.4. Since the condition ii) in Lemma 2.3 only depends on J, and not on the metric h, it follows that if conditions iii) or iv) hold for some metric h compatible with J, then they hold for any other metric conformal to h.

Remark 2.5. We will repeatedly use the following fact. Suppose q is a holomorphic quadratic differential and $B = h^{-1} \operatorname{Re}(q)$ is the corresponding traceless, h-self-adjoint, h-Codazzi tensor. Writing $q = \sigma - i\sigma(\cdot, J\cdot)$ and multiplying by $e^{i\theta}$, one checks immediately that

$$\operatorname{Re}(e^{i\theta}q) = \cos\theta \ \sigma + \sin\theta \ \sigma(\cdot, J\cdot) \ .$$

Hence the traceless, h-self-adjoint, h-Codazzi tensor $h^{-1} \operatorname{Re}(e^{i\theta}q)$ corresponding to $e^{i\theta}q$ is precisely $(\cos(\theta)\mathbb{1} - \sin(\theta)J)B$.

Based on Lemma 2.3, Krasnov and Schlenker established the following result:

Theorem 2.6 ([KS07, Lemma 3.6, Theorem 3.8]). Let Σ be a closed oriented surface of genus ≥ 2 . Given a complex structure J on Σ and a holomorphic quadratic differential q on (Σ, J) , there exists a unique Riemannian metric h compatible with J such that the pair $(h, B = h^{-1}\sigma)$ satisfies (GC), where $\sigma = \text{Re}(q)$.

Again, the map that associates to a pair (h, B) satisfying the Gauss-Codazzi equations the pair (J, q), where J is the complex structure of h and $\sigma = hB = \text{Re}(q)$, is natural with respect to the action of Diff (Σ) .

We remark that Theorem 2.6 is proved in [KS07] for a closed surface of genus ≥ 2 . However, the case of genus one holds true, and can be proved directly, provided $\sigma \neq 0$.

Proposition 2.7. Given a complex structure J on T^2 and a non-zero holomorphic quadratic differential q on (T^2, J) , there exists a unique Riemannian metric h compatible with J such that the pair $(h, B = h^{-1}\sigma)$ satisfies (GC), where $\sigma = \text{Re}(q)$.

Proof. It turns out that, given any MGHC AdS manifold diffeomorphic to $T^2 \times \mathbb{R}$, the (unique) maximal Cauchy surface is intrinsically flat, and the MGHC metric can be written (now *globally*) from the maximal surface as

$$(1+\sin(2t))dx^2 + (1-\sin(2t))dy^2 - dt^2, \qquad (2.1)$$

where:

- $t \in (-\pi/4, \pi/4)$ represents the timelike distance from the maximal surface;
- x, y are global flat coordinates for the first fundamental form $h = dx^2 + dy^2$ of the universal cover of the maximal surface.

See Section 5 of [BS20], and in particular Lemma 5.2.4, after a simple change of coordinates (translate the vertical coordinate by $\pi/4$, and then perform a change of variables on x, y). Moreover, since the second fundamental form of the maximal surface $\{t = 0\}$ equals one half the derivative at t = 0 of the metric on the level sets of t, we find that σ is expressed as $dx^2 - dy^2$, hence (in the complex coordinate z = x + iy) $q = dz^2$ is the holomorphic quadratic differential whose real part equals σ , as in Lemma 2.3.

Inspired by these observations, let us now reconstruct a MGHC AdS manifold from a pair (J, σ) on the torus. Let us realize the complex structure J as that induced by a flat

metric on T^2 , obtained as the quotient of the complex plane by a lattice Λ . In other words, we find a biholomorphism between (T^2, J) and \mathbb{C}/Λ endowed with the complex structure induced by the standard complex structure of \mathbb{C} . We stress that for the moment we do not assume any normalization on the lattice Λ , which is therefore determined up to automorphisms of \mathbb{C} .

Let us call x, y (where z = x+iy) the flat coordinates on \mathbb{C} , which induce flat coordinates on the quotient \mathbb{C}/Λ . Then the holomorphic quadratic differential q has the expression cdz^2 , for $c \neq 0$ a complex number. Multiplying the coordinate z by a square root of 1/c, we can also assume that $q = dz^2$. This has the effect of rescaling and rotating the lattice Λ , thus obtaining a new lattice Λ' inducing the same J on T^2 . Hence the expression (2.1) gives a MGHC AdS metric on $(\mathbb{C}/\Lambda') \times (-\pi/4, \pi/4)$, for which $\{t = 0\}$ is a maximal surface whose corresponding complex structure is J and quadratic differential is q.

Recalling that the cotangent bundle of Teichmüller space is identified with the bundle of holomorphic quadratic differentials, as consequence of Theorem 2.6 (for higher genus) and the above discussion (for genus one), one obtains:

Theorem 2.8. Let Σ be a closed oriented surface of genus ≥ 2 . The map sending a pair (h, B) to the pair (J, q), where J is the complex structure induced by h and Re(q) = hB, induces a $MCG(\Sigma)$ -invariant homeomorphism

$$\mathcal{F} \colon \mathcal{MGH}(\Sigma) \cong \mathcal{MS}(\Sigma) \to T^*\mathcal{T}^{\mathfrak{c}}(\Sigma)$$
.

If $\Sigma = T^2$, the same map gives a $MCG(T^2)$ -invariant homeomorphism between $\mathcal{MGH}(T^2)$ and the complement of the zero section in $T^*\mathcal{T}^{\mathfrak{c}}(T^2)$.

2.3. Mess' parameterization

Let us now focus on genus ≥ 2 and move on to another parameterization of $\mathcal{MS}(\Sigma)$, which has been essentially introduced by Mess. For this purpose, let us introduce the hyperbolic model of Teichmüller space, namely:

$$\mathcal{T}^{\mathfrak{h}}(\Sigma) := \{h \,|\, h \text{ is a hyperbolic metric on } \Sigma\}/\mathrm{Diff}_0(\Sigma)$$
 .

The "hyperbolic" Teichmüller space is $MCG(\Sigma)$ -invariantly homeomorphic to $\mathcal{T}^{\mathfrak{c}}(\Sigma)$ by the uniformization theorem. Also, the holonomy representation provides a $MCG(\Sigma)$ -invariant homeomorphism to the space of Fuchsian representations, which we denote by:

$$\mathcal{T}^{\mathfrak{rep}}(\Sigma) := \{\rho: \pi_1(\Sigma) \to \mathrm{Isom}(\mathbb{H}^2) \text{ discrete and faithful representations}\}/\mathrm{Isom}(\mathbb{H}^2) \ ,$$

where $\text{Isom}(\mathbb{H}^2)$ acts by conjugation ([Gol80]). Now, given a pair (h, B) satisfying (GC), one can construct two hyperbolic metrics by the formula

$$(h, B) \mapsto (h_l := h((\mathbb{1} - JB) \cdot, (\mathbb{1} - JB) \cdot), h_r := h((\mathbb{1} + JB) \cdot, (\mathbb{1} + JB) \cdot))$$
 (2.2)

(Here h_l and h_r stand for "left" and "right", and in fact these metrics can be interpreted as the pull-backs via the so-called *left and right Gauss maps*, see [KS07, §3], [BS20, §6],

[Bar18, §6].) It can be proved (see for instance the indicated references) that the metrics h_l and h_r are hyperbolic and that, interpreting the isometry group of Anti-de Sitter space as $\mathbb{P}SL(2,\mathbb{R}) \times \mathbb{P}SL(2,\mathbb{R})$, the map (2.2) gives the left and right components of the holonomy map

$$\text{hol}: \pi_1(\Sigma) \to \mathbb{P}\text{SL}(2,\mathbb{R}) \times \mathbb{P}\text{SL}(2,\mathbb{R})$$

of the AdS MGHC manifold determined by the embedding data (h, B) of the maximal Cauchy surface, under the isomorphism between $\mathcal{T}^{\mathfrak{h}}(\Sigma)$ and $\mathcal{T}^{\mathfrak{rep}}(\Sigma)$.

In summary, we state the following theorem:

Theorem 2.9 ([Mes07],[KS07, Theorem 3.17],[BS20, Theorem 5.5.4]). Let Σ be a closed oriented surface of genus ≥ 2 . The map sending a pair (h, B) to the pair of hyperbolic metrics (h_l, h_r) in (2.2) induces a $MCG(\Sigma)$ -invariant homeomorphism

$$\mathcal{M}^{\mathfrak{h}} \colon \mathcal{MS}(\Sigma) \to \mathcal{T}^{\mathfrak{h}}(\Sigma) \times \mathcal{T}^{\mathfrak{h}}(\Sigma)$$
.

Under the natural homeomorphism $\mathcal{T}^{\mathfrak{h}}(\Sigma) \cong \mathcal{T}^{\mathfrak{rep}}(\Sigma)$, such map coincides with the map

$$\mathcal{M}^{\mathfrak{rep}} \colon \mathcal{MGH}(\Sigma) \to \mathcal{T}^{\mathfrak{rep}}(\Sigma) \times \mathcal{T}^{\mathfrak{rep}}(\Sigma)$$
.

sending a MGHC AdS manifold to the conjugacy class of its holonomy representation.

It will be useful for future computations to express this map in the conformal model of Teichmüller space. This follows easily by observing that the complex structure of a metric of the form $h(A \cdot, A \cdot)$ equals the A-conjugate of the complex structure of h.

Lemma 2.10. Let Σ be a closed oriented surface of genus ≥ 2 . The $MCG(\Sigma)$ -invariant homeomorphism

$$\mathcal{M}^{\mathfrak{c}} \colon \mathcal{MS}(\Sigma) \to \mathcal{T}^{\mathfrak{c}}(\Sigma) \times \mathcal{T}^{\mathfrak{c}}(\Sigma)$$
.

is induced by the map

$$(h,B) \mapsto (J_l := (\mathbb{1} - JB)^{-1} J(\mathbb{1} - JB), J_r := (\mathbb{1} + JB)^{-1} J(\mathbb{1} + JB))$$
,

where J is the complex structure defined by the metric h.

The map \mathcal{M} has also an interpretation in terms of harmonic maps. Indeed, it turns out that the identity map $id:(\Sigma,J)\to(\Sigma,h_l)$ is harmonic and its Hopf differential is the holomorphic quadratic differential iq, where Re(q)=hB. Indeed, from the expression (2.2), we see that

$$h_l = (1 - \det(B)) h - 2h(\cdot, JB \cdot)$$

where we used that JB is h-self-adjoint and traceless, hence by the Cayley-Hamilton theorem $(JB)^2 = -\det(JB)\mathbb{1} = -\det(B)\mathbb{1}$. Hence the (2,0)-part of h_l with respect to the complex structure J is the quadratic differential iq, where $\sigma = \operatorname{Re}(q) = hB$ (Remark 2.5). Moreover it is holomorphic by Lemma 2.3. This implies that $id: (\Sigma, J) \to (\Sigma, h_l)$ is harmonic, since it is a diffeomorphism and its Hopf differential is holomorphic (see [Sam78, §9]). Analogously $id: (\Sigma, J) \to (\Sigma, h_r)$ is harmonic and has Hopf differential -iq. We summarize this in the following lemma.

Lemma 2.11. Let Σ be a closed oriented surface of genus ≥ 2 . Then

$$\mathcal{M}^{\mathfrak{h}} \circ \mathcal{F}^{-1}(J,q) = (h_{(J,iq)}, h_{(J,-iq)})$$
,

where h(J,q) denotes the unique hyperbolic metric on Σ such that $id:(\Sigma,J)\to(\Sigma,h_{(J,q)})$ is harmonic.

2.4. Constant curvature surfaces and circle action

Another parameterization of $\mathcal{MS}(\Sigma)$ by the product of two copies of Teichmüller space is constructed as follows. Given a maximal surface in a MGHC AdS manifold of genus ≥ 2 , a standard computation shows that the two surfaces at distance $\pi/4$ from the maximal surface have intrinsic curvature -2 (see [CT19], [BS20, Theorem 7.1.4]). Hence, multiplying the first fundamental forms of these surfaces by a factor 2, so that they become of intrinsic curvature -1, one finds two hyperbolic metrics on Σ , which are expressed by:

$$(h, B) \mapsto (h_{-} := h((\mathbb{1} - B)\cdot, (\mathbb{1} - B)\cdot), h_{+} := h((\mathbb{1} + B)\cdot, (\mathbb{1} + B)\cdot))$$
 (2.3)

By arguments similar to those leading to Theorem 2.9, one can prove that this produces again a natural homeomorphism, namely:

Theorem 2.12 ([KS07, Theorem 3.21]). Let Σ be a closed oriented surface of genus ≥ 2 . The map sending a pair (h, B) to the pair of hyperbolic metrics (h_-, h_+) in (2.3) induces a $MCG(\Sigma)$ -invariant homeomorphism

$$\mathcal{C}^{\mathfrak{h}} \colon \mathcal{MS}(\Sigma) \to \mathcal{T}^{\mathfrak{h}}(\Sigma) \times \mathcal{T}^{\mathfrak{h}}(\Sigma)$$
.

The hyperbolic metrics (h_-, h_+) are obtained as the first fundamental forms of the $\pi/4$ -equidistant surfaces from the maximal surface, after rescaling by a suitable constant.

As in Lemma 2.10, we can express this map in terms of the conformal model of Teichmüller space.

Lemma 2.13. Let Σ be a closed oriented surface of genus $g \ge 2$. The $MCG(\Sigma)$ -invariant homeomorphism

$$C^{\mathfrak{c}} : \mathcal{MS}(\Sigma) \to \mathcal{T}^{\mathfrak{c}}(\Sigma) \times \mathcal{T}^{\mathfrak{c}}(\Sigma)$$
.

is induced by the map

$$(h,B) \mapsto (J_l := (\mathbb{1} - B)^{-1} J(\mathbb{1} - B), J_r := (\mathbb{1} + B)^{-1} J(\mathbb{1} + B))$$
,

where J is the complex structure defined by the metric h.

From Theorem 2.8, we see that $\mathcal{MS}(\Sigma)$ is endowed with a circle action, which acts on a pair (J,q), where J is a complex structure and q a holomorphic quadratic differential, by multiplying the holomorphic quadratic differential by $e^{i\theta}$. By Remark 2.5, this S^1 action on $\mathcal{MS}(\Sigma)$ has the following expression in terms of the pairs (h, B),

$$e^{i\theta}\cdot(h,B)=(h,((\cos\theta)\mathbb{1}-(\sin\theta)J)B)\ ,$$

and it can be checked directly that the new pair obtained in this way still gives the embedding data of a maximal surface in a MGHC AdS manifold. Denoting by $R_{\theta}: \mathcal{MS}(\Sigma) \to \mathcal{MS}(\Sigma)$ the S^1 action, it follows immediately from (2.2) and (2.3) that:

$$\mathcal{M}^{\mathfrak{h}} = \mathcal{C}^{\mathfrak{h}} \circ R_{-\pi/2} \ . \tag{2.4}$$

Observe moreover that R_{π} has the effect of switching the left and right components under the maps $\mathcal{C}^{\mathfrak{h}}$ and $\mathcal{M}^{\mathfrak{h}}$.

By conjugating the circle action by the map $\mathcal{M}^{\mathfrak{h}}$ (or $\mathcal{C}^{\mathfrak{h}}$), one gets an induced circle action on $\mathcal{T}^{\mathfrak{h}}(\Sigma) \times \mathcal{T}^{\mathfrak{h}}(\Sigma)$, which has been defined in [BMS13] as the *landslide flow*. Motivated by this construction, we will consider the map

$$\mathcal{C}^{\mathfrak{h}}_{\theta} = \mathcal{C}^{\mathfrak{h}} \circ R_{\theta} : \mathcal{MS}(\Sigma) \to \mathcal{T}^{\mathfrak{h}}(\Sigma) \times \mathcal{T}^{\mathfrak{h}}(\Sigma) .$$

Clearly $C_0^{\mathfrak{h}} = C^{\mathfrak{h}}$ and $C_{-\pi/2}^{\mathfrak{h}} = \mathcal{M}^{\mathfrak{h}}$. Let us introduce $\mathcal{H}_{\theta} = C_{\theta}^{\mathfrak{h}} \circ \mathcal{F}^{-1}$, which is the composed map from $T^*\mathcal{T}(\Sigma) \to \mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$ used in Theorem G. Then we immediately obtain:

Lemma 2.14. Let Σ be a closed oriented surface of genus ≥ 2 . Then

$$\mathcal{H}_{\theta}(J,q) = (h_{(J,-e^{i\theta}q)}, h_{(J,e^{i\theta}q)}) ,$$

where h(J,q) denotes the unique hyperbolic metric on Σ such that $id:(\Sigma,J)\to(\Sigma,h_{(J,q)})$ is harmonic.

2.5. An equivalent model for $\mathcal{MS}(\Sigma)$

We introduce here a fundamental "change of variables", which permits us to adopt a simpler model to study $\mathcal{MS}(\Sigma)$. The basic idea is to replace the metric h (which is the first fundamental form of the maximal surface) by a suitable conformal metric g, so that the area of g is independent of the point of $\mathcal{MS}(\Sigma)$. By a standard argument in symplectic geometry, we will be allowed to assume that the area form of g is a fixed symplectic form ρ on Σ .

To make this concrete, let us introduce the function $f(t) = \sqrt{1+t^2}$, which we will always apply to $t = \|\sigma\|_g^2$, for g a Riemannian metric conformal to h. Recall that in general, if σ is a symmetric (2,0) tensor on Σ , $\|\sigma\|_g^2$ denotes the squared norm of the operator $A = g^{-1}\sigma$, namely one half the trace of A^TA , where A^T is the g-adjoint operator of A. When σ is the real part of a quadratic differential, A is g-self-adjoint and traceless by Lemma 2.3, hence $\|\sigma\|_g^2 = -\det A$.

Let Σ be a closed orientable surface of genus ≥ 2 . We now introduce the following space:

$$\mathcal{MS}_0(\Sigma) := \left\{ (g, \sigma) \middle| \begin{array}{l} g \text{ is a Riemannian metric on } \Sigma \\ \sigma \text{ is the real part of a } g\text{-quadratic differential} \\ (h = (1 + f(\|\sigma\|_g)) \, g, B = h^{-1}\sigma) \text{ satisfy (GC)} \end{array} \right\} / \mathrm{Diff}_0(\Sigma)$$

The map sending (g, σ) to (h, B), where $h = (1 + f(\|\sigma\|_g)) g$ and $B = h^{-1}\sigma$ tautologically induces a $MCG(\Sigma)$ -equivariant map from $\mathcal{MS}_0(\Sigma)$ to $\mathcal{MS}(\Sigma)$. By [KS07], the principal curvatures of a maximal Cauchy surface of genus ≥ 2 in a MGHC AdS manifold are strictly less than one in absolute value, which implies that $\det B \in (-1, 0]$. (On the other hand, if $\Sigma = T^2$, then $\det B \equiv -1$ since the maximal surface is flat.) Hence the following lemma shows that the map induced by

$$(h,B) \mapsto \left(\frac{1+\det B}{2}h, hB\right)$$

is an inverse, and $\mathcal{MS}(\Sigma)$ and $\mathcal{MS}_0(\Sigma)$ are homeomorphic, under the assumption that Σ has genus ≥ 2 .

Lemma 2.15. Given a metric g and a (2,0)-tensor σ on Σ , let $h = (1 + f(\|\sigma\|_g)) g$ and $B = h^{-1}\sigma$. Then

$$\det B = -\frac{\|\sigma\|_g^2}{(1 + f(\|\sigma\|_g))^2} \qquad and \qquad 1 + \det B = \frac{2}{1 + f(\|\sigma\|_g)} \ . \tag{2.5}$$

Proof. The first identity comes from observing that $\|\sigma\|_h^2 = -\det B$ as remarked above, and that if $h = e^u g$ then $\|\sigma\|_h^2 = e^{-2u} \|\sigma\|_g^2$. The second identity is an easy algebraic manipulation using the definition of f.

An immediate consequence is the following:

Lemma 2.16. Let $(g, \sigma) \in \mathcal{MS}_0(\Sigma)$. Then the area of g equals $-\pi \chi(\Sigma)$.

Proof. Let $h = (1 + f(\|\sigma\|_g)) g$ as usual. Using (2.5), the area forms of g and h satisfy the identity:

$$dA_g = \frac{1}{1 + f(\|\sigma\|_g)} dA_h = \frac{1 + \det B}{2} dA_h \ .$$

Since the pair (h, B) satisfy (GC) by hypothesis, $1 + \det B = -K_h$, hence

$$\int_{\Sigma} dA_g = -\frac{1}{2} \int_{\Sigma} K_h dA_h = -\pi \chi(\Sigma)$$

by Gauss-Bonnet.

From now on, we will fix an area form ρ of total area $-\pi \chi(\Sigma)$. Given an almost-complex structure J on Σ , we define the Riemannian metric

$$g_J(\cdot,\cdot) := \rho(\cdot,J\cdot)$$
.

Clearly $dA_{g_J} = \rho$. We introduce the space $\mathcal{MS}_0(\Sigma, \rho)$ of pairs (J, σ) such that (g_J, σ) satisfy the conditions in the definition of $\mathcal{MS}_0(\Sigma)$, namely the pair $(h = (1+f(\|\sigma\|_{g_J})) g_J, B = h^{-1}\sigma)$ satisfy (GC), and we quotient by the action of $\operatorname{Symp}_0(\Sigma, \rho)$ (i.e. the identity component in the group of diffeomorphisms of Σ that preserve ρ).

It remains to show that the map $(J, \sigma) \mapsto (h, B)$ induces a homeomorphism between $\mathcal{MS}_0(\Sigma, \rho)$ and $\mathcal{MS}(\Sigma)$. This follows from standard arguments relying on the Moser trick, and we only give a sketch here, see for instance [Hod05, §3.2.3] for more details. Moser's stability theorem asserts that given a smooth family ω_t of cohomologous symplectic forms on a closed manifold, there exists a family of diffeomorphisms ϕ_t such that $\phi_t^*\omega_t = \omega_0$. On a surface Σ , given two area forms ρ and ρ' of the same total area, one can apply Moser's theorem to the family $\rho_t = (1-t)\rho + t\rho'$ and deduce that there exists $\phi \in \text{Diff}_0(\Sigma)$ such that $\phi^*\rho' = \rho$. This implies that any (g, σ) as in the definition of $\mathcal{MS}(\Sigma)$ has a representative in its $\text{Diff}_0(\Sigma)$ -orbit whose area form is ρ , i.e. a representative of the form (g_J, σ) . Moreover, if ψ_t is a family of diffeomorphisms such that $\psi_0 = id$ and $\psi_1^*\rho = \rho$, by applying again Moser's theorem to the family $\rho_t = \psi_t^*\rho_1$ one can deform ψ_t to a family of symplectomorphisms ϕ_t such that $\phi_0 = id$ and $\phi_1 = \psi_1$. This shows that $\text{Diff}_0(\Sigma) \cap \text{Symp}(\Sigma, \rho) = \text{Symp}_0(\Sigma, \rho)$. In conclusion, we have:

Proposition 2.17. Let Σ be a closed oriented surface of genus ≥ 2 . The map

$$(J, \sigma) \mapsto (h = (1 + f(\|\sigma\|_{I})) g_{J}, B = h^{-1}\sigma)$$

induces a $MCG(\Sigma)$ -invariant homeomorphism between $\mathcal{MS}_0(\Sigma, \rho)$ and $\mathcal{MS}(\Sigma)$.

3. The toy model: genus 1

The purpose of this section is to provide a para-hyperKähler structure on the cotangent bundle of the space $\mathcal{J}(\mathbb{R}^2)$ of linear complex structures on \mathbb{R}^2 . Interpreting the complement of the zero section in $T^*\mathcal{J}(\mathbb{R}^2)$ as the space $\mathcal{MGH}(T^2)$, we will deduce the case $\Sigma = T^2$ in all the results stated in the introduction.

3.1. Space of linear almost-complex structures

We begin by defining the space $\mathcal{J}(\mathbb{R}^2)$. In this section, ρ denotes the standard volume form $\rho = dx \wedge dy$ on \mathbb{R}^2 .

Definition 3.1. We denote by $\mathcal{J}(\mathbb{R}^2)$ the set of endomorphisms J of \mathbb{R}^2 such that $J^2 = -1$, and satisfying $\rho(v, Jv) > 0$ for some (and consequently for every) non-zero vector $v \in \mathbb{R}^2$.

In other words, $\mathcal{J}(\mathbb{R}^2)$ is the collection of all (linear) complex structures on \mathbb{R}^2 that are compatible with its standard orientation. It turns out that $\mathcal{J}(\mathbb{R}^2)$ is a two-dimensional manifold.

It is simple to see that, given any $J \in \mathcal{J}(\mathbb{R}^2)$, the tensor $g_J := \rho(\cdot, J \cdot)$ is a positive definite scalar product on \mathbb{R}^2 , with respect to which J is an orthogonal transformation. By differentiating the identity $J^2 = -1$, we see that the tangent space of $\mathcal{J}(\mathbb{R}^2)$ can be described as

$$T_J \mathcal{J}(\mathbb{R}^2) = \{ \dot{J} \in \operatorname{End}(\mathbb{R}^2) \mid \dot{J}J + J\dot{J} = 0 \}.$$

Equivalently, $T_J \mathcal{J}(\mathbb{R}^2)$ is the space of endomorphisms \dot{J} that are traceless and g_J -self-adjoint. The tangent space $T_J \mathcal{J}(\mathbb{R}^2)$ is endowed with a natural (almost) complex structure, given by $\mathcal{I}(\dot{J}) := -J\dot{J}$.

We will represent the cotangent space of $\mathcal{J}(\mathbb{R}^2)$ as follows:

$$T_J^* \mathcal{J}(\mathbb{R}^2) = \{ \sigma \in S_2(\mathbb{R}^2) \mid J^* \sigma = \sigma(J \cdot, J \cdot) = -\sigma \},$$

where $S_2(\mathbb{R}^2)$ stands for the space of symmetric bilinear forms of \mathbb{R}^2 . An equivalent way to describe the cotangent space at J is as the set of bilinear forms on \mathbb{R}^2 that can be written as $\sigma = \operatorname{Re} \phi$, where ϕ is a symmetric \mathbb{C} -valued bilinear form that is complex-linear with respect to J. In other words, ϕ satisfies $\phi(J\cdot,\cdot) = \phi(\cdot,J\cdot) = i\phi(\cdot,\cdot)$. When this is the case, then ϕ can be expressed as follows:

$$\phi = \sigma - i\,\sigma(\cdot, J\cdot). \tag{3.1}$$

Observe also that σ belongs to $T_J^*\mathcal{J}(\mathbb{R}^2)$ if and only if $g_J^{-1}\sigma$ belongs to $T_J\mathcal{J}(\mathbb{R}^2)$ (here $g_J^{-1}\sigma$ represents the g_J -self-adjoint operator associated to σ , that is, $\sigma(\cdot,\cdot) = g_J((g_J^{-1}\sigma)\cdot,\cdot) = g_J(\cdot,(g_J^{-1}\sigma)\cdot)$). The natural pairing between the tangent and the cotangent space is the following:

$$\langle \sigma \mid \dot{J} \rangle_J = \frac{1}{2} \operatorname{tr}_{g_J}(\sigma(\cdot, \dot{J}\cdot)) = \frac{1}{2} \operatorname{tr}(g_J^{-1}\sigma\dot{J}).$$
 (3.2)

We also define the following positive definite scalar products:

$$\langle \sigma, \sigma' \rangle_J := \frac{1}{2} \operatorname{tr} \left(g_J^{-1} \sigma g_J^{-1} \sigma' \right), \qquad \langle \dot{J}, \dot{J}' \rangle_J := \frac{1}{2} \operatorname{tr} (\dot{J} \dot{J}')$$

for every $\dot{J}, \dot{J}' \in T_J \mathcal{J}(\mathbb{R}^2)$ and $\sigma, \sigma' \in T_J^* \mathcal{J}(\mathbb{R}^2)$. It is immediate to check that the almost complex structure \mathcal{I} preserves both scalar products. We also denote $\|\sigma\|_J^2 = \langle \sigma, \sigma \rangle_J$. If ϕ is a quadratic differential whose real part is equal to σ , we set $\|\phi\|_J := \|\sigma\|_J$.

3.2. The tangent space of $T^*\mathcal{J}(\mathbb{R}^2)$

We now provide a characterization of the tangent space to $T^*\mathcal{J}(\mathbb{R}^2)$.

Lemma 3.2. Let $(J, \sigma) \in T^*\mathcal{J}(\mathbb{R}^2)$. Then $(\dot{J}, \dot{\sigma}) \in \operatorname{End}(\mathbb{R}^2) \times S_2(\mathbb{R}^2)$ belongs to $T_{(J,\sigma)}T^*\mathcal{J}(\mathbb{R}^2)$ if and only if

$$\dot{J} \in T_J \mathcal{J}(\mathbb{R}^2), \ \dot{\sigma}_0 \in T_J^* \mathcal{J}(\mathbb{R}^2) \ and \ \operatorname{tr}_{g_J} \dot{\sigma} = -2\langle \sigma \mid J\dot{J} \rangle_J,$$

where $\dot{\sigma}_0$ denotes the g_J -traceless part of $\dot{\sigma}$.

Proof. We first compute

$$\dot{g}_J = \rho(\cdot, \dot{J}\cdot) = -\rho(\cdot, J^2\dot{J}\cdot) = -g_J(\cdot, J\dot{J}) . \tag{3.3}$$

Then, because σ is g_J -traceless, we have

$$0 = (\operatorname{tr}_{g_J} \sigma)' = -\operatorname{tr} (g_J^{-1} \dot{g}_J g_J^{-1} \sigma) + \operatorname{tr} (g_J^{-1} \dot{\sigma})$$

hence

$$\operatorname{tr}_{g_J} \dot{\sigma} = -\operatorname{tr} \Bigl(J \dot{J} g_J^{-1} \sigma \Bigr) = -2 \langle \sigma \mid J \dot{J} \rangle_J \; .$$

This concludes the proof.

The group $SL(2,\mathbb{R})$ naturally acts on $\mathcal{J}(\mathbb{R}^2)$ by conjugation and, more generally on its tangent and cotangent space as follows:

$$(J, \dot{J}) \in T\mathcal{J}(\mathbb{R}^2), \quad A \cdot (J, \dot{J}) := (AJA^{-1}, A\dot{J}A^{-1})$$

 $(J, \sigma) \in T^*\mathcal{J}(\mathbb{R}^2), \quad A \cdot (J, \sigma) := (AJA^{-1}, (A^{-1})^*\sigma)$
(3.4)

for any $A \in SL(2,\mathbb{R})$. The action of $SL(2,\mathbb{R})$ induces a faithful action of $\mathbb{P}SL(2,\mathbb{R}) = SL(2,\mathbb{R})/\{\pm 1\}$.

Lemma 3.3. For every $A \in SL(2, \mathbb{R})$ and $J \in \mathcal{J}(\mathbb{R}^2)$, we have:

$$\langle A \cdot \sigma \mid A \cdot \dot{J} \rangle_{A \cdot J} = \langle \sigma \mid \dot{J} \rangle_{J},$$
$$\langle A \cdot \dot{J}, A \cdot \dot{J}' \rangle_{A \cdot J} = \langle \dot{J}, \dot{J}' \rangle_{J},$$
$$\langle A \cdot \sigma, A \cdot \sigma' \rangle_{A \cdot J} = \langle \sigma, \sigma' \rangle_{J},$$

where $\dot{J}, \dot{J}' \in T_J \mathcal{J}(\mathbb{R}^2)$ and $\sigma, \sigma' \in T_J^* \mathcal{J}(\mathbb{R}^2)$.

Proof. The proof is immediate, once one checks that $g_{AJA^{-1}}^{-1}(A^{-1})^*\sigma = A(g_J^{-1}\sigma)A^{-1}$. \square

By differentiating the $\mathrm{SL}(2,\mathbb{R})$ -action on $T_J^*\mathcal{J}(\mathbb{R}^2)$, we obtain a linear isomorphism from $T_{(J,\sigma)}T^*\mathcal{J}(\mathbb{R}^2)$ to $T_{A\cdot(J,\sigma)}T^*\mathcal{J}(\mathbb{R}^2)$. By a little abuse of notation, we still denote this isomorphism by A. It is explicitly given by:

$$(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)} T^* \mathcal{J}(\mathbb{R}^2), \quad A \cdot (\dot{J}, \dot{\sigma}) := (A\dot{J}A^{-1}, (A^{-1})^* \dot{\sigma}) .$$

We remark that, using Lemma 3.3, one could verify by hands that the conditions of Lemma 3.2 are preserved by this expression.

It is also useful to provide a natural linear isomorphism between the tangent space of $T^*_{(J,\sigma)}\mathcal{J}(\mathbb{R}^2)$ and the product of two copies of $T_J\mathcal{J}(\mathbb{R}^2)$.

Proposition 3.4. The map

$$\Psi_{(J,\sigma)}: T_{(J,\sigma)}T^*\mathcal{J}(\mathbb{R}^2) \longrightarrow (T_J\mathcal{J}(\mathbb{R}^2))^2 (\dot{J},\dot{\sigma}) \longmapsto (\dot{J},g_J^{-1}\dot{\sigma}_0)$$

is a linear isomorphism, with inverse

$$\Xi_{(J,\sigma)}: (T_J \mathcal{J}(\mathbb{R}^2))^2 \longrightarrow T_{(J,\sigma)} T^* \mathcal{J}(\mathbb{R}^2) (\dot{J}, \dot{K}) \longmapsto (\dot{J}, g_J(\cdot, \dot{K} \cdot) - \langle \sigma \mid J \dot{J} \rangle_J g_J).$$

Moreover, $\Psi_{(J,\sigma)}$ commutes with the $SL(2,\mathbb{R})$ actions, meaning that

$$(A,A) \circ \Psi_{(J,\sigma)} = \Psi_{A \cdot (J,\sigma)} \circ A$$
.

Proof. The proof follows immediately from Lemma 3.2 and Lemma 3.3.

3.3. A para-hyperKähler structure on $T^*\mathcal{J}(\mathbb{R}^2)$

Throughout the paper, f will denote the function $f(t) := \sqrt{1+t^2}$ for $t \in \mathbb{R}$ which we introduced already in Section 2.5. Unless otherwise stated, we will denote by $\langle \cdot, \cdot \rangle$, $\langle \cdot | \cdot \rangle$ the scalar products $\langle \cdot, \cdot \rangle_J$ on $T_J \mathcal{J}(\mathbb{R}^2)$ and $T_J^* \mathcal{J}(\mathbb{R}^2)$ (it will be clear from the context which one of these we will refer to) and the pairing $\langle \cdot | \cdot \rangle_J$ between $T_J \mathcal{J}(\mathbb{R}^2)$ and its dual. Similarly $\| \cdot \| = \| \cdot \|_J$.

Definition 3.5. Let us define the symmetric bi-linear form \mathbf{g} on $T_{(J,\sigma)}T^*\mathcal{J}(\mathbb{R}^2)$:

$$\mathbf{g}_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) = f(\|\sigma\|)\langle\dot{J},\dot{J}'\rangle - \frac{1}{f(\|\sigma\|)}\langle\dot{\sigma}_0,\dot{\sigma}'_0\rangle$$
(3.5)

and the endomorphisms $\mathbf{I}, \mathbf{J}, \mathbf{K}$ of $T_{(J,\sigma)}T^*\mathcal{J}(\mathbb{R}^2)$:

$$\mathbf{I}_{(J,\sigma)}(\dot{J},\dot{\sigma}) = \left(-J\dot{J}, -\dot{\sigma}_0(\cdot, J\cdot) - \langle \sigma \mid \dot{J} \rangle g_J\right) \tag{3.6}$$

$$\mathbf{J}_{(J,\sigma)}(\dot{J},\dot{\sigma}) = \left(\frac{1}{f(\|\sigma\|)} g_J^{-1} \dot{\sigma}_0, \ f(\|\sigma\|) g_J(\cdot,\dot{J}\cdot) + \frac{\langle \sigma, \dot{\sigma}_0(\cdot,J\cdot) \rangle}{f(\|\sigma\|)} g_J\right)$$
(3.7)

$$\mathbf{K}_{(J,\sigma)}(\dot{J},\dot{\sigma}) = \left(-\frac{1}{f(\|\sigma\|)} J g_J^{-1} \dot{\sigma}_0, -f(\|\sigma\|) g_J(\cdot, \dot{J}J\cdot) - \frac{\langle \sigma, \dot{\sigma}_0 \rangle}{f(\|\sigma\|)} g_J\right)$$
(3.8)

where $f(t) = \sqrt{1 + t^2}$.

For future reference, we also record the expressions of the forms:

$$\omega_{\mathbf{I}} = \mathbf{g}(\cdot, \mathbf{I}\cdot), \quad \omega_{\mathbf{J}} = \mathbf{g}(\cdot, \mathbf{J}\cdot), \quad \omega_{\mathbf{K}} = \mathbf{g}(\cdot, \mathbf{K}\cdot).$$

These are given by:

$$(\omega_{\mathbf{I}})_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) = -f(\|\sigma\|)\langle\dot{J},J\dot{J}'\rangle + \frac{1}{f(\|\sigma\|)}\langle\dot{\sigma}_0,\dot{\sigma}'_0(\cdot,J\cdot)\rangle, \tag{3.9}$$

$$(\omega_{\mathbf{J}})_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) = \langle \dot{\sigma}'_0 \mid \dot{J} \rangle - \langle \dot{\sigma}_0 \mid \dot{J}' \rangle, \tag{3.10}$$

$$(\omega_{\mathbf{K}})_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) = \langle \dot{\sigma}'_0 \mid J\dot{J}\rangle - \langle \dot{\sigma}_0 \mid J\dot{J}'\rangle. \tag{3.11}$$

Theorem 3.6. The quadruple $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$ is an $SL(2, \mathbb{R})$ -invariant para-hyperKähler structure on $T^*\mathcal{J}(\mathbb{R}^2)$.

Proof. Observe that, through the linear isomorphism of Lemma 3.2, the endomorphisms $\mathbf{I}, \mathbf{J}, \mathbf{K}$ can be represented as:

$$\begin{split} &(\Xi_{(J,\sigma)}^*\mathbf{I}_{(J,\sigma)})(\dot{J},\dot{K}) = (-J\dot{J},\,J\dot{K}),\\ &(\Xi_{(J,\sigma)}^*\mathbf{J}_{(J,\sigma)})(\dot{J},\dot{K}) = (f(\|\sigma\|)^{-1}\dot{K},\,f(\|\sigma\|)\dot{J}),\\ &(\Xi_{(J,\sigma)}^*\mathbf{K}_{(J,\sigma)})(\dot{J},\dot{K}) = (-f(\|\sigma\|)^{-1}J\dot{K},\,f(\|\sigma\|)J\dot{J}). \end{split}$$

From here, it is clear that the following relations are satisfied:

$$\mathbf{I}^2 = -1, \quad \mathbf{J}^2 = \mathbf{K}^2 = 1, \quad \mathbf{K} = \mathbf{IJ}.$$

We will see in Lemma 3.8 that the 2-forms $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$ are respectively the real and imaginary part of the complex symplectic form $\omega^{\mathbb{C}}$ on $T^*\mathcal{J}(\mathbb{R}^2)$, which is defined as the differential of $-\lambda^{\mathbb{C}}$. Hence the forms $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$ are obviously exact. We will show in Corollary 3.16 that $\omega_{\mathbf{I}}$ is closed, as well. One can now apply a general argument to conclude that that $\mathbf{I}, \mathbf{J}, \mathbf{K}$ are integrable, see Lemma A.5 in Appendix A. However, we will also deduce this directly using some of the interpretations of $\mathbf{I}, \mathbf{J}, \mathbf{K}$ that we show below. More concretely, integrability of \mathbf{I} is proved in Corollary 3.10; integrability of \mathbf{J} and \mathbf{K} follows from the pull-back results of Corollary 3.16 and Corollary 3.19. This shows that the quadruple $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$ defines a para-hyperKähler structure on $T^*\mathcal{J}(\mathbb{R}^2)$. The $\mathrm{SL}(2,\mathbb{R})$ -invariance is checked in each expression by applying Lemma 3.3.

Remark 3.7. Let us observe that **I** preserves the tangent space to the 0-section of $T^*\mathcal{J}(\mathbb{R}^2)$, since $\mathbf{I}_{(J,0)}(\dot{J},0) = (-J\dot{J},0)$. We will denote by G and Ω the restrictions of \mathbf{g} and $\omega_{\mathbf{I}}$ to the 0-section of $T^*\mathcal{J}(\mathbb{R}^2)$, which is identified to $\mathcal{J}(\mathbb{R}^2)$. In particular, for every $J \in \mathcal{J}(\mathbb{R}^2)$

$$G_J(\dot{J},\dot{J}') := \langle \dot{J},\dot{J}' \rangle_J \qquad \Omega_J(\dot{J},\dot{J}') := -\langle \dot{J},J\dot{J}' \rangle_J,$$

where $\dot{J}, \dot{J}' \in T_J \mathcal{J}(\mathbb{R}^2)$. Hence (G, Ω) is a Kähler structure on $\mathcal{J}(\mathbb{R}^2)$.

Now, it turns out that $\mathcal{J}(\mathbb{R}^2)$ is diffeomorphic to the hyperbolic plane \mathbb{H}^2 . To see this, one can define a map from $\mathcal{J}(\mathbb{R}^2)$ to the upper half-space model of \mathbb{H}^2 by declaring that the standard linear complex structure

$$J_0 = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$$

is mapped to $i \in \mathbb{H}^2$. Observe that $SO(2) < SL(2,\mathbb{R})$ stabilizes both J_0 (for the $SL(2,\mathbb{R})$ -action on $\mathcal{J}(\mathbb{R}^2)$ introduced in Section 3.2) and i (for the classical action in the upper half-plane model). Since both actions are transitive, one can uniquely extend this assignment to an $SL(2,\mathbb{R})$ -equivariant diffeomorphism. It is not hard to check that this is a Kähler isometry, that is, the metric G corresponds to the hyperbolic metric of \mathbb{H}^2 , the complex structure \mathcal{I} to the standard complex structure of \mathbb{H}^2 , and therefore the symplectic form Ω to the area form of \mathbb{H}^2 . See [Tra18, Lemma 4.3.2], [Tra19, §3.1] or [Hod05, §2.2.2] for more details.

3.4. Liouville form on $T^*\mathcal{J}(\mathbb{R}^2)$

Recall that the Liouville form of any manifold M is the 1-form λ on T^*M defined by $\lambda_{(p,\alpha)}(v) = \alpha(\pi_*v)$, for $\alpha \in T_p^*M$. If M is a complex manifold, one has an induced complex structure \mathbf{I} on T^*M , and λ is the real part of the complex-valued 1-form $\lambda^{\mathbb{C}} = \lambda - i\lambda \circ \mathbf{I}$. The term $\lambda \circ \mathbf{I}$ can also be written as $\alpha(\pi_*(\mathcal{I}(v)))$, where \mathcal{I} is the complex structure of M, since π is holomorphic. In our setting, we therefore have the following expression for the (complex-valued) Liouville form on $T^*\mathcal{J}(\mathbb{R}^2)$:

$$\lambda_{(J,\sigma)}^{\mathbb{C}}(\dot{J},\dot{\sigma}) = \langle \sigma \mid \dot{J} \rangle_J + i \langle \sigma \mid J\dot{J} \rangle_J.$$

We define the complex-valued cotangent symplectic structure of $T^*\mathcal{J}(\mathbb{R}^2)$ by setting

$$\omega^{\mathbb{C}} := -d\lambda^{\mathbb{C}}$$
.

In this section we show the following:

Lemma 3.8. The \mathbb{C} -valued symplectic form $\omega^{\mathbb{C}}$ on $T^*\mathcal{J}(\mathbb{R}^2)$ equals $\omega_{\mathbf{J}} + i\omega_{\mathbf{K}}$.

In other words, we have to show that $\omega^{\mathbb{C}}$ has the following expression:

$$\omega^{\mathbb{C}}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) = (\langle \dot{\sigma}'_0 \mid \dot{J} \rangle_J - \langle \dot{\sigma}_0 \mid \dot{J}' \rangle_J) + i(\langle \dot{\sigma}'_0 \mid J\dot{J} \rangle_J - \langle \dot{\sigma}_0 \mid J\dot{J}' \rangle_J).$$

Before providing the proof, we give a useful lemma.

Lemma 3.9. For every $\dot{J}, \dot{J}' \in T_J \mathcal{J}(\mathbb{R}^2)$ we have

$$\dot{J}\dot{J}' = \langle \dot{J}, \dot{J}' \rangle_J \, \mathbb{1} - \langle J\dot{J}, \dot{J}' \rangle_J \, J.$$

Moreover, we have:

i)
$$\dot{J}^2 = -(\det \dot{J})\mathbb{1} = ||\dot{J}||_J^2\mathbb{1} \text{ for every } \dot{J} \in T_J \mathcal{J}(\mathbb{R}^2);$$

$$ii)$$
 tr $(\dot{J}\dot{J}'\dot{J}'') = 0$ for every $\dot{J}, \dot{J}', \dot{J}'' \in T_J \mathcal{J}(\mathbb{R}^2)$.

Proof. Let us first observe that

$$J\dot{J}\dot{J}' = -\dot{J}J\dot{J}' = \dot{J}\dot{J}'J.$$

Therefore, $\dot{J}\dot{J}'$ commutes with J. It is simple to check that $M \in \operatorname{End}(\mathbb{R}^2)$ commutes with J if and only if M belongs to $\operatorname{Span}(\mathbb{1},J)$, hence

$$\dot{J}\dot{J}' = \langle \dot{J}, \dot{J}' \rangle_J \, \mathbb{1} - \langle J\dot{J}, \dot{J}' \rangle_J \, J.$$

By Cayley-Hamilton theorem,

$$0 = \dot{J}^2 - (\operatorname{tr} \dot{J})\dot{J} + (\det \dot{J}) \, \mathbb{1} = \dot{J}^2 + (\det \dot{J}) \, \mathbb{1}.$$

Therefore, we have $2\|\dot{J}\|_J^2 = \operatorname{tr}(\dot{J}^2) = -2 \det \dot{J}$. For the last assertion, we apply the first part of the statement:

$$\operatorname{tr}(\dot{J}\dot{J}'\dot{J}'') = \langle \dot{J}, \dot{J}' \rangle_{J} \operatorname{tr}(\dot{J}'') - \langle J\dot{J}, \dot{J}' \rangle_{J} \operatorname{tr}(J\dot{J}'').$$

This expression vanishes because \dot{J}'' and $J\dot{J}''$ are both traceless, being elements of $T_J\mathcal{J}(\mathbb{R}^2)$.

Proof of Lemma 3.8. The set $T^*\mathcal{J}(\mathbb{R}^2)$ can be considered as a submanifold of the vector space $\operatorname{End}(\mathbb{R}^2) \times S_2(\mathbb{R}^2)$. In particular, any tangent vector $(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\mathbb{R}^2)$ can be extended to a vector field, that we continue to denote with abuse by $(\dot{J}, \dot{\sigma})$, on a neighborhood of (J, σ) in $\operatorname{End}(\mathbb{R}^2) \times S_2(\mathbb{R}^2)$, with values in $\operatorname{End}(\mathbb{R}^2) \times S_2(\mathbb{R}^2)$. Moreover, we can require that the component \dot{J} depends only on the variable $J \in \operatorname{End}(\mathbb{R}^2)$. In all the following computations, we will denote by D the standard flat connection on the vector space $\operatorname{End}(\mathbb{R}^2) \times S_2(\mathbb{R}^2)$, and we consider $\langle \cdot | \cdot \rangle$ as a pairing defined for elements

П

of the entire vector spaces $\operatorname{End}(\mathbb{R}^2)$ and $S_2(\mathbb{R}^2)$, where the extension is given by the same expression (3.2).

If $\lambda := \operatorname{Re} \lambda^{\mathbb{C}}$, then we have:

$$d\lambda ((\dot{J}, \dot{\sigma}), (\dot{J}', \dot{\sigma}')) = (\dot{J}, \dot{\sigma})(\lambda(\dot{J}', \dot{\sigma}')) - (\dot{J}', \dot{\sigma}')(\lambda(\dot{J}, \dot{\sigma})) - \lambda([(\dot{J}, \dot{\sigma}), (\dot{J}', \dot{\sigma}')])$$

$$= (\dot{J}, \dot{\sigma})(\langle \sigma \mid \dot{J}' \rangle_{J}) - (\dot{J}', \dot{\sigma}')(\langle \sigma \mid \dot{J} \rangle_{J}) - \langle \sigma \mid [\dot{J}, \dot{J}'] \rangle_{J}.$$

Observe that $\dot{J}(g_J) = -g_J(\cdot, J\dot{J}\cdot)$. Then

$$\begin{split} (\dot{J},\dot{\sigma})(\langle\sigma\mid\dot{J}'\rangle_{J}) &= \frac{1}{2}(\dot{J},\dot{\sigma})(\operatorname{tr}(g_{J}^{-1}\sigma\dot{J}')) \\ &= \frac{1}{2}(-\operatorname{tr}(g_{J}^{-1}\dot{J}(g_{J})g_{J}^{-1}\sigma\dot{J}') + \operatorname{tr}(g_{J}^{-1}\dot{\sigma}\dot{J}') + \operatorname{tr}(g_{J}^{-1}\sigma D_{\dot{J}}\dot{J}')) \\ &= \frac{1}{2}\operatorname{tr}(J\dot{J}g_{J}^{-1}\sigma\dot{J}') + \frac{1}{2}\operatorname{tr}(g_{J}^{-1}\dot{\sigma}_{0}\dot{J}') + \langle\sigma\mid D_{\dot{J}}\dot{J}'\rangle_{J} \\ &= \langle\dot{\sigma}_{0}\mid\dot{J}'\rangle_{J} + \langle\sigma\mid D_{\dot{J}}\dot{J}'\rangle_{J}. \end{split}$$

In the last step we applied point ii) of Lemma 3.9. Replacing this relation in the expression for $d\lambda$ we obtain

$$\begin{split} \mathrm{d}\lambda \left((\dot{J}, \dot{\sigma}), (\dot{J}', \dot{\sigma}') \right) &= \langle \dot{\sigma}_0 \mid \dot{J}' \rangle_J - \langle \dot{\sigma}_0' \mid \dot{J} \rangle_J + \langle \sigma \mid D_{\dot{J}} \dot{J}' - D_{\dot{J}'} \dot{J} - [\dot{J}, \dot{J}'] \rangle_J \\ &= \langle \dot{\sigma}_0 \mid \dot{J}' \rangle_J - \langle \dot{\sigma}_0' \mid \dot{J} \rangle_J. \end{split}$$

The same type of argument applies to the imaginary part of $\lambda^{\mathbb{C}}$. More concretely, setting $\mu := \operatorname{Im} \lambda^{\mathbb{C}}$, we arrive at the expression:

$$\mathrm{d}\mu\left((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')\right) = \langle \dot{\sigma}_0 \mid J\dot{J}' \rangle_J - \langle \dot{\sigma}_0' \mid J\dot{J} \rangle_J + \langle \sigma \mid D_{\dot{I}}(J\dot{J}') - D_{\dot{I}'}(J\dot{J}) - J[\dot{J},\dot{J}'] \rangle_J.$$

It remains to check that the last term vanishes. But this term equals $\langle \sigma \mid \dot{J}\dot{J}' - \dot{J}'\dot{J} \rangle$. Using Lemma 3.9, $\dot{J}\dot{J}' - \dot{J}'\dot{J}$ is proportional to J, hence this scalar product vanishes because $\langle \sigma \mid J \rangle = (1/2) \mathrm{tr}(g_J^{-1} \sigma J) = 0$.

Corollary 3.10. The almost-complex structure \mathbf{I} equals the almost-complex structure induced by \mathcal{I} on $T^*\mathcal{J}(\mathbb{R}^2)$. In particular, \mathbf{I} is an integrable almost-complex structure.

Proof. We have showed in Lemma 3.8 that $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$ are the real and imaginary parts of the complex symplectic form $\omega^{\mathbb{C}}$ of $T^*\mathcal{J}(\mathbb{R}^2)$. Denote by $\hat{\mathbf{I}}$ the almost complex structure of $T^*\mathcal{J}(\mathbb{R}^2)$. Since $\mathcal{J}(\mathbb{R}^2)$ is a complex manifold, $\omega^{\mathbb{C}}$ is a complex symplectic form, which means $\omega^{\mathbb{C}}(\cdot,\hat{\mathbf{I}}\cdot)=i\omega^{\mathbb{C}}(\cdot,\cdot)$. Taking the imaginary parts in this equality we find $\omega_{\mathbf{J}}=\omega_{\mathbf{K}}(\cdot,\hat{\mathbf{I}}\cdot)$. But from the definitions of $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$, we see that

$$\omega_{\mathbf{J}} = \mathbf{g}(\cdot, \mathbf{J} \cdot) = \mathbf{g}(\cdot, \mathbf{K} \mathbf{I} \cdot) = \omega_{\mathbf{K}}(\cdot, \mathbf{I} \cdot) \ .$$

Since $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$ are nondegenerate, this implies $\hat{\mathbf{I}} = \mathbf{I}$.

3.5. Relation with $\mathcal{MGH}(T^2)$

Theorem 2.8 furnishes a diffeomorphism $\mathcal{F}: \mathcal{MGH}(T^2) \to T^*\mathcal{T}^{\mathfrak{c}}(T^2)$ between the deformation space of MGHC anti-de Sitter manifolds diffeomorphic to $T^2 \times \mathbb{R}$ and the complement of the zero section of the cotangent space $T^*\mathcal{T}^{\mathfrak{c}}(T^2)$ to the Teichmüller space of the torus. The latter can be identified with $T^*\mathcal{J}(\mathbb{R}^2)$, as we show in the following lemma.

Lemma 3.11. There is a homeomorphism between $\mathcal{J}(\mathbb{R}^2)$ and $\mathcal{T}^{\mathfrak{c}}(T^2)$, which is equivariant with respect to the actions of $\mathrm{SL}(2,\mathbb{Z}) \cong MCG(T^2)$.

Proof. The map from $\mathcal{J}(\mathbb{R}^2)$ to $\mathcal{T}^{\mathfrak{c}}(T^2)$ is defined by considering a linear almost-complex structure J as a (constant) tensor on \mathbb{R}^2 , which therefore induces an almost-complex structure on the torus $T^2 \cong \mathbb{R}^2/\mathbb{Z}^2$. The map is a bijection because every element in $\mathcal{T}^{\mathfrak{c}}(T^2)$, namely an isotopy class of almost-complex structures on T^2 , can be represented as the conformal structure induced by J_0 on \mathbb{R}^2/Λ , for $\Lambda \cong \mathbb{Z}^2$ a (marked) lattice. One can moreover assume (up to homothety of Λ) that \mathbb{R}^2/Λ has area 1, and such representation is unique up to conjugation in SO(2). Conjugating J_0 by the unique element of $SL(2,\mathbb{R})$ that maps Λ to \mathbb{Z}^2 (as marked lattices), one finds the unique $J \in \mathcal{J}(\mathbb{R}^2)$ that is mapped to the given class in $\mathcal{T}^{\mathfrak{c}}(T^2)$. Identifying $MCG(T^2)$ with $SL(2,\mathbb{Z})$, the homeomorphism is clearly equivariant.

Remark 3.12. We observe that the symplectic form Ω introduced in Remark 3.7 coincides with $4\Omega_{WP}$, where Ω_{WP} denotes the Weil-Petersson symplectic form on the space $\mathcal{J}(\mathbb{R}^2) \cong \mathcal{T}^{\mathfrak{c}}(T^2)$ through the identification described in Lemma 3.11 (see also Lemma 4.5).

Hence we immediately obtain the proof of Theorem A in genus one.

Theorem A (genus one). The deformation space $\mathcal{MGH}(T^2)$ admits a $MCG(T^2)$ -invariant para-hyperKähler structure $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$.

Proof. By Proposition 2.2 and Theorem 2.8, $\mathcal{MGH}(T^2)$ is identified to the complement of the zero section in $T^*\mathcal{T}^{\mathfrak{c}}(T^2)$, which is in turn naturally identified to the complement of the zero section in $T^*\mathcal{J}(\mathbb{R}^2)$ by Lemma 3.11. Hence the existence of the para-hyperKähler structure follows immediately from Proposition 3.6. Since all the identifications are equivariant with respect to the action of $MCG(T^2) \cong SL(2,\mathbb{Z})$, and the para-hyperKähler structure of $T^*\mathcal{J}(\mathbb{R}^2)$ is $\mathbb{P}SL(2,\mathbb{R})$ -invariant by Proposition 3.6, it follows that the obtained para-hyperKähler structure on $\mathcal{MGH}(T^2)$ is mapping-class group invariant.

More concretely, we can see the induced map from $T^*\mathcal{J}(\mathbb{R}^2)$ to $T^*\mathcal{T}^{\mathfrak{c}}(T^2)$ as the map sending the pair (J,σ) , for σ a symmetric bilinear form on \mathbb{R}^2 satisfying $\sigma(J\cdot,J\cdot)=-\sigma$, to the pair $([J],\phi)$ where [J] is the isotopy class as in the proof of Lemma 3.11 and ϕ is the holomorphic quadratic differential whose real part is σ as in Equation (3.1).

Remark 3.13. We remark that the complex-linear quadratic differential ϕ is holomorphic with respect to J simply because, identifying (T^2, J) with a biholomorphism to the quotient

of \mathbb{R}^2 by a lattice Λ as in the proof of Lemma 3.11, the lift of ϕ to \mathbb{R}^2 is constant. As a matter of fact, every holomorphic quadratic differential on (T^2, J) lifts to a constant holomorphic quadratic differential, namely of the form adz^2 for $a \in \mathbb{C}$, on $\mathbb{R}^2 \cong \mathbb{C}$.

Remark 3.14. It will be useful to interpret the map from $\mathcal{J}(\mathbb{R}^2)$ to $\mathcal{T}^{\mathfrak{c}}(T^2)$ in terms of Beltrami differentials. Let us consider a smooth path $J_t \in \mathcal{J}(\mathbb{R}^2)$, and denote $J_{t=0} = J$. As usual, we consider these as almost-complex structures on $T^2 = \mathbb{R}^2/\mathbb{Z}^2$. Then the Beltrami differential of the identity map $id: (T^2, J) \to (T^2, J_t)$ coincides with

$$\nu_t = (\mathbb{1} - J_t J)^{-1} (\mathbb{1} + J_t J).$$

Indeed ν_t coincides with $L^{-1} \circ A$, where L and A are the complex linear and complex anti-linear parts of $d(id): (T.T^2, J) \to (T.T^2J_t)$, respectively. Hence ν_t is constant over T^2 , which means that it is a harmonic Beltrami differential (i.e. of the form $g^{-1}\overline{\psi}$ for g the flat metric and ψ a holomorphic quadratic differential, see Remark 3.13 above). A simple computation shows that the derivative of ν_t at t=0, which represents an element of $T\mathcal{T}^{\mathfrak{c}}(T^2)$, is again harmonic and has the expression

$$\dot{\nu} = \frac{1}{2}\dot{J}J\tag{3.12}$$

We are now ready to conclude the proof of Theorem B in genus one. Considering $\mathcal{MGH}(T^2)$ as the complement of the zero section in $T^*\mathcal{J}(\mathbb{R}^2)$, the map $\mathcal{F}:\mathcal{MGH}(T^2)\to T^*\mathcal{T}^{\mathfrak{c}}(T^2)$ given by Theorem 2.8 is nothing but the restriction of the map that we used in Lemma 3.11, which we still denote as $\mathcal{F}:T^*\mathcal{J}(\mathbb{R}^2)\to T^*\mathcal{T}^{\mathfrak{c}}(T^2)$ by a little abuse of notation.

Theorem B (genus one). We have

$$\mathcal{F}^*(\mathcal{I}_{T^*\mathcal{T}(T^2)}, \Omega_{T^*\mathcal{T}(T^2)}^{\mathbb{C}}) = \left(-\mathbf{I}, -\frac{i}{2}\overline{\omega}_{\mathbf{I}}^{\mathbb{C}}\right) ,$$

where $\mathcal{I}_{T^*\mathcal{T}(T^2)}$ denotes the complex structure of $T^*\mathcal{T}(T^2)$ and $\Omega^{\mathbb{C}}_{T^*\mathcal{T}(T^2)}$ its complex symplectic form.

Before the proof, we recall that (in any genus) the cotangent bundle of $\mathcal{T}^{\mathfrak{c}}(\Sigma)$ has a natural complex symplectic form $\Omega^{\mathbb{C}}_{T^*\mathcal{T}(\Sigma)}$ defined as $-d\lambda^{\mathbb{C}}_{T^*\mathcal{T}(\Sigma)}$, where $\lambda^{\mathbb{C}}_{T^*\mathcal{T}(\Sigma)}$ is the Liouville form. Given a point $([J], \phi) \in T^*\mathcal{T}^{\mathfrak{c}}(\Sigma)$, the pairing between holomorphic quadratic differentials and tangent vectors expressed as classes of Beltrami differentials $[\dot{\nu}]$ is the following:

$$\langle \phi \mid \dot{\nu} \rangle_{[J]} = \int_{\Sigma} \phi \bullet \dot{\nu} \tag{3.13}$$

where

$$(\phi \bullet \dot{\nu})(v,w) := \frac{1}{2i}(\phi(\dot{\nu}(v),w) - \phi(\dot{\nu}(w),v)) \tag{3.14}$$

(see e. g. [BMS15, Section 2.1]). The complex Liouville form is then simply expressed as:

$$(\lambda_{T^*\mathcal{T}^c}^{\mathbb{C}})_{([J],\phi)}([\dot{\nu}],\dot{\phi}) = \langle \phi \mid \dot{\nu} \rangle_{[J]}$$

We now prove Theorem B in the torus case.

Proof of Theorem B, genus g = 1. Let $(J_t)_t$ be a smooth path in $\mathcal{J}(\mathbb{R}^2)$, with $J_{t=0} = J$. If $\pi : T^*\mathcal{T}^{\mathfrak{c}}(T^2) \to \mathcal{T}^{\mathfrak{c}}(T^2)$ is the projection, Equation (3.12) in Remark 3.14 shows:

$$\mathrm{d}\pi \circ \mathrm{d}\mathcal{F}_{(J,\sigma)}\left(\dot{J},\dot{\sigma}\right) = \frac{1}{2}\dot{J}J =: \dot{\nu} \ .$$

Recall that $\mathcal{F}(J,\sigma) = ([J],\phi)$, where $\phi = \sigma - i\,\sigma(\cdot,J\cdot)$ is the holomorphic quadratic differential whose real part is σ . Let now g_J be the bilinear form $\rho(\cdot,J\cdot)$, and let $\{e_1,e_2=Je_1\}$ be a g_J -orthonormal basis of \mathbb{R}^2 . Since all the quantities that we consider lift to constant tensors on \mathbb{R}^2 , we can get rid of the integral over $T^2 \cong \mathbb{R}^2/\mathbb{Z}^2$ in Equation (3.13), and we find:

$$\begin{split} \langle \phi \mid \dot{\nu} \rangle_{[J]} &= \frac{1}{2i} \left(\phi(\dot{\nu}(e_1), e_2) - \phi(\dot{\nu}(e_2), e_1) \right) \\ &= \frac{1}{4i} \left(\sigma(\dot{J}Je_1, e_2) - i \, \sigma(\dot{J}Je_1, Je_2) - \sigma(\dot{J}Je_2, e_1) + i \, \sigma(\dot{J}Je_2, Je_1) \right) \\ &= \frac{1}{4i} \left(\sigma(\dot{J}e_2, e_2) + i \, \sigma(\dot{J}Je_1, e_1) + \sigma(\dot{J}e_1, e_1) + i \, \sigma(\dot{J}Je_2, e_2) \right) \\ &= \frac{1}{4i} \left(\mathrm{tr} \left(g_J^{-1} \sigma \dot{J} \right) + i \, \mathrm{tr} \left(g_J^{-1} \sigma \dot{J} J \right) \right) \\ &= -\frac{i}{2} \left(\langle \sigma \mid \dot{J} \rangle_J - i \langle \sigma \mid J \dot{J} \rangle_J \right) \\ &= -\frac{i}{2} \, \overline{\lambda_{(J,\sigma)}^{\mathbb{C}}(\dot{J}, \dot{\sigma})}. \end{split}$$

Therefore we have shown

$$(\mathcal{F}^*\lambda_{T^*\mathcal{T}(T^2)}^{\mathbb{C}})_{(J,\sigma)}(\dot{J},\dot{\sigma}) = -\frac{i}{2}\overline{\lambda_{(J,\sigma)}^{\mathbb{C}}(\dot{J},\dot{\sigma})}.$$

Taking differentials, we obtain

$$\mathcal{F}^*\Omega_{T^*\mathcal{T}(T^2)}^{\mathbb{C}} = -\frac{i}{2}(\omega_{\mathbf{J}} - i\,\omega_{\mathbf{K}}) = -\frac{i}{2}\overline{\omega}_{\mathbf{I}}^{\mathbb{C}}.$$

To show that \mathcal{F} pulls-back the almost-complex structure of $T^*\mathcal{T}(T^2)$ to $-\mathbf{I}$, one can argue exactly as in the proof of Corollary 3.10, observing that $\Omega^{\mathbb{C}}_{T^*\mathcal{T}(T^2)}$ is a complex symplectic form with respect to the complex structure of $T^*\mathcal{T}(T^2)$, while $\overline{\omega}^{\mathbb{C}}_{\mathbf{I}}$ is a complex symplectic form with respect to the (almost)-complex structure $-\mathbf{I}$.

3.6. A formal Mess homeomorphism

In this section we interpret the para-complex structure \mathbf{J} and the para-complex sympletic form $\omega_{\mathbf{J}}^{\mathbb{B}}$ as the pull-back of natural structures on $\mathcal{T}(T^2) \times \mathcal{T}(T^2)$ via a map

$$\mathcal{M}: T^*\mathcal{J}(\mathbb{R}^2) \to \mathcal{T}(T^2) \times \mathcal{T}(T^2)$$
,

which is formally defined essentially as the Mess homeomorphism (Section 2.3), although in the genus one case this map does not have the same geometric significance from the Anti-de Sitter viewpoint as for the higher genus case. Nevertheless, this "toy model" is essential for the higher genus case, since the map \mathcal{M} studied here will then induce Mess homeomorphism for genus ≥ 2 .

Inspired by Section 2.5, given a pair $(J, \sigma) \in T^*\mathcal{J}(\mathbb{R}^2)$, we define $h = h(J, \sigma)$ the Riemannian metric

$$h := (1 + f(\|\sigma\|))g_J = \left(1 + \sqrt{1 + \|\sigma\|^2}\right)g_J,$$

where we recall that $g_J = \rho(\cdot, J \cdot)$ and $\|\cdot\| = \|\cdot\|_J$. We also set $B := h^{-1}\sigma$. Exactly as in Lemma 2.15, we have the identities

$$\det B = -\frac{\|\sigma\|^2}{(1 + f(\|\sigma\|))^2} \quad \text{and} \quad 1 + \det B = \frac{2}{1 + f(\|\sigma\|)}. \quad (3.15)$$

The second identity shows that $\mathbb{1} \mp JB$ is invertible for every $(J, \sigma) \in T^*\mathcal{J}(\mathbb{R}^2)$. Indeed, $\det(\mathbb{1} \mp JB) = 1 + \det B$, since JB is traceless. We can thus define

$$\mathcal{M}: T^*\mathcal{J}(\mathbb{R}^2) \longrightarrow \mathcal{J}(\mathbb{R}^2) \times \mathcal{J}(\mathbb{R}^2)$$
$$\mathcal{M}(J, \sigma) := \left((\mathbb{1} - JB)^{-1} J(\mathbb{1} - JB), (\mathbb{1} + JB)^{-1} J(\mathbb{1} + JB) \right).$$

Remark 3.15. We remark that this is formally the analogue of the expression of Mess homeomorphism in terms of almost-complex structures, given in Lemma 2.10. In particular, the left and right components of \mathcal{M} are the linear complex structures associated to the metrics $h((\mathbb{1} \mp JB)\cdot, (\mathbb{1} \mp JB)\cdot)$ on \mathbb{R}^2 . As a reminder of this fact, we will often denote them by J_l and J_r , respectively, to be consistent with the notation that will be used in Section 4, where the higher genus case is discussed.

Now, the space $\mathcal{J}(\mathbb{R}^2) \times \mathcal{J}(\mathbb{R}^2)$ is naturally endowed with the almost para-complex structure \mathcal{P} coming from its product structure. This is defined at any $(J, J') \in \mathcal{J}(\mathbb{R}^2) \times \mathcal{J}(\mathbb{R}^2)$ as follows:

$$\mathcal{P}(\dot{J}, \dot{J}') := (\dot{J}, -\dot{J}').$$

Moreover, denoting by Ω_{WP} the Weil-Petersson symplectic form of $\mathcal{J}(\mathbb{R}^2)$ (see also Remark 3.12), and by $\pi_l, \pi_r : \mathcal{J}(\mathbb{R}^2) \times \mathcal{J}(\mathbb{R}^2) \to \mathcal{J}(\mathbb{R}^2)$ the projections to the left and right factor, $\mathcal{J}(\mathbb{R}^2) \times \mathcal{J}(\mathbb{R}^2)$ has two symplectic forms given by $\pi_l^* \Omega_{WP} \pm \pi_r^* \Omega_{WP}$. Together, they are

combined into a para-complex symplectic form

$$\Omega^{\mathbb{B}} := \frac{1}{2} (\pi_l^* \Omega_{WP} + \pi_r^* \Omega_{WP}) + \frac{\tau}{2} (\pi_l^* \Omega_{WP} - \pi_r^* \Omega_{WP}) .$$

It is easily checked that $\Omega^{\mathbb{B}}$ is para-complex with respect to \mathcal{P} in the sense that $\Omega^{\mathbb{B}}(\mathcal{P}\cdot,\cdot) = \Omega^{\mathbb{B}}(\cdot,\mathcal{P}\cdot) = \tau\Omega^{\mathbb{B}}(\cdot,\cdot)$.

Theorem C (baby version). We have

$$\mathcal{M}^*(\mathcal{P}, 4\Omega^{\mathbb{B}}) = (\mathbf{J}, \omega_{\mathbf{J}}^{\mathbb{B}}) .$$

where \mathcal{P} denotes the para-complex structure of $\mathcal{J}(\mathbb{R}^2) \times \mathcal{J}(\mathbb{R}^2)$ and $\Omega^{\mathbb{B}}$ its para-complex symplectic form.

Proof. Since B and JB are traceless and det J = 1, we have $B^2 = (JB)^2 = -\det B \, \mathbb{1}$. We will make use of this relation all along the current proof. A simple consequence is the following:

$$(\mathbb{1} \mp JB)^{-1} = \frac{1}{1 + \det B} (\mathbb{1} \pm JB).$$
 (3.16)

Applying this relation, we can develop the left and right complex structures J_l and J_r (the left and right components of \mathcal{M} , respectively) as follows:

$$J_{l,r} = (\mathbb{1} \mp JB)^{-1}J(\mathbb{1} \mp JB)$$

$$= \frac{1}{1 + \det B}(\mathbb{1} \pm JB)J(\mathbb{1} \mp JB)$$

$$= \frac{1}{1 + \det B}(J \pm JBJ \mp J^2B - JBJ^2B)$$

$$= \frac{1 - \det B}{1 + \det B}J \pm \frac{2}{1 + \det B}B.$$

In these relations and the ones that will follow the upper sign in \pm or \mp always refers to J_l and the lower sign to J_r . From the second identity in (3.15), we get:

$$\frac{2}{1 + \det B} = 1 + f, \qquad \frac{1 - \det B}{1 + \det B} = f , \qquad (3.17)$$

where $f = f(\|\sigma\|)$. Combining these relations with the development above, we obtain:

$$J_l = f J + g_J^{-1} \sigma, \qquad J_r = f J - g_J^{-1} \sigma.$$
 (3.18)

We are interested in computing the first order variation of these expressions along a direction $(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$. First we compute the derivative of the function $f = f(\|\sigma\|)$:

$$f' = \left(\sqrt{1 + \|\sigma\|^2}\right)'$$

$$= \frac{(\|\sigma\|^2)'}{2\sqrt{1 + \|\sigma\|^2}}$$

$$\begin{split} &= \frac{1}{2f} \operatorname{tr} \left(g_J^{-1} \sigma (g_J^{-1} \sigma)' \right) \\ &= \frac{1}{2f} \operatorname{tr} \left(g_J^{-1} \sigma (-g_J^{-1} \dot{g}_J g_J^{-1} \sigma + g_J^{-1} \dot{\sigma}) \right) \end{split}$$

As already observed in relation (3.3), the first order variation of g_J with respect to \dot{J} is equal to $-g_J(\cdot,J\dot{J}\cdot)$. Therefore Lemma 3.9 part ii implies that the trace of the product $g_J^{-1}\sigma g_J^{-1}\dot{g}_Jg_J^{-1}\sigma$ vanishes. Moreover, in the term $\mathrm{tr}(g_J^{-1}\sigma g_J^{-1}\dot{\sigma})$ there is no contribution from the trace part of $\dot{\sigma}$. In conclusion, we deduce that $f'=\frac{\langle \sigma,\dot{\sigma}_0\rangle}{f}$. For convenience, we also set

$$Q^{\pm} = Q^{\pm}(\dot{J}, \dot{\sigma}) := \frac{1}{f} g_J^{-1} \dot{\sigma}_0 \pm \dot{J}.$$

Then we have

$$\begin{split} \dot{J}_{l,r} &= (f J \pm g_J^{-1} \sigma)' \\ &= \pm f \, Q^{\pm} \pm \frac{1}{2} \operatorname{tr}_{g_J} \, \dot{\sigma} \, \mathbb{1} \mp g_J^{-1} \dot{g}_J g_J^{-1} \sigma + f' \, J \\ &= \pm f \, Q^{\pm} \mp \left\langle \sigma \mid J \dot{J} \right\rangle \, \mathbb{1} \pm J \dot{J} g_J^{-1} \sigma + \frac{\left\langle \sigma, \dot{\sigma}_0 \right\rangle}{f} \, J \\ &= \pm f \, Q^{\pm} \pm \left\langle \sigma \mid \dot{J} \right\rangle J + \frac{\left\langle \sigma \mid g_J^{-1} \dot{\sigma}_0 \right\rangle}{f} \, J \\ &= \pm f \, Q^{\pm} + \left\langle \sigma \mid Q^{\pm} \right\rangle J, \end{split} \tag{Lemma 3.9}$$

where $\langle \cdot, \cdot \rangle = \langle \cdot, \cdot \rangle_J$ and $\langle \cdot | \cdot \rangle = \langle \cdot | \cdot \rangle_J$. Then the differential of the map \mathcal{M} can be expressed as follows:

$$d\mathcal{M}_{(J,\sigma)}(\dot{J},\dot{\sigma}) = (\langle \sigma \mid Q^+ \rangle J + f Q^+, \langle \sigma \mid Q^- \rangle J - f Q^-).$$

Let us now determine the pull-back of the forms $\pi_l^*\Omega \pm \pi_r^*\Omega$ by the map \mathcal{M} , where Ω was defined in Remark 3.7. To emphasize the dependence on (J, σ) , we will now write $J_l = (\pi_l \circ \mathcal{M})(J, \sigma)$ and $J_r = (\pi_r \circ \mathcal{M})(J, \sigma)$. Given $(\dot{J}, \dot{\sigma})$ and $(\dot{J}', \dot{\sigma}')$ two tangent vectors at (J, σ) , we also set $Q^{\pm} = Q^{\pm}(\dot{J}, \dot{\sigma})$ and $R^{\pm} = Q^{\pm}(\dot{J}', \dot{\sigma}')$, to simplify the notation. Then, making use of the fact that $\operatorname{tr}(Q^{\pm}) = \operatorname{tr}(R^{\pm}) = 0$ and of Lemma 3.9, we deduce that

$$(\pi_{l,r} \circ \mathcal{M})^* \Omega((\dot{J}, \dot{\sigma}), (\dot{J}', \dot{\sigma}')) = -\langle \operatorname{d}(\pi_{l,r} \circ \mathcal{M}) (\dot{J}, \dot{\sigma}), J_{l,r} \cdot \operatorname{d}(\pi_{l,r} \circ \mathcal{M}) (\dot{J}', \dot{\sigma}') \rangle_{J_{l,r}}$$

$$= -\frac{1}{2} \operatorname{tr} \left((\langle \sigma \mid Q^{\pm} \rangle J \pm f Q^{\pm}) (fJ \pm g_J^{-1} \sigma) (\langle \sigma \mid R^{\pm} \rangle J \pm f R^{\pm}) \right)$$

$$= -\frac{f}{2} \operatorname{tr} \left(\langle \sigma \mid Q^{\pm} \rangle J g_J^{-1} \sigma R^{\pm} + \langle \sigma \mid R^{\pm} \rangle Q^{\pm} g_J^{-1} \sigma J \right) +$$

$$-\frac{f^3}{2} \operatorname{tr} \left(Q^{\pm} J R^{\pm} \right)$$

$$= -f \left(\langle g_J^{-1} \sigma, Q^{\pm} \rangle \langle J g_J^{-1} \sigma, R^{\pm} \rangle - \langle g_J^{-1} \sigma, R^{\pm} \rangle \langle Q^{\pm}, J g_J^{-1} \sigma \rangle \right) +$$

$$-f^3 \langle Q^{\pm}, J R^{\pm} \rangle$$

Now, because $Q^{\pm}, R^{\pm} \in T_I \mathcal{J}(\mathbb{R}^2)$, assuming $\sigma \neq 0$, we can write

$$Q^{\pm} = \frac{1}{\|\sigma\|} \left(\left\langle g_J^{-1} \sigma, Q^{\pm} \right\rangle g_J^{-1} \sigma + \left\langle J g_J^{-1} \sigma, Q^{\pm} \right\rangle J g_J^{-1} \sigma \right),$$

and similarly for R^{\pm} , so that the term

$$\langle g_J^{-1}\sigma, Q^{\pm}\rangle \langle Jg_J^{-1}\sigma, R^{\pm}\rangle - \langle g_J^{-1}\sigma, R^{\pm}\rangle \langle Q^{\pm}, Jg_J^{-1}\sigma\rangle$$
 (3.19)

coincides with $-\|\sigma\|^2 \langle Q^{\pm}, JR^{\pm} \rangle$ by direct computation. Note that when $\sigma = 0$, the expression (3.19) vanishes. In any case, recalling that $Q^{\pm} = Q^{\pm}(\dot{J}, \dot{\sigma})$ and $R^{\pm} = Q^{\pm}(\dot{J}', \dot{\sigma}')$, we can thus conclude that

$$(\pi_{l,r} \circ \mathcal{M})^* \Omega((\dot{J}, \dot{\sigma}), (\dot{J}', \dot{\sigma}')) = -f \langle Q^{\pm}, JR^{\pm} \rangle (-\|\sigma\|^2 + f^2)$$

$$= -\frac{f}{2} \operatorname{tr}(Q^{\pm} J R^{\pm}) \qquad (f^2 = 1 + \|\sigma\|^2)$$

$$= -f \langle \dot{J}, J\dot{J}' \rangle_J + \frac{1}{f} \langle \dot{\sigma}_0, \dot{\sigma}'_0(\cdot, J \cdot) \rangle_J \mp (\langle \dot{\sigma}_0 \mid J\dot{J}' \rangle_J - \langle \dot{\sigma}'_0 \mid J\dot{J} \rangle_J)$$

$$= (\omega_{\mathbf{I}} \pm \omega_{\mathbf{K}})((\dot{J}, \dot{\sigma}), (\dot{J}', \dot{\sigma}')) .$$

Therefore we have

$$\mathcal{M}^*(\pi_l^*\Omega + \pi_r^*\Omega) = 2\omega_{\mathbf{I}}, \qquad \mathcal{M}^*(\pi_l^*\Omega - \pi_r^*\Omega) = 2\omega_{\mathbf{K}}.$$

By what observed in Remark 3.12, this proves that $\mathcal{M}^*(4\Omega^{\mathbb{B}}) = \omega_{\mathbf{J}}^{\mathbb{B}}$.

One can then check directly that $d\mathcal{M}_{(J,\sigma)} \circ \mathbf{J} = \mathcal{P} \circ d\mathcal{M}_{(J,\sigma)}$, using that $Q^{\pm} \circ \mathbf{J}(\dot{J},\dot{\sigma}) = \pm Q^{\pm}(\dot{J},\dot{\sigma})$. However, the fact that $\mathcal{M}^*\mathcal{P} = \mathbf{J}$ also follows immediately by the same trick as in Corollary 3.10. Indeed, since $\Omega^{\mathbb{B}}$ is para-complex with respect to \mathcal{P} , if we denote $\hat{\mathbf{J}} = \mathcal{M}^*\mathcal{P}$, then $\omega_{\mathbf{J}}^{\mathbb{B}}$ is para-complex with respect to $\hat{\mathbf{J}}$, which means that $\omega_{\mathbf{K}}(\cdot, \cdot) = \omega_{\mathbf{I}}(\cdot, \hat{\mathbf{J}}\cdot)$. But since the same holds for \mathbf{J} , we deduce that $\hat{\mathbf{J}} = \mathbf{J}$.

As an immediate consequence, we obtain:

Corollary 3.16. The 2-forms $\omega_{\mathbf{I}}$ and $\omega_{\mathbf{K}}$ on $T^*\mathcal{J}(\mathbb{R}^2)$ are closed, and \mathbf{J} is an integrable almost para-complex structure.

3.7. The circle action

Let us now study the behavior of a natural circle action on $T^*\mathcal{J}(\mathbb{R}^2)$ with respect to the para-hyperKähler structure $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$. Recall that a cotangent vector $\sigma \in T_J^*\mathcal{J}(\mathbb{R}^2)$ can be seen as the real part of a complex-valued, J-complex linear symmetric form ϕ . By Remark 2.5, multiplication by $e^{i\theta}$ on complex bi-linear forms induces a circle action on $T^*\mathcal{J}(\mathbb{R}^2)$ given by

$$e^{i\theta}\cdot(J,\sigma)=\left(J,\cos(\theta)\sigma+\sin(\theta)\sigma(\cdot,J\cdot)\right)\,.$$

We will denote by $R_{\theta}: T^*\mathcal{J}(\mathbb{R}^2) \to T^*\mathcal{J}(\mathbb{R}^2)$ the action of $e^{i\theta}$. Observe that clearly R_{θ} preserves the zero section, hence it induces a circle action on $\mathcal{MGH}(T^2)$, which is identified

to the complement of the zero section in $T^*\mathcal{J}(\mathbb{R}^2)$ as in the proof of the genus one version of Theorem A in Section 3.5.

Theorem E (genus one). The circle action on $T^*\mathcal{J}(\mathbb{R}^2)$ is Hamiltonian with respect to $\omega_{\mathbf{I}}$, generated by the function $H(J,\sigma) = f(\|\sigma\|)$ and satisfies

$$R_{\theta}^* \mathbf{g} = \mathbf{g}$$
 $R_{\theta}^* \omega_{\mathbf{I}} = \omega_{\mathbf{I}}$ $R_{\theta}^* \omega_{\mathbf{I}}^{\mathbb{C}} = e^{-i\theta} \omega_{\mathbf{I}}^{\mathbb{C}}$.

Proof. The infinitesimal generator of the circle action is

$$V_{\theta}(J,\sigma) = \frac{d}{d\theta}(J,\cos(\theta)\sigma + \sin(\theta)\sigma(\cdot,J\cdot))\Big|_{\theta=0} = (0,\sigma(\cdot,J\cdot)).$$

We now compute

$$\begin{split} \iota_{V_{\theta}}\omega_{\mathbf{I}}(\dot{J},\dot{\sigma}) &= -\omega_{\mathbf{I}}((\dot{J},\dot{\sigma}),(0,\sigma(\cdot,J\cdot))) = -\mathbf{g}((\dot{J},\dot{\sigma}),\mathbf{I}(0,\sigma(\cdot,J\cdot))) \\ &= \mathbf{g}((\dot{J},\dot{\sigma}),(0,\sigma(\cdot,J^2\cdot))) = -\mathbf{g}((\dot{J},\dot{\sigma}),(0,\sigma)) \\ &= \frac{1}{f(\|\sigma\|)} \langle \dot{\sigma}_0,\sigma \rangle = \frac{1}{f(\|\sigma\|)} \langle \dot{\sigma},\sigma \rangle = dH_{(J,\sigma)}(\dot{J},\dot{\sigma}) \;, \end{split}$$

which proves the first statement. This immediately implies that $R_{\theta}^*\omega_{\mathbf{I}} = \omega_{\mathbf{I}}$, by Cartan's magic formula.

We now compute $R_{\theta}^*\omega^{\mathbb{C}}$. Let us first find an expression for the differential of R_{θ} . By definition, we have

$$(dR_{\theta})_{(J,\sigma)}(\dot{J},\cos(\theta)\dot{\sigma}) = (\dot{J},\cos(\theta)\dot{\sigma} + \sin(\theta)\dot{\sigma}(\cdot,J\cdot) + \sin(\theta)\sigma(\cdot,\dot{J}\cdot)) . \tag{3.20}$$

By Lemma 3.2, we can write

$$\dot{\sigma} = \dot{\sigma}_0 - \langle \sigma \mid J\dot{J} \rangle_J g_J$$

so that

$$\dot{\sigma}(\cdot, J \cdot) = \dot{\sigma}_0(\cdot, J \cdot) + \langle \sigma \mid J \dot{J} \rangle_J \rho . \tag{3.21}$$

Moreover, by Lemma 3.9, we have

$$g_J^{-1}\sigma\dot{J} = \langle g_J^{-1}\sigma,\dot{J}\rangle_J \mathbb{1} - \langle Jg_J^{-1}\sigma,\dot{J}\rangle_J J$$

which implies that

$$\sigma(\cdot, \dot{J}\cdot) = \langle \sigma \mid \dot{J} \rangle_J g_J - \langle \sigma \mid J \dot{J} \rangle_J \rho . \tag{3.22}$$

Combining (3.21) and (3.22) with (3.20), we obtain that the differential of R_{θ} can be expressed as follows

$$(dR_{\theta})_{(J,\sigma)}(\dot{J},\dot{\sigma}) = (\dot{J},\cos(\theta)\dot{\sigma} + \sin(\theta)\dot{\sigma}_0(\cdot,J\cdot) + \sin(\theta)\langle\sigma\mid\dot{J}\rangle g_J) \ .$$

Hence,

$$(R_{\theta}^{*}\omega^{\mathbb{C}})((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) = \langle \cos(\theta)\dot{\sigma}'_{0} + \sin(\theta)\dot{\sigma}'_{0}J \mid \dot{J}\rangle - \langle \cos(\theta)\dot{\sigma}_{0} + \sin(\theta)\dot{\sigma}_{0}J \mid \dot{J}'\rangle + i(\langle \cos(\theta)\dot{\sigma}'_{0} + \sin(\theta)\dot{\sigma}'_{0}J \mid J\dot{J}\rangle - \langle \cos(\theta)\dot{\sigma}_{0} + \sin(\theta)\dot{\sigma}_{0}J \mid J\dot{J}'\rangle) = \cos(\theta)(\langle \dot{\sigma}'_{0} \mid \dot{J}\rangle - \langle \dot{\sigma}_{0} \mid \dot{J}'\rangle) + \sin(\theta)(\langle \dot{\sigma}'_{0} \mid J\dot{J}\rangle - \langle \dot{\sigma}_{0} \mid J\dot{J}'\rangle)$$

$$+ i(\cos(\theta)(\langle \dot{\sigma}'_0 \mid J\dot{J}\rangle - \langle \dot{\sigma}_0 \mid J\dot{J}'\rangle) - \sin(\theta)(\langle \dot{\sigma}'_0 \mid \dot{J}\rangle - \langle \dot{\sigma}_0 \mid \dot{J}'\rangle)$$

$$= (\cos(\theta)\omega_{\mathbf{J}} + \sin(\theta)\omega_{\mathbf{K}}) + i(\cos(\theta)\omega_{\mathbf{K}} - \sin(\theta)\omega_{\mathbf{J}})$$

$$= e^{-i\theta}\omega^{\mathbb{C}}((\dot{J}, \dot{\sigma}), (\dot{J}', \dot{\sigma}')).$$

Let us finally check that R_{θ} preserves the metric **g**. We denote by

$$(r_{\theta})_{(J,\sigma)}(\dot{J},\dot{\sigma}) = \cos(\theta)\dot{\sigma} + \sin(\theta)\dot{\sigma}_0(\cdot,J\cdot) + \sin(\theta)\langle\sigma\mid\dot{J}\rangle g_J$$

the second component of the differential of R_{θ} at the point (J, σ) . Moreover, we remark that $\|\sigma\|_J = \|\cos(\theta)\sigma + \sin(\theta)\sigma(\cdot, J\cdot)\|_J$, in other words the circle action preserves the scalar product on $T_J^*\mathcal{J}(\mathbb{R}^2)$. We can now compute

$$(R_{\theta}^{*}\mathbf{g})_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) = \mathbf{g}_{R_{\theta}(J,\sigma)}((\dot{J},(r_{\theta})_{(J,\sigma)}(\dot{J},\dot{\sigma})),(\dot{J}',(r_{\theta})_{(J,\sigma)}(\dot{J}',\dot{\sigma}')))$$

$$= f(\|\sigma\|_{J})\langle\dot{J},\dot{J}'\rangle_{J} - \frac{1}{f(\|\sigma\|_{J})}\langle\cos(\theta)\dot{\sigma}_{0} + \sin(\theta)\dot{\sigma}_{0}(\cdot,J\cdot),\cos(\theta)\dot{\sigma}'_{0} + \sin(\theta)\dot{\sigma}'_{0}(\cdot,J\cdot)\rangle_{J}$$

$$= f(\|\sigma\|_{J})\langle\dot{J},\dot{J}'\rangle_{J} - \frac{1}{f(\|\sigma\|_{J})}\langle\dot{\sigma}_{0},\dot{\sigma}'_{0}\rangle_{J}$$

$$= \mathbf{g}_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')).$$

This concludes the proof.

We then immediately obtain the relations of (1.2) for the pull-back of $\mathbf{I}, \mathbf{J}, \mathbf{K}$, which we re-write here for completeness.

$$R_{\theta}^* \mathbf{I} = \mathbf{I}$$
 $R_{\theta}^* \mathbf{J} = \cos(\theta) \mathbf{J} + \sin(\theta) \mathbf{K}$ $R_{\theta}^* \mathbf{K} = -\sin(\theta) \mathbf{J} + \cos(\theta) \mathbf{K}$. (3.23)

We now turn to the proof of the genus one case of Theorem F, concerning a para-Kähler potential for \mathbf{g} . We recall that a smooth function f is a para-Kähler potential for a para-Kähler structure (\mathbf{g}, \mathbf{P}) if $\omega_{\mathbf{P}} = (\tau/2)\overline{\partial}_{\mathbf{P}}\partial_{\mathbf{P}}f$. We will make use of the following identity, whose proof is similar to the Kähler case and is given in Appendix A, Lemma A.4.

$$2\tau \bar{\partial}_{\mathbf{P}} \partial_{\mathbf{P}} f = d(df \circ \mathbf{P}) \tag{3.24}$$

Theorem F (genus one). The function -4H is a para-Kähler potential for the para-Kähler structures (\mathbf{g}, \mathbf{J}) and (\mathbf{g}, \mathbf{K}) on $T^*\mathcal{J}(\mathbb{R}^2)$.

Proof. From the previous result we have:

$$\mathcal{L}_{V_{\theta}}(\omega_{\mathbf{J}} + i\omega_{\mathbf{K}}) = \frac{d}{d\theta} R_{\theta}^{*}(\omega_{\mathbf{J}} + i\omega_{\mathbf{K}}) \Big|_{\theta=0} = \frac{d}{d\theta} R_{\theta}^{*} \omega_{\mathbf{I}}^{\mathbb{C}} \Big|_{\theta=0} = -i\omega_{\mathbf{I}}^{\mathbb{C}} = -i\omega_{\mathbf{J}} + \omega_{\mathbf{K}}$$

which implies

$$\mathcal{L}_{V_{\theta}}\omega_{\mathbf{J}} = \omega_{\mathbf{K}}$$
 and $\mathcal{L}_{V_{\theta}}\omega_{\mathbf{K}} = -\omega_{\mathbf{J}}$.

Therefore,

$$2\tau \bar{\partial}_{\mathbf{J}}\partial_{\mathbf{J}}(-H) = -d(dH \circ \mathbf{J}) = -d(\iota_{V_{a}}\omega_{\mathbf{I}}(\mathbf{J}\cdot)) = -d(\iota_{V_{a}}\omega_{\mathbf{K}}) = -\mathcal{L}_{V_{a}}\omega_{\mathbf{K}} = \omega_{\mathbf{J}}$$

and similarly,

$$2\tau\bar{\partial}_{\mathbf{K}}\partial_{\mathbf{K}}(-H) = -d(dH \circ \mathbf{K}) = -d(\iota_{V_{\theta}}\omega_{\mathbf{I}}(\mathbf{K}\cdot)) = d(\iota_{V_{\theta}}\omega_{\mathbf{J}}) = \mathcal{L}_{V_{\theta}}\omega_{\mathbf{J}} = \omega_{\mathbf{K}},$$

which shows that (-4H) is a para-Kähler potential for **g** with respect to **J** and **K**. \square

3.8. A one-parameter family of maps

Using the circle action, we can define for every $\theta \in S^1$ the map

$$\mathcal{C}_{\theta} = \mathcal{C} \circ R_{\theta} : T^* \mathcal{J}(\mathbb{R}^2) \to \mathcal{J}(\mathbb{R}^2) \times \mathcal{J}(\mathbb{R}^2)$$
.

Remark 3.17. Using Remark 2.5, we observe that for $\theta = -\pi/2$ the map $\mathcal{C} = \mathcal{C}_{-\pi/2}$ has the expression

$$C(J,\sigma) := ((\mathbb{1} - B)^{-1}J(\mathbb{1} - B), (\mathbb{1} + B)^{-1}J(\mathbb{1} + B))$$

and is therefore a formal analogue of the parameterization

$$C: \mathcal{MGH}(\Sigma) \to \mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$$
,

of the deformation space $\mathcal{MGH}(\Sigma)$ by means of the induced metric on the surfaces of constant curvature -2.

Remark 3.18. Exactly as in Lemma 2.14, identifying $\mathcal{J}(\mathbb{R}^2)$ with $\mathcal{T}^{\mathfrak{c}}(T^2)$ (Lemma 3.11), we obtain a map

$$T^*\mathcal{J}(\mathbb{R}^2) \to \mathcal{T}^{\mathfrak{c}}(T^2) \times \mathcal{T}^{\mathfrak{c}}(T^2)$$

which can be interpreted as the map

$$\mathcal{H}_{\theta}(J,q) = (h_{(J,-e^{i\theta}q)}, h_{(J,e^{i\theta}q)}) ,$$

where h(J,q) denotes the unique complex structure on T^2 such that the identity is harmonic with respect to the flat metric g_J on $T^2 \cong \mathbb{R}^2/\mathbb{Z}^2$ on the source and to h(J,q) on the target.

Now, using the genus one versions of Theorem C and Theorem E (see also (3.23)), which have been proved above in this section, and the identity $C_{\theta} = \mathcal{M} \circ R_{\theta + \frac{\pi}{2}}$, which follows from (2.4), we see that

$$\mathcal{C}^*_{\theta}(\mathcal{P}, 4\Omega^{\mathbb{B}}) = \left(-\sin(\theta)\mathbf{J} + \cos(\theta)\mathbf{K}, \omega_{\mathbf{I}} - \tau(\cos(\theta)\omega_{\mathbf{J}} + \sin(\theta)\omega_{\mathbf{K}})\right).$$

As an immediate consequence, we have the following "baby versions" of Theorems D and G

Theorem D (baby version). We have

$$\mathcal{C}^*(\mathcal{P}, 4\Omega^{\mathbb{B}}) = (\mathbf{K}, \omega_{\mathbf{K}}^{\mathbb{B}}) ,$$

where \mathcal{P} denotes the para-complex structure of $\mathcal{J}(\mathbb{R}^2) \times \mathcal{J}(\mathbb{R}^2)$ and $\Omega^{\mathbb{B}}$ its para-complex symplectic form.

Theorem G (baby version). We have

$$\operatorname{Im} \mathcal{H}_{\theta}^{*}(2\Omega^{\mathbb{B}}) = -\operatorname{Re}(ie^{i\theta}\Omega_{T^{*}\mathcal{I}(\mathbb{R}^{2})}^{\mathbb{C}}).$$

where $\Omega^{\mathbb{B}}$ is the para-complex symplectic form of $\mathcal{J}(\mathbb{R}^2) \times \mathcal{J}(\mathbb{R}^2)$.

As a consequence, we obtain, among other things, a direct proof of the integrability of \mathbf{K} , thus completing Theorem 3.6.

Corollary 3.19. The 2-forms $\omega_{\mathbf{I}}$ and $\omega_{\mathbf{J}}$ on $T^*\mathcal{J}(\mathbb{R}^2)$ are closed, and \mathbf{K} is an integrable almost para-complex structure.

4. The general case: genus $\geqslant 2$

In the next two sections we give a proof of Theorem A in the general case of closed surfaces of genus ≥ 2 . In this section, we realize the deformation space of MGHC anti-de Sitter structures $\mathcal{MGH}(\Sigma)$ as the quotient by $\operatorname{Symp}_0(\Sigma,\rho)$ (the group of symplectomorphisms of (Σ,ρ) isotopic to the identity, see Section 2.5) of a set $\widetilde{\mathcal{MS}}_0(\Sigma,\rho)$ sitting inside an infinite dimensional space $T^*\mathcal{J}(\Sigma)$ that is formally endowed with a para-hyperKähler structure $(\mathbf{g},\mathbf{I},\mathbf{J},\mathbf{K})$. We then give a distribution inside $TT^*\mathcal{J}(\Sigma)$ that is preserved by \mathbf{I} , and \mathbf{K} and maps isomorphically to the tangent space to $\mathcal{MS}_0(\Sigma,\rho)$, thus deducing that these structures descend to the quotient. To this aim, we characterize tangent vectors in several different ways (Proposition K), that we prove are equivalent in Section 4.6. The proof of Theorem A is then completed in Section 5, where we show that the induced symplectic forms are non-degenerate and closed, generalizing the constructions seen for the toy model in Section 3.

4.1. The group of (Hamiltonian) symplectomorphisms and its Lie algebra

Let us fix a symplectic form ρ on Σ . By Cartan's formula for every vector field V on Σ we have

$$\mathcal{L}_{V}\rho = \iota_{V} d\rho + d(\iota_{V}\rho) = d(\iota_{V}\rho),$$

since $d\rho = 0$. Therefore, the flow of V acts by symplectomorphisms on (Σ, ρ) if and only if the 1-form $\iota_V \rho$ is closed. Hence we can define the Lie algebra of the group $\operatorname{Symp}_0(\Sigma, \rho)$ of symplectomorphisms of (Σ, ρ) isotopic to the identity as follows:

$$\mathfrak{S}(\Sigma,\rho) := \mathrm{Lie}(\mathrm{Symp}_0(\Sigma,\rho)) = \{ V \in \Gamma(T\Sigma) \mid \mathrm{d}(\iota_V \rho) = 0 \} \cong_{\rho} Z^1(\Sigma),$$

where in the last step we are using the identification between $\Gamma(T\Sigma)$ and $\Lambda^1(\Sigma) = \Gamma(T^*\Sigma)$ induced by ρ , and $Z^1(\Sigma)$ denotes the space of smooth closed 1-forms. A symplectomorphism ψ is Hamiltonian if there is an isotopy $\psi_{\bullet} : [0,1] \to \operatorname{Symp}_0(\Sigma,\rho)$, with $\psi_0 = \operatorname{id}$ and $\psi_1 = \psi$, and a smooth family of functions $H_t : \Sigma \to \mathbb{R}$ such that $\iota_{V_t}\rho = dH_t$, where V_t is the infinitesimal generator of the symplectomorphism ψ_t . We denote by $\operatorname{Ham}(\Sigma,\rho)$ the group

of Hamiltonian symplectomorphisms of (Σ, ρ) . This is a normal subgroup of Symp (Σ, ρ) and its Lie algebra is defined as

$$\mathfrak{H}(\Sigma, \rho) := \operatorname{Lie}(\operatorname{Ham}(\Sigma, \rho)) = \{ V \in \Gamma(T\Sigma) \mid \iota_V \rho \text{ exact} \} \cong_{\rho} B^1(\Sigma),$$

where $B^1(\Sigma)$ denote the space of smooth exact 1-forms on Σ .

We have the following non-degenerate pairings:

$$\langle \cdot \mid \cdot \rangle_{\mathfrak{S}} : \begin{array}{ccc} \Lambda^{1}(\Sigma) /_{B^{1}(\Sigma)} \times Z^{1}(\Sigma) & \longrightarrow & \mathbb{R} \\ & ([\alpha], \beta) & \longmapsto & \int_{\Sigma} \alpha \wedge \beta, \\ \\ \langle \cdot \mid \cdot \rangle_{\mathfrak{H}} : & \Lambda^{1}(\Sigma) /_{Z^{1}(\Sigma)} \times B^{1}(\Sigma) & \longrightarrow & \mathbb{R} \\ & ([\alpha], \beta) & \longmapsto & \int_{\Sigma} \alpha \wedge \beta. \end{array}$$

Since $Z^1(\Sigma) = \ker d$, the group $\Lambda^1(\Sigma)/Z^1(\Sigma)$ identifies with $B^2(\Sigma)$ through the differential map d. In particular we have

$$\Lambda^{1}(\Sigma) /_{B^{1}(\Sigma)} \subset \mathfrak{S}(\Sigma, \rho)^{*}, \quad B^{2}(\Sigma) \cong_{d} \Lambda^{1}(\Sigma) /_{Z^{1}(\Sigma)} \subset \mathfrak{H}(\Sigma, \rho)^{*}. \tag{4.1}$$

Observe that, for every tangent vector field V and for every 1-form α , we have

$$\iota_V \alpha \, \rho = \alpha \wedge \iota_V \rho. \tag{4.2}$$

In particular, if V is a Hamiltonian vector field, with $\iota_V \rho = dH$, then

$$\langle [\alpha] \mid dH \rangle_{\mathfrak{H}} = \int_{\Sigma} \alpha \wedge dH = \int_{\Sigma} H \, d\alpha = \int_{\Sigma} \alpha(V) \, \rho,$$
 (4.3)

for every $H \in \mathscr{C}^{\infty}(\Sigma)$ and $[\alpha] \in \Lambda^{1}(\Sigma)/Z^{1}(\Sigma)$.

4.2. The Teichmüller space as a symplectic quotient

Before treating the case of $\mathcal{MS}(\Sigma)$, we recall briefly how we can recover Teichmüller space as an infinite dimensional symplectic quotient. Most of the computations of this section can already be found in [Don03] and [Tra18]: we report them here for reference purposes.

We denote by P the $\mathrm{SL}(2,\mathbb{R})$ -principal bundle over (Σ,ρ) whose fibers are linear maps $F:\mathbb{R}^2\to T_p\Sigma$ that identify the area form ρ_p with the standard area form ρ_0 on \mathbb{R}^2 via pull-back. In other words, we require that $F^*\rho_p=dx\wedge dy$. The $\mathrm{SL}(2,\mathbb{R})$ -action is defined by $A\cdot(p,F)=(p,F\circ A^{-1})$.

Observe that any symplectomorphism ϕ of (Σ, ρ) naturally lifts to a diffeomorphism $\hat{\varphi}$ of the total space P, by setting

$$\hat{\varphi}(p,F) := (\varphi(p), d\varphi_n \circ F) \in P$$

for every $(p, F) \in P$. We now consider the bundle over Σ

$$P(\mathcal{J}(\mathbb{R}^2)) := P \times_{\mathrm{SL}(2,\mathbb{R})} \mathcal{J}(\mathbb{R}^2) = P \times \mathcal{J}(\mathbb{R}^2) /_{\mathrm{SL}(2,\mathbb{R})},$$

where $\mathrm{SL}(2,\mathbb{R})$ acts diagonally on the two factors. A section of $P(\mathcal{J}(\mathbb{R}^2)) \to \Sigma$ induces a complex structure J on Σ which is compatible with ρ , in the sense that $\rho(\cdot, J\cdot)$ is positive definite: this is defined on $T_p\Sigma$ by the endomorphism $F_p \circ J_p \circ F_p^{-1}$. Recalling that $\mathrm{SL}(2,\mathbb{R})$ acts on $\mathcal{J}(\mathbb{R}^2)$ by conjugation, one easily checks that this section J is well-defined, that is, if two pairs $((p,F),J_p)$ and $((p,F'),J'_p)$ differ by the diagonal action of $\mathrm{SL}(2,\mathbb{R})$, then they induce the same complex structure on $T_p\Sigma$. We will often confuse sections of $P(\mathcal{J}(\mathbb{R}^2)) \to \Sigma$ with complex structures J.

We will denote by g_J the Riemannian metric $\rho(\cdot, J\cdot)$. By construction, the area form of g_J is equal to ρ for every complex structure J as above. We set

$$\mathcal{J}(\Sigma) := \Gamma(\Sigma, P(\mathcal{J}(\mathbb{R}^2))).$$

Given $J \in \mathcal{J}(\Sigma)$, a tangent vector $\dot{J} \in T_J \mathcal{J}(\Sigma)$ identifies with a section of the pull-back vector bundle $J^*(T^vP(\mathcal{J}(\mathbb{R}^2))) \to \Sigma$, where $T^vP(\mathcal{J}(\mathbb{R}^2))$ denotes the vertical subbundle of $TP(\mathcal{J}(\mathbb{R}^2))$ with respect to the projection over Σ . In other words, \dot{J} is a section of $\operatorname{End}(T\Sigma)$ that satisfies $\dot{J}J + J\dot{J} = 0$. The space $\mathcal{J}(\Sigma)$ is formally an infinite dimensional symplectic manifold with symplectic form Ω_J given by

$$\Omega_J(\dot{J},\dot{J}') = -\frac{1}{2} \int_{\Sigma} \operatorname{tr}(\dot{J}J\dot{J}') \rho .$$

Definition 4.1. Let (X, ω) be a symplectic manifold, and assume that a Lie group G acts on (X, ω) by symplectomorphisms. We say that the action is Hamiltonian if there exists a smooth function $\mu \colon X \to \mathfrak{g}^*$ satisfying the following properties:

i) μ is Ad*-equivariant, i.e. for every $g \in G$ and $p \in X$ we have:

$$\mu_{q\cdot p} = \operatorname{Ad}^*(q)(\mu_p) = \mu_p \circ \operatorname{Ad}(q^{-1}) \in \mathfrak{g}^*;$$

ii) given $\xi \in \mathfrak{g}$, we denote by V_{ξ} the vector field of X generating the action of the 1-parameter subgroup generated by ξ , i.e. $V_{\xi}(p) := \frac{\mathrm{d}}{\mathrm{d}t} \exp(t\xi) \cdot p|_{t=0}$. Moreover, we set μ^{ξ} to be the function $p \mapsto \mu_{p}(\xi) \in \mathbb{R}$ on X. Then, for every $\xi \in \mathfrak{g}$ we have:

$$\mathrm{d}\mu^{\xi} = \iota_{V_{\xi}}\omega = \omega(V_{\xi},\cdot).$$

A map μ satisfying the properties above is called a moment map for the action of G on (X, ω) .

In the following, we denote by $K_J \in \mathscr{C}^{\infty}(\Sigma)$ the Gaussian curvature of the metric $g_J = \rho(\cdot, J\cdot)$.

Theorem 4.2 ([Don03], [Tra18]). Set $c := \frac{2\pi\chi(\Sigma)}{\operatorname{Area}(\Sigma,\rho)}$. Then the function

$$\mu: \mathcal{J}(\Sigma) \longrightarrow \mathfrak{H}(\Sigma, \rho)^*$$
 $J \longmapsto -2(K_J - c)\rho$

is a moment map for the action of $\operatorname{Ham}(\Sigma, \rho)$ on $(\mathcal{J}(\Sigma), \Omega)$.

Here, we are using the inclusion $B^2(\Sigma) \subset \mathfrak{H}(\Sigma,\rho)^*$ introduced in (4.1). By property i) in Definition 4.1, the preimage $\mu^{-1}(0)$ is invariant by the action of the Hamiltonian group $\operatorname{Ham}(\Sigma,\rho)$. Consequently, any variation $\dot{J}=\mathcal{L}_V J$ induced by a Hamiltonian vector field V lies in the kernel of $\mathrm{d}\mu$. By property ii) in Definition 4.1, for any $J\in\mu^{-1}(0)$ the space $\operatorname{Ker}(\mathrm{d}\mu_J)$ coincides with the Ω_J -orthogonal of the tangent space $T_J(\operatorname{Ham}(\Sigma,\rho)\cdot J)$ to the orbit of J under the action of the Hamiltonian group. Therefore, there is a well-defined induced symplectic form on the quotient $\widetilde{\mathcal{T}}(\Sigma) = \mu^{-1}(0)/\operatorname{Ham}(\Sigma,\rho)$. The classical Teichmüller space $\mathcal{T}(\Sigma)$ can be identified (see [Don03, §2.2]) with the further quotient of $\widetilde{\mathcal{T}}(\Sigma)$ by

$$H := \operatorname{Symp}_0(\Sigma, \rho) / \operatorname{Ham}(\Sigma, \rho)$$
,

as briefly sketched at the end of Section 2.5. Because the orbits of H are symplectic submanifolds of $\tilde{\mathcal{T}}(\Sigma)$ (see [Don03, §2.2], [Tra18, Lemma 4.4.8]), we can define a symplectic form on $\mathcal{T}(\Sigma)$ by setting

$$\widehat{\Omega}_{[J]}([\dot{J}],[\dot{J}']) := \Omega_J(\dot{J}^h,\dot{J}'^h) \tag{4.4}$$

where $\dot{J}^h \in \operatorname{Ker}(\mathrm{d}\mu)$ denotes a lift of \dot{J} that is Ω_J -orthogonal to the orbit of $\operatorname{Symp}_0(\Sigma, \rho)$. These lifts can be described by a differential geometric property. For this purpose, we introduce the notion of divergence of an endomorphism: given $A \in \operatorname{End}(T\Sigma)$ and given G a Riemannian metric on Σ , we define $\operatorname{div}_G A$ to be the 1-form

$$(\operatorname{div}_G A)(X) := \sum_i G((\nabla_{e_i}^G A)X, e_i),$$

where $(e_i)_i$ is a local G-orthonormal frame, ∇^G is the Levi-Civita connection of G and X is a vector field on Σ . We will also denote by $\operatorname{div}_G V$ the usual divergence of a vector field V with respect to the Riemannian metric G. Whenever we are dealing with a fixed almost complex structure J, we will omit the dependence of metric $g_J = \rho(\cdot, J \cdot)$ on J and simply write g, in order to simplify the notation. In particular, if we write $\operatorname{div}_g A$, it has to be interpreted as the divergence of the endomorphism A with respect to g_J . Moreover, because J is ∇^g -parallel, i. e. $(\nabla^g_X J)Y = 0$ for every X, Y tangent vector fields on Σ , we deduce

$$\operatorname{div}_{a}(JA) = -\operatorname{div}_{a}(AJ) = -(\operatorname{div}_{a}A) \circ J \tag{4.5}$$

for any $A \in \text{End}(T\Sigma)$. Another immediate relation that we will use repeatedly is the following:

$$\operatorname{div}_{q}(X) = d(\iota_{X}\rho)(v, Jv) \tag{4.6}$$

for any unit vector v.

Proposition 4.3. Let J be in $\mu^{-1}(0) \subset \mathcal{J}(\Sigma)$. An element $\dot{J} \in T_J \mathcal{J}(\Sigma)$ lies in the kernel of $\mathrm{d}\mu$ if and only if $\mathrm{div}_g \dot{J}$ is a closed 1-form. Moreover, $\dot{J} \in \mathrm{Ker}(\mathrm{d}\mu)$ is Ω_J -orthogonal to $T_J(\mathrm{Symp}_0(\Sigma, \rho) \cdot J)$ inside $\mathrm{Ker}(\mathrm{d}\mu)$ if and only if $\mathrm{div}_g \dot{J}$ is an exact 1-form.

Proof. Let V be a vector field on Σ . We observe that

$$\frac{1}{2}\operatorname{tr}(\dot{J}J\mathcal{L}_VJ) = (\operatorname{div}_g\dot{J})(V) - \operatorname{div}_g(\dot{J}V) . \tag{4.7}$$

To see this, first we notice that

$$(\operatorname{div}_{g} \dot{J})(V) = \sum_{i} g((\nabla_{e_{i}} \dot{J})V, e_{i})$$

$$= \sum_{i} g(\nabla_{e_{i}} (\dot{J}V) - \dot{J}\nabla_{e_{i}}V, e_{i})$$

$$= \operatorname{div}_{g} (\dot{J}V) - \sum_{i} g(\dot{J}\nabla_{e_{i}}V, e_{i})$$

$$= \operatorname{div}_{g} (\dot{J}V) - \operatorname{tr}(\dot{J}A_{V}),$$

where ∇ denotes the Levi-Civita connection of g and A_V stands for the endomorphism $A_V(X) := \nabla_X V$. As shown in the proof of Lemma 4.19 below, the endomorphism $\mathcal{L}_V J$ can be expressed as $JA_V - A_V J$ (see relation (4.13)). In particular we have

$$\operatorname{tr}(\dot{J}A_{V}) = -\operatorname{tr}\left(\dot{J}JJA_{V}\right) \qquad (J^{2} = -1)$$

$$= -\frac{1}{2}\left(\operatorname{tr}\left(\dot{J}JJA_{V}\right) - \operatorname{tr}\left(J\dot{J}JA_{V}\right)\right) \qquad (\dot{J} \in T_{J}\mathcal{J}(\Sigma))$$

$$= -\frac{1}{2}\left(\operatorname{tr}\left(\dot{J}JJA_{V}\right) - \operatorname{tr}\left(\dot{J}JA_{V}J\right)\right)$$

$$= -\frac{1}{2}\operatorname{tr}\left(\dot{J}J\mathcal{L}_{V}J\right),$$

and so relation (4.7) follows. Now, applying such identity we find

$$\Omega_{J}(\dot{J}, \mathcal{L}_{V}J) = -\frac{1}{2} \int_{\Sigma} \operatorname{tr}(\dot{J}J\mathcal{L}_{V}J) \rho$$

$$= -\int_{\Sigma} \left(\frac{1}{2} \operatorname{tr}(\dot{J}J\mathcal{L}_{V}J) + \operatorname{div}_{g}(\dot{J}V)\right) \rho$$

$$= -\int_{\Sigma} (\operatorname{div}_{g}\dot{J})(V) \rho$$

$$= -\int_{\Sigma} (\operatorname{div}_{g}\dot{J}) \wedge \iota_{V}\rho.$$
(4.8)

Consider now \dot{J} in kernel of the differential of the moment map $d\mu$. By property ii) in Definition 4.1, we have $\Omega_J(\dot{J}, \mathcal{L}_V J) = 0$ for every Hamiltonian vector field V. If H denotes the Hamiltonian function of V, then by relation (4.8) we have

$$0 = \int_{\Sigma} (\operatorname{div}_{g} \dot{J}) \wedge dH = \int_{\Sigma} H d(\operatorname{div}_{g} \dot{J}),$$

where in the last step we applied Stokes' theorem on the 2-form $d(H(\operatorname{div}_g \dot{J}))$. Therefore, by letting the Hamiltonian function H vary, we deduce that $\operatorname{div}_q \dot{J}$ is a closed 1-form.

Similarly, if $\operatorname{div}_q \dot{J} = \operatorname{d} f$ is exact and $V \in \mathfrak{S}(\Sigma, \rho)$, then

$$df \wedge \iota_V \rho = df \wedge \iota_V \rho + f d(\iota_V \rho) = d(f \iota_V \rho) \qquad (\iota_V \rho \text{ is closed})$$

is exact and $\Omega_J(\dot{J}, \mathcal{L}_V J) = 0$ for every $V \in \mathfrak{S}(\Sigma, \rho)$. Vice versa, assume that \dot{J} is in Ker d μ and it satisfies $\Omega_J(\dot{J}, \mathcal{L}_V J) = 0$ for every $V \in \mathfrak{S}(\Sigma, \rho)$. Then, again by relation (4.8),

$$\int_{\Sigma} (\operatorname{div}_g \dot{J}) \wedge \alpha = 0$$

for every closed 1-form α . Since div \dot{J} is closed, and since the pairing

$$(\alpha,\beta) \mapsto \int_{\Sigma} \alpha \wedge \beta$$

between closed 1-forms is non-degenerate in $H^1(\Sigma) \times H^1(\Sigma)$, we deduce that $\operatorname{div}_g \dot{J}$ represents the trivial class inside the first de Rham cohomology group or, in other words, that $\operatorname{div}_g \dot{J}$ is exact.

Remark 4.4. The argument described in the proof of Proposition 4.3 combined with Theorem 4.2 provides us with a convenient way to express the first order variation of the curvature K_J with respect to \dot{J} , that is

$$dK_J(\dot{J}) \rho = \frac{1}{2} d(\operatorname{div}_g \dot{J}).$$

In the following, we briefly see how to deduce this relation. On the one hand, by the explicit expression of the moment map μ from Theorem 4.2, we have that

$$\langle \mathrm{d}\mu\left(\dot{J}\right) \mid V \rangle_{\mathfrak{H}} = -2 \int_{\Sigma} H \, \mathrm{d}K_{J}\left(\dot{J}\right) \rho$$

for any Hamiltonian vector field V with Hamiltonian function H. On the other hand, being μ a moment map for the action of the Hamiltonian group, it satisfies

$$\langle d\mu (\dot{J}) | V \rangle_{\mathfrak{H}} = \Omega_{J} (\dot{J}, \mathcal{L}_{V} J)$$

$$= -\int_{\Sigma} (\operatorname{div}_{g} \dot{J}) \wedge dH \qquad (\text{relation (4.8)})$$

$$= -\int_{\Sigma} H d(\operatorname{div}_{g} \dot{J}),$$

once again for any Hamiltonian vector field V with Hamiltonian function H. Combining the relations above, we find that

$$-2\int_{\Sigma} H \,\mathrm{d}K_{J}\,(\dot{J})\rho = -\int_{\Sigma} H \,\mathrm{d}\!\left(\mathrm{div}_{g}\,\dot{J}\right),$$

and by letting the Hamiltonian function vary, we deduce the desired relation.

Classically, one further re-normalizes the lift \dot{J}^h to be L^2 -orthogonal to the tangent space to the orbit. This gives the additional condition that $\operatorname{div}_g \dot{J} = 0$ ([Tra18], [Tro92]), which recovers the description of the tangent space to Teichmüller space via traceless Codazzi tensors and the formula of (a multiple of) the Weil-Petersson symplectic form, if we choose an area form ρ on Σ such that $\operatorname{Area}(\Sigma, \rho) = -2\pi\chi(\Sigma)$, which means that c = -1 in Theorem 4.2.

Lemma 4.5 ([BMS15, Section 2.1]). Let $\dot{J}, \dot{J}' \in T_{[J]}\mathcal{T}^{\mathfrak{c}}(\Sigma)$ be traceless Codazzi tensors representing tangent vectors to Teichmüller space. Then,

$$(\Omega_{WP})_{[J]}(\dot{J},\dot{J}') = -\frac{1}{8} \int_{\Sigma} \operatorname{tr}(\dot{J}J\dot{J}') da,$$

and the Weil-Petersson metric can be expressed as

$$(G_{WP})_{[J]}(\dot{J},\dot{J}') = \frac{1}{8} \int_{\Sigma} \operatorname{tr}(\dot{J}\dot{J}') da$$

where da is the volume form of the unique hyperbolic metric with conformal structure J.

Remark 4.6. The change in the sign with respect to the relation appearing in [BMS15, §2.1] is due to the fact that here we are considering the complex structure $\dot{\nu} \mapsto -J\dot{\nu}$ on the space of Beltrami differentials, which is opposite to the one used by [BMS15].

Remark 4.7. We remark, however, that any choice of a supplement W of $T_J(\operatorname{Symp}_0(\Sigma, \rho) \cdot J)$ inside $\operatorname{Ker}(d\mu)$ that is Ω_J -orthogonal to $T_J(\operatorname{Symp}(\Sigma, \rho) \cdot J)$ gives a well-defined model for the tangent space to $\mathcal{T}^{\mathfrak{c}}(\Sigma)$ with the property that $(W, \Omega_J|_W)$ is symplectomorphic to $(T_J\mathcal{T}^{\mathfrak{c}}(\Sigma), 4\Omega_{WP})$.

4.3. The construction of $\mathcal{MS}_0(\Sigma, \rho)$

Let us now consider the bundle over Σ defined by

$$P(T^*\mathcal{J}(\mathbb{R}^2)) := P \times_{\mathrm{SL}(2,\mathbb{R})} T^*\mathcal{J}(\mathbb{R}^2) = P \times T^*\mathcal{J}(\mathbb{R}^2) /_{\mathrm{SL}(2,\mathbb{R})},$$

where P is the frame bundle introduced in Section 4.2 and $\mathrm{SL}(2,\mathbb{R})$ acts diagonally. The fiber of $P(T^*\mathcal{J}(\mathbb{R}^2))$ over the point $p \in \Sigma$ identifies with $T^*\mathcal{J}(T_p\Sigma)$, i. e. the space of pairs (J_p, σ_p) where J_p is an almost complex structure of $T_p\Sigma$ compatible with ρ_p , and σ_p is a g_{J_p} -traceless and symmetric bilinear form on $T_p\Sigma$ that satisfies $\sigma_p(J_p, J_p) = -\sigma_p$. Since the para-hyperKähler structure of $T^*\mathcal{J}(\mathbb{R}^2)$ is $\mathrm{SL}(2,\mathbb{R})$ -invariant (see Theorem 3.6), each fiber $T^*\mathcal{J}(T_p\Sigma)$ is naturally endowed with a para-hyperKähler structure $(\hat{\mathbf{g}}_p, \hat{\mathbf{I}}_p, \hat{\mathbf{J}}_p, \hat{\mathbf{K}}_p)$, obtained by identifying $T_p\Sigma$ with \mathbb{R}^2 using an area-preserving isomorphism $F_p: T_p\Sigma \to \mathbb{R}^2$.

The space of smooth sections

$$T^*\mathcal{J}(\Sigma) := \Gamma(P(T^*\mathcal{J}(\mathbb{R}^2))$$

can be identified with the set of pairs (J, σ) , where J is a complex structure on Σ , and σ is a symmetric and g-traceless 2-tensor, where $g = g_J = \rho(\cdot, J \cdot)$. The element σ can be equivalently characterized as the real part of a complex valued J-complex linear and symmetric

2-tensor ϕ . We identify the tangent space of $T^*\mathcal{J}(\Sigma)$ at (J,σ) with the space of sections of the vector bundle $(J,\sigma)^*(T^vP(T^*\mathcal{J}(\mathbb{R}^2))) \to \Sigma$, where $T^vP(T^*\mathcal{J}(\mathbb{R}^2))$ stands for the vertical sub-bundle of $TP(T^*\mathcal{J}(\mathbb{R}^2))$ with respect to the projection map $P(T^*\mathcal{J}(\mathbb{R}^2)) \to \Sigma$. In particular, we can consider a tangent vector $(\dot{J},\dot{\sigma})$ at (J,σ) as the data of (see Lemma 3.2):

- a section \dot{J} of End $(T\Sigma)$ satisfying $\dot{J}J + J\dot{J} = 0$. In other words, \dot{J} is a g-self-adjoint and traceless endomorphism of $T\Sigma$;
- a symmetric 2-tensor $\dot{\sigma}$ satisfying

$$\dot{\sigma} = \dot{\sigma}_0 - \langle \sigma \mid J\dot{J} \rangle g,$$

where $\dot{\sigma}_0$ is a symmetric and g-traceless 2-tensor. Observe in particular that the g-full trace part of $\dot{\sigma}$ is uniquely determined by \dot{J} .

Formally, $T^*\mathcal{J}(\Sigma)$ is an infinite dimensional para-hyperKähler manifold, where the symplectic forms are defined as

$$(\omega_{\mathbf{X}})_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) := \int_{\Sigma} \hat{\omega}_{\mathbf{X}}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) \rho, \tag{4.9}$$

for $\mathbf{X} = \mathbf{I}, \mathbf{J}, \mathbf{K},$ and the pseudo-Riemannian metric is given by

$$\mathbf{g}_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) := \int_{\Sigma} \hat{\mathbf{g}}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) \rho , \qquad (4.10)$$

where $\hat{\omega}$ and $\hat{\mathbf{g}}$ denote the symplectic form and the pseudo-Riemannian metric obtained by identifying the fibers of $P(T^*\mathcal{J}(\mathbb{R}^2)) \to \Sigma$ with the space of linear almost-complex structures on $T.\Sigma$ as described above. Similarly we have linear endomorphisms

$$\mathbf{I}, \mathbf{J}, \mathbf{K} : T_{(J,\sigma)}T^*\mathcal{J}(\Sigma) \longrightarrow T_{(J,\sigma)}T^*\mathcal{J}(\Sigma),$$

obtained by applying pointwisely the endomorphisms $\hat{\mathbf{I}}, \hat{\mathbf{J}}, \hat{\mathbf{K}}$ to a smooth section $(\dot{J}, \dot{\sigma})$. Their definition is formally identical to the ones in relations (3.6), (3.7), and (3.8), with the only difference that now J, σ , g_J , \dot{J} , $\dot{\sigma}$ are all tensors, and $f(\|\sigma\|)$ is a smooth function over Σ .

Remark 4.8. The expression of Mess homeomorphism introduced in Section 3.6 can be formally applied to define a map

$$\mathcal{M}: T^*\mathcal{J}(\Sigma) \longrightarrow \mathcal{J}(\Sigma) \times \mathcal{J}(\Sigma)$$
$$\mathcal{M}(J,\sigma) := \left((\mathbb{1} - JB)^{-1} J(\mathbb{1} - JB), (\mathbb{1} + JB)^{-1} J(\mathbb{1} + JB) \right) ,$$

which takes as input a almost complex structure J and a g_J -traceless symmetric 2-tensor σ , and provides a pair of almost complex structures on Σ . In what follows, we will denote by J_l and J_r the left and right components of \mathcal{M} , and a tangent vector at

$$T_{(J_l,J_r)}\mathcal{J}(\Sigma) \times \mathcal{J}(\Sigma) \cong T_{J_l}\mathcal{J}(\Sigma) \times T_{J_r}\mathcal{J}(\Sigma)$$

will be given by a pair (\dot{J}_l, \dot{J}_r) .

Remark 4.9. The notation introduced is intentionally abusive, to emphasize the similarities between the toy model $T^*\mathcal{J}(\mathbb{R}^2)$ and the infinite-dimensional manifold $T^*\mathcal{J}(\Sigma)$. In what follows, we will often make use of the relations proved in Section 3, which concern $T^*\mathcal{J}(\mathbb{R}^2)$, in the context of $T^*\mathcal{J}(\Sigma)$. These arguments are legitimate because identities at the level of the toy model can be interpreted as pointwise identities at the level of smooth sections inside $T^*\mathcal{J}(\Sigma)$.

4.4. The para-hyperKähler structure of $\mathcal{MS}_0(\Sigma, \rho)$

We will now give an explicit description of the tangent bundle of the space $\mathcal{MS}_0(\Sigma, \rho)$ (introduced in Section 2.5), which is well suited to present its para-hyperKähler structure.

We recall from Section 2.5 that $\mathcal{MS}_0(\Sigma, \rho)$ is the quotient of the infinite-dimensional manifold

$$\widetilde{\mathcal{MS}_0}(\Sigma,\rho) := \left\{ (J,\sigma) \middle| \begin{array}{l} g = \rho(\cdot,J\cdot) \text{ is a Riemannian metric on } \Sigma, \\ \sigma \text{ is the real part of a J-quadratic differential,} \\ (h = (1+f(\|\sigma\|_g))\,g, B = h^{-1}\sigma) \text{ satisfy (GC)} \end{array} \right\}$$

by the action of $\operatorname{Symp}_0(\Sigma,\rho)$, where as usual $f=f(\|\sigma\|_g)=\sqrt{1+\|\sigma\|_g^2}$. We will first introduce a very specific distribution $\mathcal{V}=\{V_{(J,\sigma)}\}_{(J,\sigma)}$ tangent to $\widetilde{\mathcal{MS}}_0(\Sigma,\rho)$, presenting several characterizations of it in Proposition K. The proof of the equivalence of these equivalent characterizations requires a certain amount of computations, and it will be postponed to Section 4.6. Theorem L will then describe the identification between $V_{(J,\sigma)}$ and the tangent space to $\mathcal{MS}_0(\Sigma,\rho)$ at the equivalence class of (J,σ) by the action of $\operatorname{Symp}_0(\Sigma,\rho)$. Its proof is fragmented into several lemmas, which will constitute the main technical core of Section 4.6.

Remark 4.10. The references to the process of infinite-dimensional symplectic reduction are limited to the case of the Teichmüller space, which has been described in Section 4.2. However, it is right and proper to acknowledge the reader that the definition itself of the vector space $V_{(J,\sigma)}$, together with the ideas behind the proofs of its properties, are all results of a deeper analysis developed in analogy to the hyperKähler symplectic reduction process of Donaldson [Don03] in our context of interest (see Section 6).

Proposition K. Given $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, and $(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$, the following conditions are equivalent:

i) the pair $(\dot{J}, \dot{\sigma})$ satisfies

$$\begin{cases} \operatorname{div}_{g}(f^{-1}g^{-1}\dot{\sigma}_{0}) = -f^{-1}\langle\nabla^{g}_{J\bullet}\sigma \mid \dot{J}\rangle, \\ \operatorname{div}_{g}\dot{J} = -f^{-2}\langle\nabla^{g}_{J\bullet}\sigma, \dot{\sigma}_{0}\rangle. \end{cases}$$
(V1)

ii) the endomorphisms $Q^{\pm} = Q^{\pm}(\dot{J}, \dot{\sigma}) := f^{-1}g^{-1}\dot{\sigma}_0 \pm \dot{J}$ satisfy

$$\begin{cases} \operatorname{div}_{g}(Q^{+}JJ_{l}) = -\langle \nabla^{g}_{J\bullet}\sigma \mid Q^{+}\rangle, \\ \operatorname{div}_{g}(Q^{-}JJ_{r}) = \langle \nabla^{g}_{J\bullet}\sigma \mid Q^{-}\rangle, \end{cases}$$
(V2)

where J_l and J_r denote the components of the Mess map \mathcal{M} ;

iii) the endomorphisms Q^{\pm} satisfy

$$\begin{cases} \operatorname{div}_{g} Q^{+} = -f^{-1} \langle \nabla_{J \bullet}^{g} \sigma \mid Q^{+} \rangle, \\ \operatorname{div}_{g} Q^{-} = +f^{-1} \langle \nabla_{J \bullet}^{g} \sigma \mid Q^{-} \rangle. \end{cases}$$
(V3)

Moreover, the 1-forms $\operatorname{div}_{h_l} \dot{J}_l$ and $\operatorname{div}_{h_r} \dot{J}_r$ are exact.

Definition 4.11. Given $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, we define $V_{(J,\sigma)}$ to be the subspace of $T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$ of those elements $(\dot{J}, \dot{\sigma})$ that satisfy one of (and therefore all) the conditions in Proposition K.

Theorem L. For every pair (J, σ) lying in $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, the vector space $V_{(J,\sigma)}$ is contained inside $T_{(J,\sigma)}\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, it is invariant by the action of \mathbf{I} , \mathbf{J} and \mathbf{K} , and it defines a Symp (Σ, ρ) -invariant distribution $\mathcal{V} = \{V_{(J,\sigma)}\}_{(J,\sigma)}$ on $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$. Moreover, the natural projection $\pi \colon \widetilde{\mathcal{MS}}_0(\Sigma, \rho) \to \mathcal{MS}_0(\Sigma, \rho)$ induces a linear isomorphism $d\pi_{(J,\sigma)} \colon V_{(J,\sigma)} \to T_{[J,\sigma]}\mathcal{MS}_0(\Sigma, \rho)$.

The proof of Theorem L is postponed to Section 4.5. Lemma 4.17 shows that $V_{(J,\sigma)}$ is tangent to the locus of those (J,σ) that satisfy the Gauss-Codazzi equations, which is precisely the definition of the subset $\widetilde{\mathcal{MS}}_0(\Sigma,\rho)$ of $T^*\mathcal{J}(\Sigma)$. The invariance of $V_{(J,\sigma)}$ by the action of \mathbf{I} , \mathbf{J} and \mathbf{K} is proved in Lemma 4.13 and the invariance of the distribution \mathcal{V} by $\mathrm{Symp}(\Sigma,\rho)$ follows from Lemma 4.12. We will show in Lemma 4.21 that the differential of the projection map π is injective. Finally, in Lemma 4.18 we show that the dimension of $V_{(J,\sigma)}$ is larger than or equal to the dimension of $\mathcal{MS}_0(\Sigma,\rho)$, which is $6|\chi(\Sigma)|$, and therefore we conclude that the differential of the projection π induces a linear isomorphism between $V_{(J,\sigma)}$ and the tangent space to $\mathcal{MS}_0(\Sigma,\rho)$.

We are now ready to summarize the proof of Theorem A in genus $g \ge 2$, although some of the steps of the proof will follow from the geometric interpretations that we provide in Section 5.

Theorem A (genus ≥ 2). Let Σ be a closed oriented surface of genus ≥ 2 . Then $\mathcal{MGH}(\Sigma)$ admits a $MCG(\Sigma)$ -invariant para-hyperKähler structure $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$. Moreover the Fuchsian locus $\mathcal{F}(\Sigma)$ is totally geodesic and (\mathbf{g}, \mathbf{I}) restricts to (a multiple of) the Weil-Petersson Kähler structure of Teichmüller space.

Proof. Identifying the tangent space $T_{[J,\sigma]}\mathcal{MS}_0(\Sigma,\rho)$ with $V_{(J,\sigma)}$ (Theorem L), we can define on $\mathcal{MS}_0(\Sigma,\rho)$ para-complex structures **J** and **K**, a complex structure **I**, and a pseudo-Riemannian metric **g** by restriction from the infinite dimensional space $T^*\mathcal{J}(\Sigma)$.

The definition is well-posed, namely it does not depend on the representative in a given $\operatorname{Symp}_0(\Sigma, \rho)$ -orbit, by the invariance statement in Theorem L and the invariance of \mathbf{g} , \mathbf{I} , \mathbf{J} and \mathbf{K} (which is proved immediately, with the same tools as in the proof of Lemma 4.12).

It is clear that \mathbf{I} , \mathbf{J} and \mathbf{K} are still compatible with \mathbf{g} and satisfy the para-quaternionic relations. We also have corresponding 2-forms $\omega_{\mathbf{I}}$, $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$. A priori the metric \mathbf{g} , and consequently the 2-forms $\omega_{\mathbf{I}}$, $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$ may be degenerate when restricted to $V_{(J,\sigma)}$ and thus on $\mathcal{MS}_0(\Sigma,\rho)$. We rule out this possibility in Section 5 by identifying these forms with well-known symplectic forms (therefore closed and non-degenerate) on $T^*\mathcal{T}^{\mathfrak{c}}(\Sigma)$ and on $\mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$. From the results of Section 5, we also obtain that $\omega_{\mathbf{I}}$, $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$ are integrable. See Corollaries 5.1, 5.2, 5.3 and 5.5 for all these statements. We can then conclude that the quadruple $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$ endows $\mathcal{MS}_0(\Sigma, \rho)$ with a para-hyperKähler structure.

The mapping class group invariance of the para-hyperKähler structure follows from the $\operatorname{Symp}(\Sigma, \rho)$ -invariance of Theorem L, since we have a natural isomorphism between $MCG(\Sigma)$ and $\operatorname{Symp}(\Sigma, \rho)/\operatorname{Symp}_0(\Sigma, \rho)$, and again the $\operatorname{Symp}(\Sigma, \rho)$ -invariance of \mathbf{g} , \mathbf{I} , \mathbf{J} and \mathbf{K} .

Observe that the Fuchsian locus of $\mathcal{MGH}(\Sigma)$ corresponds to the pairs (J, σ) with $\sigma = 0$. By Proposition K, its tangent space consists of the pairs $(\dot{J}, \dot{\sigma})$ with $\dot{\sigma} = 0$ and $\operatorname{div}_g \dot{J} = 0$, hence it corresponds precisely to the model of the tangent space of Teichmüller space that we described in Section 4.2 (see Lemma 4.5 and the preceding discussion). By comparing the expression of the Weil-Petersson metric in Lemma 4.5 with the restriction of the metric \mathbf{g} (see (3.5)), we see immediately that $\mathbf{g}|_{\mathcal{F}(\Sigma)}$ coincides with $4G_{WP}$. Finally, the Fuchsian locus is the set of fixed points of the circle action, that is isometric for the metric \mathbf{g} by Theorem E, which is proved in Section 5. By a standard argument, this implies that the Fuchsian locus is totally geodesic.

4.5. The proof of Theorem L

This subsection is dedicated to the proof of Lemmas 4.12, 4.13, 4.17, 4.18 and 4.21. Together, these results prove Theorem L to identify $V_{(J,\sigma)}$ with the tangent space to $\mathcal{MS}_0(\Sigma,\rho)$ at $[(J,\sigma)]$. The last of them is definitely the most challenging, and it requires some technical ingredients, which will be described along the way and which will be useful for the part concerning the ∞ -dimensional symplectic reduction (Section 6).

4.5.1. Proof of Lemma 4.12: invariance under symplectomorphisms.

Lemma 4.12. The distribution $\mathcal{V} = \{V_{(J,\sigma)}\}_{(J,\sigma)\in\widetilde{\mathcal{MS}_0}(\Sigma,\rho)}$ is invariant under the action of $\operatorname{Symp}(\Sigma,\rho)$. In other words, for every symplectomorphism φ of (Σ,ρ) and for every $(\dot{J},\dot{\sigma}) \in V_{(J,\sigma)}$, we have $(\varphi^*\dot{J},\varphi^*\dot{\sigma}) \in V_{(\varphi^*J,\varphi^*\sigma)}$.

Proof. The main point where the condition of symplectomorphism is essential concerns the metric $g_J = \rho(\cdot, J \cdot)$. The relation $\varphi^* \rho = \rho$ implies that $g_{\varphi^* J}$, the metric with area form

equal to ρ and complex structure φ^*J , is equal to φ^*g , the pull-back of the Riemannian metric $g = g_J$ by φ . This is clearly equivalent to say that $\varphi \colon (\Sigma, g_{\varphi^*J}) \to (\Sigma, g_J)$ is an isometry. In particular, for every endomorphism of the tangent bundle A we have that

$$\varphi^*(\operatorname{div}_q A) = \operatorname{div}_{\varphi^* q}(\varphi^* A).$$

Moreover, since the differential of φ is represented by an element in $SL(2,\mathbb{R})$ at every point, Equation (3.4) implies that the inner products and pairings of the tensors (defined in 3.2) are preserved by φ . The statement then follows from the naturality of the action and the expressions defining the subspace $V_{(J,\sigma)}$.

4.5.2. Proof of Lemma 4.13: Invariance under I, J and K

Lemma 4.13. For every $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, the subspace $V_{(J,\sigma)}$ is preserved by \mathbf{I} , \mathbf{J} and \mathbf{K} .

Proof. This is a simple consequence of the description of $V_{(J,\sigma)}$ provided by Proposition K, part i). Since $\mathbf{K} = \mathbf{IJ}$, it is enough to check that $\mathbf{I}(\dot{J},\dot{\sigma})$ and $\mathbf{J}(\dot{J},\dot{\sigma})$ belong to $V_{(J,\sigma)}$, whenever $(\dot{J},\dot{\sigma})$ is in $V_{(J,\sigma)}$.

The first component of $\mathbf{J}(\dot{J},\dot{\sigma})$ is equal to $f^{-1}g^{-1}\dot{\sigma}_0$, while the g-traceless part of the second component is equal to $fg(\cdot,\dot{J}\cdot)$. If we replace \dot{J} with $f^{-1}g^{-1}\dot{\sigma}_0$, and $\dot{\sigma}_0$ with $fg(\cdot,\dot{J}\cdot)$ in the equations (V1), we obtain the invariance of $V_{(J,\sigma)}$ under the action of \mathbf{J} . For the invariance under \mathbf{I} , we observe that, for every $(\dot{J},\dot{\sigma})$ in $V_{(J,\sigma)}$ we have

$$\operatorname{div}_{g}(-J\dot{J}) = \operatorname{div}_{g}(\dot{J}J)$$

$$= (\operatorname{div}_{g}\dot{J}) \circ J \qquad (\text{rel. } (4.5))$$

$$= -f^{-2}\langle \nabla^{g}_{J^{2}\bullet}\sigma, \dot{\sigma}_{0}\rangle$$

$$= -f^{-2}\langle \nabla^{g}_{J\bullet}\sigma \mid Jg^{-1}\dot{\sigma}_{0}\rangle \qquad (\text{Lemma } 2.3 \text{ part } iv))$$

$$= -f^{-2}\langle \nabla^{g}_{J\bullet}\sigma, -\dot{\sigma}_{0}(\cdot, J\cdot)\rangle.$$

This shows that $\mathbf{I}(\dot{J}, \dot{\sigma})$ satisfies the first equation in (V1). Arguing similarly for $\operatorname{div}_g(f^{-1}g^{-1}\dot{\sigma}_0 J)$, we obtain the invariance of $V_{(J,\sigma)}$ by \mathbf{I} .

4.5.3. Proof of Lemma 4.17: $V_{(J,\sigma)}$ lies in the kernel of the linearized GC-equations

Lemma 4.14. Every element $(\dot{J}, \dot{\sigma})$ of $V_{(J,\sigma)}$ lies in the kernel of the linearized Codazzi equation $d^{\nabla^h}B = 0$.

As observed in Lemma 2.3, the Codazzi equation is equivalent to the requirement that $g^{-1}\sigma$ is a g-Codazzi tensor. In order to prove the statement above, we need to compute the first order variation of the expression $\nabla_X^g(g^{-1}\sigma Y) - \nabla_Y^g(g^{-1}\sigma X) - g^{-1}\sigma[X,Y]$, for any pair of tangent vector fields X, Y, as we vary (J, σ) in the direction $(\dot{J}, \dot{\sigma})$. To simplify the

notation, we will denote by ∇ the Levi-Civita connection of $g = g_J$ and by $\dot{\nabla}$ its variation as we change the complex structure J.

Lemma 4.15. Let $\dot{J} \in T_J \mathcal{J}(\Sigma)$ be an infinitesimal variation of complex structures on Σ . If $\dot{\nabla}$ denotes the first order variation of the Levi-Civita connection of $g := \rho(\cdot, J \cdot)$ along \dot{J} , then the following relation holds:

$$\dot{\nabla}_X Y = -\frac{1}{2}((\operatorname{div} \dot{J})(X) JY + J(\nabla_X \dot{J})Y),$$

for every pair of tangent vector fields X, Y on Σ .

Proof. The statement follows from Koszul's formula, which asserts that

$$2\,g(\nabla_X Y,Z) = X(g(Y,Z)) + Y(g(X,Z)) - Z(g(X,Y)) + g([X,Y],Z) - g([X,Z],Y) - g([Y,Z],X),$$

for every tangent vector fields X, Y, Z. Its derivative leads to the equation

$$2\dot{g}(\nabla_X Y, Z) + 2g(\dot{\nabla}_X Y, Z) = X(\dot{g}(Y, Z)) + Y(\dot{g}(X, Z)) - Z(\dot{g}(X, Y)) + \dot{g}([X, Y], Z) - \dot{g}([X, Z], Y) - \dot{g}([Y, Z], X).$$

As seen in relation (3.3), a first order variation of complex structures \dot{J} determines a variation of the Riemannian metric $g = g_J$ of the form $\dot{g} = -g(\cdot, J\dot{J}\cdot)$. By developing the above expression in terms of the covariant derivatives of \dot{g} , and by the fact that ∇ is a torsion-free connection, we can express the term $2g(\dot{\nabla}_X Y, Z)$ as follows

$$\begin{split} 2\,g(\dot{\nabla}_XY,Z) &= (\nabla_X\dot{g})(Y,Z) + (\nabla_Y\dot{g})(X,Z) - (\nabla_Z\dot{g})(X,Y) \\ &= -g(Y,J(\nabla_X\dot{J})Z) - g(X,J(\nabla_Y\dot{J})Z) + g(X,J(\nabla_Z\dot{J})Y) \quad (J \ \nabla\text{-parallel}) \\ &= -g(J(\nabla_X\dot{J})Y + J(\nabla_Y\dot{J})X,Z) + g(J(\nabla_Z\dot{J})X,Y), \end{split}$$

where in the last step we used the fact that $J(\nabla_V \dot{J})$ is g-symmetric for every tangent vector field V. Let $M_X \in \operatorname{End}(T\Sigma)$ denote the endomorphism $M_X V := (\nabla_V \dot{J}) X$, and let M_X^* be its g-adjoint. Then the relation above can be rephrased in the following terms:

$$2 g(\dot{\nabla}_X Y, Z) = -g(J(\nabla_X \dot{J})Y + JM_X Y, Z) + g(JM_X Z, Y)$$

$$= -g(J(\nabla_X \dot{J})Y + JM_X Y, Z) - g(Z, M_X^* JY) \qquad (J^* = -J)$$

$$= -g((J(\nabla_X \dot{J}) + JM_X + M_X^* J)Y, Z).$$

Since this holds for every vector field Z, we deduce that

$$\dot{\nabla}_X Y = -\frac{1}{2} (J(\nabla_X \dot{J}) + JM_X + M_X^* J) Y. \tag{4.11}$$

The endomorphism $JM_X + M_X^*J$ satisfies

$$(JM_X + M_X^*J)^* = M_X^*J^* + J^*M_X = -(JM_X + M_X^*J),$$

since the adjoint of the complex structure coincides with -J. In other words, $JM_X + M_X^*J$ is a g-skew-symmetric endomorphism of $T\Sigma$, and consequently it is of the form λJ , for

some smooth function λ over Σ (the space of g_p -skew-symmetric endomorphisms of $T_p\Sigma$ has dimension 1). The function λ can be determined in the following way:

$$\lambda = -\frac{1}{2}\operatorname{tr}((JM_X + M_X^*J)J)$$

$$= \operatorname{tr}(M_X) \qquad (\operatorname{tr}(M_X) = \operatorname{tr}(M_X^*) \text{ and } J^2 = -1)$$

$$= \sum_i g((\nabla_{e_i}\dot{J})X, e_i)$$

$$= (\operatorname{div}\dot{J})X,$$

where $(e_i)_i$ is a local g-orthonormal frame. Replacing the expression $JM_X + M_X^*J = \lambda J$ in relation (4.11), we deduce the desired assertion.

In the proof of Lemma 4.14 we will also need the following technical lemma:

Lemma 4.16. If A is a traceless endomorphism, then

$$(\nabla_X A)Y - (\nabla_Y A)X = (\operatorname{div} A)(Y)X - (\operatorname{div} A)(X)Y.$$

Proof. It is enough to check the identity for $X = e_1$, $Y = e_2$, where e_1, e_2 is a local orthonormal frame on Σ . To simplify the notation, we set $(\nabla_i A)_j^k := g((\nabla_{e_i} A)e_j, e_k)$. In particular we have that $(\nabla_i A)_1^1 = -(\nabla_i A)_2^2$ for every i. Since A is traceless, the same is true for $\nabla_X A$. With this identity in mind, we proceed to check the desired relation:

$$(\nabla_{e_1} A)e_2 - (\nabla_{e_2} A)e_1 = (\nabla_1 A)_2^1 e_1 + (\nabla_1 A)_2^2 e_2 - (\nabla_2 A)_1^1 e_1 - (\nabla_2 A)_1^2 e_2$$

$$= (\nabla_1 A)_2^1 e_1 - (\nabla_1 A)_1^1 e_2 + (\nabla_2 A)_2^2 e_1 - (\nabla_2 A)_1^2 e_2$$

$$= ((\nabla_1 A)_2^1 + (\nabla_2 A)_2^2) e_1 - ((\nabla_1 A)_1^1 + (\nabla_2 A)_1^2) e_2$$

$$= (\operatorname{div} A)(e_2) e_1 - (\operatorname{div} A)(e_1) e_2.$$

We now have all the elements to prove the statement of Lemma 4.14:

Proof of Lemma 4.14. First we compute the first order variation of the tensor $g^{-1}\sigma$ along $(\dot{J},\dot{\sigma}) \in T_{(J,\sigma)}\mathcal{J}(\Sigma)$. As seen in Lemma 3.2, the variation of the metric g is equal to $\dot{g} = -g(\cdot,J\dot{J}\cdot)$. In particular, we have

$$(g^{-1}\sigma)' = -g^{-1}\dot{g}g^{-1}\sigma + g^{-1}\dot{\sigma}$$

$$= -g^{-1}(-gJ\dot{J})g^{-1}\sigma + g^{-1}\dot{\sigma}_0 - \langle \sigma \mid J\dot{J} \rangle \mathbb{1}$$

$$= J\dot{J}g^{-1}\sigma + g^{-1}\dot{\sigma}_0 - \langle \sigma \mid J\dot{J} \rangle \mathbb{1}.$$
 (Lemma 3.2)

As aforementioned, we need to compute the derivative of the expression $(d^{\nabla}g^{-1}\sigma)(X,Y) = \nabla_X(g^{-1}\sigma Y) - \nabla_Y(g^{-1}\sigma X) - g^{-1}\sigma[X,Y]$ as we vary (J,σ) along a direction $(\dot{J},\dot{\sigma})$. The

final goal will be to show that, if $(\dot{J}, \dot{\sigma})$ belongs to our preferred space $V_{(J,\sigma)}$, then such variation is equal to 0. We observe that

$$((\mathbf{d}^{\nabla}g^{-1}\sigma)(X,Y))' = \nabla_X((g^{-1}\sigma)'Y) - \nabla_Y((g^{-1}\sigma)'X) - (g^{-1}\sigma)'[X,Y] + + \dot{\nabla}_X(g^{-1}\sigma Y) - \dot{\nabla}_Y(g^{-1}\sigma X)$$

$$= (\nabla_X(g^{-1}\sigma)')Y - (\nabla_Y(g^{-1}\sigma)')X + \dot{\nabla}_X(g^{-1}\sigma Y) - \dot{\nabla}_Y(g^{-1}\sigma X)$$

$$= \underbrace{(\operatorname{div}(g^{-1}\sigma)')(Y)X - (\operatorname{div}(g^{-1}\sigma)')(X)Y}_{\text{term 1}} + \underbrace{\dot{\nabla}_X(g^{-1}\sigma Y) - \dot{\nabla}_Y(g^{-1}\sigma X)}_{\text{term 2}}.$$
(Lemma 4.16)

First we rephrase term 1 in the expression above. Applying the expression for $(g^{-1}\sigma)'$ derived at the beginning of the proof, we have

$$(\operatorname{div}(g^{-1}\sigma)')(Y) X - (\operatorname{div}(g^{-1}\sigma)')(X) Y = \operatorname{div}(J\dot{J}g^{-1}\sigma + g^{-1}\dot{\sigma}_0)(Y) X +$$

$$-\operatorname{div}(J\dot{J}g^{-1}\sigma + g^{-1}\dot{\sigma}_0)(X) Y +$$

$$-Y(\langle \sigma \mid J\dot{J} \rangle) X + X(\langle \sigma \mid J\dot{J} \rangle) Y.$$

Assume now that $(J, \dot{\sigma})$ belongs to $V_{(J,\sigma)}$. Taking the sum of the equations in (V2) (Proposition K part ii)) and using the expressions of J_l and J_r from (3.18), we obtain the relation

$$\operatorname{div}(J \,\dot{J} \,g^{-1}\sigma + g^{-1}\dot{\sigma}_0) = \langle \nabla^g_{J\bullet}\sigma \mid \dot{J} \rangle.$$

Together with Lemma 2.3 part iv), we see that

$$(\operatorname{div}(g^{-1}\sigma)')(Y) X - (\operatorname{div}(g^{-1}\sigma)')(X) Y = \langle \nabla_{JY}\sigma \mid \dot{J} \rangle X - \langle \nabla_{JX}\sigma \mid \dot{J} \rangle Y +$$

$$- Y(\langle \sigma \mid J\dot{J} \rangle) X + X(\langle \sigma \mid J\dot{J} \rangle) Y$$

$$= \langle \sigma \mid J(\nabla_{Y}\dot{J}) \rangle Y - \langle \sigma \mid J(\nabla_{Y}\dot{J}) \rangle X.$$

This is the expression for term 1 that we will use in the final computation. Now we focus our attention on term 2. Applying the variation formula of the Levi-Civita connection given by Lemma 4.15, we obtain that

$$\begin{split} \dot{\nabla}_X(g^{-1}\sigma Y) - \dot{\nabla}_Y(g^{-1}\sigma X) &= \frac{1}{2} \left((\operatorname{div} \dot{J})(Y) J g^{-1}\sigma X + J(\nabla_Y \dot{J}) g^{-1}\sigma X + - (\operatorname{div} \dot{J})(X) J g^{-1}\sigma Y - J(\nabla_X \dot{J}) g^{-1}\sigma Y \right) \\ &= \frac{1}{2} J g^{-1}\sigma \left((\operatorname{div} \dot{J})(Y) X - (\operatorname{div} \dot{J})(X) Y \right) + \\ &+ \frac{1}{2} J \left((\nabla_Y \dot{J}) g^{-1}\sigma X - (\nabla_X \dot{J}) g^{-1}\sigma Y \right) \\ &= \frac{1}{2} J g^{-1}\sigma \left((\nabla_X \dot{J}) Y - (\nabla_Y \dot{J}) X \right) + \end{split}$$

$$+\frac{1}{2}J\left((\nabla_Y\dot{J})g^{-1}\sigma X-(\nabla_X\dot{J})g^{-1}\sigma Y\right)$$

where, in the last step, we applied Lemma 4.16 to $A = \dot{J}$. As a last remark, we apply Lemma 3.9 to $(\nabla_{\bullet}\dot{J}) g^{-1}\sigma$ and $g^{-1}\sigma(\nabla_{\bullet}\dot{J})$, deriving:

$$g^{-1}\sigma\left(\nabla_{\bullet}\dot{J}\right) - \left(\nabla_{\bullet}\dot{J}\right)g^{-1}\sigma = 2\langle\sigma\mid J(\nabla_{\bullet}\dot{J})\rangle J. \tag{4.12}$$

We finally combine the expressions found above for terms 1 and 2 with this last relation. If (J, σ) is a point of $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$ and $(\dot{J}, \dot{\sigma})$ is an element of $V_{(J,\sigma)}$, then the variation of the differential $d^{\nabla}g^{-1}\sigma$ can be expressed as follows:

$$((\mathbf{d}^{\nabla}g^{-1}\sigma)(X,Y))' = \langle \sigma \mid J(\nabla_{X}\dot{J})\rangle Y - \langle \sigma \mid J(\nabla_{Y}\dot{J})\rangle X +$$

$$+ \frac{1}{2}Jg^{-1}\sigma\left((\nabla_{X}\dot{J})Y - (\nabla_{Y}\dot{J})X\right) +$$

$$+ \frac{1}{2}J\left((\nabla_{Y}\dot{J})g^{-1}\sigma X - (\nabla_{X}\dot{J})g^{-1}\sigma Y\right)$$

$$= \frac{1}{2}\left(2\langle \sigma \mid J(\nabla_{X}\dot{J})\rangle Y + Jg^{-1}\sigma(\nabla_{X}\dot{J})Y - J(\nabla_{X}\dot{J})g^{-1}\sigma Y\right) +$$

$$- \frac{1}{2}\left(2\langle \sigma \mid J(\nabla_{Y}\dot{J})\rangle X + Jg^{-1}\sigma(\nabla_{Y}\dot{J})X - J(\nabla_{Y}\dot{J})g^{-1}\sigma X\right)$$

$$= \frac{1}{2}\left(2\langle \sigma \mid J(\nabla_{X}\dot{J})\rangle Y + 2\langle \sigma \mid J(\nabla_{X}\dot{J})\rangle J^{2}Y\right) +$$

$$- \frac{1}{2}\left(2\langle \sigma \mid J(\nabla_{Y}\dot{J})\rangle X + 2\langle \sigma \mid J(\nabla_{Y}\dot{J})\rangle J^{2}X\right)$$

$$(\text{rel. } (4.12) \text{ for } \bullet = X, Y)$$

$$= 0.$$

Therefore $(\dot{J}, \dot{\sigma})$ lies in the kernel of the linearized Codazzi equation.

Lemma 4.17. Every element $(\dot{J}, \dot{\sigma}) \in V_{(J,\sigma)}$ lies in the kernel of the linearized Gauss-Codazzi equations (GC).

Proof. We start by relating the Riemannian metrics h_l and h_r associated to the complex structures J_l and J_r with h and the endomorphism B:

$$h_{l,r} = \rho(\cdot, (\mathbb{1} \mp JB)^{-1}J(\mathbb{1} \mp JB)\cdot)$$

$$= \rho((\mathbb{1} \mp JB)^{-1}(\mathbb{1} \mp JB)\cdot, (\mathbb{1} \mp JB)^{-1}J(\mathbb{1} \mp JB)\cdot)$$

$$= \det(\mathbb{1} \mp JB)^{-1}\rho((\mathbb{1} \mp JB)\cdot, J(\mathbb{1} \mp JB)\cdot)$$

$$= \frac{f+1}{2}g((\mathbb{1} \mp JB)\cdot, (\mathbb{1} \mp JB)\cdot)$$

$$= \frac{1}{2}h((\mathbb{1} \mp JB)\cdot, (\mathbb{1} \mp JB)\cdot).$$
(rel. (3.17))

We remark that the metrics $h_{l,r}$ differ by the left and right hyperbolic metrics given by Mess' homeomorphism by a factor 2. Indeed, since the tensor $\mathbb{1} \mp JB$ is h-Codazzi, the

curvature of the metric $h_{l,r}$ can be computed as follows

$$K_{h_{l,r}} = \frac{2K_h}{\det(\mathbb{1} \mp JB)} = \frac{2K_h}{1 + \det B}.$$

A proof of this fact can be found in [KS07]. This shows that the Gauss-Codazzi equations have the following equivalent descriptions:

$$\begin{cases} d^{\nabla^h} B = 0, \\ K_h = -1 - \det B, \end{cases} \Leftrightarrow \begin{cases} d^{\nabla^g} g^{-1} \sigma = 0, \\ K_{h_l} = -2, \end{cases} \Leftrightarrow \begin{cases} d^{\nabla^g} g^{-1} \sigma = 0, \\ K_{h_r} = -2. \end{cases}$$

Assume now that $(\dot{J}, \dot{\sigma})$ is an element of $V_{(J,\sigma)}$. By Lemma 4.14, $(\dot{J}, \dot{\sigma})$ lies in the kernel of the linearized Codazzi equation. Therefore, by what previously observed, it is enough to show that the first order variation of the curvature 2-form K_{h_l} d $a_{h_l} = K_{h_l} \rho$ (or $K_{h_r} \rho$) along the direction $\dot{J}_l = d\mathcal{M}(\dot{J}, \dot{\sigma})_1$ (or $\dot{J}_r = d\mathcal{M}(\dot{J}, \dot{\sigma})_2$, respectively) is equal to 0.

The first order variation of the 2-form $K_{h_l} \rho$ coincides with $\frac{1}{2} \operatorname{d} (\operatorname{div}_{h_l} \dot{J}_l)$ (see [Don03]), and by the last statement of Proposition K, the divergence $\operatorname{div}_{h_l} \dot{J}_l$ is an exact 1-form. Since $\operatorname{d}^2 = 0$, we deduce that the derivative of $K_{h_l} \operatorname{d} a_{h_l}$ along \dot{J}_l is 0, and therefore $(\dot{J}, \dot{\sigma})$ lies in the kernel of the linearized Gauss-Codazzi equations.

4.5.4. Proof of Lemma 4.18: the dimension of $V_{(I,\sigma)}$ is $\geq 6|\chi(\Sigma)|$

Lemma 4.18. For every $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$,

$$\dim V_{(J,\sigma)} \ge 6|\chi(\Sigma)|.$$

Proof. We make use of the description provided by Proposition K part *iii*). Because the equations are decoupled, it is sufficient to show that the space of solutions of the equation

$$\operatorname{div}_{g} Q^{\pm}(\dot{J}, \dot{\sigma}) = \mp f^{-1} \langle \nabla_{J \bullet} \sigma \mid Q^{\pm}(\dot{J}, \dot{\sigma}) \rangle$$

has dimension at least $3|\chi(\Sigma)|$. We give the details for the equation concerning Q^+ , the other case being analogous. Consider the differential operator $D_J: T_J \mathcal{J}(\Sigma) \to \Lambda^1(\Sigma)$ defined by

$$D_J(Q) = \operatorname{div}_g Q + f^{-1} \langle \nabla_{J \bullet} \sigma \mid Q \rangle.$$

Its principal symbol coincides with that of the divergence operator $\operatorname{div}_g: T_J\mathcal{J}(\Sigma) \to \Lambda^1(\Sigma)$, hence D_J is Fredholm and $\operatorname{ind}(D_J) = \operatorname{ind}(\operatorname{div}_g)$, where $\operatorname{ind}(\cdot)$ denotes the index of the differential operator, i.e. the difference between the dimension of its kernel and the dimension of its co-kernel. It is well-known (see for instance [Tro92]) that $\operatorname{dim}(\operatorname{Ker}(\operatorname{div}_g)) = 3|\chi(\Sigma)|$: the space of traceless, divergence-free tensors represents the tangent space to the Teichmüller space of Σ . On the other hand, $\operatorname{div}_g: T_J\mathcal{J}(\Sigma) \to \Lambda^1(\Sigma)$ is surjective. Indeed, using the musical isomorphism $\#: \Lambda^1(\Sigma) \to \Gamma(T\Sigma)$ induced by the metric $g = g_J$, this is equivalent to proving that $\operatorname{div}_g^\#: T_J\mathcal{J}(\Sigma) \to \Gamma(T\Sigma)$ is surjective. Let $L: \Gamma(T\Sigma) \to$

 $T_J \mathcal{J}(\Sigma)$ denote the Lie derivative operator. Its L^2 -adjoint is $L^*(\dot{J}) = -J(\operatorname{div}_g \dot{J})^\#$. In fact,

$$\langle LV, \dot{J} \rangle_{L^{2}} = \int_{\Sigma} \frac{1}{2} \operatorname{tr} \left(\dot{J} \mathcal{L}_{V} J \right) \rho$$

$$= \int_{\Sigma} (\operatorname{div}_{g} \dot{J} J)(V) \rho \qquad (\text{Equation (4.7)})$$

$$= \int_{\Sigma} (\operatorname{div}_{g} \dot{J})(JV) \rho \qquad (\text{Equation (4.5)})$$

$$= -\langle J(\operatorname{div}_{g} \dot{J})^{\#}, V \rangle_{L^{2}}.$$

Therefore, the operator $L^*L:\Gamma(T\Sigma)\to\Gamma(T\Sigma)$ is self-adjoint. Moreover, if $V\in\mathrm{Ker}(L^*L)$ then

$$0 = \langle L^*L(V), V \rangle_{L^2} = \|\mathcal{L}_V J\|_{L^2}^2 ,$$

which implies that V = 0 because (Σ, J) has no biholomorphisms isotopic to the identity. A standard computation in local coordinates shows that the operator L^*L is elliptic, thus by [Voi02, Theorem 5.1] we have an L^2 -orthogonal decomposition

$$\Gamma(T\Sigma) = \operatorname{Ker}(L^*L) \oplus \operatorname{Im}(L^*L) = \operatorname{Im}(L^*L)$$
,

which shows that every vector field is in the image of $\operatorname{div}_g^\#$. Hence,

$$\dim(\operatorname{Ker}(D_J)) \geqslant \operatorname{ind}(D_J) = \operatorname{ind}(\operatorname{div}_q) = 3|\chi(\Sigma)|$$

and this concludes the proof of the assertion.

4.5.5. Proof of Lemma 4.21: transversality to the tangent space to the $\operatorname{Symp}_0(\Sigma, \rho)$ -orbit

Lemma 4.19. For every symplectic vector field X on (Σ, ρ) and for every $(J, \sigma) \in T^*\mathcal{J}(\Sigma)$, with σ that is the real part of a holomorphic quadratic differential on (Σ, J) , we have $\mathbf{I}(\mathcal{L}_X J, \mathcal{L}_X \sigma) = (-\mathcal{L}_{JX} J, -\mathcal{L}_{JX} \sigma)$.

Proof. We need to introduce some notation to prove the desired identity. Let A_V be the total derivative of a vector field V tangent to Σ , i. e. $A_VY := \nabla_Y^g V$, for any tangent vector field Y (as usual g is the Riemannian metric $\rho(\cdot, J \cdot)$). Then we can write

$$(\mathcal{L}_{V}J)Y = [V, JY] - J[V, Y]$$

$$= \nabla_{V}^{g}(JY) - \nabla_{JY}^{g}V - J\nabla_{V}^{g}Y + J\nabla_{Y}^{g}V \qquad (\nabla^{g} \text{ torsion-free})$$

$$= J\nabla_{V}^{g}Y - A_{V}(JY) - J\nabla_{V}^{g}Y + JA_{V}Y \qquad (J \nabla^{g}\text{-parallel})$$

$$= (JA_{V} - A_{V}J)Y.$$

In other words, we have that

$$\mathcal{L}_V J = J A_V - A_V J \tag{4.13}$$

A similar computation shows that

$$\mathcal{L}_V \sigma = \nabla_V^g \sigma + \sigma(A_V, \cdot) + \sigma(\cdot, A_V). \tag{4.14}$$

Now we apply the definition of the almost-complex structure **I** to the pair $(\mathcal{L}_X J, \mathcal{L}_X \sigma)$, with X symplectic vector field:

$$\mathbf{I}(\mathcal{L}_X J, \mathcal{L}_X \sigma) = (-J(\mathcal{L}_X J), -(\mathcal{L}_X \sigma)_0(\cdot, J \cdot) - \langle \sigma \mid \mathcal{L}_X J \rangle g)$$

Being $J \nabla^g$ -parallel, we have that $A_{JX} = JA_X$. Therefore

$$-J(\mathcal{L}_X J) = -J(JA_X - A_X J)$$
 (rel. (4.13) for $V = X$)
$$= -(JA_{JX} - A_{JX} J)$$
 (rel. (4.13) for $V = JX$)
$$= -\mathcal{L}_{JX} J.$$

This shows that the first component of $\mathbf{I}(\mathcal{L}_X J, \mathcal{L}_X \sigma)$ coincides with $-\mathcal{L}_{JX} J$. To study the second component we will need a few additional remarks. First we notice that it is enough to show

$$-(\mathcal{L}_X\sigma)_0(\cdot,J\cdot) = -(\mathcal{L}_{JX}\sigma)_0,$$

because the trace part of the second component of a pair $(\dot{J}, \dot{\sigma})$ is uniquely determined by \dot{J} (compare with Lemma 3.2). Given V a vector field, the endomorphism A_V can be decomposed into a sum

$$A_V = \frac{\text{tr}(A_V)}{2} \mathbb{1} - \frac{\text{tr}(JA_V)}{2} J + A_V^s,$$

where the first term is its trace part, the second is the g-skew-symmetric part, and the third is the traceless and g-symmetric part.

If X is a ρ -symplectic vector field, then the trace part of A_X vanishes. Since $A_{JX} = JA_X$ (again because J is ∇^g -parallel), the decomposition of A_{JX} is

$$A_{JX} = JA_X = \frac{\text{tr}(JA_X)}{2}\mathbb{1} + 0 + JA_X^s.$$

In particular, X symplectic implies that the g-skew-symmetric part of A_{JX} vanishes, and $A_{JX}^s = JA_X^s$. Therefore

$$(\mathcal{L}_{JX}\sigma)_0 = \nabla_{JX}^g \sigma + \sigma(A_{JX}\cdot,\cdot) + \sigma(\cdot,A_{JX}\cdot) - \frac{\operatorname{tr}(g^{-1}\sigma A_{JX} + A_{JX}g^{-1}\sigma)}{2}g \quad (\text{rel. } (4.14))$$
$$= \nabla_{JX}^g \sigma + \operatorname{tr}(JA_X)\sigma + \sigma(JA_X^s\cdot,\cdot) + \sigma(\cdot,JA_X^s\cdot) - \operatorname{tr}(g^{-1}\sigma JA_X^s)g$$

Applying Lemma 3.9 to $g^{-1}\sigma\,JA_X^s$ and $JA_X^sg^{-1}\sigma$ we obtain

$$\sigma(\cdot,JA_X^s\cdot)+\sigma(JA_X^s\cdot,\cdot)=\operatorname{tr}\!\left(g^{-1}\sigma\,JA_X^s\right)g,$$

which, combined with the expression found above, shows that

$$(\mathcal{L}_{JX}\sigma)_0 = \nabla_{JX}^g \sigma + \operatorname{tr}(JA_X) \sigma$$

$$= (\nabla_X^g \sigma)(\cdot, J\cdot) + \operatorname{tr}(JA_X) \sigma$$
 (Lemma 2.3 part iv))

On the other hand, by the decomposition of A_X , we have

$$(\mathcal{L}_X \sigma)_0 = \nabla_X^g \sigma + \sigma(A_X, \cdot) + \sigma(\cdot, A_X) - \frac{\operatorname{tr}(g^{-1} \sigma A_X + A_X g^{-1} \sigma)}{2} g \qquad (\text{rel } (4.14))$$

$$= \nabla_X^g \sigma - \frac{\operatorname{tr}(J A_X)}{2} (\sigma(J, \cdot) + \sigma(\cdot, J, \cdot)) + \sigma(A_X^s, \cdot) + \sigma(\cdot, A_X^s, \cdot) - \operatorname{tr}(g^{-1} \sigma A_X) g$$

$$= \nabla_X^g \sigma - \operatorname{tr}(J A_X) \sigma(\cdot, J, \cdot) + \sigma(A_X^s, \cdot) + \sigma(\cdot, A_X^s, \cdot) - \operatorname{tr}(g^{-1} \sigma A_X^s) g,$$

where in the last step we used that $\operatorname{tr}(g^{-1}\sigma A_X^s) = \operatorname{tr}(g^{-1}\sigma A_X)$. Now, applying Lemma 3.9 to $g^{-1}\sigma A_X^s$ and $A_X^s g^{-1}\sigma$ we obtain

$$\sigma(\cdot, A_X^s \cdot) + \sigma(A_X^s \cdot, \cdot) = \operatorname{tr}(g^{-1}\sigma A_X^s) g,$$

which reduces the expression above to the equality

$$(\mathcal{L}_X \sigma)_0 = \nabla_X^g \sigma - \operatorname{tr}(J A_X) \, \sigma(\cdot, J \cdot).$$

The identity $(\mathcal{L}_X \sigma)_0(\cdot, J \cdot) = (\mathcal{L}_{JX} \sigma)_0$ is now immediate.

Lemma 4.20. Let G be the operator $G(J, \sigma) := K_h + 1 + \det B$, defined over the space $T^*\mathcal{J}(\Sigma)$ and with values in $\mathscr{C}^{\infty}(\Sigma)$. Assume that (J, σ) satisfies the Gauss-Codazzi equations and let U be a vector field on Σ . Then

$$dG_{(J,\sigma)}(\mathcal{L}_U J, \mathcal{L}_U \sigma) = \frac{1}{2} \Delta_h(f^{-1} \operatorname{div}_g U) - (1 - \det B) f^{-1} \operatorname{div}_g U.$$

In particular, if $(\mathcal{L}_U J, \mathcal{L}_U \sigma)$ belongs to the kernel of the differential of G, then U is a ρ -symplectic vector field, i. e. $d(\iota_U \rho) = 0$.

Proof. The final goal will be to compute $\frac{d}{dt}G(J_t, \sigma_t)|_{t=0}$, where $(J_t, \sigma_t) = (\psi_t^* J, \psi_t^* \sigma)$. We first determine the Riemannian metric $g_t = \rho(\cdot, J_t \cdot)$ associated to the complex structure $J_t := d\psi_t^{-1} J d\psi_t$, where $(\psi_t)_t$ represents the flow of U.

$$g_t = \rho(\cdot, (\mathrm{d}\psi_t^{-1} J \, \mathrm{d}\psi_t) \cdot) = \rho((\mathrm{d}\psi_t^{-1} \, \mathrm{d}\psi_t) \cdot, (\mathrm{d}\psi_t^{-1} J \, \mathrm{d}\psi_t) \cdot)$$

$$= (\det(\mathrm{d}\psi_t^{-1}) \circ \psi_t) \ \rho(\mathrm{d}\psi_t \cdot, J \, \mathrm{d}\psi_t \cdot) = (\det(\mathrm{d}\psi_t^{-1}) \circ \psi_t) \ g(\mathrm{d}\psi_t \cdot, \mathrm{d}\psi_t \cdot)$$

$$= (\det(\mathrm{d}\psi_t^{-1}) \circ \psi_t) \ \psi_t^* g,$$

where $g = g_0$ and $\det(\mathrm{d}\psi_t^{-1}) = ((\psi_t^{-1})^*\rho)/\rho$. In particular, g_t is conformal to ψ_t^*g with conformal factor given by $u_t := \det(\mathrm{d}\psi_t^{-1}) \circ \psi_t$. We now determine the Riemannian metric h_t associated to the pair (J_t, σ_t) as described at the beginning of Section 4:

$$h_{t} = \left(1 + \sqrt{1 + \|\psi_{t}^{*}\sigma\|_{g_{t}}^{2}}\right) g_{t} = \left(1 + \sqrt{1 + \|\psi_{t}^{*}\sigma\|_{g_{t}}^{2}}\right) u_{t} \psi_{t}^{*} g$$

$$= \frac{1 + \sqrt{1 + \|\psi_{t}^{*}\sigma\|_{g_{t}}^{2}}}{1 + \sqrt{1 + \|\sigma\|_{q}^{2} \circ \psi_{t}}} u_{t} \psi_{t}^{*} h = v_{t} \psi_{t}^{*} h.$$

Therefore, the metric h_t also differs from ψ_t^*h by a conformal factor, here denoted by v_t and defined from the relation above (observe that $v_0 \equiv 1$). Using the classical expression for the curvature under conformal change of the metric, we deduce that

$$K_{h_t} = K_{v_t \psi_t^* h}$$

$$= v_t^{-1} \left(K_{\psi_t^* h} - \frac{1}{2} \Delta_{\psi_t^* h} \ln v_t \right)$$

$$= v_t^{-1} \left(K_h \circ \psi_t - \frac{1}{2} (\Delta_h \ln(v_t \circ \psi_t^{-1})) \circ \psi_t \right).$$

The last term of the operator G that we need to analyze is the determinant of the endomorphism B_t associated to the pair (J_t, σ_t) :

$$\det B_t = \det(h_t^{-1}\sigma_t)$$

$$= v_t^{-2} \det((\psi_t^* h)^{-1} \psi_t^* \sigma)$$

$$= v_t^{-2} (\det B) \circ \psi_t.$$

We can finally deduce an expression for the term

$$K_{h_t} + \det B_t = v_t^{-1} \left(K_h \circ \psi_t - \frac{1}{2} (\Delta_h \ln(v_t \circ \psi_t^{-1})) \circ \psi_t + v_t^{-1} (\det B) \circ \psi_t \right).$$

Combining the relations found above, we can compute the first order variation of operator G along the path $t \mapsto (\psi_t^* J, \psi_t^* \sigma)$, obtaining

$$(K_{h_t} + 1 + \det B_t)' = -\dot{v}(K_h + \det B) + U(K_h) - \frac{1}{2}\Delta_h \dot{v} - \dot{v}\det B + U(\det B)$$
$$= \dot{v}(1 - \det B) - \frac{1}{2}\Delta_h \dot{v},$$

where, in the last line, we used the fact that (J, σ) satisfies $G(J, \sigma) = 0$. The final statement will now follow by computing \dot{v} :

$$\dot{v} = \frac{\mathrm{d}v_t}{\mathrm{d}t} \Big|_{t=0} = \frac{\mathrm{d}(v_t \circ \psi_t^{-1})}{\mathrm{d}t} \Big|_{t=0}$$

$$= \frac{\mathrm{d}}{\mathrm{d}t} \frac{1 + \sqrt{1 + \|\sigma_t\|_{g_t}^2 \circ \psi_t^{-1}}}{1 + \sqrt{1 + \|\sigma\|_g^2}} u_t \Big|_{t=0}$$

$$= \frac{(\|\sigma_t\|_{g_t}^2 \circ \psi_t^{-1})'}{2f(1+f)} + \dot{u}.$$

$$(v_0 \equiv 1)$$

We now observe that

$$\|\sigma_t\|_{g_t}^2 = \frac{1}{2} \operatorname{tr} \left(g_t^{-1} \sigma_t g_t^{-1} \sigma_t \right) = \frac{1}{2u_t^2} \operatorname{tr} \left((\psi_t^* g)^{-1} \psi_t^* \sigma (\psi_t^* g)^{-1} \psi_t^* \sigma \right)$$
$$= u_t^{-2} (\|\sigma\|_g^2 \circ \psi_t),$$

which implies the following:

$$(\|\sigma_t\|_{q_t}^2 \circ \psi_t^{-1})' = ((u_t \circ \psi_t^{-1})^{-2})' \|\sigma\|_q^2 = -2\dot{u} \|\sigma\|_q^2 = -2\dot{u}(f^2 - 1)$$

since $u_0 = \det(\operatorname{d}(id)) \equiv 1$. A simple computation now shows that $\dot{u} = \frac{\operatorname{d}u_t}{\operatorname{d}t}|_{t=0} = -\operatorname{div}_g U$. To conclude, we have

$$\dot{v} = \left(-\frac{f^2 - 1}{f(1+f)} + 1\right)\dot{u} = -f^{-1}\operatorname{div}_g U.$$

For the second part of the statement, observe that, being B traceless and h-self-adjoint, its determinant is a non-positive function. In particular $(1 - \det B) \ge 1$. Consequently the linear operator

$$T: \lambda \longmapsto -\frac{1}{2}\Delta_h \lambda + (1 - \det B)\lambda$$

is self-adjoint and positive definite in $L^2(\Sigma, da_h)$, in particular injective. Therefore, if $(\mathcal{L}_U J, \mathcal{L}_U \sigma)$ lies in the kernel of the differential of G, then the function $\lambda = f^{-1} \operatorname{div}_g U$ is sent to 0 by the operator T, and so $\operatorname{div}_g U = 0$. By relation (4.6) this is equivalent to say that U is a ρ -symplectic vector field.

Lemma 4.21. For every $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, we have

$$V_{(J,\sigma)} \cap T_{(J,\sigma)}(\operatorname{Symp}_0(\Sigma, \rho) \cdot (J, \sigma)) = \{0\}$$
.

Proof. Assume X to be a symplectic vector field such that $(\mathcal{L}_X J, \mathcal{L}_X \sigma)$ belongs to $V_{(J,\sigma)}$. By equivariance of Mess homeomorphism, this is equivalent to the fact that $(\mathcal{L}_X J_l, \mathcal{L}_X J_r)$ are exact 1-forms. In particular we must have that the h_l -divergence of $\mathcal{L}_X J_l$ is an exact 1-form. By Proposition 4.3, this implies that $\mathcal{L}_X J_l$ is Ω_{J_l} -orthogonal to T_{J_l} Symp $_0(\Sigma, \rho) \cdot J_l$. Therefore, for every ρ -symplectic vector field Y we have $\Omega_{J_l}(\mathcal{L}_X J_l, \mathcal{L}_Y J_l) = 0$. Consequently, inside the quotient $\widetilde{\mathcal{T}}(\Sigma)$ of $\mathcal{J}(\Sigma)$ by the action of the Hamiltonian group $\operatorname{Ham}(\Sigma, \rho)$, the class $[\mathcal{L}_X J_l]$ is $\widehat{\Omega}_{[J_l]}$ -orthogonal to the H-orbit of $[J_l]$ (compare with Section 4.2, and in particular with Equation (4.4)). As observed by Donaldson [Don03] (see also [Tra19]), the H-orbits are symplectic submanifolds of $\widetilde{\mathcal{T}}(\Sigma)$, and the class of $\mathcal{L}_X J_l$ is tangent to the orbit $H \cdot [J_l]$. By what previously observed, we deduce that the class of $\mathcal{L}_X J_l$ is equal to zero inside $T_{J_l}H \cdot [J_l]$ or, in other words, that $\mathcal{L}_X J_l$ is tangent to the orbit of J_l by the Hamiltonian group inside $\mathcal{J}(\Sigma)$. Since the Lie derivative operator $X \mapsto \mathcal{L}_X J$ is injective, because (Σ, J) has no biholomorphisms isotopic to the identity, the vector field X is actually ρ -Hamiltonian, i. e. $\iota_X \rho = \mathrm{d} f$ for some $f \in \mathscr{C}^{\infty}(\Sigma)$.

On the other hand, if $(\mathcal{L}_X J, \mathcal{L}_X \sigma)$ belongs to $V_{(J,\sigma)}$, the same has to hold for $\mathbf{I}(\mathcal{L}_X J, \mathcal{L}_X \sigma) = -(\mathcal{L}_{JX} J, \mathcal{L}_{JX} \sigma)$ (see Lemma 4.19). In particular, the differential of the function G considered in Lemma 4.20 applied to $(\mathcal{L}_{JX} J, \mathcal{L}_{JX} \sigma)$ must vanish, by what observed in Lemma 4.17. By the second part of Lemma 4.20, we deduce that JX is ρ -symplectic, i.e. $d(\iota_{JX}\rho) = 0$. This implies that the 1-form $-df \circ J = -(\iota_X \rho) \circ J = \iota_{JX} \rho$ is closed, and therefore that the function f is g-harmonic (since $d(df \circ J) = -\Delta_g f \rho$). Being Σ compact, we deduce that f is constant, and therefore that the vector field X is equal to 0, as desired.

4.6. The proof of Proposition K

This subsection is dedicated to the proof of Proposition K, which provides a series of equivalent descriptions of the tangent space to $\mathcal{MS}_0(\Sigma, \rho)$, the model for the deformation space of MGHC AdS structures introduced in Section 2.5. The proof of the result is technically involved, and none of the tools developed here will be used in the rest of the exposition. In particular, the reader who is willing to trust the statement of Proposition K can skip this part without losing necessary ingredients for the remainder of the paper.

We will first focus on the equivalence of the first three descriptions appearing in the statement of Proposition K. This part is mainly algebraic and it follows from explicit manipulations of the equations.

Proof of $i) \Leftrightarrow ii) \Leftrightarrow iii$). Recalling that $Q^{\pm} = Q^{\pm}(\dot{J}, \dot{\sigma}) := f^{-1}g^{-1}\dot{\sigma}_0 \pm \dot{J}$ (these terms formally appeared already in the study of the differential of Mess' map in Proposition C (baby version)), one readily checks that $i) \Leftrightarrow ii$, since taking the sum and difference of the equations in (V1) one obtains the equations in (V3), and vice versa.

Let us now prove $ii) \Leftrightarrow iii)$. As a preliminary step, we claim that, for every $J' \in T_J \mathcal{J}(\Sigma)$ and for every tangent vector field V, we have

$$df(\dot{J}'V) = f^{-1} \langle \nabla^g_{(g^{-1}\sigma)V} \sigma \mid \dot{J}' \rangle = f^{-1} \langle \nabla^g_{\bullet} \sigma \mid \dot{J}' \rangle ((g^{-1}\sigma)V). \tag{4.15}$$

Assuming temporarily this relation, and using that the components J_l and J_r of the map \mathcal{M} can be expressed as follows (see (3.18))

$$J_l = fJ + g^{-1}\sigma, \qquad J_r = fJ - g^{-1}\sigma,$$

we obtain the following identities in local coordinates x_i :

$$(\operatorname{div}_{g} Q^{+})JJ_{l}V = \operatorname{d}x^{i} ((\nabla_{\partial_{i}}^{g} Q^{+})JJ_{l}V)$$

$$= \operatorname{d}x^{i} ((\nabla_{\partial_{i}}^{g} Q^{+}JJ_{l})V) - \operatorname{d}x^{i} (Q^{+}J(\nabla_{\partial_{i}}^{g} (fJ+g^{-1}\sigma))V)$$

$$= \operatorname{div}_{g} (Q^{+}JJ_{l})V - \left(\partial_{i}f \operatorname{d}x^{i} (Q^{+}J^{2}V) + \operatorname{d}x^{i} (Q^{+}J(\nabla_{\partial_{i}}^{g}g^{-1}\sigma)V)\right)$$

$$= \operatorname{div}_{g} (Q^{+}JJ_{l})V + \operatorname{d}f (Q^{+}V) - \operatorname{d}x^{i} (Q^{+}J(\nabla_{V}^{g}g^{-1}\sigma)\partial_{i}) \quad (g^{-1}\sigma \operatorname{Codazzi})$$

$$= \operatorname{div}_{g} (Q^{+}JJ_{l})V + \operatorname{d}f (Q^{+}V) - 2\langle \nabla_{V}^{g}\sigma \mid Q^{+}J\rangle \qquad (\nabla^{g}g^{-1} = 0)$$

$$= \operatorname{div}_{g} (Q^{+}JJ_{l})V + \operatorname{d}f (Q^{+}V) + 2\langle (\nabla_{V}^{g}\sigma)(\cdot, J\cdot) \mid Q^{+}\rangle \qquad (Q^{+} \in T_{J}\mathcal{J}(\Sigma))$$

$$= \operatorname{div}_{g} (Q^{+}JJ_{l})V + \operatorname{d}f (Q^{+}V) + 2\langle \nabla_{JV}^{g}\sigma \mid Q^{+}\rangle \qquad (\operatorname{Lemma 2.3 part } iv)$$

$$= \operatorname{div}_{g} (Q^{+}JJ_{l})V + f^{-1}\langle \nabla_{(g^{-1}\sigma)V}^{g}\sigma \mid Q^{+}\rangle + 2\langle \nabla_{JV}^{g}\sigma \mid Q^{+}\rangle, \quad (\text{rel. } (4.15))$$

$$= \operatorname{div}_{g} (Q^{+}JJ_{l})V + f^{-1}\langle \nabla_{(g^{-1}\sigma+fJ)V}^{g}\sigma \mid Q^{+}\rangle + \langle \nabla_{JV}^{g}\sigma \mid Q^{+}\rangle,$$

where in the second to last line we used relation (4.15). This can be rewritten as:

$$(\operatorname{div}_g Q^+ + f^{-1} \langle \nabla_{J \bullet}^g \sigma \mid Q^+ \rangle) \circ JJ_l = \operatorname{div}_g(Q^+ JJ_l) + \langle \nabla_{J \bullet}^g \sigma \mid Q^+ \rangle.$$

This shows that the first equations in (V2) and (V3) are equivalent. Proceeding in an analogous way for the term $\operatorname{div}_g Q^-$, we can deduce that the second equations are equivalent too, and therefore conclude that $ii) \Leftrightarrow iii$.

It remains to prove relation (4.15). Since $(g^{-1}\sigma)^2 = ||\sigma||^2 \mathbb{1}$ by Cayley-Hamilton theorem, we have

$$V(\|\sigma\|^2) \,\mathbb{1} = (\nabla_V^g g^{-1}\sigma)g^{-1}\sigma + g^{-1}\sigma(\nabla_V^g g^{-1}\sigma). \tag{4.16}$$

Now we observe that

$$\begin{split} 2\langle \nabla^g_{(g^{-1}\sigma)V} \sigma \mid \dot{J}' \rangle &= \mathrm{d} x^i \, (\dot{J}'(\nabla^g_{(g^{-1}\sigma)V} g^{-1}\sigma) \partial_i) & (\nabla^g g^{-1} = 0) \\ &= \mathrm{d} x^i \, (\dot{J}'(\nabla^g_{\partial_i} g^{-1}\sigma)(g^{-1}\sigma)V) & (g^{-1}\sigma \, \mathrm{Codazzi}) \\ &= -\, \mathrm{d} x^i \, (\dot{J}' g^{-1}\sigma(\nabla^g_{\partial_i} g^{-1}\sigma)V) + \partial_i (\|\sigma\|^2) \, \mathrm{d} x^i \, (\dot{J}'V) & (\mathrm{relation} \, (4.16)) \\ &= -\, \mathrm{d} x^i \, (\dot{J}' g^{-1}\sigma(\nabla^g_V g^{-1}\sigma)\partial_i) + \mathrm{d} (\|\sigma\|^2) \, (\dot{J}'V) & (g^{-1}\sigma \, \mathrm{Codazzi}) \\ &= -\, \mathrm{tr} \Big(\dot{J}' g^{-1}\sigma(\nabla^g_V g^{-1}\sigma)\Big) + \mathrm{d} (\|\sigma\|^2) \, (\dot{J}'V) & \\ &= \mathrm{d} (\|\sigma\|^2) \, (\dot{J}'V), \end{split}$$

where, in the last step, we applied Lemma 3.9 to the triple \dot{J}' , $g^{-1}\sigma$, $\nabla_V^g g^{-1}\sigma \in T_J \mathcal{J}(\Sigma)$. Finally, we have

$$\mathrm{d}f\left(\dot{J}'V\right) = \frac{1}{2\sqrt{1+\left\|\sigma\right\|^2}}\,\mathrm{d}(\left\|\sigma\right\|^2)\,\dot{J}'V = f^{-1}\langle\nabla^g_{(g^{-1}\sigma)V}\,\sigma\mid\dot{J}'\rangle,$$

which concludes the proof of relation (4.15), and hence of the statement.

The last statement of Proposition K is slightly more elaborated, because it requires a "conversion" in the linear connection: while the first three characterizations are expressed in terms of the Levi-Civita connection ∇^g of g, the last one involves the Levi-Civita connections ∇^l and ∇^r of the Riemannian metrics h_l and h_r , respectively. For this reason we will need some additional ingredients, described in Lemmas 4.22, 4.23, 4.24. The transition from ∇^g to ∇^l , ∇^r is done passing through the Levi-Civita connection of the Riemannian metric h: in Lemma 4.22 we express the derivative $\nabla^h_{\bullet}B$ in terms of $\nabla^g_{\bullet}\sigma$, and in Lemma 4.23 we compute the h-divergence operator in terms of div_g . With these tools, we will be able to determine the expressions for the divergence with respect to h_l and h_r in terms of div_g and the derivative $\nabla^g_{\bullet}\sigma$, as described in Lemma 4.24, which will make the equivalence iii $\Leftrightarrow iv$ simpler to handle.

Lemma 4.22. Let $(J, \sigma) \in T^*\mathcal{J}(\Sigma)$, where σ is the real part of a J-holomorphic quadratic differential. Let h denote the Riemannian metric (1+f)g, with Levi-Civita connection ∇^h , and let $B = h^{-1}\sigma$ be the h-self-adjoint operator associated to σ . For every tangent vector field X we have

$$\nabla_X^h B = (1+f)^{-1} f^{-1} \nabla_X^g (g^{-1} \sigma),$$

where
$$f = f(\|\sigma\|_g) = \sqrt{1 + \|\sigma\|_g^2}$$
, and ∇^g is the Levi-Civita connection of $g = \rho(\cdot, J \cdot)$.

Proof. First we observe that, if σ is zero, then the relation is obviously satisfied. In what follows we will assume that σ is not identically zero.

The tensor $\nabla_X^g(g^{-1}\sigma)$ is a symmetric and traceless endomorphism of the tangent space of Σ . For every $p \in \Sigma$ outside the set of zeros of σ (which is a finite set), the elements $(g^{-1}\sigma)_p$ and $(Jg^{-1}\sigma)_p$ form a basis of the space of traceless symmetric endomorphisms of $T_p\Sigma$. In particular, using the scalar product $\langle \cdot, \cdot \rangle$ we can represent $\nabla_X^g(g^{-1}\sigma)$ in terms of such basis, obtaining the following expression:

$$\begin{split} \nabla_X^g(g^{-1}\sigma)Y &= \frac{1}{\|\sigma\|_g^2} \left(\langle g^{-1}\sigma, \nabla_X^g(g^{-1}\sigma) \rangle g^{-1}\sigma + \langle Jg^{-1}\sigma, \nabla_X^g(g^{-1}\sigma) \rangle Jg^{-1}\sigma \right) \\ &= \frac{1}{\|\sigma\|_g^2} \left(\frac{1}{2} \operatorname{d}(\|\sigma\|_g^2) \left(X \right) g^{-1}\sigma + \langle \sigma, \nabla_{JX}^g \sigma \rangle Jg^{-1}\sigma \right) \\ &= \frac{1}{2\|\sigma\|_g^2} \left(\operatorname{d}(\|\sigma\|_g^2) \left(X \right) g^{-1}\sigma + \operatorname{d}(\|\sigma\|_g^2) \left(JX \right) Jg^{-1}\sigma \right). \end{split}$$

From the first to the second line, we are making use of the definitions of the scalar products $\langle \cdot, \cdot \rangle$ and Lemma 2.3 part iv). By definition of f, we have $\|\sigma\|_g^2 = f^2 - 1$, therefore $d(\|\sigma\|_g^2) = 2f df$. Combining this with the chain of equalities above, we obtain that

$$\nabla_X^g (g^{-1}\sigma) Y = (f^2 - 1)^{-1} f \left(df(X) g^{-1}\sigma + df(JX) J g^{-1}\sigma \right).$$

The exact same observations made to express $\nabla_X^g(g^{-1}\sigma)Y$ allow us to deduce that

$$(\nabla_X^h B)Y = \frac{1}{2\|B\|^2} \left(d(\|B\|^2) (X) B + d(\|B\|^2) (JX) JB \right)$$

(remember that B is Codazzi with respect to h by Lemma 2.3 part i)). Unraveling the definitions of f and B, we see that $||B||^2 = (1+f)^{-1}(f-1)$, and consequently $d(||B||^2) = 2(1+f)^{-2} df$. In particular we obtain that

$$\begin{split} (\nabla_X^h B) Y &= (1+f)^{+1-2} (f-1)^{-1} \left(\mathrm{d} f \left(X \right) B + \mathrm{d} f \left(J X \right) J B \right) \\ &= (f^2 - 1)^{-1} (1+f)^{-1} \left(\mathrm{d} f \left(X \right) g^{-1} \sigma + \mathrm{d} f \left(J X \right) J g^{-1} \sigma \right), \end{split}$$

where in the last step we used that $B = h^{-1}\sigma = (1+f)^{-1}g^{-1}\sigma$. Outside the zero locus of σ , the statement now follows by comparing the two expressions found for $\nabla_X^g(g^{-1}\sigma)Y$ and $\nabla_X^h B$, and the identity holds on the whole surface by continuity.

Lemma 4.23. Let T be a smooth section of endomorphisms of $T\Sigma$. Then

$$\operatorname{div}_h T = \operatorname{div}_g T + (1+f)^{-1} \operatorname{d} f \circ T_0^s,$$

where T_0^s is the g-symmetric and traceless part of T.

Proof. By Koszul's formula we have

$$2h(\nabla_X^h Y, Z) = X(h(Y, Z)) + Y(h(X, Z)) - Z(h(X, Y)) + h([X, Y], Z) - h([X, Z], Y) - h([Y, Z], X)$$

= $df(X)g(Y, Z) + df(Y)g(X, Z) - df(Z)g(X, Y) + 2h(\nabla_X^g Y, Z),$

where in the last step we used the relation h = (1 + f)g and the Koszul's formula for g. In particular, we deduce that

$$\nabla_X^h Y = \nabla_X^g Y + \frac{1}{2} (1+f)^{-1} (df(X) Y + df(Y) X - g(X, Y) \operatorname{grad}_g f),$$

where $\operatorname{grad}_q f = g^{-1} df$ is the g-gradient of f. Then we have

$$\begin{split} (\operatorname{div}_h T)X &= \operatorname{d} x^i \left((\nabla^h_{\partial_i} T) X \right) = \operatorname{d} x^i \left(\nabla^h_{\partial_i} (TX) - T (\nabla^h_{\partial_i} X) \right) \\ &= \operatorname{d} x^i \left(\nabla^g_{\partial_i} (TX) - T (\nabla^g_{\partial_i} X) \right) + \frac{1}{2} (1+f)^{-1} \operatorname{d} x^i \left(\partial_i f \, TX + \operatorname{d} f \, (TX) \, \partial_i + \right. \\ &\quad - g(\partial_i, TX) \operatorname{grad}_g f - T (\partial_i f \, X + \operatorname{d} f \, (X) \, \partial_i - g(\partial_i, X) \operatorname{grad}_g f)) \\ &= (\operatorname{div}_g T) X + \frac{1}{2} (1+f)^{-1} (\operatorname{d} f \, (TX) + 2 \operatorname{d} f \, (TX) - \operatorname{d} f \, (TX) - \operatorname{d} f \, (TX) + \right. \\ &\quad - \operatorname{d} f \, (X) \operatorname{tr} T + g(T \operatorname{grad}_g f, X)) \\ &= (\operatorname{div}_g T) X + \frac{1}{2} (1+f)^{-1} (\operatorname{d} f \, (TX) - \operatorname{d} f \, (X) \operatorname{tr} T + \operatorname{d} f \, (T^*X)) \\ &= (\operatorname{div}_g T) X + (1+f)^{-1} \operatorname{d} f \circ \left(\frac{T+T^*}{2} - \frac{\operatorname{tr} T}{2} \, \mathbbm{1} \right) (X) \\ &= (\operatorname{div}_g T) X + (1+f)^{-1} \operatorname{d} f \circ T_0^s(X). \end{split}$$

From the first to the second line we used the relation found above between the Levi-Civita connections of g and h; in the forth line T^* is denoting the g-adjoint of T; in the last line we observed that the endomorphism $(T + T^*)/2$ is the g-symmetric part of T, and $(\operatorname{tr} T)/2 \mathbb{1}$ is its full-trace part.

Let A_l and A_r be the tensors

$$A_l := \mathbb{1} - JB, \qquad A_r := \mathbb{1} + JB.$$

Then the metrics h_l and h_r coincide with $h(A_l, A_l)$ and $h(A_r, A_r)$, respectively (see for instance the proof of Lemma 4.17). If σ is the real part of a J-holomorphic quadratic differential, then the tensors A_l and A_r are h-Codazzi (B is h-Codazzi by Lemma 2.3, and $\mathbb{1}$ and J are ∇^h -parallel) and h-symmetric. In particular, we can express the Levi-Civita connection of h_l and h_r respectively as follows:

$$\nabla_X^{h_l} Y = A_l^{-1} \nabla_X^{h} (A_l Y), \qquad \nabla_X^{h_r} Y = A_r^{-1} \nabla_X^{h} (A_r Y). \tag{4.17}$$

These relations can be proved by checking that the connections defined on the right-hand sides are torsion free (which follows from A_l and A_r being h-Codazzi), and h_l - and h_r -

symmetric, respectively (which follows from A_l and A_r being h-symmetric and from the description of h_l and h_r given above). See also [KS07].

Lemma 4.24. Let T be a smooth section of traceless endomorphisms of $T\Sigma$. Then

$$(\operatorname{div}_{h_l} T)X = (\operatorname{div}_g T)X - f^{-1} \langle \nabla_X^g \sigma \mid JT^s \rangle$$
$$(\operatorname{div}_{h_r} T)X = (\operatorname{div}_g T)X + f^{-1} \langle \nabla_X^g \sigma \mid JT^s \rangle$$

where T^s stands for the g-symmetric part of T.

Proof. Recalling relations (3.17) and (3.16), we have that

$$A_{l,r}^{-1} = \frac{1+f}{2}(\mathbb{1} \pm JB). \tag{4.18}$$

In this expression and the ones that will follow, we consider the sign above in \pm or \mp to be the one appearing for the expression of A_l , and the one on the bottom for A_r . Being 1 and $J \nabla^h$ -parallel, we have $\nabla^h_X A_{l,r} = \mp J \nabla^h_X B$. Applying the expressions (4.17) for the Levi-Civita connections of h_l and h_r , respectively, we find

$$\begin{split} (\operatorname{div}_{h_{l,r}}T)X &= \operatorname{d}x^{i}\left((\nabla_{\partial_{i}}^{l,r}T)X\right) = \operatorname{d}x^{i}\left(\nabla_{\partial_{i}}^{l,r}(TX) - T(\nabla_{\partial_{i}}^{l,r}X)\right) \\ &= \operatorname{d}x^{i}\left(A_{l,r}^{-1}\nabla_{\partial_{i}}^{h}(A_{l,r}TX) - TA_{l,r}^{-1}\nabla_{\partial_{i}}^{h}(A_{l,r}X)\right) \\ &= \operatorname{d}x^{i}\left(A_{l,r}^{-1}(\nabla_{\partial_{i}}^{h}A_{l,r})TX + A_{l,r}^{-1}A_{l,r}(\nabla_{\partial_{i}}^{h}T)X + A_{l,r}^{-1}A_{l,r}T(\nabla_{\partial_{i}}^{h}X) + \\ &- TA_{l,r}^{-1}(\nabla_{\partial_{i}}^{h}A_{l,r})X - TA_{l,r}^{-1}A_{l,r}(\nabla_{\partial_{i}}^{h}X)\right) \\ &= \operatorname{d}x^{i}\left(A_{l,r}^{-1}(\nabla_{\partial_{i}}^{h}A_{l,r})TX\right) + \left(\operatorname{div}_{h}T\right)X - \operatorname{d}x^{i}\left(TA_{l,r}^{-1}(\nabla_{\partial_{i}}^{h}A_{l,r})X\right) \\ &= \left(\operatorname{div}_{h}T\right)X + \operatorname{tr}\left(A_{l,r}^{-1}(\nabla_{TX}^{h}A_{l,r}) - TA_{l,r}^{-1}(\nabla_{X}^{h}A_{l,r})\right). \quad (A_{l,r}\ h_{l,r}\text{-Codazzi}) \end{split}$$

Lemma 4.23 allows us to express the first term of the sum in terms of the divergence with respect to g, so now we focus on the other two terms. For the first one, we express it as follows:

$$\operatorname{tr}\left(A_{l,r}^{-1}(\nabla_{TX}^{h}A_{l,r})\right) = \frac{1+f}{2}\operatorname{tr}\left((\mathbb{1} \pm JB)(\nabla_{TX}^{h}A_{l,r})\right) \qquad (\text{rel. } (4.18))$$

$$= \mp \frac{1+f}{2}\operatorname{tr}\left((\mathbb{1} \pm JB)J\nabla_{TX}^{h}B\right) \qquad (\nabla_{X}^{h}A_{l,r} = \mp J\nabla_{X}^{h}B)$$

$$= -\frac{1+f}{2}\operatorname{tr}\left(JBJ\nabla_{TX}^{h}B\right) \qquad (J\nabla_{TX}^{h}B\operatorname{ traceless})$$

$$= -\frac{1+f}{2}\operatorname{tr}\left(B\nabla_{TX}^{h}B\right) \qquad (B \in T_{J}\mathcal{J}(\Sigma))$$

$$= -(1+f)^{-1}\operatorname{d}f\left(TX\right). \qquad (\operatorname{d}(\|B\|^{2}) = 2(1+f)^{-2}\operatorname{d}f\right)$$

We proceed similarly for the third term:

$$\operatorname{tr}\left(TA_{l,r}^{-1}(\nabla_X^h A_{l,r})\right) = \frac{1+f}{2}\operatorname{tr}\left(T(\mathbb{1} \pm JB)(\nabla_X^h A_{l,r})\right)$$
 (rel. (4.18))

$$= \mp \frac{1+f}{2} \operatorname{tr} \left(T(\mathbb{1} \pm JB) J \nabla_X^h B \right) \qquad (\nabla_X^h A_{l,r} = \mp J \nabla_X^h B)$$

$$= \frac{1}{2} f^{-1} (\mp \operatorname{tr} \left(T J \nabla_X^g (g^{-1}\sigma) \right) - \operatorname{tr} \left(T J B J \nabla_X^g (g^{-1}\sigma) \right)) \quad \text{(Lemma 4.22)}$$

$$= f^{-1} (\pm \langle \nabla_X^g \sigma \mid J T^s \rangle - (1+f)^{-1} \langle \nabla_X^g \sigma \mid T^a g^{-1}\sigma \rangle).$$

In the last step, we expressed B as $(1+f)^{-1}g^{-1}\sigma$, and we observed that, if $T=T^a+T^s$, where T^a and T^s denote the g-anti-symmetric and g-symmetric parts of T, respectively, then T^a does not contribute to the term $\operatorname{tr}(TJ\nabla_X^g(g^{-1}\sigma))$ since $\nabla_X^g(g^{-1}\sigma)$ is traceless, and similarly T^s does not contribute to the term $\operatorname{tr}(TJBJ\nabla_X^g(g^{-1}\sigma))$ by Lemma 3.9, part ii (observe that T^s is traceless since T is, by hypothesis).

Combining the relations obtained above with Lemma 4.23 we see that

$$(\operatorname{div}_{h_{l,r}}T)X = (\operatorname{div}_{h}T)X + \operatorname{tr}\left(A_{l,r}^{-1}(\nabla_{TX}^{h}A_{l,r}) - A_{l,r}^{-1}T(\nabla_{X}^{h}A_{l,r})\right)$$

$$= (\operatorname{div}_{g}T)X + (1+f)^{-1}\operatorname{d}f(T^{s}X) - (1+f)^{-1}\operatorname{d}f(TX) +$$

$$- f^{-1}(\pm\langle\nabla_{X}^{g}\sigma\mid JT^{s}\rangle - (1+f)^{-1}\langle\nabla_{X}^{g}\sigma\mid T^{a}g^{-1}\sigma\rangle)$$

$$= (\operatorname{div}_{g}T)X - (1+f)^{-1}\operatorname{d}f(T^{a}X) + (1+f)^{-1}f^{-1}\langle\nabla_{X}^{g}\sigma\mid T^{a}g^{-1}\sigma\rangle +$$

$$\mp f^{-1}\langle\nabla_{X}^{g}\sigma\mid JT^{s}\rangle.$$

Since the space of g-anti-symmetric endomorphisms of $T_p\Sigma$ has real dimension 1, we can write $T^a = u J$ for some smooth function u. Therefore

$$\begin{split} \left\langle \nabla_{X}^{g} \sigma \mid T^{a} g^{-1} \sigma \right\rangle &= u \left\langle \nabla_{X}^{g} \sigma \mid J g^{-1} \sigma \right\rangle \\ &= u \left\langle \nabla_{JX}^{g} \sigma, \sigma \right\rangle_{g} \\ &= \frac{u}{2} \operatorname{d}(\|\sigma\|_{g}^{2}) \left(JX\right) \\ &= u f \operatorname{d} f \left(JX\right), \end{split} \tag{$f^{2} - 1 = \|\sigma\|_{g}^{2}$}$$

where, from the first to the second line, we used Lemma 2.3 part iv). Expressing again T^a as u J in the relation we found above for $(\operatorname{div}_{h_{l,r}} T)X$, and combining it with what just shown, we obtain the desired statement.

We are finally ready to prove the last statement of Proposition K:

Proof of the last statement. We will apply the previous lemma to $T = \dot{J}_l$, \dot{J}_r . To do so, we express the g-symmetric and g-anti-symmetric parts of \dot{J}_l , \dot{J}_r . From the relation described in the proof of Theorem C (baby version) for the differential of the Mess homeomorphism \mathcal{M} , we see that \dot{J}_l and \dot{J}_r can be expressed in terms of $(\dot{J}, \dot{\sigma})$ as follows:

$$\dot{J}_{l,r} = \langle \sigma \mid Q^{\pm} \rangle J \pm f Q^{\pm}, \tag{4.19}$$

where $Q^{\pm} = Q^{\pm}(\dot{J}, \dot{\sigma})$ is defined as in Proposition K part *iii*) (we will not need the actual definition of Q^{\pm} , but only the fact that $Q^{\pm} \in T_J \mathcal{J}(\Sigma)$ and the expression (V3) for the

equations of $V_{(J,\sigma)}$). In particular we have that the g-symmetric parts of \dot{J}_l , \dot{J}_r are:

$$(\dot{J}_{l,r})^s = \pm f Q^{\pm} \tag{4.20}$$

The very last ingredient needed is the following convenient way to express the terms \dot{J}_l and \dot{J}_r :

$$\dot{J}_{l,r} = \mp Q^{\pm} J J_{l,r} - \langle \sigma \mid J Q^{\pm} \rangle \mathbb{1}, \tag{4.21}$$

whose proof goes as follows:

$$\dot{J}_{l,r} = \langle \sigma \mid Q^{\pm} \rangle J \pm f Q^{\pm}$$

$$= \pm f Q^{\pm} + J Q^{\pm} g^{-1} \sigma - \langle \sigma \mid J Q^{\pm} \rangle \mathbb{1}$$

$$= \mp Q^{\pm} J (f J \pm g^{-1} \sigma) - \langle \sigma \mid J Q^{\pm} \rangle \mathbb{1}$$

$$= \mp Q^{\pm} J J_{l,r} - \langle \sigma \mid J Q^{\pm} \rangle \mathbb{1},$$
(rel. (3.18))

where in the second line we applied Lemma 3.9 to $JQ^{\pm}, g^{-1}\sigma \in T_J\mathcal{J}(\Sigma)$. Now, applying Lemma 4.24, we obtain

$$(\operatorname{div}_{h_{l,r}} \dot{J}_{l,r}) X = (\operatorname{div}_g \dot{J}_{l,r}) X \mp \langle \nabla_X^g \sigma \mid J(\pm Q^{\pm}) \rangle$$
 (rel. (4.20))
$$= \mp \operatorname{div}_g (Q^{\pm} J J_{l,r}) X - \operatorname{d}(\langle \sigma \mid J Q^{\pm} \rangle) X - \langle \nabla_X^g \sigma \mid J Q^{\pm} \rangle.$$
 (rel. (4.21))

Using once again Lemma 2.3 part iv), this identity can be rewritten as

$$\operatorname{div}_{h_{l,r}} \dot{J}_{l,r} + \operatorname{d}(\langle \sigma \mid JQ^{\pm} \rangle) = \mp \operatorname{div}_{q}(Q^{\pm}JJ_{l,r}) - \langle \nabla^{g}_{I\bullet}\sigma \mid Q^{\pm} \rangle.$$

This finally concludes the proof of the last part of Proposition 3.4. By a straightforward computation using relations (3.18) and (4.19), we can see that

$$\langle \sigma \mid JQ^{\pm} \rangle = -\frac{\langle [J_l, J_r], \dot{J}_l \rangle}{8(1 - \langle J_l, J_r \rangle)} = \frac{\langle [J_l, J_r], \dot{J}_r \rangle}{8(1 - \langle J_l, J_r \rangle)} \ .$$

5. Geometric interpretations

In this section we conclude the study of the para-hyperKähler structure on the deformation space $\mathcal{MGH}(\Sigma)$ for Σ a closed surface of genus ≥ 2 , giving interpretations in terms of anti-de Sitter geometry to the elements that constitute the para-hyperKähler structure $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$. As a byproduct, we will deduce that the symplectic forms $\omega_{\mathbf{I}}, \omega_{\mathbf{J}}, \omega_{\mathbf{K}}$ are non-degenerate and closed, which will conclude the proof of Theorem A. Finally, we study the relation between $\omega_{\mathbf{I}}, \omega_{\mathbf{J}}, \omega_{\mathbf{K}}$ and Goldman symplectic form $\Omega_{Gol}^{\mathbb{B}}$ on the $\mathbb{P}\mathrm{SL}(2,\mathbb{B})$ -character variety.

5.1. The cotangent bundle parametrization

Recall from Section 2.2 that Krasnov and Schlenker introduced in [KS07] a way of parametrizing the deformation space $\mathcal{MGH}(\Sigma) \cong \mathcal{MS}(\Sigma)$ via the cotangent bundle $T^*\mathcal{T}^{\mathfrak{c}}(\Sigma)$ to the Teichmüller space of Σ . Precisely, they produced a mapping-class group invariant homeomorphism

$$\mathcal{F} \colon \mathcal{MS}(\Sigma) \to T^*\mathcal{T}^{\mathfrak{c}}(\Sigma)$$
,

see Theorem 2.8. Using this map, we can identify, up to a multiplicative factor, the natural symplectic structure on $T^*\mathcal{T}^{\mathfrak{c}}(\Sigma)$ with the complex symplectic form $\overline{\omega}_{\mathbf{I}}^{\mathbb{C}}$, where $\omega_{\mathbf{I}}^{\mathbb{C}} = \omega_{\mathbf{J}} + i\omega_{\mathbf{K}}$.

Theorem B (genus ≥ 2). Let Σ be a closed oriented surface of genus ≥ 2 . Then

$$\mathcal{F}^*(\mathcal{I}_{T^*\mathcal{T}(\Sigma)}, \Omega_{T^*\mathcal{T}(\Sigma)}^{\mathbb{C}}) = \left(-\mathbf{I}, -\frac{i}{2}\overline{\omega}_{\mathbf{I}}^{\mathbb{C}}\right) ,$$

where $\mathcal{I}_{T^*\mathcal{T}(\Sigma)}$ denotes the complex structure of $T^*\mathcal{T}(\Sigma)$ and $\Omega^{\mathbb{C}}_{T^*\mathcal{T}(\Sigma)}$ its complex symplectic form.

Proof. The proof is an adaptation of the arguments of the genus one version of Theorem B, proved in Section 3.5. A computation identical to Remark 3.14 shows that if $(J_t)_t$ is a 1-parameter family of complex structures on Σ , with $J_0 = J$, then the Beltrami differential of the identity map $id: (\Sigma, J) \to (\Sigma, J_t)$ is

$$\nu_t = (\mathbb{1} - J_t J)^{-1} (\mathbb{1} + J_t J)$$

and that $\dot{\nu} = \frac{1}{2}\dot{J}J$. Hence given a pair $(\dot{J}, \dot{\sigma})$ in our model of the tangent space $T_{(J,\sigma)}\mathcal{MS}_0(\Sigma)$ (see Proposition K and Theorem L), we have

$$\mathrm{d}\pi \circ \mathrm{d}\mathcal{F}_{(J,\sigma)}\left(\dot{J},\dot{\sigma}\right) = \frac{1}{2}\dot{J}J.$$

Now let g_J be the Riemannian metric $\rho(\cdot, J\cdot)$, let $\{e_1, e_2 = Je_1\}$ be a local g_J -orthonormal frame and let $\phi = \sigma - i \sigma(\cdot, J\cdot)$ as usual. The same computation as in the genus one case, using the definition of the pairing in $T^*\mathcal{T}^{\mathfrak{c}}(\Sigma)$ (see (3.13) and (3.14)), shows that

$$(\phi \bullet \dot{\nu})(e_1, e_2) = \frac{1}{2i} \left(\phi(\dot{\nu}(e_1), e_2) - \phi(\dot{\nu}(e_2), e_1) \right)$$
$$= -\frac{i}{2i} \overline{\lambda_{(J,\sigma)}^{\mathbb{C}}(\dot{J}, \dot{\sigma})}.$$

Hence we obtain

$$\mathcal{F}^*\Omega^{\mathbb{C}}_{T^*\mathcal{T}(\Sigma)} = -\frac{i}{2}(\omega_{\mathbf{J}} - i\,\omega_{\mathbf{K}}) = -\frac{i}{2}\overline{\omega}^{\mathbb{C}}_{\mathbf{I}} \ ,$$

which proves one part of the statement. For the pull-back of the complex structure $\mathcal{I}_{T^*\mathcal{T}(\Sigma)}$, one argues exactly as in the genus one case to conclude that $\mathcal{F}^*\mathcal{I}_{T^*\mathcal{T}(\Sigma)} = -\mathbf{I}$.

As a consequence, we immediately obtain:

Corollary 5.1. The almost complex structure I on $\mathcal{MS}_0(\Sigma)$ is integrable; the 2-forms $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$ are symplectic forms.

Proof. Since $\mathcal{I}_{T^*\mathcal{T}(\Sigma)}$ is an integrable almost-complex structure and $\Omega^{\mathbb{C}}_{T^*\mathcal{T}(\Sigma)}$ is a complex symplectic form, the two statements follow immediately from Theorem B.

5.2. The Mess homeomorphism

In Section 2.3 we explained that, under the identification between $\mathcal{MS}(\Sigma)$ and $\mathcal{MGH}(\Sigma)$, Mess homeomorphism

$$\mathcal{M}^{\mathfrak{c}} \colon \mathcal{MS}(\Sigma) \to \mathcal{T}^{\mathfrak{c}}(\Sigma) \times \mathcal{T}^{\mathfrak{c}}(\Sigma)$$

is expressed by the formula of Lemma 2.10, which is formally the same expression as the map $\mathcal{M}: T^*\mathcal{J}(\mathbb{R}^2) \to \mathcal{J}(\mathbb{R}^2) \times \mathcal{J}(\mathbb{R}^2)$ defined in Section 3.6. This implies that $\mathcal{M}^{\mathfrak{c}}: \mathcal{MS}(\Sigma) \to \mathcal{T}^{\mathfrak{c}}(\Sigma) \times \mathcal{T}^{\mathfrak{c}}(\Sigma)$ is induced by the map (see Remark 4.8) that we introduced in the finite dimensional context.

Recall also that $\mathcal{T}^{\mathfrak{c}}(\Sigma) \times \mathcal{T}^{\mathfrak{c}}(\Sigma)$ is naturally endowed with a para-complex structure $\mathcal{P}_{\mathcal{T}^{\mathfrak{c}}(\Sigma) \times \mathcal{T}^{\mathfrak{c}}(\Sigma)}$, which is the endomorphism of the cotangent bundle for which the integral submanifolds of the distribution of 1-eigenspaces are the slices $\mathcal{T}^{\mathfrak{c}}(\Sigma) \times \{*\}$, and those for the (-1)-eigenspaces are the slices $\{*\} \times \mathcal{T}^{\mathfrak{c}}(\Sigma)$. Plus, it has a para-complex symplectic form

$$\Omega_{\mathcal{T}^{\mathfrak{c}}(\Sigma) \times \mathcal{T}^{\mathfrak{c}}(\Sigma)}^{\mathbb{B}} := \frac{1}{2} (\pi_{l}^{*} \Omega_{WP} + \pi_{r}^{*} \Omega_{WP}) + \frac{\tau}{2} (\pi_{l}^{*} \Omega_{WP} - \pi_{r}^{*} \Omega_{WP})$$

where Ω_{WP} is the Weil-Petersson symplectic form and π_l, π_r denote the projections on the left and right factor. Here we show the relation of these structures with the parahyperKähler structure ($\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K}$), via Mess' diffeomorphism.

Theorem C (genus ≥ 2). Let Σ be a closed oriented surface of genus ≥ 2 . Then

$$\mathcal{M}^*(\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}, 4\Omega^{\mathbb{B}}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}) = (\mathbf{J}, \omega^{\mathbb{B}}_{\mathbf{J}}) ,$$

where $\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$ denotes the para-complex structure of $\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)$ and $\Omega^{\mathbb{B}}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$ its para-complex symplectic form.

Proof. Let $W_{(J_l,J_r)}$ denote the image of $V_{(J,\sigma)}$ under $d\mathcal{M}$. Because $\operatorname{div}_{h_l}\dot{J}_l$ and $\operatorname{div}_{h_r}\dot{J}_r$ are exact 1-forms by the last statement in Proposition K, the vector space $W_{(J_l,J_r)}$ is $(\Omega_{J_l} \oplus (\pm \Omega_{J_r}))$ -orthogonal to the tangent space to the orbit of $\operatorname{Symp}_0(\Sigma,\rho)$ (see [Don03] or Proposition 4.3). Moreover, by Lemma 4.21 and equivariance of the map \mathcal{M} , the space $W_{(J_l,J_r)}$ is in direct sum with the tangent space to the orbit. Therefore, $(W_{(J_l,J_r)}, \frac{1}{4}(\Omega_{J_l} \oplus (\pm \Omega_{J_r})))$ is symplectomorphic to $(T_{[J_l]}\mathcal{T}^{\mathfrak{c}}(\Sigma) \times T_{[J_r]}\mathcal{T}^{\mathfrak{c}}(\Sigma), \pi_l^*\Omega_{WP} \pm \pi_r^*\Omega_{WP})$. Finally, the computations in Section 3.6 (see the proof of the baby version of Theorem C) can be carried out word-by-word in this context and show that

$$\frac{1}{2}\mathcal{M}^*((\Omega_{J_l} \oplus \Omega_{J_r}) + \tau(\Omega_{J_l} \oplus (-\Omega_{J_r}))) = \omega_{\mathbf{I}} + \tau\omega_{\mathbf{K}} = \omega_{\mathbf{J}}^{\mathbb{B}}.$$

The fact that $\mathcal{M}^*\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)} = \mathbf{J}$ then follows by the usual argument, as in the conclusion of the baby version of Theorem C, at the end of Section 3.6.

As a consequence, we obtain:

Corollary 5.2. The almost para-complex structure \mathbf{J} on $\mathcal{MS}_0(\Sigma)$ is integrable; the 2-forms $\omega_{\mathbf{I}}$ and $\omega_{\mathbf{K}}$ are symplectic forms.

Proof. This follows immediately from Theorem C, the integrability of $\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$ and the closedness and non-degeneracy of $\Omega^{\mathbb{B}}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$.

Corollary 5.3. The metric \mathbf{g} on $\mathcal{MS}_0(\Sigma)$ is non-degenerate.

Proof. The proof follows by observing (for instance) that $\mathbf{g}(\cdot, \cdot) = \omega_{\mathbf{J}}(\cdot, \mathbf{J})$, together with the non-degeneracy of $\omega_{\mathbf{J}}$ and the invertibility of \mathbf{J} .

5.3. The circle action on $\mathcal{MS}(\Sigma)$

We now consider a circle action on $\mathcal{MS}(\Sigma)$, which is simply defined, under the diffeomorphism

$$\mathcal{F}: \mathcal{MS}(\Sigma) \to T^*\mathcal{T}^{\mathfrak{c}}(\Sigma)$$
,

by $e^{i\theta} \cdot ([J], q) = ([J], e^{i\theta}q)$, for q a holomorphic quadratic differential. As in Remark 2.5, we see that the circle action is induced by the following expression in terms of pairs (h, B):

$$R_{\theta}(J,\sigma) = (J,\cos(\theta)\sigma + \sin(\theta)\sigma(\cdot,J\cdot))$$
,

and it is easily checked that R_{θ} descends to the quotient $\mathcal{MS}(\Sigma)$. As done in the introduction, we denote by \mathcal{C}_{θ} the composition $\mathcal{C} \circ R_{\theta}$, for every $\theta \in S^1$. We then prove the following results:

Theorem E (genus ≥ 2). Let Σ be a closed oriented surface of genus ≥ 2 . The circle action on $\mathcal{MGH}(\Sigma)$ is Hamiltonian with respect to $\omega_{\mathbf{I}}$, and satisfies

$$R_{\theta}^* \mathbf{g} = \mathbf{g}$$
 $R_{\theta}^* \omega_{\mathbf{I}} = \omega_{\mathbf{I}}$ $R_{\theta}^* \omega_{\mathbf{I}}^{\mathbb{C}} = e^{-i\theta} \omega_{\mathbf{I}}^{\mathbb{C}}$.

A Hamiltonian is given by the function

$$\mathcal{A}: \mathcal{MS}(\Sigma) \to \mathbb{R}$$
 $\mathcal{A}([J, \sigma]) = \int_{\Sigma} \left(1 + \sqrt{1 + \|\sigma\|_J^2}\right) \rho$.

that gives the area of the maximal surface.

Remark 5.4. In [BMS13] and [BMS15], the authors studied the landslide flow on $\mathcal{T}^{\mathfrak{c}}(\Sigma) \times \mathcal{T}^{\mathfrak{c}}(\Sigma)$ that in our notations corresponds to the 1-parameter family of maps $\mathcal{M} \circ R_{\theta/2} \circ \mathcal{M}^{-1}$. They showed that the landslide flow is Hamiltonian with respect to the symplectic form $\pi_l^* \Omega_{WP} + \pi_r^* \Omega_{WP}$. Our Theorem E recovers this result, including it in a more general context.

Theorem F (genus ≥ 2). Let Σ be a closed oriented surface of genus ≥ 2 . Then the function -4A is a para-Kähler potential for the para-Kähler structures (\mathbf{g}, \mathbf{J}) and (\mathbf{g}, \mathbf{K}) .

The proofs are straightforward adaptations of those provided in Section 3.7 for the genus one case. We only remark that we modified the natural Hamiltonian that one obtains by integrating the function H over Σ , namely the function

$$[J, \sigma] \mapsto \int_{\Sigma} f(\|\sigma\|_J) \rho = \int_{\Sigma} \sqrt{1 + \|\sigma\|_J^2} \rho$$

by adding a constant, so that $\mathcal{A}([J,\sigma])$ can be interpreted as the area of the maximal surface in Anti-de Sitter space corresponding to the point $[J,\sigma]$ under the identification between $\mathcal{MS}(\Sigma)$ and the space of equivariant maximal surfaces in Anti-de Sitter space. Indeed, recalling that the first fundamental form of the maximal surface is given by the metric $h = (1 + f(\|\sigma\|_J))g_J$ where $g_J = \rho(\cdot, J\cdot)$, the area form of h is

$$dA_h = (1 + f(\|\sigma\|_J))dA_{g_J} = \left(1 + \sqrt{1 + \|\sigma\|_J^2}\right)\rho.$$

Theorem C and Theorem E have other direct consequences. Recalling from Section 2.4 the definition of the map

$$C_{\theta} = C \circ R_{\theta} : \mathcal{MS}(\Sigma) \to \mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$$
,

we see that

$$C_{\theta}^{*}(\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}, 4\Omega_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}^{\mathbb{B}}) = (\cos(\theta)\mathbf{K} - \sin(\theta)\mathbf{J}, \omega_{\mathbf{I}} - \tau(\cos(\theta)\omega_{\mathbf{J}} + \sin(\theta)\omega_{\mathbf{K}})) . (5.1)$$

As an immediate consequence, we conclude the proofs of Theorems D and G. For the former, it suffices to observe that for $\theta = 0$ the parameterization $\mathcal{C}_{\theta} = \mathcal{C} \circ R_{\theta} : \mathcal{MS}(\Sigma) \to \mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$, given by the induced metric on the two Cauchy surfaces of constant curvature -2, is simply $\mathcal{C} = \mathcal{C}_0$.

Theorem D (genus ≥ 2). Let Σ be a closed oriented surface of genus ≥ 2 . Then

$$\mathcal{C}^*(\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}, 4\Omega^{\mathbb{B}}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}) = (\mathbf{K}, \omega^{\mathbb{B}}_{\mathbf{K}}) \ ,$$

where $\mathcal{P}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$ denotes the para-complex structure of $\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)$ and $\Omega^{\mathbb{B}}_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}$ its para-complex symplectic form.

In particular, we deduce:

Corollary 5.5. The almost para-complex structure **K** on $\mathcal{MS}_0(\Sigma)$ is integrable.

Finally, we have the proof of Theorem G, which is expressed purely in terms of Teichmüller theory. Namely, recalling that the map

$$\mathcal{H}_{\theta}: T^*\mathcal{T}(\Sigma) \to \mathcal{T}(\Sigma) \times \mathcal{T}(\Sigma)$$

associates to a pair ([J], q) the pair $(h_{(J, -e^{i\theta}q)}, h_{(J, e^{i\theta}q)})$ of hyperbolic metrics on Σ , where $h_{(J,q)}$ has the property that the (unique) harmonic map $(\Sigma, J) \to (\Sigma, h)$ isotopic to the

identity has Hopf differential equal to q. By Lemma 2.14, this map is identified to $\mathcal{C}_{\theta}^{\mathfrak{h}} \circ \mathcal{F}^{-1}$. Using (5.1) and Theorem B, we obtain Theorem G:

Theorem G (genus ≥ 2). Let Σ be a closed oriented surface of genus ≥ 2 . Then

$$\operatorname{Im} \mathcal{H}_{\theta}^{*}(2\Omega_{\mathcal{T}(\Sigma)\times\mathcal{T}(\Sigma)}^{\mathbb{B}}) = -\operatorname{Re}(ie^{i\theta}\Omega_{T^{*}\mathcal{T}(\Sigma)}^{\mathbb{C}}).$$

5.4. Para-complex geometry of the $\mathbb{P}SL(2,\mathbb{B})$ -character variety

Let \mathbb{B} be the algebra of para-complex numbers, i.e $\mathbb{B} = \mathbb{R} \oplus \tau \mathbb{R}$ with $\tau^2 = 1$. In this section we study the para-complex geometry of $\mathcal{MGH}(\Sigma)$ seen as a component of the $\mathbb{P}\mathrm{SL}(2,\mathbb{B})$ -character variety. We show that multiplication by τ on \mathbb{B} induces a para-complex structure on the $\mathbb{P}\mathrm{SL}(2,\mathbb{B})$ -character variety that makes Goldman symplectic form $\Omega^{\mathbb{B}}_{Gol}$ para-holomorphic, and that the Goldman form $\Omega^{\mathbb{B}}_{Gol}$ coincides with the para-complex symplectic form $\omega^{\mathbb{B}}_{J}$ up to a multiplicative factor (Corollary H). Moreover, we give a formula for Goldman symplectic form based on anti-de Sitter geometry and show that the \mathbb{B} -valued Fenchel-Nielsen coordinates defined in [Tam20] are para-holomorphic Darboux coordinates for $\Omega^{\mathbb{B}}_{Gol}$.

5.4.1. Para-complex structure on the character variety

Let us recall the construction of the isometry group of Anti-de Sitter space in terms of the para-complex numbers, following [Dan13]. We denote by $SL(2,\mathbb{B})$ the set of 2-by-2 matrices with coefficients in \mathbb{B} and determinant 1. Any matrix $A \in SL(2,\mathbb{B})$ can be written uniquely as $A = A_+e^+ + A_-e^-$, where $A_{\pm} \in SL(2,\mathbb{R})$, $e^{\pm} = \frac{1\pm\tau}{2}$ (see Appendix A). The map

$$\mathrm{SL}(2,\mathbb{B}) \to \mathrm{SL}(2,\mathbb{R}) \times \mathrm{SL}(2,\mathbb{R})$$

 $A \mapsto (A_+,A_-)$

induces an isomorphism between $\mathbb{P}SL(2,\mathbb{B})$ and $\mathbb{P}SL(2,\mathbb{R}) \times \mathbb{P}SL(2,\mathbb{R})$, where by $\mathbb{P}SL(2,\mathbb{B})$ we mean

$$\mathbb{P}SL(2,\mathbb{B}) = SL(2,\mathbb{B})/\{\pm 1,\pm \tau\}$$
.

We define the $\mathbb{P}SL(2,\mathbb{B})$ -character variety as follows:

$$\chi(\Sigma, \mathbb{P}SL(2, \mathbb{B})) = \{\rho : \pi_1(\Sigma) \to \mathbb{P}SL(2, \mathbb{B})\}/\mathbb{P}SL(2, \mathbb{B}),$$

namely the set of conjugacy classes of representations $\rho : \pi_1(\Sigma) \to \mathbb{P}\mathrm{SL}(2,\mathbb{B})$. The aformentioned isomorphism between $\mathbb{P}\mathrm{SL}(2,\mathbb{B})$ and $\mathbb{P}\mathrm{SL}(2,\mathbb{R}) \times \mathbb{P}\mathrm{SL}(2,\mathbb{R})$ identifies $\chi(\Sigma,\mathbb{P}\mathrm{SL}(2,\mathbb{B}))$ with $\chi(\Sigma,\mathbb{P}\mathrm{SL}(2,\mathbb{R})) \times \chi(\Sigma,\mathbb{P}\mathrm{SL}(2,\mathbb{R}))$ by associating to ρ the pair of representations $\rho_{\pm} : \pi_1(\Sigma) \to \mathbb{P}\mathrm{SL}(2,\mathbb{R})$ defined by the property $\rho(\gamma) = \rho_+(\gamma)e^+ + \rho_-(\gamma)e^-$ for every $\gamma \in \pi_1(\Sigma)$.

By the work of Mess ([Mes07]), the moduli space $\mathcal{MGH}(\Sigma)$ is diffeomorphic, under the holonomy map, to a connected component of $\chi(\Sigma, \mathbb{P}SL(2, \mathbb{B}))$, corresponding to pairs of representations in $\mathbb{P}SL(2,\mathbb{R})$ that are discrete and faithful, and that induce hyperbolic structures on Σ compatible with the fixed orientation of Σ . Let us denote by $\chi_0(\Sigma, \mathbb{P}SL(2,\mathbb{B}))$ this connected component. By the work of Goldman [Gol84], the tangent space of $\chi_0(\Sigma, \mathbb{P}SL(2,\mathbb{B}))$ at $[\rho] \in \chi_0(\Sigma, \mathbb{P}SL(2,\mathbb{B}))$ is isomorphic to the first cohomology group $H^1(\Sigma, \mathfrak{sl}_2(\mathbb{B})_{\mathrm{Ad}\,\rho})$. We recall that elements of $H^1(\Sigma, \mathfrak{sl}_2(\mathbb{B})_{\mathrm{Ad}\,\rho})$ are equivalence classes of closed 1-forms on Σ with values in the flat bundle $\mathfrak{sl}_2(\mathbb{B})_{\mathrm{Ad}\,\rho}$ defined by

$$\mathfrak{sl}_2(\mathbb{B})_{\mathrm{Ad}\,\rho} = (\tilde{\Sigma} \times \mathfrak{sl}_2(\mathbb{B}))/\sim$$

where $(\tilde{x}, v) \sim (\gamma \tilde{x}, \operatorname{Ad}(\rho(\gamma))v)$ for every $\tilde{x} \in \tilde{\Sigma}$, $\gamma \in \pi_1(\Sigma)$ and $v \in \mathfrak{sl}_2(\mathbb{B})$. As usual, two 1-forms are equivalent if their difference is exact. Here, the exterior differential is the \mathbb{B} -linear extension of the usual differential for $\mathfrak{sl}_2(\mathbb{R})$ -valued forms.

We can then endow $\chi_0(\Sigma, \mathbb{P}SL(2,\mathbb{B}))$ with a natural para-complex structure \mathcal{T} that multiplies by τ an $\mathfrak{sl}_2(\mathbb{B})_{\mathrm{Ad}\,\rho}$ -valued 1-form.

5.4.2. Goldman symplectic form

A general construction by Goldman endows every character variety of a semi-simple Lie group with a symplectic form Ω_{Gol} ([Gol84]). In the setting of $\mathbb{P}SL(2,\mathbb{B})$ this can be obtained as follows. The pairing

$$B: \mathfrak{sl}_2(\mathbb{B}) \otimes \mathfrak{sl}_2(\mathbb{B}) \to \mathbb{B}$$
$$(X, Y) \mapsto \operatorname{tr}(XY)$$

is a non-degenerate \mathbb{B} -bi-linear form that is invariant under conjugation. Pre-composing B with the standard cup-product in co-homology, we obtain a bi-linear pairing

$$H^{1}(\Sigma, \mathfrak{sl}_{2}(\mathbb{B})_{\mathrm{Ad}\,\rho}) \times H^{1}(\Sigma, \mathfrak{sl}_{2}(\mathbb{B})_{\mathrm{Ad}\,\rho}) \to \mathbb{B}$$
$$([\sigma \otimes \phi], [\sigma' \otimes \phi']) \mapsto \int_{\Sigma} B(\phi, \phi')(\sigma \wedge \sigma')$$

which is non-degenerate by Poincaré duality and skew-symmetric. By general arguments of Goldman ([Gol84]) and Atiyah-Bott ([AB83]) the resulting \mathbb{B} -valued 2-form on $\chi_0(\Sigma, \mathbb{P}SL(2, \mathbb{B}))$, which we denote by $\Omega_{Gol}^{\mathbb{B}}$, is closed.

Lemma 5.6. The \mathbb{B} -valued symplectic form $\Omega_{Gol}^{\mathbb{B}}$ is para-holomorphic with respect to \mathcal{T} .

Proof. Recall that a \mathbb{B} -valued 2-form ω in a para-complex manifold (M, \mathbf{P}) is para-complex if $\omega(X, \mathbf{P}Y) = \omega(\mathbf{P}X, Y) = \tau \omega(X, Y)$. In our setting, for every closed $\mathfrak{sl}_2(\mathbb{B})_{\mathrm{Ad} \rho}$ -valued 1-forms $\sigma \otimes \phi$ and $\sigma' \otimes \phi'$, we have

$$\Omega_{Gol}^{\mathbb{B}}(\sigma \otimes \phi, \mathcal{T}(\sigma' \otimes \phi')) = \Omega_{Gol}^{\mathbb{B}}(\sigma \otimes \phi, \sigma' \otimes \tau \phi') = \int_{\Sigma} B(\phi, \tau \phi')(\sigma \wedge \sigma')
= \tau \int_{\Sigma} B(\phi, \phi')(\sigma \wedge \sigma') = \tau \Omega_{Gol}^{\mathbb{B}}(\sigma \otimes \phi, \sigma' \otimes \phi') .$$

Skew-symmetry and bi-linearity of $\Omega_{Gol}^{\mathbb{B}}$ imply that $\Omega_{Gol}^{\mathbb{B}}$ is para-complex. Because $\Omega_{Gol}^{\mathbb{B}}$ is para-complex and closed, and the exterior differential decomposes as $d = \bar{\partial}_T + \partial_T$ (see Appendix A), we deduce that $\bar{\partial}_T \Omega_{Gol}^{\mathbb{B}} = 0$, hence $\Omega_{Gol}^{\mathbb{B}}$ is para-holomorphic.

Because $\Omega_{Gol}^{\mathbb{B}}$ is \mathbb{B} -valued and symplectic, its real and imaginary parts are closed 2-forms and thus define two symplectic structures on $\chi_0(\Sigma, \mathbb{P}SL(2,\mathbb{B}))$. The space $\chi_0(\Sigma, \mathbb{P}SL(2,\mathbb{B}))$ is identified with the product $\mathcal{T}^{\mathsf{rep}}(\Sigma) \times \mathcal{T}^{\mathsf{rep}}(\Sigma)$, where $\mathcal{T}^{\mathsf{rep}}(\Sigma)$ is meant as the space of discrete and faithful representations with values in $\mathbb{P}SL(2,\mathbb{R})$, and is therefore endowed with a real Goldman form $\Omega_{Gol}^{\mathbb{R}}$, defined in the analogous way. It is known from the work of Goldman that, if

$$\operatorname{hol}: \mathcal{T}^{\mathfrak{h}}(\Sigma) \to T^{\mathfrak{rep}}(\Sigma)$$

is the holonomy map for hyperbolic structures on Σ , then

$$hol^* \Omega_{Gol}^{\mathbb{R}} = \Omega_{WP} , \qquad (5.2)$$

where Ω_{WP} is the Weil-Petersson symplectic form on $\mathcal{T}^{\mathfrak{h}}(\Sigma)$. We now express $\Omega_{Gol}^{\mathbb{B}}$ in terms of the real Goldman forms on each component $\mathcal{T}^{\mathfrak{rep}}(\Sigma)$.

Proposition 5.7. Given a closed oriented surface of genus ≥ 2 , we have the following identity on $\chi_0(\Sigma, \mathbb{P}SL(2,\mathbb{B})) \cong \mathcal{T}^{\mathfrak{rep}}(\Sigma) \times \mathcal{T}^{\mathfrak{rep}}(\Sigma)$:

$$\Omega_{Gol}^{\mathbb{B}} = \frac{1}{2} (\pi_l^* \Omega_{Gol}^{\mathbb{R}} + \pi_r^* \Omega_{Gol}^{\mathbb{R}}) + \frac{\tau}{2} (\pi_l^* \Omega_{Gol}^{\mathbb{R}} - \pi_r^* \Omega_{Gol}^{\mathbb{R}}).$$

Proof. The isomorphism $\mathbb{P}SL(2,\mathbb{B}) \cong \mathbb{P}SL(2,\mathbb{R}) \times \mathbb{P}SL(2,\mathbb{R})$ induces an isomorphism of Lie algebras $\mathfrak{sl}_2(\mathbb{B}) \cong \mathfrak{sl}_2(\mathbb{R}) \times \mathfrak{sl}_2(\mathbb{R})$ given by decomposing $X \in \mathfrak{sl}_2(\mathbb{B})$ into $X = X_+e^+ + X_-e^-$ with $X_{\pm} \in \mathfrak{sl}_2(\mathbb{R})$. Moreover, the adjoint action of $\rho = \rho_-e^- + \rho_+e^+$ on $\mathfrak{sl}_2(\mathbb{B})$ induces the action of (ρ_+, ρ_-) on $\mathfrak{sl}_2(\mathbb{R}) \times \mathfrak{sl}_2(\mathbb{R})$ given by the adjoint action on each factor. As a consequence, an $\mathfrak{sl}_2(\mathbb{B})_{\mathrm{Ad}\,\rho}$ -valued 1-form $\sigma \otimes \phi$ can be uniquely written as $(\sigma \otimes \phi_+e^+) + (\sigma \otimes \phi_-e^-)$, and $\sigma \otimes \phi_\pm$ are $\mathfrak{sl}_2(\mathbb{R})_{\mathrm{Ad}\,\rho_\pm}$ -valued 1-forms. Finally, we observe that $\sigma \otimes \phi$ is closed (respectively exact) if and only if $\sigma \otimes \phi_\pm$ are closed (respectively exact). Therefore, by Lemma 5.6,

$$\begin{split} &\Omega_{Gol}^{\mathbb{B}}(\sigma\otimes\phi,\sigma'\otimes\phi') = \Omega_{Gol}^{\mathbb{B}}\left(\sigma\otimes(\phi_{+}e^{+}+\phi_{-}e^{-}),\sigma'\otimes(\phi'_{+}e^{+}+\phi'_{-}e^{-})\right) \\ &= \frac{1}{4}\Omega_{Gol}^{\mathbb{B}}\left(\sigma\otimes((\phi_{+}+\phi_{-})+\tau(\phi_{+}-\phi_{-})),\sigma'\otimes((\phi'_{+}+\phi'_{-})+\tau(\phi'_{+}-\phi'_{-}))\right) \\ &= \frac{1}{4}\left(\Omega_{Gol}^{\mathbb{R}}\left(\sigma\otimes(\phi_{+}+\phi_{-}),\sigma'\otimes(\phi'_{+}+\phi'_{-})\right)+\Omega_{Gol}^{\mathbb{R}}\left(\sigma\otimes(\phi_{+}-\phi_{-}),\sigma'\otimes(\phi'_{+}-\phi'_{-})\right)\right) \\ &+ \frac{\tau}{4}\left(\Omega_{Gol}^{\mathbb{R}}\left(\sigma\otimes(\phi_{+}+\phi_{-}),\sigma'\otimes(\phi'_{+}-\phi'_{-})\right)+\Omega_{Gol}^{\mathbb{R}}\left(\sigma\otimes(\phi_{+}-\phi_{-}),\sigma'\otimes(\phi'_{+}+\phi'_{-})\right)\right) \\ &= \frac{1}{2}\left(\Omega_{Gol}^{\mathbb{R}}\left(\sigma\otimes\phi_{+},\sigma'\otimes\phi'_{+}\right)+\Omega_{Gol}^{\mathbb{R}}\left(\sigma\otimes\phi_{-},\sigma'\otimes\phi'_{-}\right)\right) \\ &+ \frac{\tau}{2}\left(\Omega_{Gol}^{\mathbb{R}}\left(\sigma\otimes\phi_{+},\sigma'\otimes\phi'_{+}\right)-\Omega_{Gol}^{\mathbb{R}}\left(\sigma\otimes\phi_{-},\sigma'\otimes\phi'_{-}\right)\right) \end{split}$$

and the result follows.

Now, recall that on $\mathcal{MGH}(\Sigma)$ we have the holonomy map

$$\mathcal{H}ol: \mathcal{MGH}(\Sigma) \to \chi_0(\Sigma, \mathbb{P}SL(2, \mathbb{B}))$$
,

and moreover Mess homeomorphism

$$\mathcal{M}^{\mathfrak{h}}: \mathcal{MGH}(\Sigma) \to \mathcal{T}^{\mathfrak{h}}(\Sigma) \times \mathcal{T}^{\mathfrak{h}}(\Sigma)$$
.

The work of Mess, combined with the expression of \mathcal{M} that we gave in (2.2) (see Theorem 2.9) showed that the left and right components of \mathcal{M} coincide with the holonomy of the MGHC AdS manifold, under the identification $\mathcal{T}^{\mathfrak{h}}(\Sigma) \cong \mathcal{T}^{\mathfrak{rep}}(\Sigma)$. In other words, we have $\mathcal{H}ol = (\text{hol}, \text{hol}) \circ \mathcal{M}$. Combining Proposition 5.7 with Theorem C, and using Goldman's fundamental identity (5.2), we obtain that $\mathcal{H}ol^*(\mathcal{T}, 4\Omega_{Gol}^{\mathbb{B}}) = \omega_{\mathbf{J}}^{\mathbb{B}}$. Since $\Omega_{Gol}^{\mathbb{B}}$ is paracomplex with respect to \mathcal{T} , and $\omega_{\mathbf{J}}^{\mathbb{B}}$ is para-complex with respect to \mathbf{J} , the usual argument shows that $\mathcal{H}ol^*\mathcal{T} = \mathbf{J}$. This concludes the proof of the following result.

Corollary H. Let Σ be a closed oriented surface of genus ≥ 2 . Then

$$\mathcal{H}ol^*(\mathcal{T}, 4\Omega_{Gol}^{\mathbb{B}}) = (\mathbf{J}, \omega_{\mathbf{J}}^{\mathbb{B}}) .$$

5.4.3. The para-complex Fenchel-Nielsen coordinates

Goldman symplectic forms on Teichmüller space and on the space of quasi-Fuchsian representations are intimately related to hyperbolic geometry because, for instance, the (real or complex) length and twist parameters ([Wol83],[Pla01]) provide Darboux coordinates. Here we show that an analogous result holds for $\Omega_{Gol}^{\mathbb{B}}$ using the \mathbb{B} -valued Fenchel-Nielsen coordinates introduced in [Tam20] to describe the deformation space of MGHC anti-de Sitter structures.

We first introduce some facts on Anti-de Sitter geometry and recall the definition of these coordinates. The model of Anti-de Sitter space that we will use is simply the Lie group $\mathbb{P}SL(2,\mathbb{R})$ endowed with the bi-invariant Lorentzian metric which is induced by (1/8) the Killing form on the Lie algebra, where the factor (1/8) serves to normalize the sectional curvature to be equal to -1. The group of orientation-preserving, time-preserving isometries is isomorphic to $\mathbb{P}SL(2,\mathbb{R}) \times \mathbb{P}SL(2,\mathbb{R})$, acting by left and right multiplication, and the boundary of AdS space identifies to $\mathbb{RP}^1 \times \mathbb{RP}^1$, in such a way that the action of the isometry group extends to the obvious component-wise action of $\mathbb{P}SL(2,\mathbb{R}) \times \mathbb{P}SL(2,\mathbb{R})$ on $\mathbb{RP}^1 \times \mathbb{RP}^1$.

Now, consider an isometry of the form $(\gamma_+, \gamma_-) \in \mathbb{P}SL(2, \mathbb{R}) \times \mathbb{P}SL(2, \mathbb{R})$, where both γ_- and γ_+ are loxodromic. Then (γ_+, γ_-) acts on the boundary $\mathbb{RP}^1 \times \mathbb{RP}^1$ by fixing four points, of which one is attracting and one repelling. It turns out (see for instance the first part of the proof of Lemma 5.9 below) that there exists a spacelike geodesic connecting these two points (a priori the other possibility, that is however excluded by Lemma 5.9, is that the segment connecting the two points is lightlike and contained in the boundary). This spacelike geodesic is what we will call the principal axis. More formally:

Definition 5.8. Given a pair of loxodromic elements $(\gamma_+, \gamma_-) \in \mathbb{P}SL(2, \mathbb{R}) \times \mathbb{P}SL(2, \mathbb{R})$, we call principal axis of (γ_+, γ_-) the unique spacelike geodesic Axis (γ_+, γ_-) of Anti-de Sitter space whose endpoints on $\mathbb{RP}^1 \times \mathbb{RP}^1$ are $(\gamma_+^{rep}, \gamma_-^{rep})$ and $(\gamma_+^{att}, \gamma_-^{att})$, where γ_\pm^{rep} and γ_\pm^{att} denote the repelling and attracting fixed points on \mathbb{RP}^1 .

Now fix a pair of pants decomposition $\mathcal{P} = \{\gamma_1, \dots, \gamma_n\}$ of Σ , for $n = (3/2)|\chi(\Sigma)|$. If $\rho : \pi_1(\Sigma) \to \mathbb{P}\mathrm{SL}(2,\mathbb{B})$ is the holonomy of a MGHC Anti-de Sitter manifold, then $\rho(\gamma_j)$ is loxodromic, i.e. conjugated to a diagonal matrix, for every $\gamma_j \in \mathcal{P}$. The \mathbb{B} -valued length is defined as

 $\ell_{\rho}^{\mathbb{B}}(\gamma_j) = 2 \operatorname{arccosh}\left(\frac{\operatorname{tr}(\rho(\gamma_j))}{2}\right) ,$

where the hyperbolic arccosine is computed using its power series expansion. This quantity is related to how a loxodromic isometry acts on anti-de Sitter space. Indeed, the real part of $\ell_{\rho}^{\mathbb{B}}(\gamma_{j})$ represents the translation length of $\rho(\gamma_{j})$ on the principal axis under $\rho(\gamma_{j})$, whereas the imaginary part is the rotation angle of $\rho(\gamma_{j})$ acting on the orthogonal of such a geodesic. The \mathbb{B} -valued lengths completely determine a $\mathbb{P}\mathrm{SL}(2,\mathbb{B})$ -representation of a pair of pants. The \mathbb{B} -twist parameter $\mathrm{tw}_{\rho}^{\mathbb{B}}(\gamma_{j})$ indicates how the pair of pants are glued together along the boundary curve γ_{j} : the real part describes the shear parameter and the imaginary part the bending between the two pairs of pants in anti-de Sitter space.

It turns out that $\ell_{\rho}^{\mathbb{B}}(\gamma_{j})$ and $\operatorname{tw}_{\rho}^{\mathbb{B}}(\gamma_{j})$ are related to the classical Fenchel-Nielsen coordinates ℓ_{\bullet} and $\operatorname{tw}_{\bullet}$ as follows. Under the isomorphism $\mathbb{P}\operatorname{SL}(2,\mathbb{B}) \cong \mathbb{P}\operatorname{SL}(2,\mathbb{R}) \times \mathbb{P}\operatorname{SL}(2,\mathbb{R})$, the holonomy $\rho: \pi_{1}(\Sigma) \to \mathbb{P}\operatorname{SL}(2,\mathbb{B})$ of a MGHC Anti-de Sitter structure corresponds to a pair of faithful and discrete representations $\rho_{\pm}: \pi_{1}(\Sigma) \to \mathbb{P}\operatorname{SL}(2,\mathbb{R})$, which are thus the holonomies of hyperbolic metrics on Σ . We have the following relations:

$$\ell_{\rho}^{\mathbb{B}}(\gamma_{j}) = \ell_{\rho_{+}}(\gamma_{j})e^{+} + \ell_{\rho_{-}}(\gamma_{j})e^{-} = \left(\frac{\ell_{\rho_{+}}(\gamma_{j}) + \ell_{\rho_{-}}(\gamma_{j})}{2}\right) + \tau\left(\frac{\ell_{\rho_{+}}(\gamma_{j}) - \ell_{\rho_{-}}(\gamma_{j})}{2}\right)$$

and

$$\operatorname{tw}_{\rho}^{\mathbb{B}}(\gamma_{j}) = \operatorname{tw}_{\rho_{+}}(\gamma_{j})e^{+} + \operatorname{tw}_{\rho_{-}}(\gamma_{j})e^{-} = \left(\frac{\operatorname{tw}_{\rho_{+}}(\gamma_{j}) + \operatorname{tw}_{\rho_{-}}(\gamma_{j})}{2}\right) + \tau\left(\frac{\operatorname{tw}_{\rho_{+}}(\gamma_{j}) - \operatorname{tw}_{\rho_{-}}(\gamma_{j})}{2}\right).$$

Theorem I. The \mathbb{B} -valued Fenchel-Nielsen coordinates are para-holomorphic for \mathcal{T} , and are Darboux coordinates with respect to the para-complex symplectic form $\Omega_{Gol}^{\mathbb{B}}$.

Proof. Let us show the first statement. As a consequence of the definition of \mathcal{T} , \mathcal{T} acts as the identity on the first factor on the tangent space to $\mathcal{T}^{\mathfrak{rep}}(\Sigma) \times \mathcal{T}^{\mathfrak{rep}}(\Sigma)$, and as minus the identity on the second factor. Hence for any curve $\gamma_i \in \mathcal{P}$, we have

$$\mathcal{T}\left(\frac{\partial}{\partial \ell_{\rho_{\pm}}^{j}}\right) = \pm \frac{\partial}{\partial \ell_{\rho_{\pm}}^{j}} \quad \text{and} \quad \mathcal{T}\left(\frac{\partial}{\partial \mathrm{tw}_{\rho_{\pm}}^{j}}\right) = \pm \frac{\partial}{\partial \mathrm{tw}_{\rho_{\pm}}^{j}} \ .$$

Therefore,

$$d\ell_{\rho}^{\mathbb{B}}\left(\mathcal{T}\left(\frac{\partial}{\partial \ell_{\rho_{\pm}}^{j}}\right)\right) = \frac{\pm 1 + \tau}{2}$$

and

$$\tau d\ell_{\rho}^{\mathbb{B}} \left(\frac{\partial}{\partial \ell_{\rho_{\pm}}^{j}} \right) = \tau \left(\frac{1 \pm \tau}{2} \right) = \frac{\pm 1 + \tau}{2} .$$

On the other hand, it is clear that

$$d\ell_{\rho}^{\mathbb{B}}\left(\mathcal{T}\left(\frac{\partial}{\partial \mathrm{tw}_{\rho_{\pm}}^{j}}\right)\right) = 0 = \tau d\ell_{\rho}^{\mathbb{B}}\left(\frac{\partial}{\partial \mathrm{tw}_{\rho_{\pm}}^{j}}\right) ,$$

hence the \mathbb{B} -lengths are para-holomorphic. A similar computation holds for the \mathbb{B} -twist parameters.

For the second statement, let $\ell_{\rho}^{\mathbb{B},j}$ and $\operatorname{tw}_{\rho}^{\mathbb{B},j}$ denote the \mathbb{B} -length and \mathbb{B} -twist parameters along the curve $\gamma_j \in \mathcal{P}$. We use an analogous notation for the Fenchel-Nielsen coordinates associated to the representations ρ_{\pm} into $\mathbb{P}\operatorname{SL}(2,\mathbb{R})$ such that $\rho = \rho_+ e^+ + \rho_- e^-$. By Proposition 5.7, and the fact that the classical Fenchel-Nielsen coordinates are Darboux for Ω_{WP} , we have

$$\operatorname{Re}(\Omega_{Gol}^{\mathbb{B}})\left(\frac{\partial}{\partial \ell_{\rho}^{\mathbb{B},i}}, \frac{\partial}{\partial \ell_{\rho}^{\mathbb{B},j}}\right) = \frac{1}{2}\left(\Omega_{WP}\left(\frac{\partial}{\partial \ell_{\rho_{+}}^{i}}, \frac{\partial}{\partial \ell_{\rho_{+}}^{j}}\right) + \Omega_{WP}\left(\frac{\partial}{\partial \ell_{\rho_{-}}^{i}}, \frac{\partial}{\partial \ell_{\rho_{-}}^{j}}\right)\right) = 0,$$

and similarly

$$\operatorname{Im}(\Omega_{Gol}^{\mathbb{B}})\left(\frac{\partial}{\partial \ell_{\rho}^{\mathbb{B},i}}, \frac{\partial}{\partial \ell_{\rho}^{\mathbb{B},j}}\right) = \frac{1}{2}\left(\Omega_{WP}\left(\frac{\partial}{\partial \ell_{\rho_{+}}^{i}}, \frac{\partial}{\partial \ell_{\rho_{+}}^{j}}\right) - \Omega_{WP}\left(\frac{\partial}{\partial \ell_{\rho_{-}}^{i}}, \frac{\partial}{\partial \ell_{\rho_{-}}^{j}}\right)\right) = 0.$$

For the same reason

$$\Omega_{Gol}^{\mathbb{B}} \left(\frac{\partial}{\partial \mathrm{tw}_{\rho}^{\mathbb{B},i}}, \frac{\partial}{\partial \mathrm{tw}_{\rho}^{\mathbb{B},j}} \right) = 0.$$

On the other hand,

$$\operatorname{Re}(\Omega_{Gol}^{\mathbb{B}})\left(\frac{\partial}{\partial \ell_{\rho}^{\mathbb{B},i}}, \frac{\partial}{\partial \operatorname{tw}_{\rho}^{\mathbb{B},j}}\right) = \frac{1}{2}\left(\Omega_{WP}\left(\frac{\partial}{\partial \ell_{\rho_{+}}^{i}}, \frac{\partial}{\partial \operatorname{tw}_{\rho_{+}}^{j}}\right) + \Omega_{WP}\left(\frac{\partial}{\partial \ell_{\rho_{-}}^{i}}, \frac{\partial}{\partial \operatorname{tw}_{\rho_{-}}^{j}}\right)\right) = \delta_{i,j}$$

and

$$\operatorname{Im}(\Omega_{Gol}^{\mathbb{B}})\left(\frac{\partial}{\partial \operatorname{tw}_{\rho}^{\mathbb{B},i}}, \frac{\partial}{\partial \ell_{\rho}^{\mathbb{B},j}}\right) = \frac{1}{2}\left(\Omega_{WP}\left(\frac{\partial}{\partial \operatorname{tw}_{\rho_{+}}^{i}}, \frac{\partial}{\partial \ell_{\rho_{+}}^{j}}\right) - \Omega_{WP}\left(\frac{\partial}{\partial \operatorname{tw}_{\rho_{-}}^{i}}, \frac{\partial}{\partial \ell_{\rho_{-}}^{j}}\right)\right) = 0$$

We conclude that

$$\Omega_{Gol}^{\mathbb{B}} = \sum_{j=1}^{n} d\ell_{\rho}^{\mathbb{B},j} \wedge dtw_{\rho}^{\mathbb{B},j}$$
(5.3)

as claimed. \Box

5.4.4. The para-complex cosine formula

Let us now move on to the proof of Theorem J, namely a B-valued version of the cosine formula for the Weil-Petersson symplectic form. For this purpose, it will suffice to focus on the case where there is at least a point of intersection. The fundamental geometric computation is contained in the following lemma.

Lemma 5.9. Let (α_+, α_-) and (β_+, β_-) be two pairs of loxodromic elements in $\mathbb{P}SL(2, \mathbb{R}) \times \mathbb{P}SL(2, \mathbb{R})$. Suppose that the actions of (α_+, β_+) and (α_-, β_-) on \mathbb{RP}^1 are topologically conjugated (i.e. there exists an orientation-preserving diffeomorphism f of \mathbb{RP}^1 such that $f\alpha_-f^{-1} = \alpha_+$ and $f\beta_-f^{-1} = \beta_+$) and that the axes of α_\pm and β_\pm intersect in \mathbb{H}^2 with counterclockwise angle equal to $\varphi_\pm \in [0, \pi)$. Denote $\tilde{\alpha} = \operatorname{Axis}(\alpha_+, \alpha_-)$ and $\tilde{\beta} = \operatorname{Axis}(\beta_+, \beta_-)$ the principal axes in Anti-de Sitter space. Then

- (1) There is a unique complete timelike geodesic σ that intersects both $\tilde{\alpha}$ and $\tilde{\beta}$ orthogonally. Call the points of intersection $q_{\tilde{\alpha}}$ and $q_{\tilde{\beta}}$. Orient σ so that its tangent vector is future-directed.
- (2) The signed timelike distance along σ between $q_{\tilde{\alpha}}$ and $q_{\tilde{\beta}}$ equals $(\varphi_+ \varphi_-)/2 \in (-\pi/2, \pi/2)$.
- (3) The counterclockwise angle of intersection between $\tilde{\alpha}$ and the parallel transport of $\tilde{\beta}$ to $q_{\tilde{\alpha}}$ along the unique orthogonal timelike geodesic σ equals $(\varphi_+ + \varphi_-)/2$.

Let us clarify the meaning of the second statement. Given two points p and q in AdS space connected by a timelike geodesic, the timelike distance between p and q is the length of the shortest geodesic segment from p and q. Recalling that timelike geodesics are closed and have length π , there are two such geodesic segments, whose sum equals π . If the points p and q are not antipodal (i.e. the timelike distance is not $\pi/2$), then we can define the signed timelike distance d(p,q), which is just the timelike distance introduced above with positive sign if the realizing geodesic segment is oriented towards the future, and with negative sign if it is oriented towards the past. We remark that d(p,q) = -d(q,p).

Proof. Denote by $\tilde{\alpha}_{\pm}$ the axes of α_{\pm} in \mathbb{H}^2 , and analogously $\tilde{\beta}_{\pm}$ are the axes of β_{\pm} . Applying the action of the group $\mathbb{P}\mathrm{SL}(2,\mathbb{R}) \times \mathbb{P}\mathrm{SL}(2,\mathbb{R})$, we can assume that, in the upper half-space model of \mathbb{H}^2 , the repelling fixed point of α_{\pm} is 0, and the attracting fixed point of α_{\pm} is ∞ (hence both axes $\tilde{\alpha}_{-}$ and $\tilde{\alpha}_{+}$ coincide with the geodesic ℓ connecting 0 and ∞), and moreover the intersection point of the axes $\tilde{\beta}_{-}$ and $\tilde{\beta}_{+}$ with ℓ is i. Then the set of order two isometries in $\mathbb{P}\mathrm{SL}(2,\mathbb{R})$ having fixed point on ℓ is an Anti-de Sitter spacelike geodesic with endpoints (0,0) and (∞,∞) in $\mathbb{RP}^1 \times \mathbb{RP}^1$, and is therefore the principal axis $\tilde{\alpha}$ of (α_{+},α_{-}) . Indeed, the Lie group and the Riemannian exponential maps coincide for a bi-invariant metric on a Lie group, and this is a left translate of a one-parameter group of hyperbolic transformations in $\mathbb{P}\mathrm{SL}(2,\mathbb{R})$. To see that the endpoints are precisely (0,0) and (∞,∞) , one can apply the criterion in [BS20, Lemma 3.2.2]; anyway this follows from the more general observation ([BS20, Lemma 3.5.1], that we will apply below) that the surface \mathcal{P}

of order two isometries is a totally geodesic spacelike plane in Anti-de Sitter space, whose boundary at infinity is the diagonal in $\mathbb{RP}^1 \times \mathbb{RP}^1$, for which the map $\mathcal{P} \to \mathbb{H}^2$ sending an order two element to its fixed point is an isometry, equivariant for the action of $\mathbb{P}SL(2,\mathbb{R})$ (by conjugation on \mathcal{P} , and the obvious action on \mathbb{H}^2). En passant, we have showed that the principal axis in Definition 5.8 is well-defined.

Now, by our assumption the axis $\tilde{\beta}_{\pm}$ is obtained by applying to $\tilde{\alpha}_{\pm}$ a rotation $R_{\varphi_{\pm}}$ of angle φ_{\pm} fixing i. Since the actions are topologically conjugated, the attracting and repelling fixed points of α_{\pm} and β_{\pm} appear with the same cyclic order on \mathbb{RP}^1 . This implies that either the rotations $R_{\varphi_{+}}$ and $R_{\varphi_{-}}$ both map attracting (resp. repelling) fixed points of α_{\pm} to attracting (resp. repelling) fixed points of β_{\pm} , or they both map the attracting fixed point to the repelling one and vice versa. This implies that the principal axis $\tilde{\beta}$ equals $(R_{\varphi_{+}}, R_{\varphi_{-}})\tilde{\alpha}$. Now, set

$$d = \frac{\varphi_+ - \varphi_-}{2}$$
 and $\theta = \frac{\varphi_+ + \varphi_-}{2}$.

Observe that $\varphi_{\pm} \in (0,\pi)$, hence $\theta \in (0,\pi)$ and $d \in (-\pi/2,\pi/2)$. Then we have

$$(R_{\varphi_+}, R_{\varphi_-}) = (R_d, R_{-d}) \circ (R_\theta, R_\theta) .$$

By the above considerations on the totally geodesic spacelike plane \mathcal{P} (in particular the equivariant isometry with \mathbb{H}^2), (R_{θ}, R_{θ}) acts as a rotation of angle θ . Moreover, by a general fact that we prove in Proposition 5.10 below for completeness, the parallel transport along the geodesic $t \mapsto R_{2t}$ (which is parameterized by arclength, future-directed, and orthogonal to the totally geodesic plane \mathcal{P} by a simple symmetry argument) equals the differential of the isometry (R_t, R_{-t}) . In conclusion, we have $q_{\tilde{\alpha}} = R_{\pi}$, $q_{\tilde{\beta}} = (R_d, R_{-d})q_{\tilde{\alpha}}$, so their signed timelike distance is d, and finally the angle at $q_{\tilde{\alpha}}$ between the parallel transport of $\tilde{\beta}$ at $q_{\tilde{\alpha}}$ and $\tilde{\alpha}$ equals θ . This concludes the proof of the second and third items.

Proposition 5.10. Given a bi-invariant metric on a Lie group G and $X, Y \in \mathfrak{g}$, the parallel transport of Y along the geodesic $g_t = \exp(tX)$ at the point g_t is given by $(L_{g_{t/2}})_*(R_{g_{t/2}})_*(Y)$.

Proof. The Levi-Civita connection of a bi-invariant metric on G is given by

$$\nabla_V W = \frac{1}{2} (D_V^l W + D_V^r W) ,$$

where D^l and D^r are the left and right invariant connections, which are defined in the following way: given a path $\gamma(t)$ such that $\gamma(0) = g$ and $\gamma'(0) = V \in T_gG$, then D_V^lW (resp. D_V^rW) is the derivative at time 0 of the left (resp. right) translated of $W_{\gamma(t)}$ at g, namely $(L_{q\gamma(t)^{-1}})_*W_{\gamma(t)}$ (resp. $(R_{q\gamma(t)^{-1}})_*W_{\gamma(t)}$).

Let us apply this to the geodesic $\gamma(t) = g_t = \exp(tX)$ and the vector field along γ defined by $W_{g_t} = (L_{g_{t/2}})_*(R_{g_{t/2}})_*(Y)$. In order to check that W is parallel along g_t , it suffices to check it at t = 0 (namely that $\nabla_X W = 0$), because multiplication on the left

and on the right by $g_{t/2}$ is an isometry that preserves the vector field W and the geodesic γ by definition. We have

$$\begin{split} D_X^l W &= \left. \frac{d}{dt} \right|_{t=0} (L_{g_t^{-1}})_* (L_{g_{t/2}})_* (R_{g_{t/2}})_* (Y) \\ &= \left. \frac{d}{dt} \right|_{t=0} (L_{g_{-t/2}})_* (R_{g_{t/2}})_* (Y) = -\mathrm{ad}_X Y \; . \end{split}$$

A similar computation shows $D_X^rW=\operatorname{ad}_XY$. Hence $\nabla_XW=0$, which concludes the claim.

Motivated by Lemma 5.9 above, we give the following definition.

Definition 5.11. Given two spacelike geodesics $\tilde{\alpha}$ and $\tilde{\beta}$ in AdS space that admit a common orthogonal timelike geodesic σ , let $q_{\tilde{\alpha}}$ and $q_{\tilde{\beta}}$ the intersection points of $\tilde{\alpha}$ and $\tilde{\beta}$ with σ . Then we define the \mathbb{B} -valued angle $d^{\mathbb{B}}(\tilde{\alpha}, \tilde{\beta})$ between $\tilde{\alpha}$ and $\tilde{\beta}$ as the para-complex number whose imaginary part equals $d(q_{\tilde{\alpha}}, q_{\tilde{\beta}})$, and the real part equals the counterclockwise angle between the parallel transport of $\tilde{\alpha}$ at $q_{\tilde{\beta}}$ along σ , and $\tilde{\beta}$.

Remark 5.12. Let $d^{\mathbb{B}}(\tilde{\alpha}, \tilde{\beta}) = \theta + \tau d$ be the \mathbb{B} -valued angle introduced above. If we invert the role of $\tilde{\alpha}$ and $\tilde{\beta}$, clearly the real part of the \mathbb{B} -valued angle becomes $\pi - \theta$. The imaginary part instead only changes sign, so

$$d^{\mathbb{B}}(\tilde{\beta}, \tilde{\alpha}) = (\pi - \theta) - \tau d.$$

Hence $\cos\left(d^{\mathbb{B}}(\tilde{\beta}, \tilde{\alpha})\right) = -\cos\left(d^{\mathbb{B}}(\tilde{\alpha}, \tilde{\beta})\right)$ because

$$\cos(\theta + \tau d) = \cos((\theta + d)e^{+} + (\theta - d)e^{-}) = \cos(\theta + d)e^{+} + \cos(\theta - d)e^{-}$$

where the last step is justified by the power series expansion of cosine. This is consistent with the skew-symmetry in the generalized cosine formula of Theorem J.

Hence Lemma 5.9 can be restated by the following formula:

$$d^{\mathbb{B}}(\tilde{\alpha}, \tilde{\beta}) = \frac{\varphi_{+} + \varphi_{-}}{2} + \tau \left(\frac{\varphi_{+} - \varphi_{-}}{2}\right) , \qquad (5.4)$$

We are finally ready to prove:

Theorem J. Let $\rho = (\rho_+, \rho_-) : \pi_1(\Sigma) \to \mathbb{P}SL(2, \mathbb{R}) \times \mathbb{P}SL(2, \mathbb{R})$ be the holonomy of a MGHC AdS manifold, and let α, β be non-trivial simple closed curves. Then

$$\Omega_{Gol}^{\mathbb{B}} \left(\frac{\partial}{\partial t \mathbf{w}_{\rho}^{\mathbb{B}, \alpha}}, \frac{\partial}{\partial t \mathbf{w}_{\rho}^{\mathbb{B}, \beta}} \right) = \sum_{p \in \alpha \cap \beta} \cos \left(d^{\mathbb{B}} (\tilde{\alpha}_{\rho}, \tilde{\beta}_{\rho}) \right) ,$$

where $\tilde{\alpha}_{\rho}$ and $\tilde{\beta}_{\rho}$ are the principal axes of $\rho(\alpha)$ and $\rho(\beta)$ on AdS space.

Proof. If the closed curves $\alpha, \beta \in \pi_1(\Sigma)$ are disjoint, then equality holds, as the LHS vanishes as a consequence of the formula (5.3), and the RHS vanishes by definition. Let us suppose α and β intersect, and let us evaluate Goldman symplectic form on the twist deformations along α and β . As before, we write the representation $\rho : \pi_1(\Sigma) \to \mathbb{P}SL(2,\mathbb{B})$ as $\rho = \rho_+ e^+ + \rho_- e^-$ with $\rho_{\pm} : \pi_1(\Sigma) \to \mathbb{P}SL(2,\mathbb{R})$. For any $p \in \alpha \cap \beta$, we denote by $\varphi_{\pm}(p) \in (0,\pi)$ the angle between the geodesic representatives of α and β for the hyperbolic metrics with holonomies ρ_{\pm} measured from α . By the work of Wolpert ([Wol83]), we have

$$\operatorname{Re}(\Omega_{Gol}^{\mathbb{B}})\left(\frac{\partial}{\partial \operatorname{tw}_{\rho}^{\mathbb{B},\alpha}}, \frac{\partial}{\partial \operatorname{tw}_{\rho}^{\mathbb{B},\beta}}\right) = \frac{1}{2}\left(\Omega_{WP}\left(\frac{\partial}{\partial \operatorname{tw}_{\rho_{+}}^{\alpha}}, \frac{\partial}{\partial \operatorname{tw}_{\rho_{+}}^{\beta}}\right) + \Omega_{WP}\left(\frac{\partial}{\partial \operatorname{tw}_{\rho_{-}}^{\alpha}}, \frac{\partial}{\partial \operatorname{tw}_{\rho_{-}}^{\beta}}\right)\right) \\
= \frac{1}{2}\sum_{p \in \alpha \cap \beta} \cos(\varphi_{+}(p)) + \cos(\varphi_{-}(p))$$

and similarly,

$$\operatorname{Im}(\Omega_{Gol}^{\mathbb{B}}) = \left(\frac{\partial}{\partial \operatorname{tw}_{\rho}^{\mathbb{B},\alpha}}, \frac{\partial}{\partial \operatorname{tw}_{\rho}^{\mathbb{B},\beta}}\right) = \frac{1}{2} \left(\Omega_{WP} \left(\frac{\partial}{\partial \operatorname{tw}_{\rho_{+}}^{\alpha}}, \frac{\partial}{\partial \operatorname{tw}_{\rho_{+}}^{\beta}}\right) - \Omega_{WP} \left(\frac{\partial}{\partial \operatorname{tw}_{\rho_{-}}^{\alpha}}, \frac{\partial}{\partial \operatorname{tw}_{\rho_{-}}^{\beta}}\right)\right)$$
$$= \frac{1}{2} \sum_{p \in \alpha \cap \beta} \cos(\varphi_{+}(p)) - \cos(\varphi_{-}(p)) .$$

Therefore,

$$\Omega_{Gol}^{\mathbb{B}} \left(\frac{\partial}{\partial t w_{\rho}^{\mathbb{B}, \alpha}}, \frac{\partial}{\partial t w_{\rho}^{\mathbb{B}, \beta}} \right) = \frac{1}{2} \sum_{p \in \alpha \cap \beta} (\cos(\varphi_{+}(p)) + \cos(\varphi_{-}(p)) + \tau(\cos(\varphi_{+}(p)) - \cos(\varphi_{-}(p)))$$

$$= \sum_{p \in \alpha \cap \beta} \cos(\varphi_{+}(p)) e^{+} + \cos(\varphi_{-}(p)) e^{-}$$

$$= \sum_{p \in \alpha \cap \beta} \cos(\varphi_{+}(p)) e^{+} + \varphi_{-}(p) e^{-}$$

$$= \sum_{p \in \alpha \cap \beta} \cos\left(\frac{\varphi_{+} + \varphi_{-}}{2} + \tau \frac{\varphi_{+} - \varphi_{-}}{2}\right)$$

$$= \sum_{p \in \alpha \cap \beta} \cos\left(d^{\mathbb{B}}(\tilde{\alpha}_{\rho}, \tilde{\beta}_{\rho})\right),$$

where the second to last step can be formally justified by considering the definition of cosine as power series, and the last step is obtained by applying Equation (5.4). Indeed we can apply Lemma 5.9, because the representations ρ_{-} and ρ_{+} are both in the Teichmüller component $\mathcal{T}^{\mathfrak{rep}}(\Sigma)$, hence their actions on \mathbb{RP}^1 are topologically conjugated. The proof is then concluded.

6. Symplectic reduction

In this section we explain the process that led us to the definition of the para-hyperKähler structure on $\mathcal{MS}_0(\Sigma, \rho)$ and to the explicit model of its tangent space described in Section 4. The main tool we use is Donaldson's construction of moment maps on infinite dimensional spaces.

6.1. The moment maps on $T^*\mathcal{J}(\mathbb{R}^2)$

To apply Donaldson's construction, which we recall in the next section, we first need to provide moment maps for the action of $\mathbb{P}SL(2,\mathbb{R})$ on $T^*\mathcal{J}(\mathbb{R}^2)$, with respect to the symplectic forms that we introduced in Section 3.3.

Recalling that $\mathfrak{sl}_2(\mathbb{R}) = \{X \in \operatorname{End}(\mathbb{R}^2) \mid \operatorname{tr} X = 0\}$, we define the maps $\eta_{\mathbf{I}}, \eta_{\mathbf{J}}, \eta_{\mathbf{K}} : T^*\mathcal{J}(\mathbb{R}^2) \to \mathfrak{sl}_2(\mathbb{R})^*$:

$$\eta_{\mathbf{I}}(J,\sigma) = f(\|\sigma\|_J) \operatorname{tr}(J\cdot)
\eta_{\mathbf{J}}(J,\sigma) = \langle \sigma \mid [\cdot, J] \rangle_J = -\operatorname{tr}(g_J^{-1}\sigma J\cdot)
\eta_{\mathbf{K}}(J,\sigma) = \langle \sigma \mid J[\cdot, J] \rangle_J = \operatorname{tr}(g_J^{-1}\sigma\cdot)$$

where $[X,Y] = XY - YX \in \mathfrak{sl}_2(\mathbb{R})$. Observe that, for every $X \in \mathfrak{sl}_2(\mathbb{R})$, the element [X,J] belongs to $T_J \mathcal{J}(\mathbb{R}^2)$.

Theorem 6.1. The maps $\eta_{\mathbf{I}}$, $\eta_{\mathbf{J}}$ and $\eta_{\mathbf{K}}$ are moment maps for the action of $\mathbb{P}\mathrm{SL}(2,\mathbb{R})$ over $T^*\mathcal{J}(\mathbb{R}^2)$ with respect to the symplectic structures $\omega_{\mathbf{I}}$, $\omega_{\mathbf{J}}$ and $\omega_{\mathbf{K}}$ introduced in (3.9), (3.10) and (3.11) respectively.

Proof. We start by noticing some properties of our action. If $\varphi_A \colon T^*\mathcal{J}(\mathbb{R}^2) \to T^*\mathcal{J}(\mathbb{R}^2)$ denotes the transformation $(J, \sigma) \mapsto A \cdot (J, \sigma)$, then a simple computation shows that, for every $A \in \mathbb{P}\mathrm{SL}(2, \mathbb{R})$ and $X \in \mathfrak{sl}_2(\mathbb{R})$,

$$V_X(\varphi_A(J,\sigma)) = d(\varphi_A)_{(J,\sigma)} (V_{\mathrm{Ad}(A^{-1})X}(J,\sigma)), \tag{6.1}$$

where $V_X(J,\sigma) := \frac{\mathrm{d}}{\mathrm{d}t} \exp(tX) \cdot (J,\sigma)|_{t=0} \in T_{(J,\sigma)}T^*\mathcal{J}(\mathbb{R}^2)$. Actually, this relation is true whenever a Lie group acts by diffeomorphisms on a smooth manifold. For future convenience, we give also an explicit description of V_X :

$$V_X(J,\sigma) = \frac{\mathrm{d}}{\mathrm{d}t} \left(\exp(tX) J \exp(-tX), \exp(-tX)^* \sigma \right) \Big|_{t=0} = ([X,J], -\sigma(X\cdot, \cdot) - \sigma(\cdot, X\cdot)).$$
(6.2)

We also notice that the action of $\mathbb{P}SL(2,\mathbb{R})$ on $\mathcal{J}(\mathbb{R}^2)$ is by biholomorphisms with respect to the complex structure \mathcal{I} (defined in Section 3.1), and its extension on $T^*\mathcal{J}(\mathbb{R}^2)$ is natural, in the sense that it preserves the complex Liouville form $\lambda^{\mathbb{C}}$ of $T^*\mathcal{J}(\mathbb{R}^2)$ (see Section 3.4 for the definition of $\lambda^{\mathbb{C}}$). This in particular implies that $\mathcal{L}_{V_X}\lambda^{\mathbb{C}} = 0$ for every $X \in \mathfrak{sl}_2(\mathbb{R})$.

Property (i) from Definition 4.1. Using the relation $\|\sigma\|_J = \|A \cdot \sigma\|_{A \cdot J}$ (see Lemma 3.3) and the invariance of the trace by conjugation, it is straightforward to check that the maps $\eta_{\mathbf{I}}$, $\eta_{\mathbf{J}}$ and $\eta_{\mathbf{K}}$ are Ad*-equivariant.

Property (ii) from Definition 4.1. From relation (6.2) and the definition of $\lambda^{\mathbb{C}}$, the maps $\eta_{\mathbf{J}}$ and $\eta_{\mathbf{K}}$ satisfy:

$$(\eta_{\mathbf{J}} + i\eta_{\mathbf{K}})_{(J,\sigma)}(X) = (\iota_{V_X} \lambda^{\mathbb{C}})_{(J,\sigma)}.$$

This relation and the observations above are enough to show that $\eta_{\mathbf{J}}$, $\eta_{\mathbf{K}}$ are moment maps with respect to the symplectic structures $\omega_{\mathbf{J}}$, $\omega_{\mathbf{K}}$, respectively. Indeed we have:

$$d(\eta_{\mathbf{J}}^{X} + i\eta_{\mathbf{K}}^{X}) = d(\iota_{V_{X}}\lambda^{\mathbb{C}})$$

$$= \mathcal{L}_{V_{X}}\lambda^{\mathbb{C}} - \iota_{V_{X}} d\lambda^{\mathbb{C}} \qquad (Cartan's formula)$$

$$= \iota_{V_{X}}\omega^{\mathbb{C}} \qquad (def. of \omega^{\mathbb{C}} \text{ and } \mathcal{L}_{V_{X}}\lambda^{\mathbb{C}} = 0)$$

$$= \iota_{V_{X}}\omega_{\mathbf{J}} + i\,\iota_{V_{X}}\omega_{\mathbf{K}}. \qquad (\omega^{\mathbb{C}} = \omega_{\mathbf{J}} + i\omega_{\mathbf{K}})$$

It remains to check that, for every $X \in \mathfrak{sl}_2(\mathbb{R})$, we have $\iota_{V_X}\omega_{\mathbf{I}} = \mathrm{d}\eta_{\mathbf{I}}^X$. To see this, we show that

$$(\|\sigma\|_J^2)' = 2\langle \sigma, \dot{\sigma}_0 \rangle_J. \tag{6.3}$$

Indeed,

$$(\|\sigma\|_{J}^{2})' = \frac{1}{2}\operatorname{tr}((g_{J}^{-1}\sigma)^{2})' = \operatorname{tr}(g_{J}^{-1}\sigma(g_{J}^{-1}\sigma)')$$

$$= \operatorname{tr}(g_{J}^{-1}\sigma(-g_{J}^{-1}\dot{g}_{J}g_{J}^{-1}\sigma + g_{J}^{-1}\dot{\sigma}))$$

$$= \operatorname{tr}((g_{J}^{-1}\sigma)^{2}J\dot{J}) + \operatorname{tr}(g_{J}^{-1}\sigma g_{J}^{-1}\dot{\sigma}_{0}) \qquad (rel. (3.3) \text{ and } \operatorname{tr}_{g_{J}}\sigma = 0)$$

$$= \|\sigma\|_{J}^{2}\operatorname{tr}(J\dot{J}) + \operatorname{tr}(g_{J}^{-1}\sigma g_{J}^{-1}\dot{\sigma}_{0}) \qquad (Lemma 3.9)$$

$$= 2\langle \sigma, \dot{\sigma}_{0}\rangle_{J}. \qquad (\operatorname{tr}(J\dot{J}) = 0)$$

Therefore we have:

$$d\eta_{\mathbf{I}}^{X}(\dot{J},\dot{\sigma}) = \frac{\langle \sigma, \dot{\sigma}_{0} \rangle_{J}}{f(\|\sigma\|_{J})} \operatorname{tr}(JX) + f(\|\sigma\|_{J}) \operatorname{tr}(\dot{J}X).$$

On the other side, we need to determine $(\iota_{V_X}\omega_{\mathbf{I}})(\dot{J},\dot{\sigma})$. Let X^* denote the adjoint of X with respect to the metric g_J . From the definition of $\omega_{\mathbf{I}}$ in equation (3.9) and the expression (6.2), we see that:

$$(\iota_{V_X}\omega_{\mathbf{I}})(\dot{J},\dot{\sigma}) = -f(\|\sigma\|_J)\langle [X,J],J\dot{J}\rangle_J - \frac{1}{f(\|\sigma\|_J)}\langle (\sigma(X\cdot,\cdot) + \sigma(\cdot,X\cdot))_0,\dot{\sigma}_0(\cdot,J\cdot)\rangle_J$$

$$= -\frac{f(\|\sigma\|_J)}{2}\operatorname{tr}\left([X,J]J\dot{J}\right) - \frac{1}{2f(\|\sigma\|_J)}\operatorname{tr}\left(g_J^{-1}\dot{\sigma}_0J(g_J^{-1}\sigma X + X^*g_J^{-1}\sigma)_0\right). \tag{6.4}$$

In order to simplify the last term of the expression, we decompose the endomorphism X as sum of its skew-symmetric part $X^a = -\frac{\operatorname{tr}(JX)}{2}J$ and its symmetric and traceless part X^s (recall that X is traceless, since it lies in $\mathfrak{sl}_2(\mathbb{R})$). A simple application of Lemma 3.9 shows that

$$(g_J^{-1}\sigma X^s + X^s g_J^{-1}\sigma)_0 = 0,$$

while the term in the skew-symmetric part contributes with

$$(g_J^{-1}\sigma X^a + (X^a)^*g_J^{-1}\sigma)_0 = (g_J^{-1}\sigma X^a - X^ag_J^{-1}\sigma)_0 = \operatorname{tr}(JX)Jg_J^{-1}\sigma.$$

As a result we obtain $(g_J^{-1}\sigma X + X^*g_J^{-1}\sigma)_0 = \operatorname{tr}(JX)Jg_J^{-1}\sigma$. Making use of this identity in relation (6.4), we find that

$$(\iota_{V_X}\omega_{\mathbf{I}})(\dot{J},\dot{\sigma}) = -\frac{f(\|\sigma\|_J)}{2}\operatorname{tr}\left([X,J]J\dot{J}\right) - \frac{\operatorname{tr}(JX)}{2f(\|\sigma\|_J)}\operatorname{tr}\left(g_J^{-1}\dot{\sigma}_0J^2g_J^{-1}\sigma\right)$$
$$= f(\|\sigma\|_J)\operatorname{tr}\left(\dot{J}X\right) + \frac{\langle\sigma,\dot{\sigma}_0\rangle_J}{f(\|\sigma\|_J)}\operatorname{tr}(JX),$$

which proves the desired equality $(\iota_{V_X}\omega_{\mathbf{I}})(\dot{J},\dot{\sigma}) = \mathrm{d}\eta_{\mathbf{I}}^X(\dot{J},\dot{\sigma}).$

6.2. Donaldson's construction

Let us now recall briefly the setting of Section 4. We defined $T^*\mathcal{J}(\Sigma)$ as the space of smooth sections of the bundle

$$P(T^*\mathcal{J}(\mathbb{R}^2)) \longrightarrow \Sigma,$$

and an element of $T^*\mathcal{J}(\Sigma)$ identifies with a pair (J, σ) , in which J is a complex structure on Σ , and σ is a symmetric and g_J -traceless 2-tensor. Moreover, a tangent vector $(\dot{J}, \dot{\sigma})$ at (J, σ) can be considered as the data of:

- a section \dot{J} of End $(T\Sigma)$ satisfying $\dot{J}J + J\dot{J} = 0$;
- a symmetric 2-tensor $\dot{\sigma}$ satisfying $\dot{\sigma} = \dot{\sigma}_0 \langle \sigma \mid J\dot{J}\rangle g$

We will often denote by $s = (J, \sigma)$ an element of $T^*\mathcal{J}(\Sigma)$, and by \dot{s} a tangent vector.

Now, given an $SL(2,\mathbb{R})$ -invariant symplectic form ω on $T^*\mathcal{J}(\mathbb{R}^2)$, every vertical space of $P(T^*\mathcal{J}(\mathbb{R}^2))$ inherits a symplectic structure, which we denote by $\hat{\omega}_{s(p)}$. In particular, given $\dot{s}, \dot{s}' \in T_s T^*\mathcal{J}(\Sigma)$ two tangent vectors, we define

$$\omega_s(\dot{s}, \dot{s}') := \int_{\Sigma} \hat{\omega}_s(\dot{s}, \dot{s}') \, \rho. \tag{6.5}$$

This gives a formal symplectic structure on $T^*\mathcal{J}(\Sigma)$, which is preserved by the action of $\operatorname{Symp}_0(\Sigma, \rho)$. Suppose the natural $\operatorname{SL}(2, \mathbb{R})$ -action on $T^*\mathcal{J}(\mathbb{R}^2)$ is Hamiltonian, with moment map $\eta: T^*\mathcal{J}(\mathbb{R}^2) \to \mathfrak{sl}_2(\mathbb{R})^*$. Given any section $s \in T^*\mathcal{J}(\Sigma)$, the moment map η induces a section η_s of the bundle $\operatorname{End}_0(T\Sigma)^*$. Then the action of $\operatorname{Ham}(\Sigma, \rho) < \operatorname{Symp}_0(\Sigma, \rho)$ is Hamiltonian, according to the following result of Donaldson.

Theorem 6.2 (Donaldson's map, [Don03, Theorem 9]). Let ∇ be any torsion-free connection on Σ satisfying $\nabla \rho = 0$. Define the map $\mu \colon T^* \mathcal{J}(\Sigma) \to \Lambda^2(\Sigma)$ as follows:

$$\mu(s) = \hat{\omega}(\nabla_{\bullet} s, \nabla_{\bullet} s) + \langle \eta_s \mid R^{\nabla} \rangle - d(c(\nabla_{\bullet} \eta_s)).$$

Then

- i) $\mu(s)$ is closed for every $s \in T^*\mathcal{J}(\Sigma)$;
- ii) μ is equivariant with respect to the action of $\operatorname{Ham}(\Sigma, \rho)$;
- iii) Given V a vector field in $\mathfrak{H}(\Sigma, \rho)$, and β_V a primitive of $\iota_V \rho$, the differential of the map

$$T^*\mathcal{J}(\Sigma)\ni s\longmapsto \int_{\Sigma}\beta_V\mu(s)\in\mathbb{R}$$

equals

$$\omega_s(\dot{s}, \mathcal{L}_V s) = \int_{\Sigma} \hat{\omega}(\dot{s}, \mathcal{L}_V s) \, \rho.$$

Let us clarify the notation of the theorem above. We set $\hat{\omega}(\nabla_{\bullet}s, \nabla_{\bullet}s)$ to be the 2-form on Σ given by

$$\hat{\omega}(\nabla_{\bullet}s, \nabla_{\bullet}s)(u, v) := \hat{\omega}(\nabla_{u}s, \nabla_{v}s).$$

Moreover, we define

$$c(\nabla_{\bullet}\eta_s)(v) := \sum_{j} \langle \nabla_{e_j} \eta_s \mid (v \otimes e_j^*)_0 \rangle ,$$

where $(e_j)_j$ is a local orthonormal frame and $(e_j^*)_j$ is the associated dual frame in $T^*\Sigma$. In particular $c(\nabla_{\bullet}\eta_s) \in \Gamma(T^*\Sigma)$ is a 1-form on Σ , because $\nabla_{\bullet}\eta_s \in \Gamma(T^*\Sigma \otimes \operatorname{End}_0(T\Sigma)^*)$.

Finally, the curvature tensor R^{∇} of the torsion free connection ∇ is defined as

$$R^{\nabla}(U,V)W := \nabla_V \nabla_U W - \nabla_U \nabla_V W - \nabla_{[V,U]} W ,$$

for all tangent vector fields U, V, W on Σ . Since $R^{\nabla}(U, V)W = -R^{\nabla}(V, U)W$, the tensor R^{∇} can be considered as a section of $\Lambda^2(\Sigma) \otimes \operatorname{End}_0(T\Sigma)$. In particular, for every U, V, we can evaluate the tensor η_s on $R^{\nabla}(U, V) \in \operatorname{End}_0(T\Sigma)$. This determines a 2-form on Σ , that will be denoted by $\langle \eta_s \mid R^{\nabla} \rangle$.

Remark 6.3. Donaldson's construction is actually much more general. It can be applied to the context of an n-dimensional manifold endowed with a volume form (which in our case is Σ), and a symplectic manifold X, together with a moment map for an action of $SL(n,\mathbb{R})$ on X (which in our case is $T^*\mathcal{J}(\mathbb{R}^2)$). Then one constructs a bundle P(X) similarly to our case, and one obtains a moment map for the action of the group of exact diffeomorphisms (which in dimension two correspond to Hamiltonian diffeomorphisms) on the space of sections of the bundle P(X). The general statement of Donaldson's theorem can be found in [Don03, Theorem 9], see also [Tra19]. We decided to state the result only in the situation of our paper, since this reduces remarkably the necessary preliminaries.

We also recall the following fact, which of course can be stated in much larger generality as explained in Remark 6.3 above:

Lemma 6.4 ([Don03, Lemma 13]). There exists a natural closed 2-form $\hat{\omega}_{P(T^*\mathcal{J}(\mathbb{R}^2))}$ on $P(T^*\mathcal{J}(\mathbb{R}^2))$ such that, for every section $s \in \Gamma(\Sigma, P(T^*\mathcal{J}(\mathbb{R}^2)))$ we have

$$s^* \hat{\omega}_{P(T^* \mathcal{J}(\mathbb{R}^2))} = \hat{\omega}(\nabla_{\bullet} s, \nabla_{\bullet} s) + \langle \eta_s \mid R^{\nabla} \rangle.$$

In particular, since $T^*\mathcal{J}(\mathbb{R}^2)$ is contractible, the de Rham cohomology class of $\mu(s)$ in $H^2(\Sigma)$ is independent of the chosen section.

6.3. The moment maps on $T^*\mathcal{J}(\Sigma)$

In Section 4, we defined a formal para-hyperKähler structure $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$ on $T^*\mathcal{J}(\Sigma)$. Note that the corresponding symplectic forms

$$(\omega_{\mathbf{X}})_{(J,\sigma)}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) := \int_{\Sigma} \hat{\omega}_{\mathbf{X}}((\dot{J},\dot{\sigma}),(\dot{J}',\dot{\sigma}')) \,\rho, \tag{6.6}$$

for $\mathbf{X} = \mathbf{I}, \mathbf{J}, \mathbf{K}$ (here $\hat{\omega}_{\mathbf{X}}$ denotes the symplectic forms induced on each fiber by an areapreserving identification of the tangent space of Σ with \mathbb{R}^2), can be obtained from the general theory of Donaldson, specifically from Equation (6.5) by integrating fiberwise the three symplectic forms introduced in the toy model. Therefore, the group $\operatorname{Symp}_0(\Sigma, \rho)$ acts on $T^*\mathcal{J}(\Sigma)$ preserving all symplectic forms $\omega_{\mathbf{X}}$ and the action of $\operatorname{Ham}(\Sigma, \rho)$ is actually Hamiltonian. In order to describe the moment map $\mu_{\mathbf{X}}$ for the action of $\operatorname{Ham}(\Sigma, \rho)$ with respect to each of the symplectic form $\omega_{\mathbf{X}}$ it is convenient to introduce the following operator:

$$r: \quad \Gamma(T^*_{(1,0)}\Sigma \otimes (T^*_{(0,1)}\Sigma)^{2\otimes}) \quad \longrightarrow \quad \Gamma(T^*_{(0,1)}\Sigma)$$

$$\psi \qquad \qquad \longmapsto \qquad \frac{\psi(v,v,\cdot)}{\|v\|_J^2}$$

for some $v \neq 0$. Since ψ is \mathbb{C} -linear in the first component and anti- \mathbb{C} -linear in the second one, the 1-form $\frac{\psi(v,v,\cdot)}{\|v\|_J^2}$ does not depend on the choice of $v \neq 0$.

Theorem 6.5. Donaldson's maps for the action of $\operatorname{Ham}(\Sigma, \rho)$ over $T^*\mathcal{J}(\Sigma)$ can be expressed as:

$$\mu_{\mathbf{I}}(J,\sigma) = \frac{\|\bar{\partial}\phi\|^2 - \|\partial\phi\|^2}{f(\|\sigma\|)} \rho - 2f(\|\sigma\|)K_J \rho - 2i\bar{\partial}\partial f(\|\sigma\|),$$
$$(\mu_{\mathbf{J}} + i\mu_{\mathbf{K}})(J,\sigma) = -2i\,\partial r(\partial\bar{\phi}),$$

where ϕ is the quadratic differential whose real part is equal to σ , and $\partial = \partial_J$, $\bar{\partial} = \bar{\partial}_J$.

Proof. In order to determine the expressions for Donaldson's maps $\mu_{\mathbf{I}}, \mu_{\mathbf{J}}, \mu_{\mathbf{K}}$, we apply Theorem 6.2 starting from the moment maps $\eta_{\mathbf{I}}, \eta_{\mathbf{J}}, \eta_{\mathbf{K}}$ introduced in Section 6.1. We will focus on each term of Donaldson's maps individually, and then combine the resulting expressions to determine the evaluations of $\mu_{\mathbf{I}}, \mu_{\mathbf{J}}, \mu_{\mathbf{K}}$ at $(J, \sigma) \in T^* \mathcal{J}(\Sigma)$. As a torsion-free connection we will use the Levi-Civita connection of the metric g_J , which satisfies $\nabla_{\bullet} \rho \equiv 0$, since ρ equals the Riemannnian volume form of g_J .

The term $\hat{\omega}_{\mathbf{I}}(\nabla_{\bullet} s, \nabla_{\bullet} s)$: Since ∇ is the Levi-Civita connection of g_J , we have $\nabla_{\bullet} J \equiv 0$. In particular $\nabla_{\bullet} s = (0, \nabla_{\bullet} \sigma)$. Then we have:

$$(\hat{\omega}_{\mathbf{I}})_{(J,\sigma)}((0, \nabla_{e_{1}}\sigma), (0, \nabla_{e_{2}}\sigma)) = \frac{1}{2f(\|\sigma\|)} \operatorname{tr} \left(g_{J}^{-1}(\nabla_{e_{1}}\sigma)g_{J}^{-1}(\nabla_{e_{2}}\sigma)J\right)$$

$$= \frac{1}{2f(\|\sigma\|)} \left(\nabla_{1}\sigma_{11}\nabla_{2}\sigma_{12} - \nabla_{1}\sigma_{22}\nabla_{2}\sigma_{21} + \nabla_{1}\sigma_{12}\nabla_{2}\sigma_{22} - \nabla_{1}\sigma_{21}\nabla_{2}\sigma_{11}\right)$$

$$= \frac{1}{f(\|\sigma\|)} \left(\nabla_{1}\sigma_{11}\nabla_{2}\sigma_{12} - \nabla_{1}\sigma_{12}\nabla_{2}\sigma_{11}\right),$$

where $\nabla_i \sigma_{jk} = (\nabla_{e_i} \sigma)(e_j, e_k)$. In the last step we used the fact that $\nabla_v \sigma$ is symmetric and g_J -traceless for every v. The operators ∂_J and $\bar{\partial}_J$ are defined as follows:

$$(\partial_J \phi)(v,\cdot,\cdot) = \frac{1}{2} \left(\nabla_v \phi - i \nabla_{Jv} \phi \right), \qquad (\bar{\partial}_J \phi)(v,\cdot,\cdot) = \frac{1}{2} \left(\nabla_v \phi + i \nabla_{Jv} \phi \right).$$

A simple but tedious computation shows that

$$\|\bar{\partial}_{J}\phi(e_{1},\cdot,\cdot)\|^{2} - \|\partial_{J}\phi(e_{1},\cdot,\cdot)\|^{2} = \nabla_{1}\sigma_{11}\nabla_{2}\sigma_{12} - \nabla_{1}\sigma_{12}\nabla_{2}\sigma_{11}.$$

In the end, we get

$$\hat{\omega}_{\mathbf{I}}(\nabla_{\bullet}(J,\sigma),\nabla_{\bullet}(J,\sigma)) = \frac{\left\|\bar{\partial}_{J}\phi\right\|^{2} - \left\|\bar{\partial}_{J}\phi\right\|^{2}}{f(\|\sigma\|)}\rho$$

where $\|\partial_J \phi\| := \|\partial_J \phi(v, \cdot, \cdot)\|$, $\|\bar{\partial}_J \phi\| := \|\bar{\partial}_J \phi(v, \cdot, \cdot)\|$ for some unit vector v (the norm is independent of the choice of such v).

The term $\langle \eta_{\mathbf{I}} \mid R^{\nabla} \rangle$: Since we are considering the Levi-Civita connection of g_J , the tensor R^{∇} coincides with the Riemann tensor of g_J . Then we have $R_J = K_J J \otimes \rho$, where K_J denotes the Gaussian curvature of g_J . In particular $\langle \eta_{\mathbf{I}}(J,\sigma) \mid R_J \rangle = -2f(\|\sigma\|)K_J \rho$.

The term $d(c(\nabla_{\bullet}\eta_{\mathbf{I}}))$: First we observe that

$$(\nabla_v \eta_{\mathbf{I}(J,\sigma)})(X) = d(f(\|\sigma\|))(v)\operatorname{tr}(JX),$$

where $X \in \mathfrak{sl}(T.\Sigma, \rho) = \operatorname{End}_0(T.\Sigma)$. Then we have

$$c(\nabla_{\bullet}\eta_{\mathbf{I}(J,\sigma)})(w) := \langle \nabla_{e_{1}}\eta_{\mathbf{I}(J,\sigma)} \mid (w \otimes e_{1}^{*})_{0} \rangle + \langle \nabla_{e_{2}}\eta_{\mathbf{I}(J,\sigma)} \mid (w \otimes e_{2}^{*})_{0} \rangle$$

$$= d(f(\|\sigma\|)) (e_{1}) \operatorname{tr}(J(w \otimes e_{1}^{*})_{0}) + d(f(\|\sigma\|)) (e_{2}) \operatorname{tr}(J(w \otimes e_{2}^{*})_{0})$$

$$= d(f(\|\sigma\|)) (e_{1}) e_{1}^{*}(Jw) + d(f(\|\sigma\|)) (e_{2}) e_{2}^{*}(Jw)$$

$$= (d(f(\|\sigma\|)) \circ J)(w).$$

Therefore $c(\nabla_{\bullet}\eta_{\mathbf{I}(J,\sigma)}) = d(f(\|\sigma\|)) \circ J$. It is immediate to check that, for every function $\varphi \in \mathscr{C}^{\infty}(\Sigma)$, we have $d(d\varphi \circ J) = -\Delta_{q_J} \varphi \, \rho = -2i\partial_J \bar{\partial}_J \varphi = 2i\bar{\partial}_J \partial_J \varphi$.

The combination of the terms found above provides the desired description of the Donaldson's map $\mu_{\mathbf{I}}$. We now focus on the addends appearing in the expression of $(\mu_{\mathbf{J}} + i\mu_{\mathbf{K}})(J, \sigma)$.

The term $(\hat{\omega}_{\mathbf{J}} + i \hat{\omega}_{\mathbf{K}})(\nabla_{\bullet} s, \nabla_{\bullet} s)$: Since $\nabla_{\bullet} J = 0$, we have

$$\hat{\omega}^{\mathbb{C}}(\nabla_{\bullet}(J,\sigma),\nabla_{\bullet}(J,\sigma)) = (\hat{\omega}_{\mathbf{J}} + i\hat{\omega}_{\mathbf{K}})((0,\nabla_{\bullet}\sigma),(0,\nabla_{\bullet}\sigma)) = 0$$

The term $\langle \eta_{\mathbf{J}} + i \eta_{\mathbf{K}} \mid R^{\nabla} \rangle$: Observe that $\langle (\eta_{\mathbf{J}} + i \eta_{\mathbf{K}})_{(J,\sigma)} \mid K_J J \otimes \rho \rangle = 0$ because [J, J] = 0.

The term $d(c(\nabla_{\bullet}\eta_{\mathbf{J}} + i\eta_{\mathbf{K}}))$: From the definition of the moment maps $\eta_{\mathbf{J}}$ and $\eta_{\mathbf{K}}$, we observe the following:

$$\langle (\eta_{\mathbf{J}} + i \, \eta_{\mathbf{K}})_{(J,\sigma)} \mid X \rangle = -\operatorname{tr} \left(g_J^{-1} \sigma J X \right) + i \operatorname{tr} \left(g_J^{-1} \sigma X \right)$$
$$= i \operatorname{tr} \left(g_J^{-1} (\sigma + i \, \sigma(\cdot, J \cdot)) X \right)$$
$$= i \operatorname{tr} \left(g_J^{-1} \bar{\phi} X \right), \tag{rel. (3.1)}$$

where we are denoting by ϕ the quadratic differential whose real part is equal to σ . Since ∇ is the Levi-Civita connection of g_J , we have $\nabla_{\bullet}g_J \equiv 0$ and, as already mentioned, $\nabla_{\bullet}J \equiv 0$. In particular we deduce that:

$$c(\nabla_{\bullet}(\eta_{\mathbf{J}} + i \eta_{\mathbf{K}}))(v) = \sum_{j} \langle \nabla_{e_{j}}(\eta_{\mathbf{J}} + i \eta_{\mathbf{K}}) \mid (v \otimes e_{j}^{*})_{0} \rangle$$

$$= i \sum_{j} \operatorname{tr}(g_{J}^{-1}(\nabla_{e_{j}}\bar{\phi})(v \otimes e_{j}^{*})_{0})$$

$$= i \sum_{j} \operatorname{tr}(g_{J}^{-1}(\nabla_{e_{j}}\bar{\phi})(v \otimes e_{j}^{*}))$$

$$= i \sum_{j} (\nabla_{e_{j}}\bar{\phi})(v, e_{j})$$

$$= i (\operatorname{div}_{g_{J}} g_{J}^{-1}\bar{\phi})(v).$$

On the other hand, we see that:

$$r(\partial_J \bar{\phi})(v) = (\partial_J \bar{\phi})(e_1, e_1, v)$$

$$= \frac{1}{2}((\nabla_{e_1} \bar{\phi})(e_1, v) - i(\nabla_{Je_1} \bar{\phi})(e_1, v))$$

$$= \frac{1}{2}((\nabla_{e_1} \bar{\phi})(e_1, v) + (\nabla_{Je_1} \bar{\phi})(Je_1, v))$$

$$= \frac{1}{2}(\operatorname{div}_{g_J} g_J^{-1} \bar{\phi})(v).$$

Therefore $c(\nabla_{\bullet}(\eta_{\mathbf{J}} + i \eta_{\mathbf{K}})) = 2i \, r(\partial_J \bar{\phi}) \in \Gamma(T^*_{(0,1)}\Sigma)$. Since $d = \partial_J + \bar{\partial}_J$ and $\bar{\partial}_J(r(\partial_J \bar{\phi})) = 0$, we obtain

$$d(c(\nabla_{\bullet}(\eta_{\mathbf{J}} + i \eta_{\mathbf{K}}))) = 2i \,\partial_{J} r(\partial_{J} \bar{\phi}),$$

which, combined with Theorem 6.2, proves the expression for Donaldson's map $\mu_{\mathbf{J}} + i \mu_{\mathbf{K}}$.

Proposition 6.6. Let ρ be a fixed volume form on Σ , and let $C := -\frac{4\pi\chi(\Sigma)}{\operatorname{Area}(\Sigma,\rho)}$.

• The map

$$\begin{array}{cccc} \tilde{\mu}_{\mathbf{I}}: & T^{*}\mathcal{J}(\Sigma) & \longrightarrow & \mathfrak{H}(\Sigma,\rho)^{*} \\ & & (J,\sigma) & \longmapsto & \mu_{\mathbf{I}}(J,\sigma) - C\,\rho \end{array}$$

П

is a moment map for the action of $\operatorname{Ham}(\Sigma, \rho)$ over the space of smooth sections $T^*\mathcal{J}(\Sigma)$, endowed with the symplectic form $\omega_{\mathbf{I}}$.

• The map

$$\tilde{\mu}_{\mathbf{J}} + i \, \tilde{\mu}_{\mathbf{K}} : T^* \mathcal{J}(\Sigma) \longrightarrow \mathfrak{S}(\Sigma, \rho)^* \otimes \mathbb{C}$$

$$(J, \sigma) \longmapsto [-2i \, r(\partial_J \bar{\phi})]$$

is a moment map for the action of $\operatorname{Symp}_0(\Sigma, \rho)$ over the space of smooth sections $T^*\mathcal{J}(\Sigma)$, endowed with the symplectic form $\omega_{\mathbf{J}} + i\,\omega_{\mathbf{K}}$ (where ϕ denotes the quadratic differential whose real part is equal to σ).

Proof. By Lemma 6.4, the integral of the 2-form $\mu_{\mathbf{I}}(J,\sigma)$ does not depend on the choice of the section $(J,\sigma) \in T^*\mathcal{J}(\Sigma)$. If $\sigma \equiv 0$, then by Gauss-Bonnet theorem we have

$$\int_{\Sigma} \mu_{\mathbf{I}}(J,0) = -4\pi \chi(\Sigma).$$

Therefore, the integral of the 2-form $\mu_{\mathbf{I}}(J,\sigma) - C\rho$ vanishes. By Poincaré's duality, a 2-form on Σ is exact if and only if its integral over Σ vanishes. This in particular proves that $\tilde{\mu}_{\mathbf{I}}$ takes values in $B^2(\Sigma)$, which is contained in the dual of the Lie algebra of $\operatorname{Ham}(\Sigma,\rho)$, as observed in Section 4.2. Following Definition 4.1, the analogous properties i) and ii) for $\tilde{\mu}_{\mathbf{I}}$ are guaranteed by Theorem 6.2, since the term $C\rho$ is independent of the section $(J,\sigma) \in T^*\mathcal{J}(\Sigma)$.

Our moment map $\mu_{\mathbf{J}} + i \, \mu_{\mathbf{K}}$ is exactly the same as the map $\mu_2 + i \, \mu_3$ appearing in the original process of symplectic reduction developed by [Don03] to describe the hyperKähler structure on the space of almost-Fuchsian manifolds. In particular, the argument in [Don03, Section 3.1] to $\mu_2 + i \, \mu_3$ applies verbatim to our context, showing that $\mu_{\mathbf{J}} + i \, \mu_{\mathbf{K}}$ can be promoted to a moment map $\tilde{\mu}_{\mathbf{J}} + i \, \tilde{\mu}_{\mathbf{K}}$ for the action of $\operatorname{Symp}_0(\Sigma, \rho)$ on $T^*\mathcal{J}(\Sigma)$. For a more detailed exposition of this phenomenon, we refer to [Tra19, Theorem 4.10].

6.4. The symplectic quotient

We are interested in the symplectic quotient

$$\tilde{\mu}_{\mathbf{I}}^{-1}(0) \cap \tilde{\mu}_{\mathbf{J}}^{-1}(0) \cap \tilde{\mu}_{\mathbf{K}}^{-1}(0) / \operatorname{Symp}_{0}(\Sigma, \rho) \ .$$

The aim of this section is to show that this quotient can be identified with $\mathcal{MS}(\Sigma)$ so that the para-hyperKähler structure on $\mathcal{MS}(\Sigma)$ defined in Section 4 is inherited from the infinite dimensional space $T^*\mathcal{J}(\Sigma)$. Although the arguments are inspired by similar constructions in hyperKähler geometry ([Don03], [Tra19], [Hod05]), substantial difficulties arise when dealing with a pseudo-Riemannian metric, which we now explain.

In the more common setting of infinite dimensional hyperKähler quotients, one starts with the data of an infinite dimensional manifold M endowed with three complex structures \mathbf{I}, \mathbf{J} and \mathbf{K} satisfying the relations of quaternionic numbers and a Riemannian metric \mathbf{g} compatible with each of the complex structures. This defines three symplectic forms $\omega_{\mathbf{X}} = \mathbf{g}(\cdot, \mathbf{X})$ for $\mathbf{X} = \mathbf{I}, \mathbf{J}, \mathbf{K}$ on M. Assume now that a group G acts on M by Hamiltonian

symplectomorphisms with respect to each of the symplectic forms with moment map $\zeta_{\mathbf{X}}$ and let N be the submanifold

$$N = \zeta_{\mathbf{I}}^{-1}(0) \cap \zeta_{\mathbf{J}}^{-1}(0) \cap \zeta_{\mathbf{K}}^{-1}(0)$$
.

The properties of the moment maps and the Riemannian metric \mathbf{g} allow to orthogonally decompose the tangent space of M at every point $p \in N$ as

$$T_pM = T_pN \oplus \mathbf{I}(T_p(G \cdot p)) \oplus \mathbf{J}(T_p(G \cdot p)) \oplus \mathbf{K}(T_p(G \cdot p))$$

and the **g**-orthogonal of $T_p(G \cdot p)$ inside T_pN furnishes a model for the tangent space of the quotient N/G that is invariant under \mathbf{I}, \mathbf{J} and \mathbf{K} . This is sufficient to conclude that the quotient inherits a hyperKähler structure ([Hit87]).

In our pseudo-Riemannian setting, however, the absence of a Hilbert space structure on the tangent space prevents us from obtaining a similar orthogonal decomposition using only the properties of the moment maps. However, the hyperKähler construction suggests that the tangent space at (J, σ) to the zero locus of the three moment maps is the largest subspace $V_{(J,\sigma)} \subset T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$ that is **g**-orthogonal to $T_{(J,\sigma)}(\operatorname{Symp}_0(\Sigma,\rho)\cdot(J,\sigma))$ and invariant under **I**, **J** and **K**, inside the kernel of the differential of the three moment maps. These are the properties that led us to the equations defining $V_{(J,\sigma)}$ in Section 4. In what follows we explain how to derive the equations in Proposition K starting from these geometric conditions.

6.4.1. Identifying $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$ as the zero locus of the moment maps

Let us start by characterizing the pairs (J, σ) that lie in the zero locus of the three moment maps.

Lemma 6.7. A pair $(J, \sigma) \in T^*\mathcal{J}(\Sigma)$ satisfies $(\tilde{\mu}_{\mathbf{J}} + i \, \tilde{\mu}_{\mathbf{K}})(J, \sigma) = 0$ if and only if σ is the real part of a holomorphic quadratic differential ϕ on the Riemann surface (Σ, J) .

Proof. Let us consider the subalgebra of $\mathfrak{S}(\Sigma, \rho)$ defined by

$$\mathfrak{h}_J = \{ V \in \Gamma(T\Sigma) \mid d(\iota_V \rho) = d(\iota_{JV} \rho) = 0 \} .$$

Because $(\tilde{\mu}_{\mathbf{J}} + i \, \tilde{\mu}_{\mathbf{K}})(J, \sigma) = 0$, by definition of moment map, we have

$$0 = -2i \int_{\Sigma} r(\partial_J \bar{\phi}) \wedge \iota_V \rho$$

for every $V \in \mathfrak{h}_J$. Note that the form $r(\partial_J \bar{\phi})$ is anti-holomorphic, being it in the zero locus of $\mu_{\mathbf{J}} + i \, \mu_{\mathbf{K}}$. Since $\{\iota_V \rho \mid V \in \mathfrak{h}_J\}$ parameterizes harmonic 1-forms on (Σ, J) , by Poincaré duality the real and imaginary parts of $r(\partial_J \bar{\phi})$ are exact. Hence, $r(\partial_J \bar{\phi})$ is exact and identically zero because the only anti-holomorphic functions on (Σ, J) are constant. This shows that $\partial_J \bar{\phi} = 0$, thus ϕ is a holomorphic quadratic differential on (Σ, J) .

Lemma 6.8. Let $(J, \sigma) \in T^*\mathcal{J}(\Sigma)$. If σ is the real part of a holomorphic quadratic differential, then

$$\tilde{\mu}_{\mathbf{I}}(J,\sigma) = -\frac{4K_h}{1 + \det B} \rho - C\rho,$$

where

$$h := (1 + f(\|\sigma\|)) g_J, \qquad B := h^{-1}\sigma \qquad C := -\frac{4\pi\chi(\Sigma)}{\operatorname{Area}(\Sigma, \rho)}.$$

Proof. If σ is the zero quadratic differential, then the statement is immediate. From now on, we will assume σ to be non-zero. We make use of the computations developed by Trautwein in [Tra19]. In particular, we will need the following relations: if λ denotes the function $\|\sigma\|^2$, then, outside the zeros of σ ,

$$-\frac{i}{2\lambda}\bar{\partial}\lambda \wedge \partial\lambda = \|\partial\phi\|^2 \rho, \qquad \bar{\partial}\partial(\ln\lambda) = 2i K_J \rho,$$

where $\partial = \partial_J$ and $\bar{\partial} = \bar{\partial}_J$. In particular we can write $\tilde{\mu}_{\mathbf{I}}$ as follows:

$$\tilde{\mu}_{\mathbf{I}}(J,\sigma) = i \left(\frac{\bar{\partial}\lambda \wedge \bar{\partial}\lambda}{2\lambda\sqrt{1+\lambda}} + \sqrt{1+\lambda} \; \bar{\partial}\partial(\ln\lambda) - 2\bar{\partial}\partial(\sqrt{1+\lambda}) \right) - C\rho$$

From here, the strategy is the same as the one in [Tra18, Proposition 4.5.16]. In particular, we can find the following sequence of identities:

$$\widetilde{\mu}_{\mathbf{I}}(J,\sigma) = i \left(\frac{\overline{\partial}\lambda \wedge \partial\lambda}{2\lambda\sqrt{1+\lambda}} + \sqrt{1+\lambda} \,\overline{\partial}\partial(\ln\lambda) - 2\overline{\partial}\partial(\sqrt{1+\lambda}) \right) - C\rho$$

$$= i \left(\overline{\partial}(\sqrt{1+\lambda}) \wedge \partial(\ln\lambda) + \sqrt{1+\lambda} \,\overline{\partial}\partial(\ln\lambda) - 2\overline{\partial}\partial(\sqrt{1+\lambda}) \right) - C\rho$$

$$= i\overline{\partial} \left(\sqrt{1+\lambda} \,\partial(\ln\lambda) - 2\partial(\sqrt{1+\lambda}) \right) - C\rho$$

$$= i\overline{\partial} \left(-2\partial \left(\ln\left(1+\sqrt{1+\lambda}\right) \right) + \partial(\ln\lambda) \right) - C\rho$$

$$= \left(\Delta_{g_J} \ln\left(1+\sqrt{1+\|\sigma\|^2}\right) - 2K_J - C \right) \rho,$$

$$= \left(\Delta_{g_J} \ln(1+f(\|\sigma\|)) - 2K_J - C \right) \rho,$$

where, in the second to last step, we used again the identity $\bar{\partial}\partial(\ln \lambda) = 2i K_J \rho$, and the fact that $-2i\bar{\partial}\partial\varphi = \Delta_{g_J}\varphi \rho$.

From the formula of the curvature under conformal change, we obtain that the Gaussian curvature K_h of the metric h satisfies:

$$K_h = \frac{1}{1 + f(\|\sigma\|)} \left(K_J - \frac{1}{2} \Delta_{g_J} \ln \left(1 + f(\|\sigma\|) \right) \right)$$

A computation exactly as in Lemma 2.15 shows the identity

$$\frac{2}{1+f(\|\sigma\|)} = 1 + \det B.$$

If we combine this relation with the expression for the Gaussian curvature of h and the description we found above for $\tilde{\mu}_{\mathbf{I}}$, we see that

$$\tilde{\mu}_{\mathbf{I}}(J,\sigma) = -\left(\frac{4K_h}{1+\det B} + C\right)\,\rho.$$

This proves the desired relation outside of the zero locus of σ , which is a finite set. The statement then follows by continuity of the expression.

Combining Lemma 6.7 and Lemma 6.8, we can then identify the zero locus of the three moment maps with $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$. Recall that the symplectic form ρ that we fixed in Section 2.5 has area $-\pi\chi(\Sigma)$.

Corollary 6.9. Let $(J, \sigma) \in T^*\mathcal{J}(\Sigma)$. Then $\widetilde{\mu}_{\mathbf{I}}(J, \sigma) = \widetilde{\mu}_{\mathbf{J}}(J, \sigma) = \widetilde{\mu}_{\mathbf{K}}(J, \sigma) = 0$ if and only if $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$.

Proof. Recall from Section 2.5 that $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$ is the space of pairs (J, σ) such that $h = (1 + f(\|\sigma\|))g_J$ and $B = h^{-1}\sigma$ satisfy the Gauss-Codazzi equations. By Lemma 6.7, we know that σ is the real part of a holomorphic quadratic differential, hence B is Codazzi by Lemma 2.3. Now, from our choice of ρ , we see that

$$C = -\frac{4\pi\chi(\Sigma)}{\operatorname{Area}(\Sigma, \rho)} = 4$$

so by Lemma 6.8 we have that $\tilde{\mu}_{\mathbf{I}}(J,\sigma) = 0$ if and only if $K_h = -1 - \det(B)$, which is exactly the Gauss equation for space-like surfaces in Anti-de Sitter space.

6.4.2. The differential of the map $\mu_{\rm I}$

One subtlety we have to take care of concerns the fact that the action of $\operatorname{Symp}_0(\Sigma, \rho)$ is not Hamiltonian with respect to $\omega_{\mathbf{I}}$. However, we can show by an explicit computation that (up to a sign) property ii) in Definition 4.1

$$\langle d\tilde{\mu}_{\mathbf{I}}(\dot{J},\dot{\sigma}) \mid V \rangle_{\mathfrak{S}} = \omega_{\mathbf{I}}((\dot{J},\dot{\sigma}), (\mathcal{L}_{V}J, \mathcal{L}_{V}\sigma))$$

holds for every $(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$ and for every $V \in \mathfrak{S}(\Sigma, \rho)$, not only for $V \in \mathfrak{H}(\Sigma, \rho)$.

Proposition 6.10. For every $(J, \sigma) \in T^*\mathcal{J}(\Sigma)$ such that σ is the real part of a holomorphic quadratic differential, and for every tangent vector $(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$ we have

$$\mathrm{d}\tilde{\mu}_{\mathbf{I}}(\dot{J},\dot{\sigma}) = -\,\mathrm{d}\Big(f\,\mathrm{div}_{g_J}\,\dot{J} + \mathrm{d}\dot{f}\circ J + \mathrm{d}f\circ\dot{J} - f^{-1}\beta\Big) \ ,$$

where $f = f(\|\sigma\|_J) = \sqrt{1 + \|\sigma\|_J^2}$ and β is the 1-form on Σ defined by $\beta(V) := \langle \dot{\sigma}_0, (\nabla_V \sigma)(\cdot, J \cdot) \rangle$.

Proof. The main bulge of the proof is to show that

$$\left(\frac{\left\|\bar{\partial}\phi\right\|^2 - \left\|\partial\phi\right\|^2}{f(\|\sigma\|)}\rho\right)' = d(f^{-1}\beta) + 2\dot{f}K_J\rho + \operatorname{div}_{g_J}\dot{J} \wedge df, \tag{6.7}$$

where we are taking the derivative of the expression with respect to (J, σ) in the direction $(\dot{J}, \dot{\sigma})$. Assuming this relation for now, we see how to conclude. The variations of the other two terms appearing in the expression for $\tilde{\mu}_{\mathbf{I}}$ (see Theorem 6.5) are simpler to handle. We have that

$$(-2fK_J\rho)' = -2\dot{f}K_J\rho - 2fdK_J(\dot{J})\rho$$
$$= -2\dot{f}K_J\rho - fd\left(\operatorname{div}_{g_J}\dot{J}\right),$$

where in the last step we used the expression for the derivative of the curvature described in Remark 4.4, and

$$(\Delta_{g_J} f)' \rho = -\operatorname{d}((\operatorname{d} f \circ J)') = -\operatorname{d}(\operatorname{d} \dot{f} \circ J + \operatorname{d} f \circ \dot{J}).$$

By combining these relations, we find

$$d\tilde{\mu}_{\mathbf{I}}(\dot{J},\dot{\sigma}) = d(f^{-1}\beta) + 2\dot{f}K_{J}\rho + \operatorname{div}_{g_{J}}\dot{J} \wedge df - 2\dot{f}K_{J}\rho - f d(\operatorname{div}_{g_{J}}\dot{J}) - d(d\dot{f} \circ J + df \circ \dot{J})$$

$$= -d(f \operatorname{div}_{g_{J}}\dot{J} + df \circ \dot{J} + d\dot{f} \circ J - f^{-1}\beta),$$

which proves the desired formula.

Let us now focus on (6.7). As derived in the proof of Theorem 6.5, we have that

$$\frac{\left\|\bar{\partial}\phi\right\|^2 - \|\partial\phi\|^2}{f(\|\sigma\|)} = \hat{\omega}_{\mathbf{I}}((0, \nabla_{e_1}\sigma), (0, \nabla_{e_2}\sigma)), \tag{6.8}$$

for any choice of a local frame e_1, e_2 satisfying $\rho(e_1, e_2) = 1$. We will determine relation (6.7) by computing the derivative

$$\left(\frac{\|\bar{\partial}\phi\|^2 - \|\partial\phi\|^2}{f(\|\sigma\|)}\right)' = (\hat{\omega}_{\mathbf{I}}((0, \nabla_{e_1}\sigma), (0, \nabla_{e_2}\sigma)))'$$

$$= (f^{-1}\langle \nabla_{e_1}\sigma, (\nabla_{e_2}\sigma)(\cdot, J\cdot)\rangle)'$$

$$= \left(\frac{1}{2}f^{-1}\operatorname{tr}\left(g_J^{-1}(\nabla_{e_1}\sigma)g_J^{-1}(\nabla_{e_2}\sigma)J\right)\right)'$$

In order to simplify the development of this first order variation, we need a few preliminary observations. Since equation (6.8) holds for any local frame e_1, e_2 with ρ -volume equal to 1, we can assume e_1, e_2 to be a g_J -orthonormal frame and to not change under the variation \dot{J} . Moreover, in light of Lemma 3.9 part ii) and relation (3.3), the traces of the terms involving the derivatives of g_J^{-1} and of J vanish. This allows us to reduce the number of addends to study. In particular we have that

$$\left(\frac{\|\bar{\partial}\phi\|^2 - \|\partial\phi\|^2}{f(\|\sigma\|)}\right)' = -f^{-2}\dot{f}\langle\nabla_{e_1}\sigma, (\nabla_{e_2}\sigma)(\cdot, J\cdot)\rangle + f^{-1}\langle(\nabla_{e_1}\sigma)', (\nabla_{e_2}\sigma)(\cdot, J\cdot)\rangle + f^{-1}\langle\nabla_{e_1}\sigma, (\nabla_{e_2}\sigma)'(\cdot, J\cdot)\rangle.$$

$$+ f^{-1}\langle\nabla_{e_1}\sigma, (\nabla_{e_2}\sigma)'(\cdot, J\cdot)\rangle.$$
(6.9)

Now, using Lemma 4.15, we can obtain an expression for $(\nabla_{e_i}\sigma)'$. In fact, observing that

$$(\nabla_V \sigma)' = \nabla_V \dot{\sigma} + \dot{\nabla}_V \sigma , \qquad (6.10)$$

we compute

$$(\dot{\nabla}_{V}\sigma)(X,Y) = -\sigma(\dot{\nabla}_{V}X,Y) - \sigma(X,\dot{\nabla}_{V}Y)$$

$$= \frac{1}{2} \left((\operatorname{div}_{g_{J}} \dot{J})V(\sigma(JX,Y) + \sigma(X,JY)) + \sigma(J(\nabla_{V}\dot{J})X,Y) + \sigma(X,J(\nabla_{V}\dot{J})Y) \right)$$

$$= (\operatorname{div}_{g_{J}} \dot{J})V \sigma(JX,Y) + \langle \sigma \mid J(\nabla_{V}\dot{J}) \rangle g_{J}(X,Y) ,$$

where, in the last line, we applied Lemma 3.9 to the products $g_J^{-1}\sigma J(\nabla_V \dot{J})$ and $J(\nabla_V \dot{J})g_J^{-1}\sigma$. Moreover we have

$$\nabla_{V}\dot{\sigma} = \nabla_{V}(\dot{\sigma}_{0} - \langle \sigma \mid J\dot{J}\rangle g_{J}) = \nabla_{V}\dot{\sigma}_{0} - V(\langle \sigma \mid J\dot{J}\rangle) g_{J}$$
$$= \nabla_{V}\dot{\sigma}_{0} - (\langle \nabla_{V}\sigma \mid J\dot{J}\rangle + \langle \sigma \mid J\nabla_{V}\dot{J}\rangle) g_{J}.$$

Combining the previous two equations with (6.10), we get

$$(\nabla_{V}\sigma)' = \nabla_{V}\dot{\sigma}_{0} - (\langle\nabla_{V}\sigma \mid J\dot{J}\rangle + \langle\sigma \mid J\nabla_{V}\dot{J}\rangle)g_{J} + (\operatorname{div}_{g_{J}}\dot{J})V\,\sigma(\cdot,J\cdot) + \langle\sigma \mid J\nabla_{V}\dot{J}\rangle g_{J}$$

$$= \nabla_{V}\dot{\sigma}_{0} - \langle\nabla_{V}\sigma \mid J\dot{J}\rangle g_{J} + (\operatorname{div}_{g_{J}}\dot{J})V\,\sigma(\cdot,J\cdot) . \tag{6.11}$$

In light of these observations, we can combine relations (6.11) and (6.9) to deduce that

$$\left(\frac{\|\bar{\partial}\phi\|^{2} - \|\partial\phi\|^{2}}{f(\|\sigma\|)}\right)' = -f^{-1}\dot{f}\,\hat{\omega}_{\mathbf{I}}((0,\nabla_{e_{1}}\sigma),(0,\nabla_{e_{2}}\sigma)) + f^{-1}\langle\nabla_{e_{1}}\dot{\sigma}_{0},(\nabla_{e_{2}}\sigma)(\cdot,J\cdot)\rangle +
+ f^{-1}\langle\nabla_{e_{1}}\sigma,(\nabla_{e_{2}}\dot{\sigma}_{0})(\cdot,J\cdot)\rangle + f^{-1}(\operatorname{div}_{g_{J}}\dot{J})(e_{1})\langle\sigma(\cdot,J\cdot),(\nabla_{e_{2}}\sigma)(\cdot,J\cdot)\rangle
+ f^{-1}(\operatorname{div}_{g_{J}}\dot{J})(e_{2})\langle\nabla_{e_{1}}\sigma,\sigma(\cdot,J^{2}\cdot)\rangle
= -f^{-1}\dot{f}\,\hat{\omega}_{\mathbf{I}}((0,\nabla_{e_{1}}\sigma),(0,\nabla_{e_{2}}\sigma)) + \hat{\omega}_{\mathbf{I}}((0,\nabla_{e_{1}}\dot{\sigma}_{0}),(0,\nabla_{e_{2}}\sigma))
+ \hat{\omega}_{\mathbf{I}}((0,\nabla_{e_{1}}\sigma),(0,\nabla_{e_{2}}\dot{\sigma}_{0})) + f^{-1}(\operatorname{div}_{g_{J}}\dot{J})(e_{1})\,e_{2}(\|\sigma\|^{2}/2) +
- f^{-1}(\operatorname{div}_{g_{J}}\dot{J})(e_{2})\,e_{1}(\|\sigma\|^{2}/2)
= -f^{-1}\dot{f}\,\hat{\omega}_{\mathbf{I}}((0,\nabla_{e_{1}}\sigma),(0,\nabla_{e_{2}}\sigma)) + \hat{\omega}_{\mathbf{I}}((0,\nabla_{e_{1}}\dot{\sigma}_{0}),(0,\nabla_{e_{2}}\sigma))
+ \hat{\omega}_{\mathbf{I}}((0,\nabla_{e_{1}}\sigma),(0,\nabla_{e_{2}}\dot{\sigma}_{0})) + (\operatorname{div}_{g_{J}}\dot{J}\wedge\operatorname{d}f)(e_{1},e_{2})$$
(6.12)

where in the first step we used the fact that the trace part in the expression for $(\nabla_V \sigma)'$ does not give a contribution. Comparing relation (6.7) with (6.12), the proof is complete if we show that

a)
$$(df \wedge \beta)(e_1, e_2) = f\dot{f}\,\hat{\omega}_{\mathbf{I}}((0, \nabla_{e_1}\sigma), (0, \nabla_{e_2}\sigma))$$

b) $d\beta(e_1, e_2) = f\hat{\omega}_{\mathbf{I}}((0, \nabla_{e_1}\dot{\sigma}_0), (0, \nabla_{e_2}\sigma)) + \hat{\omega}_{\mathbf{I}}((0, \nabla_{e_1}\sigma), (0, \nabla_{e_2}\dot{\sigma}_0))) - 2K_J\langle\dot{\sigma}_0, \sigma\rangle.$

Proof of relation a). First we observe that, if σ is zero, then the relation is obviously satisfied. In what follows we will assume that σ is not identically zero. By definition of β and of wedge product

$$df \wedge \beta(e_1, e_2) = e_1(f) \langle \dot{\sigma}_0, (\nabla_{e_2} \sigma)(\cdot, J \cdot) \rangle - e_2(f) \langle \dot{\sigma}_0, (\nabla_{e_1} \sigma)(\cdot, J \cdot) \rangle$$

= $f^{-1} (\langle \sigma, \nabla_{e_1} \sigma \rangle \langle \dot{\sigma}_0, (\nabla_{e_2} \sigma)(\cdot, J \cdot) \rangle - \langle \sigma, \nabla_{e_2} \sigma \rangle \langle \dot{\sigma}_0, (\nabla_{e_1} \sigma)(\cdot, J \cdot) \rangle)$.

As observed in the proof of Lemma 4.22, for every $p \in \Sigma$ outside the set of zeros of σ , the elements $(g^{-1}\sigma)_p$ and $(Jg^{-1}\sigma)_p$ form a basis of the space of traceless symmetric endomorphisms of $T_p\Sigma$. In particular, using the scalar product $\langle \cdot, \cdot \rangle$ we can represent $\dot{\sigma}_0$ in terms of such basis, obtaining the expression $\dot{\sigma}_0 = \frac{1}{\|\sigma\|^2} (\langle \dot{\sigma}_0, \sigma \rangle \sigma + \langle \dot{\sigma}_0, \sigma(\cdot, J \cdot) \rangle \sigma(\cdot, J \cdot))$. Replacing this identity in the previous equation, we get

$$\begin{split} \mathrm{d}f \wedge \beta(e_{1},e_{2}) &= f^{-1}\left(\langle \sigma, \nabla_{e_{1}}\sigma \rangle \langle \dot{\sigma}_{0}, (\nabla_{e_{2}}\sigma)(\cdot,J\cdot) \rangle - \langle \sigma, \nabla_{e_{2}}\sigma \rangle \langle \dot{\sigma}_{0}, (\nabla_{e_{1}}\sigma)(\cdot,J\cdot) \rangle \right) \\ &= \frac{1}{\|\sigma\|^{2}f}\left(\langle \sigma, \nabla_{e_{1}}\sigma \rangle (\langle \dot{\sigma}_{0},\sigma \rangle \langle \sigma, (\nabla_{e_{2}}\sigma)(\cdot,J\cdot) \rangle + \langle \dot{\sigma}_{0},\sigma(\cdot,J\cdot) \rangle \langle \sigma(\cdot,J\cdot), (\nabla_{e_{2}}\sigma)(\cdot,J\cdot) \rangle \right) + \\ &- \langle \sigma, \nabla_{e_{2}}\sigma \rangle (\langle \dot{\sigma}_{0},\sigma \rangle \langle \sigma, (\nabla_{e_{1}}\sigma)(\cdot,J\cdot) \rangle + \langle \dot{\sigma}_{0},\sigma(\cdot,J\cdot) \rangle \langle \sigma(\cdot,J\cdot), (\nabla_{e_{1}}\sigma)(\cdot,J\cdot) \rangle)) \\ &= \frac{1}{\|\sigma\|^{2}f}\left(\langle \sigma, \nabla_{e_{1}}\sigma \rangle (\langle \dot{\sigma}_{0},\sigma \rangle \langle \sigma, (\nabla_{e_{2}}\sigma)(\cdot,J\cdot) \rangle + \langle \dot{\sigma}_{0},\sigma(\cdot,J\cdot) \rangle \langle \sigma, \nabla_{e_{2}}\sigma \rangle \right) + \\ &- \langle \sigma, \nabla_{e_{2}}\sigma \rangle (\langle \dot{\sigma}_{0},\sigma \rangle \langle \sigma, (\nabla_{e_{1}}\sigma)(\cdot,J\cdot) \rangle + \langle \dot{\sigma}_{0},\sigma(\cdot,J\cdot) \rangle \langle \sigma, \nabla_{e_{1}}\sigma \rangle)) \\ &= \frac{\langle \dot{\sigma}_{0},\sigma \rangle}{\|\sigma\|^{2}f}\left(\langle \sigma, \nabla_{e_{1}}\sigma \rangle \langle \sigma, (\nabla_{e_{2}}\sigma)(\cdot,J\cdot) \rangle - \langle \sigma, \nabla_{e_{2}}\sigma \rangle \langle \sigma, (\nabla_{e_{1}}\sigma)(\cdot,J\cdot) \rangle \right) \\ &= \frac{\langle \dot{\sigma}_{0},\sigma \rangle}{\|\sigma\|^{2}f}\langle \langle \sigma, \nabla_{e_{1}}\sigma \rangle \sigma + \langle \sigma(\cdot,J\cdot), \nabla_{e_{1}}\sigma \rangle \sigma(\cdot,J\cdot), (\nabla_{e_{2}}\sigma)(\cdot,J\cdot) \rangle \end{split}$$

At this point, we use once again the fact that $g^{-1}\sigma$ and $Jg^{-1}\sigma$ form a basis of the space of traceless symmetric endomorphisms of $T\Sigma$ (outside the zeros of σ) to express $\nabla_{e_1}\sigma$ as $\frac{1}{\|\sigma\|^2}(\langle \nabla_{e_1}\sigma,\sigma\rangle\sigma+\langle \nabla_{e_1}\sigma,\sigma(\cdot,J\cdot)\rangle\sigma(\cdot,J\cdot))$. Combining this observation with the computations above, we deduce that

$$df \wedge \beta(e_1, e_2) = f^{-1} \langle \dot{\sigma}_0, \sigma \rangle \langle \nabla_{e_1} \sigma, (\nabla_{e_2} \sigma)(\cdot, J \cdot) \rangle$$
$$= \dot{f} \langle \nabla_{e_1} \sigma, (\nabla_{e_2} \sigma)(\cdot, J \cdot) \rangle$$
$$= f \dot{f} \hat{\omega}_{\mathbf{I}} ((0, \nabla_{e_1} \sigma), (0, \nabla_{e_2} \sigma)) ,$$

which finally proves the identity a) outside the zeros of σ . In fact, since both terms of the equality extends continuously at the zeros of σ (which form a finite set), we conclude that the identity $df \wedge \beta = f \dot{f} \hat{\omega}_{\mathbf{I}}((0, \nabla_{\bullet} \sigma), (0, \nabla_{\bullet} \sigma))$ holds everywhere on Σ .

Proof of relation b). By definition of exterior differential we have

$$d\beta(e_1, e_2) = e_1 \left(\left\langle \dot{\sigma}_0, (\nabla_{e_2} \sigma)(\cdot, J \cdot) \right\rangle \right) - e_2 \left(\left\langle \dot{\sigma}_0, (\nabla_{e_1} \sigma)(\cdot, J \cdot) \right\rangle \right) - \left\langle \dot{\sigma}_0, (\nabla_{[e_1, e_2]} \sigma)(\cdot, J \cdot) \right\rangle$$

$$= \left\langle \nabla_{e_1} \dot{\sigma}_0, (\nabla_{e_2} \sigma)(\cdot, J \cdot) \right\rangle - \left\langle \nabla_{e_2} \dot{\sigma}_0, (\nabla_{e_1} \sigma)(\cdot, J \cdot) \right\rangle + \left\langle \dot{\sigma}_0, R(e_1, e_2) \sigma(\cdot, J \cdot) \right\rangle$$

$$= f \left(\hat{\omega}_{\mathbf{I}} ((0, \nabla_{e_1} \dot{\sigma}_0), (0, \nabla_{e_2} \sigma)) - \hat{\omega}_{\mathbf{I}} ((0, \nabla_{e_2} \dot{\sigma}_0), (0, \nabla_{e_1} \sigma)) + \left\langle \dot{\sigma}_0, R(e_1, e_2) \sigma(\cdot, J \cdot) \right\rangle$$

where $R(e_1, e_2)\sigma = \nabla_{e_1}\nabla_{e_2}\sigma - \nabla_{e_2}\nabla_{e_1}\sigma - \nabla_{[e_1, e_2]}\sigma$. The same proof that relates the Riemann curvature tensor with the Gaussian curvature adapts to the operator $R(e_1, e_2)\sigma$ and shows that

$$R(e_1, e_2)\sigma = 2K_J\sigma(\cdot, J\cdot)$$
.

Relation b) follows.

Remark 6.11. In what follows, we will fix a primitive of the two form $d\tilde{\mu}_{\mathbf{I}}$ found in Proposition 6.10, and define the linear map $F_{(J,\sigma)}: T_{(J,\sigma)}T^*\mathcal{J}(\Sigma) \to \Lambda^1(\Sigma)/B^1(\Sigma) \subset \mathfrak{S}(\Sigma,\rho)^*$ so that $F_{(J,\sigma)}(\dot{J},\dot{\sigma})$ equals this primitive (modulo exact 1-forms). Note that $\mathrm{Ker}(F_{(J,\sigma)})$ is a subspace of $T_{(J,\sigma)}\tilde{m}u_{\mathbf{I}}^{-1}(0)$, and a priori the inclusion might be strict. By an abuse of notation, for the rest of the paper we will simply denote $(d\tilde{\mu}_{\mathbf{I}})_{(J,\sigma)} = F_{(J,\sigma)}$.

We can finally show the connection between $\omega_{\mathbf{I}}$ and $\mathrm{d}\tilde{\mu}_{\mathbf{I}}$:

Proposition 6.12. Let $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$. For every $(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$ and for every $V \in \mathfrak{S}(\Sigma, \rho)$, we have

$$\omega_{\mathbf{I}}((\mathcal{L}_{V}J,\mathcal{L}_{V}\sigma),(\dot{J},\dot{\sigma})) = -\langle d\tilde{\mu}_{\mathbf{I}}(\dot{J},\dot{\sigma}) | V \rangle_{\mathfrak{S}}.$$

Proof. By definition of $\omega_{\mathbf{I}}$, we have

$$\omega_{\mathbf{I}}((\mathcal{L}_{V}J,\mathcal{L}_{V}\sigma),(\dot{J},\dot{\sigma})) = \int_{\Sigma} \left(-f\langle \mathcal{L}_{V}J,J\dot{J}\rangle + f^{-1}\langle (\mathcal{L}_{V}\sigma)_{0},\dot{\sigma}_{0}(\cdot,J\cdot)\rangle \right)\rho. \tag{6.13}$$

In order to simplify the second term inside the integral, we make use of the following relation the proof of which will be postponed to the end:

$$\langle \dot{\sigma}_0(\cdot, J \cdot), (\mathcal{L}_V \sigma)_0 \rangle = \langle \dot{\sigma}_0(\cdot, J \cdot), \nabla_V \sigma \rangle - \operatorname{div}_{\sigma_J}(JV) \langle \sigma, \dot{\sigma}_0 \rangle \tag{6.14}$$

Using this fact and recalling that $\dot{f} = f^{-1} \langle \sigma, \dot{\sigma}_0 \rangle$ (by relation (6.3)), we have

$$f^{-1}\langle (\mathcal{L}_{V}\sigma)_{0}, \dot{\sigma}_{0}(\cdot, J \cdot) \rangle = f^{-1}\langle \dot{\sigma}_{0}(\cdot, J \cdot), \nabla_{V}\sigma \rangle - f^{-1}\operatorname{div}_{g_{J}}(JV)\langle \sigma, \dot{\sigma}_{0} \rangle$$

$$= f^{-1}\langle \dot{\sigma}_{0}(\cdot, J \cdot), \nabla_{V}\sigma \rangle - \operatorname{div}_{g_{J}}(f^{-1}\langle \sigma, \dot{\sigma}_{0} \rangle JV) + \operatorname{d}(f^{-1}\langle \sigma, \dot{\sigma}_{0} \rangle) JV$$

$$= f^{-1}\langle \dot{\sigma}_{0}(\cdot, J \cdot), \nabla_{V}\sigma \rangle - \operatorname{div}_{g_{J}}(f^{-1}\langle \sigma, \dot{\sigma}_{0} \rangle JV) + \operatorname{d}\dot{f}(JV)$$

and integrating over Σ we find

$$\int_{\Sigma} \left(f^{-1} \langle (\mathcal{L}_{V}\sigma)_{0}, \dot{\sigma}_{0}(\cdot, J \cdot) \rangle \right) \rho = \int_{\Sigma} \left(-f^{-1} \langle \dot{\sigma}_{0}, (\nabla_{V}\sigma)(\cdot, J \cdot) \rangle + d\dot{f}(JV) \right) \rho$$

$$= \int_{\Sigma} (d\dot{f}(JV) - f^{-1}\beta(V)) \rho , \tag{6.15}$$

Using Equation (4.7), we can rewrite the first term in (6.13) as

$$\begin{split} \int_{\Sigma} -f \langle \mathcal{L}_{V} J, J \dot{J} \rangle \rho &= \int_{\Sigma} \left(f(\operatorname{div}_{g_{J}} \dot{J}) V - f \operatorname{div}_{g_{J}} (\dot{J} V) \right) \rho \\ &= \int_{\Sigma} \left(f(\operatorname{div}_{g_{J}} \dot{J}) V + \operatorname{d} f(\dot{J} V) - \operatorname{div}_{g_{J}} (f \dot{J} V) \right) \rho \\ &= \int_{\Sigma} \left(f(\operatorname{div}_{g_{J}} \dot{J}) V + \operatorname{d} f(\dot{J} V) \right) \rho \;. \end{split}$$

Combining this last relation with (6.13) and (6.15), we find

$$\omega_{\mathbf{I}}((\mathcal{L}_{V}J, \mathcal{L}_{V}\sigma), (\dot{J}, \dot{\sigma})) = \int_{\Sigma} \left(f(\operatorname{div}_{g_{J}} \dot{J})V + \operatorname{d}f \circ \dot{J}V - f^{-1}\beta(V) + \operatorname{d}\dot{f}(JV) \right) \rho$$

$$= \int_{\Sigma} \iota_{V} \left(f \operatorname{div}_{g_{J}} \dot{J} + \operatorname{d}f \circ \dot{J} + \operatorname{d}\dot{f} \circ J - f^{-1}\beta \right) \rho$$

$$= \int_{\Sigma} (f \operatorname{div}_{g_{J}} \dot{J} + \operatorname{d}f \circ \dot{J} + \operatorname{d}\dot{f} \circ J - f^{-1}\beta) \wedge \iota_{V}\rho \qquad \text{(by (4.2))}$$

$$= -\langle \operatorname{d}\tilde{\mu}_{\mathbf{I}}(\dot{J}, \dot{\sigma}) \mid V \rangle_{\mathfrak{S}} .$$

We are left to prove Equation (6.14). Let A_V denote the endomorphism of $T\Sigma$ given by $A_V(X) = \nabla_X V$. It is easy to verify using the definition of Lie derivative that

$$\mathcal{L}_V \sigma = \nabla_V \sigma + A_V^t \sigma + \sigma A_V .$$

Because $V \in \mathfrak{S}(\Sigma, \rho)$, we know that $\operatorname{tr}(A_V) = 0$. In particular, $A_V \in \operatorname{Span}(J, \dot{J}, J\dot{J})$ and we can write

$$A_{V} = -\frac{1}{2}\operatorname{tr}(JA_{V})J + \frac{1}{2\operatorname{tr}(\dot{J}^{2})}\left(\operatorname{tr}(\dot{J}A_{V})\dot{J} + \operatorname{tr}(J\dot{J}A_{V})J\dot{J}\right)$$

$$A_{V}^{*} = \frac{1}{2}\operatorname{tr}(JA_{V})J + \frac{1}{2\operatorname{tr}(\dot{J}^{2})}\left(\operatorname{tr}(\dot{J}A_{V})\dot{J} + \operatorname{tr}(J\dot{J}A_{V})J\dot{J}\right)$$

$$\dot{J}A_{V}^{*} + A_{V}\dot{J} = -\operatorname{tr}(JA_{V})J\dot{J} + \frac{1}{2}\operatorname{tr}(\dot{J}A_{V})\mathbb{1}$$

where A_V^* denotes the adjoint of A_V with respect to g_J . Then for every $\dot{J} \in T_J \mathcal{J}(\Sigma)$, we have

$$\langle \dot{J}, (\mathcal{L}_{V}\sigma)_{0} \rangle = \frac{1}{2} \operatorname{tr} \left(\dot{J} g_{J}^{-1} \mathcal{L}_{V} \sigma \right)$$

$$= \frac{1}{2} \left(\operatorname{tr} \left(\dot{J} g_{J}^{-1} \nabla_{V} \sigma \right) + \operatorname{tr} \left(\dot{J} g_{J}^{-1} A_{V}^{t} \sigma \right) + \operatorname{tr} \left(\dot{J} g_{J}^{-1} \sigma A_{V} \right) \right)$$

$$= \frac{1}{2} \left(\operatorname{tr} \left(\dot{J} g_{J}^{-1} \nabla_{V} \sigma \right) + \operatorname{tr} \left(\dot{J} A_{V}^{*} g_{J}^{-1} \sigma \right) + \operatorname{tr} \left(A_{V} \dot{J} g_{J}^{-1} \sigma \right) \right)$$

$$= \langle \dot{J}, \nabla_{V} \sigma \rangle + \langle \sigma \mid \dot{J} A_{V}^{*} + A_{V} \dot{J} \rangle$$

$$= \langle \dot{J}, \nabla_{V} \sigma \rangle - \operatorname{tr} (J A_{V}) \langle \sigma \mid J \dot{J} \rangle$$

$$= \langle \dot{J}, \nabla_V \sigma \rangle - \operatorname{div}_{g_J}(JV) \langle \sigma \mid J \dot{J} \rangle$$

and Equation (6.14) follows by taking $\dot{J} = \dot{\sigma}_0(\cdot, J\cdot)$.

Remark 6.13. Note that, because $\tilde{\mu}_{\mathbf{I}}$ and $\mu_{\mathbf{I}}$ differ by a constant, Proposition 6.12 holds for $\mu_{\mathbf{I}}$ as well.

6.4.3. The differential of the map $\tilde{\mu}_{\mathbf{J}} + i \, \tilde{\mu}_{\mathbf{K}}$

With similar techniques, we can compute the differential of the other two moment maps:

Proposition 6.14. For every $(J, \sigma) \in T^*\mathcal{J}(\Sigma)$ and for every $(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$ we have:

$$d(\tilde{\mu}_{\mathbf{J}} + i\,\tilde{\mu}_{\mathbf{K}})\,(\dot{J}, \dot{\sigma}) = -i\,\left(\operatorname{div}_{g_{J}}(g^{-1}\dot{\bar{\phi}}_{0}) + \langle\nabla_{J\bullet}\bar{\phi}\mid\dot{J}\rangle\right) - \langle\bar{\phi}\mid\nabla_{\bullet}\dot{J} + i\,\nabla_{J\bullet}\dot{J}\rangle \in \mathfrak{S}(\Sigma, \rho)^{*}\otimes\mathbb{C},$$

where ϕ is the quadratic differential whose real part is equal to σ , $\dot{\bar{\phi}}_0$ denotes the g_J -traceless part of $\dot{\bar{\phi}}$.

Proof. To simplify the notation, we set $\operatorname{div}(T) = \operatorname{div}_{g_J}(g^{-1}T)$ for any symmetric 2-tensor T. In the proof of Theorem 6.5, we showed that $r(\partial_J \bar{\phi}) = \frac{1}{2}(\operatorname{div} \bar{\phi})$. In particular we have $(\tilde{\mu}_{\mathbf{J}} + i\,\tilde{\mu}_{\mathbf{K}})(J,\sigma) = [-i\,\operatorname{div}\bar{\phi}]$, where ϕ is the quadratic differential whose real part coincides with σ . Therefore, in order to compute the differential of the moment map $\tilde{\mu}_{\mathbf{J}} + i\,\tilde{\mu}_{\mathbf{K}}$, we need to understand the variation of the quantity $\operatorname{div}\bar{\phi}$ along $(\dot{J},\dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$.

Let $C_j^i S$ denote the contraction of the tensor S in its i-th covariant and j-th contravariant entries. Then the g_J -divergence of a symmetric 2-tensor T can be expressed as

$$\mathcal{C}_1^1 \mathcal{C}_1^1(g^* \otimes \nabla_{\bullet} T),$$

where g^* denotes the metric induced by $g = g_J$ on $T^*\Sigma$. As seen in relation (3.3), the first order variation of $g = \rho(\cdot, J\cdot)$ along \dot{J} can be expressed as $\dot{g} = -g(\cdot, J\dot{J}\cdot)$. It is simple to check that the corresponding variation of g^* satisfies $\dot{g}^* = g^*(\cdot, (J\dot{J})^t\cdot)$, where $(J\dot{J})^t$ is the transpose of $J\dot{J}$. In particular, for every symmetric 2-tensor T we have

$$(\operatorname{div} T)' = \mathcal{C}_{1}^{1} \mathcal{C}_{1}^{1} (\dot{g}^{*} \otimes \nabla_{\bullet} T + g^{*} \otimes \dot{\nabla}_{\bullet} T + g^{*} \otimes \nabla_{\bullet} \dot{T})$$

$$= \underbrace{\sum_{i} (\nabla_{e_{i}} T) (J \dot{J} e_{i}, \cdot)}_{\text{term 1}} + \underbrace{\sum_{i} (\dot{\nabla}_{e_{i}} T) (e_{i}, \cdot)}_{\text{term 2}} + \underbrace{\operatorname{div} \dot{T}}_{\text{term 3}},$$

where $(e_i)_i$ is a local g-orthonormal frame. In order to compute the differential of $\tilde{\mu}_{\mathbf{J}} + i \, \tilde{\mu}_{\mathbf{K}}$, we will study each term of this expression for $T = \bar{\phi}$.

Term 1. For every $\dot{J} \in T_J \mathcal{J}(\Sigma)$, for every J-quadratic differential ϕ and for every tangent vector field V on Σ , we have

$$(\nabla_X \bar{\phi})(J\dot{J}\cdot,\cdot) = \langle \nabla_X \bar{\phi} \mid J\dot{J}\rangle \ g + \langle \nabla_X \bar{\phi} \mid \dot{J}\rangle \ g(\cdot,J\cdot).$$

This relation is a simple application of Lemma 3.9, where the endomorphisms in $T_J \mathcal{J}(\Sigma)$ are the real (or imaginary) part of $g^{-1} \nabla_X \bar{\phi}$, and $J\dot{J}$ (we are implicitly extending the bilinear pairing $\langle \cdot | \cdot \rangle$ to the complexified bundles by requiring the \mathbb{C} -linearity in its arguments, i. e. $i\langle \cdot | \cdot \rangle = \langle i \cdot | \cdot \rangle = \langle \cdot | i \cdot \rangle$). From this expression we deduce that

$$\sum_{i} (\nabla_{e_{i}} \bar{\phi}) (J \dot{J} e_{i}, V) = \sum_{i} (\langle \nabla_{e_{i}} \bar{\phi} \mid J \dot{J} \rangle \ g(e_{i}, V) + \langle \nabla_{e_{i}} \bar{\phi} \mid \dot{J} \rangle \ g(e_{i}, J V))$$

$$= \langle \nabla_{V} \bar{\phi} \mid J \dot{J} \rangle + \langle \nabla_{J V} \bar{\phi} \mid \dot{J} \rangle$$

$$= -i \langle \nabla_{V} \bar{\phi} \mid \dot{J} \rangle + \langle \nabla_{J V} \bar{\phi} \mid \dot{J} \rangle,$$

where in the last step we used the fact that $\nabla_V \bar{\phi}$ is *J*-antilinear in its arguments. **Term 2.** Applying Lemma 4.15, we see that

$$\begin{split} (\dot{\nabla}_V \bar{\phi})(X,Y) &= -\bar{\phi}(\dot{\nabla}_V X,Y) - \bar{\phi}(X,\dot{\nabla}_V Y) \\ &= \frac{1}{2} \left((\operatorname{div} \dot{J}) V(\bar{\phi}(JX,Y) + \bar{\phi}(X,JY)) + \bar{\phi}(J(\nabla_V \dot{J})X,Y) + \right. \\ &\left. + \bar{\phi}(X,J(\nabla_V \dot{J})Y) \right) \\ &= -i \left(\operatorname{div} \dot{J} \right) V \; \bar{\phi}(X,Y) - \frac{i}{2} \left(\bar{\phi}((\nabla_V \dot{J})X,Y) + \bar{\phi}(X,(\nabla_V \dot{J})Y) \right). \end{split}$$

In the last line, we used the fact that $\bar{\phi}$ is *J*-antilinear. Applying Lemma 3.9 to the real (and imaginary, separately) part of $\bar{\phi}$ and to $\nabla_V \dot{J}$, we deduce the following relation:

$$\bar{\phi}((\nabla_V \dot{J})\cdot, \cdot) + \bar{\phi}(\cdot, (\nabla_V \dot{J})\cdot) = 2\langle \bar{\phi} \mid \nabla_V \dot{J} \rangle g,$$

This, combined with the previous computation leads us to the following expression:

$$(\dot{\nabla}_V \bar{\phi})(X, Y) = -i (\operatorname{div} \dot{J}) V \ \bar{\phi}(X, Y) - i \langle \bar{\phi} \mid \nabla_V \dot{J} \rangle \ g(X, Y). \tag{6.16}$$

Moreover, the following equality holds:

$$\sum_{i} (\operatorname{div} \dot{J}) e_{i} \ \bar{\phi}(e_{i}, V) = \langle \bar{\phi} \mid \nabla_{V} \dot{J} + i \nabla_{JV} \dot{J} \rangle. \tag{6.17}$$

We will temporarily assume this fact, the proof is postponed to the end of the current argument. We can now express the second term of our initial expression as follows:

$$\sum_{i} (\dot{\nabla}_{e_{i}} \bar{\phi})(e_{i}, V) = -i \sum_{i} (\operatorname{div} \dot{J}) e_{i} \ \bar{\phi}(e_{i}, V) - i \sum_{i} \langle \bar{\phi} \mid \nabla_{e_{i}} \dot{J} \rangle \ g(e_{i}, V) \qquad \text{(relation (6.16))}$$

$$= -i \langle \bar{\phi} \mid \nabla_{V} \dot{J} + i \nabla_{JV} \dot{J} \rangle - i \sum_{i} \langle \bar{\phi} \mid \nabla_{e_{i}} \dot{J} \rangle \ g(e_{i}, V) \qquad \text{(relation (6.17))}$$

$$= -i \langle \bar{\phi} \mid \nabla_{V} \dot{J} + i \nabla_{JV} \dot{J} \rangle - i \langle \bar{\phi} \mid \nabla_{V} \dot{J} \rangle.$$

Term 3. Following the same argument of the proof of Lemma 3.2, we see that the first order variation of a quadratic differential ϕ is of the form $\dot{\phi} = \dot{\phi}_0 - \langle \phi \mid J\dot{J} \rangle g$, where $\dot{\phi}_0$ denotes the g-traceless part of $\dot{\phi}$. In particular, we deduce that

$$(\operatorname{div} \dot{\bar{\phi}})V = \operatorname{div}(\dot{\bar{\phi}}_{0} - \langle \bar{\phi} \mid J\dot{J} \rangle g)V$$

$$= (\operatorname{div} \dot{\bar{\phi}}_{0})V - \operatorname{d}(\langle \bar{\phi} \mid J\dot{J} \rangle)V$$

$$= (\operatorname{div} \dot{\bar{\phi}}_{0})V + i \operatorname{d}(\langle \bar{\phi} \mid \dot{J} \rangle)V \qquad (\bar{\phi} \text{ J-antilinear})$$

$$= (\operatorname{div} \dot{\bar{\phi}}_{0})V + i \langle \nabla_{V}\bar{\phi} \mid \dot{J} \rangle + i \langle \bar{\phi} \mid \nabla_{V}\dot{J} \rangle.$$

Finally, we combine the expressions of the three terms involved in the derivative $(\operatorname{div} \bar{\phi})'$ that we developed above, obtaining:

$$\begin{split} \mathrm{d}(\tilde{\mu}_{\mathbf{J}} + i\,\tilde{\mu}_{\mathbf{K}})\,(\dot{J},\dot{\sigma}) &= -i\,(\mathrm{div}\,\bar{\phi})' \\ &= -i\,\left(\sum_{i}(\nabla_{e_{i}}\bar{\phi})(J\dot{J}e_{i},\cdot) + \sum_{i}(\dot{\nabla}_{e_{i}}\bar{\phi})(e_{i},\cdot) + \mathrm{div}\,\dot{\bar{\phi}}\right) \\ &= -i\,\left(-i\,\langle\nabla_{\bullet}\bar{\phi}\mid\dot{J}\rangle + \langle\nabla_{J\bullet}\bar{\phi}\mid\dot{J}\rangle - i\,\langle\bar{\phi}\mid\nabla_{\bullet}\dot{J} + i\,\nabla_{J\bullet}\dot{J}\rangle + \\ &\qquad \qquad -i\,\langle\bar{\phi}\mid\nabla_{\bullet}\dot{J}\rangle + \mathrm{div}\,\dot{\bar{\phi}}_{0} + i\,\langle\nabla_{\bullet}\bar{\phi}\mid\dot{J}\rangle + i\,\langle\bar{\phi}\mid\nabla_{\bullet}\dot{J}\rangle\right) \\ &= -i\,\left(\mathrm{div}\,\dot{\bar{\phi}}_{0} + \langle\nabla_{J\bullet}\bar{\phi}\mid\dot{J}\rangle - i\,\langle\bar{\phi}\mid\nabla_{\bullet}\dot{J} + i\,\nabla_{J\bullet}\dot{J}\rangle\right), \end{split}$$

which proves our statement.

Finally, we provide a proof of Equation (6.17). Since \dot{J} is symmetric with respect to g, the same is true for $\nabla_X \dot{J}$ for any tangent vector field X. In particular, if $(\nabla_i \dot{J})_{jk}$ denotes $g((\nabla_{e_i} \dot{J})e_j, e_k)$, we must have $(\nabla_i \dot{J})_{jk} = (\nabla_i \dot{J})_{kj}$ for every i, j, k. Unraveling the definition of the divergence of \dot{J} , we see that

$$(\operatorname{div} \dot{J})(e_1) e_1 + (\operatorname{div} \dot{J})(e_2) e_2 = ((\nabla_1 \dot{J})_{11} + (\nabla_2 \dot{J})_{12}) e_1 + ((\nabla_1 \dot{J})_{21} + (\nabla_2 \dot{J})_{22}) e_2$$

$$= ((\nabla_1 \dot{J})_{11} + (\nabla_2 \dot{J})_{21}) e_1 + ((\nabla_1 \dot{J})_{12} + (\nabla_2 \dot{J})_{22}) e_2$$

$$= (\nabla_{e_1} \dot{J}) e_1 + (\nabla_{e_2} \dot{J}) e_2.$$

Hence we have

$$\sum_{i} (\operatorname{div} \dot{J}) e_{i} \ \bar{\phi}(e_{i}, V) = \bar{\phi}((\nabla_{e_{1}} \dot{J}) e_{1} + (\nabla_{e_{2}} \dot{J}) e_{2}, V)
= V^{1} \bar{\phi}((\nabla_{e_{1}} \dot{J}) e_{1} + (\nabla_{e_{2}} \dot{J}) e_{2}, e_{1}) + V^{2} \bar{\phi}((\nabla_{e_{1}} \dot{J}) e_{1} + (\nabla_{e_{2}} \dot{J}) e_{2}, e_{2}),$$

where $V^i = g(V, e_i)$. We will now make multiple use of the following elementary properties:

•
$$e_2 = Je_1, V^1 = g(JV, e_2) = (JV)^2, V^2 = -g(JV, e_1) = -(JV)^1;$$

- $(\nabla_{e_i}\dot{J})$ is g-symmetric and traceless. In particular, it anticommutes with the complex structure J;
- the tensor $\bar{\phi}$ is anti-bilinear in its entries, therefore $\bar{\phi}(J\cdot,\cdot)=\bar{\phi}(\cdot,J\cdot)=-i\bar{\phi}(\cdot,\cdot)$.

From the previous expression we deduce

$$\sum_{i} (\operatorname{div} \dot{J}) e_{i} \ \bar{\phi}(e_{i}, V) = V^{1} \bar{\phi}((\nabla_{e_{1}} \dot{J}) e_{1}, e_{1}) + (JV)^{2} \bar{\phi}((\nabla_{e_{2}} \dot{J}) J e_{1}, e_{1}) + \\
- (JV)^{1} \bar{\phi}((\nabla_{e_{1}} \dot{J}) e_{1}, J e_{1}) + V^{2} \bar{\phi}((\nabla_{e_{2}} \dot{J}) J e_{1}, J e_{1}) \\
= V^{1} \bar{\phi}((\nabla_{e_{1}} \dot{J}) e_{1}, e_{1}) + i V^{2} \bar{\phi}(J(\nabla_{e_{2}} \dot{J}) e_{1}, e_{1}) + \\
- (JV)^{2} \bar{\phi}(J(\nabla_{e_{2}} \dot{J}) e_{1}, e_{1}) + i (JV)^{1} \bar{\phi}((\nabla_{e_{1}} \dot{J}) e_{1}, e_{1}) \\
= V^{1} \bar{\phi}((\nabla_{e_{1}} \dot{J}) e_{1}, e_{1}) + V^{2} \bar{\phi}((\nabla_{e_{2}} \dot{J}) e_{1}, e_{1}) + \\
+ i (JV)^{2} \bar{\phi}((\nabla_{e_{2}} \dot{J}) e_{1}, e_{1}) + i (JV)^{1} \bar{\phi}((\nabla_{e_{1}} \dot{J}) e_{1}, e_{1}) \\
= \bar{\phi}((\nabla_{V} \dot{J}) e_{1}, e_{1}) + i \bar{\phi}((\nabla_{JV} \dot{J}) e_{1}, e_{1})$$
(6.18)

In order to derive the desired relation, we are left to show that

$$\bar{\phi}((\nabla_V \dot{J})e_1, e_1) + i\,\bar{\phi}((\nabla_{JV} \dot{J})e_1, e_1) = \langle \bar{\phi} \mid \nabla_V \dot{J} + i\nabla_{JV} \dot{J} \rangle$$

This equality can be deduced from the properties of $\bar{\phi}$ and $\nabla_X \dot{J}$ previously mentioned. Indeed we have

$$\langle \bar{\phi} \mid \nabla_V \dot{J} \rangle = \frac{1}{2} \operatorname{tr} \left(g^{-1} \bar{\phi} \nabla_V \dot{J} \right)$$

$$= \frac{1}{2} \left(\bar{\phi} ((\nabla_V \dot{J}) e_1, e_1) + \bar{\phi} ((\nabla_V \dot{J}) J e_1, J e_1) \right)$$

$$= \frac{1}{2} \left(\bar{\phi} ((\nabla_V \dot{J}) e_1, e_1) - \bar{\phi} (J(\nabla_V \dot{J}) e_1, J e_1) \right)$$

$$= \bar{\phi} ((\nabla_V \dot{J}) e_1, e_1).$$

By replacing the role of V with JV, we obtain also that $\langle \bar{\phi} \mid \nabla_{JV} \dot{J} \rangle = \bar{\phi}((\nabla_{JV} \dot{J})e_1, e_1)$. This cocludes the proof of relation (6.17).

6.4.4. Model for the tangent space to $\mathcal{MS}(\Sigma)$

We finally come to the main statement of this section.

Theorem M. For every $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$, $V_{(J,\sigma)}$ is the largest subspace of $T_{(J,\sigma)}\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$ that is:

- invariant under I, J and K;
- g-orthogonal to $T_{(J,\sigma)}(\operatorname{Symp}_0(\Sigma,\rho)\cdot (J,\sigma))$

The proof of Theorem M is completed in the rest of this section, by means of three lemmas that simplify the statement in several steps.

Lemma 6.15. Let $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$ and $(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$. The following conditions are equivalent:

a)
$$(\dot{J}, \dot{\sigma}), \mathbf{I}(\dot{J}, \dot{\sigma}), \mathbf{J}(\dot{J}, \dot{\sigma})$$
 and $\mathbf{K}(\dot{J}, \dot{\sigma})$ belong to $\operatorname{Ker}(d\tilde{\mu}_{\mathbf{I}}) \cap \operatorname{Ker}(d\tilde{\mu}_{\mathbf{J}}) \cap \operatorname{Ker}(d\tilde{\mu}_{\mathbf{K}});$
b) $(\dot{J}, \dot{\sigma})$ and $\mathbf{J}(\dot{J}, \dot{\sigma})$ belong to $\operatorname{Ker}(d\tilde{\mu}_{\mathbf{J}}) \cap \operatorname{Ker}(d\tilde{\mu}_{\mathbf{K}}).$

Proof. We only need to prove that b) implies a). By Proposition 6.12 (plus Remark 6.13) and the quaternionic relations between \mathbf{I} , \mathbf{J} and \mathbf{K} , we find

$$\langle d\tilde{\mu}_{\mathbf{I}}(\mathbf{J}(\dot{J},\dot{\sigma})) \mid V \rangle_{\mathfrak{S}} = -\omega_{\mathbf{I}}(\mathbf{J}(\dot{J},\dot{\sigma}), (\mathcal{L}_{V}J,\mathcal{L}_{V}\sigma)) = \mathbf{g}(\mathbf{IJ}(\dot{J},\dot{\sigma}), (\mathcal{L}_{V}J,\mathcal{L}_{V}\sigma))$$

$$= \mathbf{g}(\mathbf{K}(\dot{J},\dot{\sigma}), (\mathcal{L}_{V}J,\mathcal{L}_{V}\sigma)) = -\omega_{\mathbf{K}}((\dot{J},\dot{\sigma}), (\mathcal{L}_{V}J,\mathcal{L}_{V}\sigma))$$

$$= \langle d\tilde{\mu}_{\mathbf{K}}(\dot{J},\dot{\sigma}) \mid V \rangle_{\mathfrak{S}}.$$

Therefore, $(\dot{J}, \dot{\sigma}) \in \operatorname{Ker}(\operatorname{d}\tilde{\mu}_{\mathbf{K}})$ if and only if $\mathbf{J}(\dot{J}, \dot{\sigma}) \in \operatorname{Ker}(\operatorname{d}\tilde{\mu}_{\mathbf{I}})$. With a similar computation, one can also show that $(\dot{J}, \dot{\sigma}) \in \operatorname{Ker}(\operatorname{d}\tilde{\mu}_{\mathbf{K}})$ if and only if $\mathbf{I}(\dot{J}, \dot{\sigma}) \in \operatorname{Ker}(\operatorname{d}\tilde{\mu}_{\mathbf{J}})$. It follows that if we start from a pair $(\dot{J}, \dot{\sigma})$ satisfying b), then

- $(\dot{J}, \dot{\sigma}) \in \operatorname{Ker}(d\tilde{\mu}_{\mathbf{K}}) \Rightarrow \mathbf{I}(\dot{J}, \dot{\sigma}) \in \operatorname{Ker}(d\tilde{\mu}_{\mathbf{J}}) \text{ and } \mathbf{J}(\dot{J}, \dot{\sigma}) \in \operatorname{Ker}(d\tilde{\mu}_{\mathbf{I}});$
- $(\dot{J}, \dot{\sigma}) \in \operatorname{Ker}(d\tilde{\mu}_{\mathbf{J}}) \Rightarrow \mathbf{I}(\dot{J}, \dot{\sigma}) \in \operatorname{Ker}(d\tilde{\mu}_{\mathbf{K}}) \text{ and } \mathbf{K}(\dot{J}, \dot{\sigma}) \in \operatorname{Ker}(d\tilde{\mu}_{\mathbf{I}});$
- $\mathbf{J}(\dot{J}, \dot{\sigma}) \in \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{K}}) \Rightarrow \mathbf{K}(\dot{J}, \dot{\sigma}) \in \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{J}}) \text{ and } (\dot{J}, \dot{\sigma}) \in \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{I}});$
- $\mathbf{J}(\dot{J}, \dot{\sigma}) \in \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{J}}) \Rightarrow \mathbf{K}(\dot{J}, \dot{\sigma}) \in \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{K}}) \text{ and } \mathbf{I}(\dot{J}, \dot{\sigma}) \in \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{I}})$

which implies a).

Lemma 6.16. Let $(J, \sigma) \in \widetilde{\mathcal{MS}}_0(\Sigma, \rho)$ and $(J, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$. The following conditions are equivalent:

- 1) $(\dot{J}, \dot{\sigma})$ is **g**-orthogonal to $T_{(J,\sigma)}(\operatorname{Symp}_0(\Sigma, \rho) \cdot (J, \sigma))$;
- 2) for some $\mathbf{L} \in \{\mathbf{I}, \mathbf{J}, \mathbf{K}\}$, we have $\mathbf{L}(\dot{J}, \dot{\sigma}) \in \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{L}})$;
- 3) for all $\mathbf{L} \in \{\mathbf{I}, \mathbf{J}, \mathbf{K}\}$, we have $\mathbf{L}(\dot{J}, \dot{\sigma}) \in \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{L}})$.

Proof. We first show that 1) is equivalent to 2) for $L = \mathbf{J}$, the other cases being analogous. The properties of the moment maps imply

$$\mathbf{g}((\mathcal{L}_{V}J, \mathcal{L}_{V}\sigma), (\dot{J}, \dot{\sigma})) = \mathbf{g}((\mathcal{L}_{V}J, \mathcal{L}_{V}\sigma), \mathbf{J}^{2}(\dot{J}, \dot{\sigma}))$$

$$= \omega_{\mathbf{J}}((\mathcal{L}_{V}J, \mathcal{L}_{V}\sigma), \mathbf{J}(\dot{J}, \dot{\sigma}))$$

$$= -\langle \mathrm{d}\tilde{\mu}_{\mathbf{J}} (\mathbf{J}(\dot{J}, \dot{\sigma})) \mid V \rangle_{\mathfrak{S}} .$$

$$(\mathbf{J}^{2} = \mathbb{1})$$

$$(\omega_{\mathbf{J}} = \mathbf{g}(\cdot, \mathbf{J}\cdot))$$

so $(\dot{J}, \dot{\sigma})$ is **g**-orthogonal to $(\mathcal{L}_V J, \mathcal{L}_V \sigma)$ for every $V \in \mathfrak{S}$ if and only if $\mathbf{J}(\dot{J}, \dot{\sigma}) \in \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{J}})$. Note that the above relation proves that 1) and 3) are equivalent as well.

Combining Lemma 6.15 and Lemma 6.16, we see that the subspace $V_{(J,\sigma)} \subset T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$ we are interested in can be described as

$$V_{(J,\sigma)} = \{ (\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)} T^* \mathcal{J}(\Sigma) \mid (\dot{J}, \dot{\sigma}), \mathbf{J}(\dot{J}, \dot{\sigma}) \in \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{J}}) \cap \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{K}}) \}$$

Finally, the following result allows to describe $V_{(J,\sigma)}$ as the solution of the system of partial differential equations appearing in Proposition K part ii).

Lemma 6.17. Let $(J, \sigma) \in T^*\mathcal{J}(\Sigma)$ be such that σ is the real part of a holomorphic quadratic differential ϕ on (Σ, J) , and let $(\dot{J}, \dot{\sigma}) \in T_{(J,\sigma)}T^*\mathcal{J}(\Sigma)$. Then $(\dot{J}, \dot{\sigma})$ and $\mathbf{J}(\dot{J}, \dot{\sigma})$ belong to $\ker d\tilde{\mu}_{\mathbf{J}} \cap \ker d\tilde{\mu}_{\mathbf{K}}$ if and only if

$$\begin{cases}
\operatorname{div}_{g}(g^{-1}\dot{\sigma}_{0} + J\dot{J}g^{-1}\sigma) = \langle \nabla^{g}_{J\bullet}\sigma \mid \dot{J}\rangle, \\
\operatorname{div}_{g}(f\dot{J} + f^{-1}Jg^{-1}\dot{\sigma}_{0}g^{-1}\sigma) = \langle \nabla^{g}_{J\bullet}\sigma \mid f^{-1}g^{-1}\dot{\sigma}_{0}\rangle.
\end{cases}$$
(6.19)

Proof. Applying Proposition 6.14 and dividing $d(\tilde{\mu}_{\mathbf{J}} + i \, \tilde{\mu}_{\mathbf{K}})$ into real and imaginary part, respectively, we see that the following relations hold:

$$\begin{split} \mathrm{d}\tilde{\mu}_{\mathbf{J}}\left(\dot{J},\dot{\sigma}\right) &= \left[\mathrm{div}_{g}(g^{-1}\dot{\sigma}_{0}(\cdot,J\cdot)) + \left\langle \nabla_{J\bullet}\sigma \mid J\dot{J}\right\rangle - \left\langle \sigma \mid \nabla_{\bullet}\dot{J} - J\nabla_{J\bullet}\dot{J}\right\rangle \right] \\ &= \left[J^{*}(\mathrm{div}_{g}\,g^{-1}\dot{\sigma}_{0} + \left\langle \nabla_{\bullet}\sigma \mid J\dot{J}\right\rangle + \left\langle \sigma \mid J(\nabla_{\bullet}\dot{J} - J\nabla_{J\bullet}\dot{J})\right\rangle)\right] \in \Lambda^{1}(\Sigma)/B^{1}(\Sigma), \end{split}$$

$$d\tilde{\mu}_{\mathbf{K}}(\dot{J},\dot{\sigma}) = -\left[\operatorname{div}_{q} g^{-1}\dot{\sigma}_{0} + \langle \nabla_{J\bullet}\sigma \mid \dot{J}\rangle + \langle \sigma \mid J(\nabla_{\bullet}\dot{J} - J\nabla_{J\bullet}\dot{J})\rangle\right] \in \Lambda^{1}(\Sigma)/B^{1}(\Sigma).$$

Since ϕ is a holomorphic quadratic differential, it satisfies $\partial_J \bar{\phi} \equiv 0$, which is equivalent to require $\nabla_{J \bullet} \sigma = (\nabla_{\bullet} \sigma)(\cdot, J \cdot)$. In light of this relation, if we set α to be the 1-form

$$\alpha := \operatorname{div}_g g^{-1} \dot{\sigma}_0 + \langle \nabla_{\bullet} \sigma \mid J \dot{J} \rangle + \langle \sigma \mid J (\nabla_{\bullet} \dot{J} - J \nabla_{J \bullet} \dot{J}) \rangle,$$

then we can express $d\tilde{\mu}_{\mathbf{J}}(\dot{J},\dot{\sigma})$ and $d\tilde{\mu}_{\mathbf{K}}(\dot{J},\dot{\sigma})$ as follows:

$$d\tilde{\mu}_{\mathbf{J}}(\dot{J},\dot{\sigma}) = [\alpha \circ J], \ d\tilde{\mu}_{\mathbf{K}}(\dot{J},\dot{\sigma}) = [-\alpha] \in \Lambda^{1}(\Sigma)/B^{1}(\Sigma) \subseteq \mathfrak{S}(\Sigma,\rho)^{*}.$$

Assume now that $(\dot{J}, \dot{\sigma})$ satisfies $d\tilde{\mu}_{\mathbf{J}}(\dot{J}, \dot{\sigma}) = d\tilde{\mu}_{\mathbf{K}}(\dot{J}, \dot{\sigma}) = [0] \in \Lambda^1(\Sigma)/B^1(\Sigma)$ or, equivalently, that the forms α and $\alpha \circ J$ are exact. In particular there exists a smooth function over Σ such that $\alpha = df$. Since $\alpha \circ J$ is exact, we also have

$$-(\Delta_{g_J} f) \rho = d(df \circ J) = d(\alpha \circ J) = 0.$$

In other words, the function f has to be harmonic with respect to g_J and therefore constant, since Σ is compact without boundary. This proves in particular that the 1-form α vanishes identically if and only if $(\dot{J}, \dot{\sigma})$ belongs to $\ker d\tilde{\mu}_{\mathbf{J}} \cap \ker d\tilde{\mu}_{\mathbf{K}}$. The form α can be expressed as follows:

$$\alpha = \operatorname{div}_{g}(g^{-1}\dot{\sigma}_{0}) + \langle \nabla_{\bullet}\sigma \mid J\dot{J}\rangle + \langle \sigma \mid J(\nabla_{\bullet}\dot{J} - J\nabla_{J\bullet}\dot{J})\rangle$$

$$= \operatorname{div}_{g}(g^{-1}\dot{\sigma}_{0} + \langle \sigma \mid J\dot{J}\rangle\mathbb{1} + \langle \sigma \mid \dot{J}\rangle J) - \langle \nabla_{J\bullet}\sigma \mid \dot{J}\rangle$$

$$= \operatorname{div}_{g}(g^{-1}\dot{\sigma}_{0} + J\dot{J}g^{-1}\sigma) - \langle \nabla_{J\bullet}\sigma \mid \dot{J}\rangle.$$
 (Lemma 3.9)

If we apply the same argument to $\mathbf{J}(\dot{J},\dot{\sigma})$ (see Section 3.3 for the definition of \mathbf{J}), we obtain that

$$(\dot{J}, \dot{\sigma}) \in \ker d\tilde{\mu}_{\mathbf{J}} \cap \ker d\tilde{\mu}_{\mathbf{K}} \Leftrightarrow \operatorname{div}_{g}(g^{-1}\dot{\sigma}_{0} + J\dot{J}g^{-1}\sigma) = \langle \nabla_{J\bullet}\sigma \mid \dot{J}\rangle,$$
 (6.20)

$$\mathbf{J}(\dot{J},\dot{\sigma}) \in \ker \mathrm{d}\tilde{\mu}_{\mathbf{J}} \cap \ker \mathrm{d}\tilde{\mu}_{\mathbf{K}} \Leftrightarrow \mathrm{div}_{g}(f\dot{J} + f^{-1}Jg^{-1}\dot{\sigma}_{0}g^{-1}\sigma) = \langle \nabla_{J\bullet}\sigma \mid f^{-1}g^{-1}\dot{\sigma}_{0} \rangle, \ (6.21)$$

where
$$f = f(\|\sigma\|) = \sqrt{1 + \|\sigma\|^2}$$
, thus obtaining the desired statement.

Proof of Theorem M. By Corollary 6.9, the zero locus of the moment maps $\tilde{\mu}_{\mathbf{I}}$, $\tilde{\mu}_{\mathbf{J}}$ and $\tilde{\mu}_{\mathbf{K}}$ coincides precisely with $\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$. Observe, however, that $T_{(J,\sigma)}\widetilde{\mathcal{MS}}_0(\Sigma, \rho)$ is larger than $\mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{I}}) \cap \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{J}}) \cap \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{K}})$ by Remark 6.11. Nonetheless, the largest subspace W of $T_{(J,\sigma)}\widetilde{\mathcal{MS}}_0(\Sigma,\rho)$ that is **g**-orthogonal to $T_{(J,\sigma)}(\mathrm{Symp}_0(\Sigma,\rho)\cdot(J,\sigma))$ and invariant under \mathbf{I} , \mathbf{J} and \mathbf{K} is contained in $\mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{I}}) \cap \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{J}}) \cap \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{K}})$. Indeed, if $(\dot{J},\dot{\sigma})$ is in W, then the same is true for $\mathbf{I}(\dot{J},\dot{\sigma})$. Since $\mathbf{I}(\dot{J},\dot{\sigma})$ is **g**-orthogonal to the tangent of the orbit by $\mathrm{Symp}_0(\Sigma,\rho)$, Lemma 6.16 implies that $\mathbf{I}^2(\dot{J},\dot{\sigma}) = -(\dot{J},\dot{\sigma})$ lies in $\mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{I}})$. Being $(\dot{J},\dot{\sigma})$ arbitrary, we deduce that W is contained in $\mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{I}}) \cap \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{J}}) \cap \mathrm{Ker}(\mathrm{d}\tilde{\mu}_{\mathbf{K}})$.

Now, by Lemma 6.15, Lemma 6.16, and Lemma 6.17 the subspace W is described by the Equation (6.19). Taking the sum and the difference of the two equations, and using that $J_l = f J + g_J^{-1} \sigma$, $J_r = f J - g_J^{-1} \sigma$, it is straightforward to verify that (6.19) is equivalent to (V2) of Proposition K.

APPENDIX A. PARA-COMPLEX GEOMETRY

In this appendix we introduce the algebra \mathbb{B} of para-complex numbers, para-complex coordinates, and \mathbb{B} -valued symplectic forms. These will allow us to give an equivalent characterization of para-Kähler potential (Lemma A.4) and prove a criterion to show that a manifold is para-hyperKähler (Lemma A.5).

Para-complex numbers. Let \mathbb{B} be the \mathbb{R} -algebra generated by τ with $\tau^2 = 1$. Elements of \mathbb{B} are called para-complex numbers. Notice that \mathbb{B} is a two-dimensional real vector space and we will often denote $\mathbb{B} = \mathbb{R} \oplus \tau \mathbb{R}$. Borrowing terminology from the complex numbers, we talk about real and imaginary part of an element of \mathbb{B} and we define a conjugation

$$\overline{a+\tau b}=a-\tau b$$
.

This induces a norm on \mathbb{B} by taking

$$|a + \tau b|^2 = (a + \tau b)\overline{(a + \tau b)} = a^2 - b^2$$
.

Note that elements of \mathbb{B} may have negative norm. In fact, the bi-linear extension of this norm defines a Minkowski inner product on \mathbb{R}^2 with orthonormal basis $\{1, \tau\}$. It is also convenient to work with the basis of idempotents

$$e^{+} = \frac{1+\tau}{2}$$
 and $e^{-} = \frac{1-\tau}{2}$,

because the map

$$ae^+ + be^- \mapsto (a, b)$$

gives an isomorphism of \mathbb{R} -algebras between \mathbb{B} and $\mathbb{R} \oplus \mathbb{R}$, where in the latter operations are carried out component by component.

Para-complex structures. Let $\mathbf{P}: V \to V$ be a para-complex structure on a real vector space V. We denote by V^+ and V^- the eigenspaces of \mathbf{P} relative to the eigenvalues +1 and -1 respectively.

Remark A.1. The multiplication by τ on \mathbb{B} is a para-complex structure on \mathbb{B} with eigenspaces $V^+ = \mathbb{R}e^+$ and $V^- = \mathbb{R}e^-$.

Given a para-complex structure \mathbf{P} on V, we can define the para-complexification of V as the \mathbb{B} -module $V^{\mathbb{B}} = V \otimes_{\mathbb{R}} \mathbb{B}$. We can then extend the para-complex structure \mathbf{P} to $V^{\mathbb{B}}$ by \mathbb{B} -linearity and define

$$V^{1,0} = \{ v \in V^{\mathbb{B}} \mid \mathbf{P}v = \tau v \} = \{ v + \tau \mathbf{P}v \mid v \in V \}$$
$$V^{0,1} = \{ v \in V^{\mathbb{B}} \mid \mathbf{P}v = -\tau v \} = \{ v - \tau \mathbf{P}v \mid v \in V \}$$

so that $V^{\mathbb{B}} = V^{1,0} \oplus V^{0,1}$.

A para-complex structure \mathbf{P} induces a para-complex structure \mathbf{P}^* on the dual space V^* by requiring $(\mathbf{P}^*\alpha)(v) = \alpha(\mathbf{P}v)$ for any $\alpha \in V^*$ and any $v \in V$. As before, we can decompose $V^{*\mathbb{B}} = V_{1,0} \oplus V_{0,1}$ where

$$V_{1,0} = \{ \alpha \in V^{*\mathbb{B}} \mid \mathbf{P}^* \alpha = \tau \alpha \} = \{ \alpha + \tau \mathbf{P}^* \alpha \mid \alpha \in V^* \}$$
$$V_{0,1} = \{ \alpha \in V^{*\mathbb{B}} \mid \mathbf{P}^* \alpha = -\tau \alpha \} = \{ \alpha - \tau \mathbf{P}^* \alpha \mid \alpha \in V^* \} .$$

More in general, we have a decomposition of \mathbb{B} -valued n-forms into types

$$\bigwedge^{n} V^{*\mathbb{B}} = \bigoplus_{p+q=n} \bigwedge^{p,q} V^{*\mathbb{B}} ,$$

where $\bigwedge^{p,q} V^{*\mathbb{B}}$ denotes the vector space spanned by $\alpha \wedge \beta$, with $\alpha \in \bigwedge^p V_{1,0}$ and $\beta \in \bigwedge^q V_{0,1}$.

Para-complex coordinates. Recall that an almost para-complex structure **P** on M is a bundle endomorphism $\mathbf{P}: TM \to TM$ such that $\mathbf{P}^2 = \mathbb{1}$ and the eigenspaces $T^{\pm}M$ of **P** relative to the eigenvalues ± 1 have the same dimension.

Definition A.2. Let (M, \mathbf{P}) and (N, \mathbf{P}') be two almost para-complex manifolds. A smooth map $f: M \to N$ is para-holomorphic if $\mathbf{P}' \circ df = df \circ \mathbf{P}$. In particular, a function $f: (M, \mathbf{P}) \to \mathbb{B}$ is para-holomorphic if for every vector field $V \in \Gamma(TM)$ we have $df(\mathbf{P}V) = \tau df(V)$.

An almost para-complex structure \mathbf{P} on M is said to be integrable if the eigen-distributions $T^{\pm}M$ of \mathbf{P} are involutive. This is equivalent to the existence of local charts $\phi_{\alpha}:U_{\alpha}\subset$

 $M \to \mathbb{B}^n$ such that change of coordinates are para-holomorphic functions. If z^i is a \mathbb{B} -valued coordinate, its real and imaginary part

$$\operatorname{Re}(z^{i}) = x^{i} = \frac{1}{2}(z^{i} + \bar{z}^{i}) \qquad \operatorname{Im}(z^{i}) = y^{i} = \frac{1}{2\tau}(z^{i} - \bar{z}^{i})$$

define usual \mathbb{R} -valued coordinates on M.

The para-complex tangent bundle of a para-complex manifold (M, \mathbf{P}) is $T^{\mathbb{B}}M = TM \otimes_{\mathbb{R}}$ \mathbb{B} . Extending the endomorphism \mathbf{P} on $T^{\mathbb{B}}M$ by \mathbb{B} -linearity, we have a decomposition $T^{\mathbb{B}}M = T^{1,0}M \oplus T^{0,1}M$ where

$$T^{1,0}M = \{X \in T^{\mathbb{B}}M \mid \mathbf{P}X = \tau X\} = \{X + \tau \mathbf{P}X \mid X \in TM\}$$
$$T^{0,1}M = \{X \in T^{\mathbb{B}}M \mid \mathbf{P}X = -\tau X\} = \{X - \tau \mathbf{P}X \mid X \in TM\}.$$

In local \mathbb{B} -valued coordinates $z^i = x^i + \tau y^i$, the vector spaces $T^{1,0}M$ and $T^{0,1}M$ are spanned by the vector fields

$$\frac{\partial}{\partial z^i} = \frac{1}{2} \left(\frac{\partial}{\partial x^i} + \tau \frac{\partial}{\partial y^i} \right) \qquad \frac{\partial}{\partial \bar{z}^i} = \frac{1}{2} \left(\frac{\partial}{\partial x^i} - \tau \frac{\partial}{\partial y^i} \right) \ .$$

Similarly, the induced para-complex structure \mathbf{P}^* on T^*M induces a decomposition $T^{*\mathbb{B}}M = \Lambda^{1,0}(M) \oplus \Lambda^{0,1}(M)$ where

$$\Lambda^{1,0}(M) = \{ \alpha \in T^{*\mathbb{B}}M \mid \mathbf{P}^*\alpha = \tau\alpha \} = \{ \alpha + \tau \mathbf{P}^*\alpha \mid \alpha \in T^*M \}$$
$$\Lambda^{0,1}(M) = \{ \alpha \in T^{*\mathbb{B}}M \mid \mathbf{P}^*\alpha = -\tau\alpha \} = \{ \alpha - \tau \mathbf{P}^*\alpha \mid \alpha \in T^*M \} .$$

In local \mathbb{B} -valued coordinates $z^i=x^i+\tau y^i$, the vector spaces $\Lambda^{1,0}(M)$ and $\Lambda^{0,1}(M)$ are spanned by the \mathbb{B} -valued forms

$$dz^i = dx^i + \tau dy^i$$
 and $d\bar{z}^i = dx^i - \tau dy^i$.

More in general, we have a decomposition of \mathbb{B} -valued n-forms on M into types

$$\Lambda^n(M) = \bigoplus_{p+q=n} \Lambda^{p,q}(M) .$$

This induces a splitting of the \mathbb{B} -linear exterior differential $d: \Lambda^n(M) \to \Lambda^{n+1}(M)$ as $d = \partial_{\mathbf{P}} + \bar{\partial}_{\mathbf{P}}$ with

$$\partial_{\mathbf{P}}: \Lambda^{p,q}(M) \to \Lambda^{p+1,q}(M)$$
 and $\bar{\partial}_{\mathbf{P}}: \Lambda^{p,q}(M) \to \Lambda^{p,q+1}(M)$.

Para-Kähler potential. Recall that a para-Kähler structure is the data of an almost para-complex structure \mathbf{P} and a (non-degenerate) pseudo-Riemannian metric \mathbf{g} on M such that

- $\mathbf{g}(\mathbf{P}X, \mathbf{P}Y) = -\mathbf{g}(X, Y)$ for every $X, Y \in \Gamma(TM)$,
- **P** is parallel for the Levi-Civita connection of **g**.

This second condition is equivalent ([AMT09b]) to the simultaneous integrability of the eigendistributions of **P** and the closedness of the 2-form $\omega_{\mathbf{P}} = \mathbf{g}(\cdot, \mathbf{P}\cdot)$. It is easy to see that the eigendistributions of **P** are isotropic for **g**, thus **g** has signature (n, n).

Definition A.3. We say that $f: M \to \mathbb{R}$ is a para-Kähler potential of the closed 2-form $\omega_{\mathbf{P}}$ if $(\tau/2)\bar{\partial}_{\mathbf{P}}\partial_{\mathbf{P}}f = \omega_{\mathbf{P}}$.

Lemma A.4. For every smooth function $f: M \to \mathbb{R}$ we have $2\tau \bar{\partial}_{\mathbf{P}} \partial_{\mathbf{P}} f = d(df \circ \mathbf{P})$.

Proof. In local para-holomorphic coordinates $z^i = x^i + \tau y^i$ we have

$$df \circ \mathbf{P}\left(\frac{\partial}{\partial x^i}\right) = df\left(\frac{\partial}{\partial y^i}\right) = \frac{\partial f}{\partial y^i} \quad \text{and} \quad df \circ \mathbf{P}\left(\frac{\partial}{\partial y^i}\right) = df\left(\frac{\partial}{\partial x^i}\right) = \frac{\partial f}{\partial x^i}$$

thus

$$df \circ \mathbf{P} = \sum_{i=1}^{n} \frac{\partial f}{\partial y^{i}} dx^{i} + \frac{\partial f}{\partial x^{i}} dy^{i}$$

and

$$d(df \circ \mathbf{P}) = \sum_{i=1}^{n} \left(\frac{\partial^{2} f}{\partial x^{i}} - \frac{\partial^{2} f}{\partial y^{i}} \right) dx^{i} \wedge dy^{i} .$$

On the other hand

$$2\tau\bar{\partial}_{\mathbf{P}}\partial_{\mathbf{P}}f = \tau\bar{\partial}_{\mathbf{P}}\sum_{i=1}^{n} \left(\frac{\partial f}{\partial x^{i}} + \tau\frac{\partial f}{\partial y^{i}}\right) (dx^{i} + \tau dy^{i})$$

$$= \frac{1}{2}\tau\sum_{i=1}^{n} \left(\frac{\partial^{2} f}{\partial x^{i}} - \frac{\partial^{2} f}{\partial y^{i}}\right) (dx^{i} - \tau dy^{i}) \wedge (dx^{i} + \tau dy^{i})$$

$$= \sum_{i=1}^{n} \left(\frac{\partial^{2} f}{\partial x^{i}} - \frac{\partial^{2} f}{\partial y^{i}}\right) dx^{i} \wedge dy^{i}.$$

Criterion for a para-hyperKähler structure. Recall that a para-hyperKähler structure on M is a quadruple $(\mathbf{g}, \mathbf{I}, \mathbf{J}, \mathbf{K})$ where (\mathbf{g}, \mathbf{K}) and (\mathbf{g}, \mathbf{J}) are para-Kähler structures on M and $\mathbf{I} = \mathbf{K}\mathbf{J}$ is an almost complex structure on M that is compatible with \mathbf{g} in the sense that

$$g(IX, IY) = g(X, Y)$$
 for all $X, Y \in \Gamma(TM)$

and that is parallel for the Levi-Civita connection of **g**.

Lemma A.5. Let \mathbf{J}, \mathbf{K} be almost para-complex structures and let $\mathbf{I} = \mathbf{KJ}$ be an almost complex structure on M. Assume that there is pseudo-Riemannian metric \mathbf{g} compatible with \mathbf{I}, \mathbf{J} and \mathbf{K} . If the 2-forms $\omega_{\mathbf{L}} = \mathbf{g}(\cdot, \mathbf{L}\cdot)$ are closed for every $\mathbf{L} = \mathbf{I}, \mathbf{J}, \mathbf{K}$, then $(M, \mathbf{I}, \mathbf{J}, \mathbf{K}, \mathbf{g})$ is para-hyperKähler.

Proof. We first extend the differential forms $\omega_{\mathbf{L}}$ to \mathbb{B} -valued differential forms by \mathbb{B} -linearity. Because

$$\omega_{\mathbf{I}}(X,Y) = \mathbf{g}(X,\mathbf{I}Y) = -\mathbf{g}(X,\mathbf{J}\mathbf{K}Y) = \mathbf{g}(\mathbf{J}X,\mathbf{K}Y) = \omega_{\mathbf{K}}(\mathbf{J}X,Y)$$

for any section X of $T^{\mathbb{B}}M$, we have $\mathbf{J}X = \tau X$ if and only if $\iota_X \omega_{\mathbf{I}} = \tau \iota_X \omega_{\mathbf{K}}$. Suppose that $\mathbf{J}X = \tau X$ and $\mathbf{J}Y = \tau Y$, then

$$\iota_{[X,Y]}\omega_{\mathbf{I}} = \mathcal{L}_{X}(\iota_{Y}\omega_{\mathbf{I}}) - \iota_{Y}\mathcal{L}_{X}\omega_{\mathbf{I}}
= \mathcal{L}_{X}(\tau\iota_{Y}\omega_{\mathbf{K}}) - \iota_{Y}\mathcal{L}_{X}\omega_{\mathbf{I}}
= \mathcal{L}_{X}(\tau\iota_{Y}\omega_{\mathbf{K}}) - \iota_{Y}d(\iota_{X}\omega_{\mathbf{I}})$$
 (Cartan's formula with $\omega_{\mathbf{I}}$ closed)

$$= \mathcal{L}_{X}(\tau\iota_{Y}\omega_{\mathbf{K}}) - \iota_{Y}d(\tau\iota_{X}\omega_{\mathbf{K}})
= \mathcal{L}_{X}(\tau\iota_{Y}\omega_{\mathbf{K}}) - \tau\iota_{Y}\mathcal{L}_{X}\omega_{\mathbf{K}}$$
 (Cartan's formula with $\omega_{\mathbf{K}}$ closed)

$$= \tau\iota_{[X,Y]}\omega_{\mathbf{K}} .$$

Hence $\mathbf{J}[X,Y] = \tau[X,Y]$ and and the eigendistributions for \mathbf{J} are involutive. The integrability of \mathbf{J} and the closedness of $\omega_{\mathbf{J}}$ imply that \mathbf{J} is parallel for the Levi-Civita connection of \mathbf{g} . Repeating a similar computation for \mathbf{I} and \mathbf{K} we obtain the result.

References

- [AB83] M. F. Atiyah and R. Bott. The Yang-Mills equations over Riemann surfaces. Philos. Trans. Roy. Soc. London Ser. A, 308(1505):523-615, 1983.
- [ABB⁺07] Lars Andersson, Thierry Barbot, Riccardo Benedetti, Francesco Bonsante, William M. Goldman, François Labourie, Kevin P. Scannell, and Jean-Marc Schlenker. Notes on a paper of Mess. *Geom. Dedicata*, 126:47–70, 2007.
- [ABBZ12] Lars Andersson, Thierry Barbot, François Béguin, and Abdelghani Zeghib. Cosmological time versus CMC time in spacetimes of constant curvature. *Asian J. Math.*, 16(1):37–88, 2012.
- [AMT09a] D. V. Alekseevsky, C. Medori, and A. Tomassini. Homogeneous para-Kähler Einstein manifolds. Russ. Math. Surv., 64(1):1–43, 2009.
- [AMT09b] Dimitri V. Alekseevsky, Costantino Medori, and Adriano Tomassini. Para-Kähler Einstein metrics on homogeneous manifolds. C. R. Math. Acad. Sci. Paris, 347(1-2):69–72, 2009.
- [Bar15] Thierry Barbot. Deformations of Fuchsian AdS representations are quasi-Fuchsian. J. Differ. Geom., 101(1):1-46, 2015.
- [Bar18] Thierry Barbot. Lorentzian Kleinian groups. In *Handbook of group actions. Vol. III*, volume 40 of *Adv. Lect. Math. (ALM)*, pages 311–358. Int. Press, Somerville, MA, 2018.
- [BB09] Riccardo Benedetti and Francesco Bonsante. Canonical Wick rotations in 3-dimensional gravity. Mem. Amer. Math. Soc., 198(926):viii+164, 2009.
- [BBZ07] Thierry Barbot, François Béguin, and Abdelghani Zeghib. Constant mean curvature foliations of globally hyperbolic spacetimes locally modelled on AdS₃. *Geom. Dedicata*, 126:71–129, 2007.
- [BBZ11] Thierry Barbot, François Béguin, and Abdelghani Zeghib. Prescribing Gauss curvature of surfaces in 3-dimensional spacetimes: application to the Minkowski problem in the Minkowski space.

 Ann. Inst. Fourier (Grenoble), 61(2):511–591, 2011.
- [BE81] John K. Beem and Paul E. Ehrlich. Global Lorentzian geometry, volume 67 of Monographs and Textbooks in Pure and Applied Math. Marcel Dekker, Inc., New York, 1981.

- [BMS13] Francesco Bonsante, Gabriele Mondello, and Jean-Marc Schlenker. A cyclic extension of the earthquake flow I. Geom. Topol., 17(1):157–234, 2013.
- [BMS15] Francesco Bonsante, Gabriele Mondello, and Jean-Marc Schlenker. A cyclic extension of the earthquake flow II. Ann. Sci. Éc. Norm. Supér. (4), 48(4):811–859, 2015.
- [BS03] Antonio N. Bernal and Miguel Sánchez. On smooth Cauchy hypersurfaces and Geroch's splitting theorem. Comm. Math. Phys., 243(3):461–470, 2003.
- [BS10] Francesco Bonsante and Jean-Marc Schlenker. Maximal surfaces and the universal Teichmüller space. *Invent. Math.*, 182(2):279–333, 2010.
- [BS12] Francesco Bonsante and Jean-Marc Schlenker. Fixed points of compositions of earthquakes. *Duke Math. J.*, 161(6):1011–1054, 2012.
- [BS18] Francesco Bonsante and Andrea Seppi. Area-preserving diffeomorphisms of the hyperbolic plane and K-surfaces in anti-de Sitter space. J. Topol., 11(2):420–468, 2018.
- [BS19] Francesco Bonsante and Andrea Seppi. Equivariant maps into anti-de Sitter space and the symplectic geometry of $\mathbb{H}^2 \times \mathbb{H}^2$. Trans. Am. Math. Soc., 371(8):5433–5459, 2019.
- [BS20] Francesco Bonsante and Andrea Seppi. Anti-de Sitter geometry and Teichmüller theory. In: In the tradition of Thurston (to appear, V. Alberge, K. Oshika and A. Papadopoulos ed.). Springer Verlag, 2020.
- [BST17] Francesco Bonsante, Andrea Seppi, and Andrea Tamburelli. On the volume of anti-de Sitter maximal globally hyperbolic three-manifolds. Geom. Funct. Anal., 27(5):1106-1160, 2017.
- [CFG96] V. Cruceanu, P. Fortuny, and P. M. Gadea. A survey on paracomplex geometry. Rocky Mt. J. Math., 26(1):83–115, 1996.
- [CT19] Qiyu Chen and Andrea Tamburelli. Constant mean curvature foliation of globally hyperbolic (2+1)-spacetimes with particles. Geom. Dedicata, 201:281–315, 2019.
- [Dan13] Jeffrey Danciger. A geometric transition from hyperbolic to anti-de Sitter geometry. *Geom. Topol.*, 17(5):3077–3134, 2013.
- [Dan14] Jeffrey Danciger. Ideal triangulations and geometric transitions. J. Topol., 7(4):1118–1154, 2014.
- [Don03] S. K. Donaldson. Moment maps in differential geometry. In Surveys in differential geometry, Vol. VIII (Boston, MA, 2002), volume 8 of Surv. Differ. Geom., pages 171–189. Int. Press, Somerville, MA, 2003.
- [Ger70] Robert Geroch. Domain of dependence. J. Mathematical Phys., 11:437–449, 1970.
- [GMV01] Eduardo García-Río, Yasuo Matsushita, and Ramón Vázquez-Lorenzo. Paraquaternionic Kähler manifolds. Rocky Mt. J. Math., 31(1):237–260, 2001.
- [Gol80] William Mark Goldman. Discontinuous groups and the Euler class. PhD thesis, 1980. University of California, Berkeley.
- [Gol84] William M. Goldman. The symplectic nature of fundamental groups of surfaces. Adv. in Math., 54(2):200–225, 1984.
- [Hit87] N. J. Hitchin. The self-duality equations on a Riemann surface. *Proc. London Math. Soc.* (3), 55(1):59–126, 1987.
- [Hod05] Thomas Wolf Stephen Hodge. HyperKähler Geometry and Teichmüller Space. PhD thesis, Imperial College London, 2005.
- [Hop51] Heinz Hopf. Über Flächen mit einer Relation zwischen den Hauptkrümmungen. *Math. Nachr.*, 4:232–249, 1951.
- [KS07] Kirill Krasnov and Jean-Marc Schlenker. Minimal surfaces and particles in 3-manifolds. Geom. Dedicata, 126:187–254, 2007.
- [Mes07] Geoffrey Mess. Lorentz spacetimes of constant curvature. Geom. Dedicata, 126:3–45, 2007.
- [Min92] Yair N. Minsky. Harmonic maps, length, and energy in Teichmüller space. J. Differ. Geom., 35(1):151–217, 1992.

- [Pla01] Ioannis D. Platis. Complex symplectic geometry of quasi-Fuchsian space. Geom. Dedicata, 87(1-3):17-34, 2001.
- [PP08] John R. Parker and Ioannis D. Platis. Complex hyperbolic fenchel-nielsen coordinates. *Topology*, 47(2):101–135, 2008.
- [Sam78] Joseph H. Sampson. Some properties and applications of harmonic mappings. Ann. Sci. École Norm. Sup. (4), 11(2):211–228, 1978.
- [Sep19] Andrea Seppi. Maximal surfaces in Anti-de Sitter space, width of convex hulls and quasiconformal extensions of quasisymmetric homeomorphisms. J. Eur. Math. Soc. (JEMS), 21(6):1855–1913, 2019.
- [SS18] Carlos Scarinci and Jean-Marc Schlenker. Symplectic Wick rotations between moduli spaces of 3-manifolds. Ann. Sc. Norm. Super. Pisa, Cl. Sci. (5), 18(3):781–829, 2018.
- [Tam19] Andrea Tamburelli. Constant mean curvature foliation of domains of dependence in AdS₃. Trans. Am. Math. Soc., 371(2):1359–1378, 2019.
- [Tam20] Andrea Tamburelli. Fenchel-Nielsen coordinates on the augmented moduli space of anti-de Sitter structures. arXiv:1906.03715. To appear in Math. Zeitschrift., 2020.
- [Tau04] Clifford Henry Taubes. Minimal surfaces in germs of hyperbolic 3-manifolds. In *Proceedings of the Casson Fest*, volume 7 of *Geom. Topol. Monogr.*, pages 69–100. Geom. Topol. Publ., Coventry, 2004.
- [Tra18] Samuel Trautwein. Infinite dimensional GIT and moment maps in differential geometry. PhD thesis, ETH Zürich, 2018.
- [Tra19] Samuel Trautwein. The hyperkähler metric on the almost-Fuchsian moduli space. EMS Surv. Math. Sci., 6(1):83–131, 2019.
- [Tro92] Anthony J. Tromba. *Teichmüller theory in Riemannian geometry*. Lectures in Mathematics ETH Zürich. Birkhäuser Verlag, Basel, 1992. Lecture notes prepared by Jochen Denzler.
- [Vac12] Massimo Vaccaro. (Para-)Hermitian and (para-)Kähler submanifolds of a para-quaternionic Kähler manifold. *Differ. Geom. Appl.*, 30(4):347–364, 2012.
- [Voi02] Claire Voisin. Hodge theory and complex algebraic geometry, I. Cambridge studies in advanced mathematics, 2002.
- [Wol83] Scott A. Wolpert. On the symplectic geometry of deformations of a hyperbolic surface. Ann. of Math. (2), 117(2):207–234, 1983.
- [Wol89] Michael Wolf. The Teichmüller theory of harmonic maps. J. Differential Geom., 29(2):449–479, 1989.
- [Wol91a] Michael Wolf. High energy degeneration of harmonic maps between surfaces and rays in Teichmüller space. *Topology*, 30(4):517–540, 1991.
- [Wol91b] Michael Wolf. Infinite energy harmonic maps and degeneration of hyperbolic surfaces in moduli space. J. Differ. Geom., 33(2):487–539, 1991.
 - FM: Department of Mathematics, University of Virginia, Charlottesville (VA), USA. *Email address*: filippomazzoli@me.com
- AS: Institut Fourier, UMR 5582, Laboratoire de Mathématiques, Université Grenoble Alpes, CS 40700, 38058 Grenoble cedex 9, France.

Email address: andrea.seppi@univ-grenoble-alpes.fr

AT: Department of Mathematics, Rice University, Houston (TX), USA.

 $Email\ address: \ {\tt andrea.tamburelli@libero.it}$